

**Project Manual** 

for

# Multi-Campus Building Envelop Repairs (Package 4)

for the

# GALVESTON INDEPENDENT SCHOOL DISTRICT

14 September 2020

BEAM Professionals Project No.: 18267R Galveston ISD RFCSP #2020-01

Issue for Proposal



Project Manual

for

# Multi-Campus Building Envelop Repairs (Package 4)

for the

# GALVESTON INDEPENDENT SCHOOL DISTRICT

14 September 2020

BEAM Professionals Project No.: 18267R Galveston ISD RFCSP #2020-01

# Board of Trustees

Mr. Anthony Brown Mr. Johnny Smecca Ms. Monica Wagner Mr. David H. O'Neal, Jr. Mr. Shae Jobe Dr. Matthew Hay Ms. Mindy Lakin

# Administration

Dr. Kelli Moulton Mr. Timothy Bargeron Dr. Annette Scott Mr. Paul Byers Superintendent Assistant Superintendent for Business and Operations Assistant Superintendent for Curriculum & Instruction

President Vice President Secretary Trustee Trustee Trustee Trustee Trustee

**Director of Facilities** 

# PROJECT MANUAL TABLE OF CONTENTS

#### I. PROPOSAL DOCUMENTS AND CONTRACT FORMS

#### DIVISION 00 – PROCUREMENT AND CONTRACTING REQUIREMENTS

AA - REQUEST FOR COMPETITIVE SEALED PROPOSALS

**AB - INSTRUCTIONS TO OFFERORS** 

AC - COMPETITIVE SEALED PROPOSAL FORM

- AD PROPOSAL BOND
- **AE FELONY CONVICTION NOTIFICATION**
- AF AFFIDAVIT OF NON-DISCRIMINATORY EMPLOYMENT
- AG LIST OF SUBCONTRACTORS & MANUFACTURERS
- AH AFFIDAVIT OF NON-ASBESTOS, LEAD, AND PCB
- AI FINAL PAYMENT
- AJ CONTRACTOR SITE RULES
- **AK CERTIFICATE OF RESIDENCY**
- AL AFFIDAVIT OF NON-COLLUSION
- AM NOTICE OF NO BID/PROPOSAL
- AN CERTIFICATE OF CRIMINAL HISTORY
- AO CONFLICT OF INTEREST QUESTIONNAIRE
- BA AIA DOCUMENT 101 STANDARD FORM OF AGREEMENT BETWEEN OWNER AND CONTRACTOR.
- BB TEXAS STATUTORY PERFORMANCE BOND
- BC TEXAS STATUTORY PAYMENT BOND
- BD INSURANCE AND BONDS REQUIREMENTS FOR CONTRACTORS AND FACILITY RENTERS
- CA GENERAL CONDITIONS OF THE CONTRACT FOR CONSTRUCTION
- **CB SUPPLEMENTARY CONDITIONS**
- CC TERMS AND CONDITIONS
- CD SPECIAL TERMS AND CONDITIONS
- **CE INSURANCE**
- CF GENERAL REQUIREMENTS
- CG PREVAILING WAGE RATE DETERMINATION INFORMATION

#### II. SPECIFICATIONS

#### **DIVISION 1 - GENERAL REQUIREMENTS**

- SECTION 01 11 00 SUMMARY
  - 01 21 00 ALLOWANCES
  - 01 22 00 MEASUREMENT AND PAYMENT (UNIT PRICES)
  - 01 23 00 ALTERNATES
  - 01 25 00 SUBSTITUTION PROCEDURES
  - 01 26 00 CONTRACT MODIFICATION PROCEDURES
  - 01 29 00 PAYMENT PROCEDURES
  - 01 29 73 SCHEDULE OF VALUES SAMPLE
  - 01 31 13 PROJECT COORDINATION
  - 01 31 19 PROJECT MEETINGS
  - 01 32 16 CONSTRUCTION PROGRESS DOCUMENTATION
  - 01 33 00 SUBMITTAL PROCEDURES
  - 01 35 16 ALTERATION PROJECT PROCEDURES
  - 01 41 00 REGULATORY REQUIREMENTS
  - 01 45 00 QUALITY CONTROL
  - 01 50 00 TEMPORARY FACILITIES AND CONTROLS

#### PROJECT MANUAL TABLE OF CONTENTS

01 73 29 – CUTTING AND PATCHING 01 77 00 – CLOSEOUT PROCEDURES CLOSEOUT FORM A CLOSEOUT FORM B CLOSEOUT FORM C CLOSEOUT FORM D 01 78 39 – PROJECT RECORD DOCUMENTS

#### **DIVISION 2 DEMOLITION**

SECTION 02 50 00 – SELECTIVE DEMOLITION

# **DIVISION 3 CONCRETE**

SECTION 03 47 13 – TILT-UP CONCRETE PANELS

#### **DIVISION 4 MASONRY – NOT USED**

#### DIVISION 5 – METALS

SECTION 05 50 00 – METAL FABRICATIONS

# **DIVISION 6 – WOODS, PLASTICS AND COMPOSITES**

SECTION 06 10 00 – ROUGH CARPENTRY

#### **DIVISION 7 – THERMAL AND MOISTURE PROTECTION**

SECTION 07 19 00 – WATER REPELLENTS 07 56 00 – ACRYLIC FLUID APPLIED ELASTOMERIC COATING SYSTEM 07 63 00 – ROOF RELATED SHEET METAL 07 92 00 – JOINT SEALANTS

#### **DIVISION 9 – FINISHES**

SECTION 09 91 00 – PAINTING

#### DIVISION 10-32 - NOT USED

# END OF TABLE OF CONTENTS

# REQUEST FOR COMPETITIVE SEALED PROPOSALS GALVESTON INDEPENDENT SCHOOL DISTRICT

District Roof & Envelope Repairs (Package 4) GISD Request for Competitive Sealed Proposal No. 2020-01 PBK Project No. 18267R

Competitive sealed proposals will be received by Galveston Independent School District for the District Roof and Envelope Repairs (Package 4) project.

Pre-Proposal & Site Walk thru Date & Time:	Tuesday, September 15, 2020 at 10:00am
Pre- Proposal Location: (Pre-proposal meeting & start of Tour)	Galveston Independent School District Administration Building 3904 Avenue T. Galveston, TX 77550 (409) 766-5158
Proposal Date & Time:	Tuesday, September 29, 2020 at 2:00pm
Proposal Location:	Galveston Independent School District Administration Building 3904 Avenue T. Galveston, TX 77550 (409) 766-5158
Scope of Work:	Work to include roof and envelope repairs at the Administration Building, Annex Building, LA Morgan Elementary School, Parker Elementary School and Weis Collegiate Academy.
Proposal Documents:	Proposers will be expected to comply with the Proposal Evaluation Criteria published in the "Instructions to Offerors" section of the proposal documents. Plans and Specifications will be available to General Contractors on Monday, September 14, 2020. Plans and specifications can be obtained with a \$25.00 deposit (per set), two (2) set maximum and will be expected to submit a fully executed Contractor's Qualification Statement, AIA Document A305, to the Engineer/Architect of record at:
Engineer/Architect:	PBK Architects (dba BEAM Professionals) Attn. James Roberts // <u>James.Roberts@beamprof.com</u> 11 Greenway Plaza, 22 <sup>nd</sup> Floor Houston, Texas 77046 Phone: 713-965-0608 Office Hours: 8:30 AM to 5:30 PM

#### DOCUMENT AB

#### ROOF & ENVELOPE REPAIRS AT ROSENBERG ELEMENTARY SCHOOL INSTRUCTIONS TO OFFERORS

# 1.1 GENERAL

- A. Competitive Sealed Proposals will be accepted from qualified Offerors (General Contractors) only for the entire scope of work described in the Contract Documents. As a prerequisite to an Offeror's qualifying for the award of contract on this work, the Offeror must complete each item of the Contractor's Qualification Statement (AIA Document A305). The Statement forms may be obtained from the Houston Chapter of the American Institute of Architects, 315 Capitol, Suite 120, Houston, Texas 77002, (713) 520-0155. In addition to the information contained in the Statement form, offerors shall also address the selection criteria issues listed under the paragraph below for Determination of Successful Respondent and Award of Contract. The Statement and other requested information shall be submitted at the time of receipt of Proposals in three (3) copies, two (2) for the Owner and one (1) for the Architect. Qualification statements submitted by FAX transmission will not be accepted.
- B. The primary purposes of the evaluation process will be to:
  - 1. Gather information for the Owner's evaluation procedure.
  - 2. Enable the Owner and/or Architect to evaluate the Offeror's qualifications.
- C. After review of Proposals and Contractor's qualifications evaluation the Owner will make his decision and each Offeror will be notified.
- D. In arriving at his opinion concerning the Offeror's qualifications, the Architect will use the same criteria that the Owner will use in determination of the successful Offeror as detailed hereinafter.
- E. In the event a proposed Offeror fails to submit the specified Contractor's Qualification Statement at time of receipt for Proposals, such noncompliance shall be considered by both the Owner and Architect as a negative factor in the determination of the successful Offeror.

# 1.2 OFFEROR'S PRESENTATION

- A. Each Offeror by making his Proposal represents that:
  - 1. He has read and understands the Proposal Documents and his Proposal is made in accordance therewith.
  - 2. He has thoroughly familiarized themselves with Division 01 General Requirements as they are applicable to subsequent specification sections.
  - 3. He has visited the site, has familiarized himself with the local conditions under which the work is to be performed and has correlated his observations with the requirements of the proposed Contract Documents.
  - 4. He agrees to comply with the requirements of the following paragraph. Any Offeror who subsequently does not agree to comply with these requirements will automatically disqualify himself from proposing or receiving award of the contract.
- B. He agrees that:
  - 1. Work on the project will begin immediately upon receipt of signed Contract or Notice to Proceed.

- 2. Offeror will participate as a team member in cooperation with the Project Architect, Engineers, Owner, and Owners agents and/or consultants.
- 3. The Offeror will assign a competent full-time superintendent, to the project, and that superintendent shall be maintained on the project for the duration of the project, subject only to his continuous employment.
- 4. The Offeror will furnish and pay for a proposal bond in the amount of ten percent (10%) of the contract amount.
- 5. If awarded, the Offeror shall furnish and pay for a Performance Bond and a Payment Bond each in the full contract amount.
- 6. Offeror shall carry and keep in full force for the duration of the Project, insurance coverage for builder's risk, workmen's compensation, comprehensive general liability, and automobile liability as required by the General Conditions and/or Supplementary General Conditions of the Specifications.
- 7. Each Offeror by making his Proposal represents that his Proposal includes only material and equipment specified in the Proposal Documents and supplemented, if necessary, for a complete and operating system.
- 8. Where subcontract work is involved and where Acceptable Subcontractors are designated for particular sections or phases of the Work, each Offeror by making his Proposal represents that his Proposal includes only firms designated as Acceptable Subcontractors.

# 1.3 PROPOSAL DOCUMENTS

- A. Proposal Documents include the Request for Competitive Sealed Proposals, Instructions to Offerors, the Proposal Form, and the proposed Contract Documents, including any Addenda issued prior to receipt of proposals.
- B. contract Documents for the work consist of the Owner-Contractor Agreement, the Conditions of the Contract (General, Supplementary and other Conditions), the Drawings, the Specifications, and all Addenda issued prior to receipt of proposals.

# 1.4 PROPOSAL PROCEDURES

- A. A proposal is invalid if it has not been received at the designated location prior to the time and date for receipt of proposals indicated in the Request for Competitive Sealed Proposals, or prior to any extension thereof issued to the Offerors by Addenda.
- B. All requested Alternates shall be proposed. If no change in the Base Proposal is required, enter "No Change".
- C. Prior to the receipt of Proposals, Addenda will be forwarded by the Architect and will be available for inspection wherever the proposal documents are kept available for that purpose.
- D. Proposals will be received in duplicate only on the Owner's Form of Proposal for the work as indicated by the Proposal Documents, filled in, and enclosed in a sealed envelope addressed as follows:

Name of Offeror (General Contractor) Multi-Campus Building Envelope Repairs (Package 4) Galveston Independent School District Attn: Mrs. Connie Morgenroth Assistant Superintendent for Business and Operations 3904 Avenue T Galveston, Texas 77550

> INSTRUCTIONS TO OFFERORS AB - 2

- E. The Proposal Form must be accompanied by Proposal Bond or Certified Check in the amount of 10% of the proposal.
- F. All proposals must be delivered sealed to the above address at or before the time and date set. Proposals will be received at no other place. If Proposal is sent by U.S. Mail, it must be sent Registered Mail.
- G. A proposal may be withdrawn only upon request by the Offeror or his duly authorized representative, provided such request is received by the Owner at the place designated for receipt of proposals and prior to the time fixed for the opening of proposals. A withdrawal of a proposal shall not be effective unless a written confirmation of the withdrawal is received by the Owner at said place within 48 hours before the time fixed for the opening of proposals. The Proposal Bond will be returned with the proposals if withdrawn in accordance with the above. The withdrawal of a proposal does not prejudice the right of the Offeror to file a new proposal at the time and place stated. No proposal may be withdrawn after the time fixed for the opening of proposals for a period of 30 days.

# 1.5 INTERPRETATION OF PROPOSAL DOCUMENTS

- A. Offerors and sub-offerors requiring clarification or interpretation of the Proposal Documents shall make a written or verbal request which shall reach the Architect at least ten (10) days prior to the date for receipt of proposals.
- B. Any interpretation, correction or change of the Proposal Documents will be made by Addendum. Interpretations, corrections or changes of the Proposal Documents made in any other manner will not be binding.

#### 1.6 **REJECTION OF PROPOSALS**

- A. The Owner shall have the right to reject any or all proposals and to reject a proposal not accompanied by any required proposal security, or by other data required by the Proposal Documents, or to reject a proposal which is in any way incomplete or irregular.
- B. The Owner reserves the right to reject any or all proposals and to waive any formalities or irregularities and to make the award of the contract in the best interest of the Owner.
- C. The Owner reserves the right to reject any proposal if the evidence submitted by, or investigation of, such offeror fails to satisfy the Owner that such offeror is properly qualified to carry out the obligations of the contract and to complete the work therein. Award may be made to other than the low-dollar offeror and given the one offering the "best value" to the school district, in addition to the purchase price, based on the published selection criteria and on its ranking evaluation.
- D. Do not submit voluntary alternates. The Owner reserves the right to reject any proposal which is accompanied by conditional or qualifying statements, or "voluntary alternates".

# 1.7 INSURANCE

A. Each Offeror shall include in his proposal the complete cost and shall carry and keep in full force for the duration of the project, insurance coverage required under the General Conditions and Document CB - Supplementary Conditions.

# 1.8 PERFORMANCE BOND AND PAYMENT BOND

A. Each Offeror shall include in this proposal the premium costs for 100% Performance Bond and 100% Payment Bond. These bonds shall cover the faithful performance of the contract and payment of all obligations arising thereunder in such form as the Owner may prescribe. The bonding companies must be acceptable to the Owner. The selected Offeror shall deliver the required bonds to the Owner not later than the date of execution of the Contract.

# 1.9 **PROPOSAL SECURITY**

A. No proposal will be considered unless it is accompanied by a Certified or Cashier's Check or Proposal Bond executed on the form attached. In either case the amount shall be not less than ten percent (10%) of the greatest amount proposed (considering alternates, if any). The proposal security shall insure the execution of the contract and the furnishing of an acceptable Performance Bond and Payment Bond by the successful Offeror within ten (10) days after notification of award to such Offeror and that this proposal will not be withdrawn within 30 days after date of opening of proposals without the consent of the Owner. Proposal Bond shall be prepared in the identical form of AIA Document A310 or the form attached.

#### 1.10 AWARD OF CONTRACT

- A. The Offeror to whom the award is made will be promptly notified. If an Offeror (a) withdraws his proposal within 30 days after the date of time fixed for the opening of proposals in the Request for Competitive Sealed Proposals, or (b) fails or refuses to execute the Agreement, or other required forms within ten (10) days after the same are presented to him for signature, or (c) fails or refuses to furnish properly executed Performance Bond and Certification of Required Insurance within 15 calendar days of execution date of the Agreement, the Owner may award the work to another Offeror or Offerors or may call for new proposals.
- B. The Offeror will be required to (a) submit his Proposal and Proposal Bond, (b) execute Contract and Performance and Payment Bonds, and (c) submit Certification of required insurances, all using the Owner's own forms for such respective purposes.
- C. Proposal Bond is forfeited if proposal is withdrawn after the proposal opening, or Contract Documents are not executed in accordance with the above.

# 1.11 NOTICE TO PROCEED

A. The Offeror shall not commence work under this Contract until he receives the written Notice to Proceed, or the Contract is duly signed by the Owner.

# 1.12 COMPLETION TIME

- A. Offerors shall familiarize themselves with the Owners requirements concerning the project schedule as described in Section 01 31 00 of this Project Manual and throughout the contract documents.
- B. Having thoroughly familiarized himself with the conditions as they exist at the building sites and acquainted him with the labor supply and the material market, the Offeror, by submitting a proposal for work, agrees to be substantially complete with the work by the dated stated above.

It is therefore expressly agreed as a part of the consideration inducing the Owner to C. execute this contract that the Owner may deduct liquidated damages from the final payment made to the Contractor for each and every calendar day beyond the agreed date which the Contractor shall require for Substantial Completion of the work included in this contract. It is expressly understood that the said sum per day is agreed upon as a fair estimate of the pecuniary damages which will be sustained by the Owner in the event that the work is not completed within the agreed time, or within the legally extended time, if any, otherwise provided for herein. Said sum shall be considered as liquidated damages only and in no sense shall be considered a penalty, said damage being caused by additional compensation to personnel, for loss of interest on money and other miscellaneous increased costs, all of which are difficult of exact ascertainment. Also, any disruption of Owner's use of the existing facilities or newly completed facilities will also be subject to liquidated damages. Refer to Supplementary Conditions for additional requirements. Delays, disruption of use, failures to complete, and liquidated damages are fully described under Article 8.3 of the Supplementary Conditions.

The definition of Substantial Completion is found in Article 9.8.1 of the AIA General Conditions and Supplementary Conditions bound herein.

# 1.13 FELONY CONVICTION NOTIFICATION

- A. Section 44.034, of the Texas Education Code requires a person or business entity that enters into a contract with a school district must give advance notice to the district if the person or an owner or operator of the business entity has been convicted of a felony. The notice must include a general description of the conduct resulting in the conviction of a felony. Subsection (b) states "a school district may terminate a contract with a person or business entity if the district determines that the person or business entity failed to give notice as required by Subsection (a) or misrepresented the conduct resulting in the conviction. The district must compensate the person or business entity for services performed before the termination of the contract." Subsection (c) states "this section does not apply to a publicly held corporation."
- B. The Offeror <u>must</u> execute Document AE, Statement of Affirmation and submit with proposal.

#### 1.14 AFFIDAVIT OF NON-DISCRIMINATORY EMPLOYMENT

- A. The Offeror, and sub-offerors, shall agree to refrain from discrimination in terms and conditions of employment to the basis of race, color, religion, sex, or national origin, and agrees to take affirmative action as required by Federal Statutes and Rules and Regulations issued in order to maintain and insure non-discriminatory employment practices.
- B. The Offerors must execute Document AF, Affidavit of Non-Discriminatory Employment and submit with Proposal. The sub-offerors shall execute Document AF, Affidavit of Non-Discriminatory Employment before commencing work on this Project. Offerors and sub-offerors who have not executed this document will not be eligible to work on this project.

# 1.15 LIST OF SUBCONTRACTORS & MANUFACTURERS

- A. The Offeror shall supply a list of the following major:
  - 1. Mechanical
  - 2. Electrical
  - 3. Plumbing
  - 4. Masonry
  - 5. Any all other prudent subcontractor

- 6. Certified manufactures
- B. The Offeror <u>must</u> execute Document AG, List of Subcontractors & Manufacturers and submit with their Proposal.

# 1.16 CONFLICT OF INTEREST QUESTIONNAIRE

- A. According to Local Government Code, Chapter 176, a person or an agent of a person who contracts or seeks to contract for the sale or purchase of property, goods, or services with any government agency must file a completed Conflict of Interest Questionnaire with the records administrator of the local government not later than the seventh business day after the date that the person begins contract discussions or negotiations with the District or submits to the District an application, response to a request for proposals or bids, correspondence, or another writing related to a potential agreement with the District.
- B. All Offeror's and sub-offeror's proposing to do work with the District must execute Document AO, Conflict of Interest Questionnaire and submit to the District's Legal Department within seven (7) days of the Proposal Date. This requirement will be waived if the Offeror or sub-offeror has previously submitted such document to the District within the last year. In such case, provide written notification and attach to the Proposal.

# 1.17 CRIMINAL HISTORY RECORDS

- A. Prior to commencing any work on this Project, the Selected Contractor shall certify, on the form provided herein as Document AN, that for each of its employee who will have direct contact with students, the Selected Contractor has obtained, as required by Texas Education Code Section 22.0835:
  - 1. national criminal history record information from a law enforcement or criminal justice agency for each employee of the Selected Contractor hired before January 1, 2008 who will have direct contact with students; and
  - 2. national criminal history record information from the Texas Department of Safety for each employee of the Selected Contractor hired on or after January 1, 2008 who will have direct contact with students; Fingerprinting is required and shall be provided by the contractor (applicant) and administered through FAST (Fingerprint Applicant Services of Texas) which will be recorded by the District in the FACT (Fingerprint-based Applicant Clearinghouse of Texas). Currently applicant must obtain fingerprinting from L-1 Identity Solutions Company, (888) 467-2080, or schedule an appointment online at: https://tx.ibtfingerprint.com/.
- B. Any personnel who will have direct contact with students must not have been convicted of an offense identified in Texas Education Code Section 22.085.
- C. At this time, Senate Bill 9 applies only to contractors with direct contracts with the District. This requirement does not apply to sub-contractors of the Contractor, material suppliers, or a one-time service provider such as a service technician, delivery person, testing agent, code official, or similar personnel. However, changes to these requirements are anticipated and may require the acquisition and submittal of additional background checks to the District during the course of the Work.
- D. The Selected Contractor shall execute and submit Document AN, Certification of Criminal History Record Information Review by Contractor-Employer along with required Schedule 'A' documenting proposed employees to be working on site, within 10 days after receipt of Notice To Proceed and prior to commencement of Work.
- E. Furthermore, an updated Schedule 'B' shall be submitted weekly to the District indicating changes to contractor personnel with accompanying certifications and criminal history

records. Any fingerprinting and photographing required by the aforementioned code will be the responsibility of the Contractor-Employer.

# 1.18 DETERMINATION OF SUCCESSFUL RESPONDENT AND AWARD OF CONTRACT

A. In determining the Selected Offeror, the Owner will evaluate the information submitted on the criteria including, but not limited to the following:

- B. Proposal Form and other selection may commence immediately following receipt of the Offeror's list of subcontractors the day following receipt of proposals.
- C. It is the intent of the Owner to award the Contract to the Selected Offeror at the regularly scheduled Galveston ISD Board Meeting held; with issuance of a notice to proceed to follow immediately.
- D. By submitting a proposal, each Offeror agrees to waive any claim it has or may have against the Owner and its respective employees, the Architect/Engineer and consultants, and their respective employees, arising out of or in connection with the administration, evaluation, or recommendation of any proposal; waiver of any requirements under the Proposal Documents, acceptance or rejection of any proposals; and award of the contract.
- E. The Owner reserves the right to review all Offeror's and sub-offeror's qualification as deemed necessary, and shall have the final decision in the selection process. The Owner will make such investigations as it deems necessary to determine the ability of the Offeror to perform the Work, and the Offeror shall furnish all such information and data for this

purpose as may be requested. The Owner reserves the right to reject any proposal if the evidence submitted by, or investigation of, such Offeror fails to satisfy the District that such Offeror is properly qualified to carry out the obligations of the Contract and to complete the Work contemplated therein.

F. The Owner reserves the right to reject any or all proposals and to waive any informalities or irregularities and to make the award of the contract in the best interest of the Owner.

# END OF DOCUMENT AB

#### DOCUMENT AC

#### COMPETITIVE SEALED PROPOSAL FORM

#### 2020 MULTI-CAMPUS BUILDING ENVELOPE REPAIRS (PACKAGE 4) GALVESTON INDEPENDENT SCHOOL DISTRICT

Submitted by: \_\_\_\_\_

Date: \_\_\_\_\_ Phone No.: \_\_\_\_\_

To: Galveston Independent School District Administration Building 3904 Avenue T Galveston, TX 77550

Having examined Proposal and Contract Documents prepared by BEAM Professionals, dated September 14, 2020 and having examined site conditions, the undersigned proposes to furnish all labor, equipment and materials and perform all work for the completion of the above-named project for the sum indicated below.

In submitting his Proposal, the undersigned agrees to the following:

- 1. Hold proposal open for acceptance 30 days.
- 2. Accept right of Owner to reject any or all proposals, to waive formalities and to accept proposal which Owner considers most advantageous.
- 3. Enter into and execute the contract, if awarded, for the Base Proposal and accepted Alternate Proposals.
- 4. Complete work in accordance with the Contract Documents within the stipulated contract time.
- 5. By signing, the undersigned affirms that, to the best of his knowledge, the Proposals have been arrived at independently and is submitted without collusion with anyone to obtain information or gain any favoritism that would in any way limit competition or give an unfair advantage over respondents in the award of this proposal.

#### I. BASE PROPOSAL

Base Proposal No.1: **MULTI-CAMPUS BUILDING ENVELOPE REPAIRS** Undersigned agrees to complete the Work for the lump sum amount of:

Dollars \$\_

(Amount in figures)

(Amount written in words governs)

# II. ALTERNATE PROPOSAL

A. Alternate Proposal No.1: POWER WASH, PREP, PRIME, AND PAINT BRICK AT GISD ANNEX

Undersigned agrees to complete the Work for the lump sum amount of:

Dollars \$

(Amount written in words governs)

(Amount in figures)

NOTE: THIS DOCUMENT MUST BE SUBMITTED BY 2:00 PM ON SEPTEMBER 24, 2020

#### B. Alternate Proposal No.2: CLEAR ELASTOMERIC SEALANT TO BE APPLIED TO ENTIRE CAMPUS AT LA MORGAN ELEMENTARY. Undersigned agrees to complete the Work for the lump sum amount of:

Dollars \$ (Amount in figures) (Amount written in words governs) C. Alternate Proposal No.3: CLEAR ELASTOMERIC SEALANT TO BE APPLIED TO ENTIRE CAMPUS AT PARKER ELEMENTARY. Undersigned agrees to complete the Work for the lump sum amount of: Dollars \$ (Amount written in words governs) (Amount in figures) ALLOWANCES Undersigned certifies that the allowances specified below and in Section 01 21 00 are included in the Proposals and agrees that unexpended balance of allowance sums will revert to Owner in the final settlement of the contract: A. Contingency Allowance for Base Proposal 1 ......\$50,000.00 CONTRACT TIME

Undersigned agrees to begin on November 30, 2020 and be Substantially Complete as follows: All Work shall be Substantially Complete by **February 22, 2020.** 

#### V. UNIT PRICES

III.

IV.

A. **Unit Price No.1:** Remove and replace wood nailers *(include 400 lf in each dimension in Base Proposal):* 

Unit Price No. 1a (2 x 4)	Dollars \$		_/LF
		(Amount in figures)	
Unit Price No. 1b (2 x 6)	Dollars \$		_/LF
		(Amount in figures)	
Unit Price No. 1c (2 x 8)	Dollars \$		_/LF
		(Amount in figures)	
Unit Price No. 1d (2 x 12)	Dollars \$		_/LF
		(Amount in figures)	

#### VI. ADDENDA

Undersigned acknowledges receipt of Addenda	Nos
dated	, 20

# VII. CHANGES IN THE WORK

Undersigned understands that changes in the work shall be performed in accordance with the Supplementary Conditions.

# NOTE: THIS DOCUMENT MUST BE SUBMITTED BY 2:00 PM ON SEPTEMBER 24, 2020

# VIII. LIQUIDATED DAMAGES

Undersigned understands that liquidated damages as defined in the Supplementary Conditions will be included in the form of Agreement between Owner and Contractor and that the contractor will be bound thereto.

It is understood that the right is reserved by the Owner to reject any or all proposals, or waive any informalities in the proposal process.

Authorized Signature

Title

(Seal, if a Corporation) State whether Corporation, Partnership or Individual

Name of Contracting Firm

Address

Telephone

Date

END OF DOCUMENT AC

# DOCUMENT AD - PROPOSAL BOND

#### KNOW ALL MEN BY THESE PRESENTS, that we

as Principal, and \_\_\_\_\_\_, as Surety, are held and firmly bound unto the Galveston Independent School District, Galveston County, Texas, hereinafter called the Owner, in the penal sum of \_\_\_\_\_\_ Dollars (\$\_\_\_\_\_\_) lawful money of the United States, for the payment of which sum well and truly to be made, we bind ourselves, our heirs, executors, administrators and successors jointly and severally, firmly by these presents.

**THE CONDITION OF THIS OBLIGATION IS SUCH**, that whereas the Principal has submitted the accompanying Proposal, dated \_\_\_\_\_\_, being for the Roof & Envelope Repairs at Rosenberg Elementary School project, the kind and extent of work involved being set forth in detail in the proposed Contract Documents cited herein.

**NOW, THEREFORE**, if the Principal shall not withdraw the accompanying proposal within 30 days after the date set for opening thereof, and shall within ten (10) days after the prescribed forms are presented to him for signature, enter into a written contract with the Owner in accordance with the Proposal as accepted; and give Bond and good and sufficient surety for the faithful performance and proper fulfillment of such contract including payment of all persons supplying labor or materials therefor, or in the event of the withdrawal of said proposal within the period specified, or the failure to enter into such contract and give such bond within the time specified, if the Principal shall pay to the Owner the difference between the aggregate amount for which the Owner may enter into a contract for the same work with another Respondent; if the latter amount be in excess of the former, then the above obligation shall be void and of no effect, otherwise to remain in full force and virtue.

**IN WITNESS WHEREOF**, the above bonded parties have executed this instrument under their several seals this \_\_\_\_\_ day of \_\_\_\_\_\_, 20\_\_, the name and Corporate Seal of each corporate party being hereto affixed and these presents duly signed by its undersigned representatives, pursuant to authority of its governing body.

Business Address		Individual Principal
Business Address		Individual Principal
ATTEST:		
Secretary President	BY:	
Business Address	<u> </u>	Corporate Surety
ATTEST:	BY:	
	END OF DOCUMENT AD	

# **DOCUMENT AE - FELONY CONVICTION NOTIFICATION**

#### Note: The Statement of Affirmation Must Be Notarized

#### STATEMENT OF AFFIRMATION

"The undersigned affirms that he/she is duly authorized to provide this information by the person(s) or business entity making the proposal, and the information provided below concerning felony convictions has been personally and thoroughly reviewed, and verified, and is, therefore, current, true and accurate to the best of my knowledge."

Firms Name: <sub>.</sub>		Address:	
"a "b "c	My firm is a publicly held corporation, therefore, this reporting requirement is not applicable." My firm is not owned nor operated by anyone who has been convicted of a felony." My firm is owned or operated by the following individual(s) who has/have been convicted of felony:"		
	Name of Felon(s)		
	Details of Conviction(s)		
PLEAS	E CHECK a, b, or c ABOVE AND SI	IGN BELOW	
Offeror' Name _	's	Position/Title	
Offeror' Signatu		Date	
Subscri	ibed and sworn to me on this	day of	, 20
Notary	Public		
My Con	nmission Expires	_	
	EN	ND OF DOCUMENT AE	

# DOCUMENT AF - AFFIDAVIT OF NON-DISCRIMINATORY EMPLOYMENT

)))

STATE OF TEXAS	
COUNTY OF	

#### AFFIDAVIT

This Company, Contractor, or Subcontractor agrees to refrain from discrimination in terms and conditions of employment to the basis of race, color, religion, sex, or national origin, and agrees to take affirmative action as required by Federal Statutes and rules and Regulations issued pursuant thereto in order to maintain and insure non-discriminatory employment practices.

_	Company
_	Printed Name
_	Signature
STATE OF TEXAS ) ) COUNTY OF )	
Sworn to and subscribed before me at, 20	, Texas, this the day of

Notary Public in and for \_\_\_\_\_ County, Texas

# END OF DOCUMENT AF

# DOCUMENT AG - LIST OF SUBCONTRACTORS AND MANUFACTURERS

PROJECT: (Name, address)	GISD A 3904 A Galvest	dministration venue T on, Texas 775	g Envelope Rep 550 nt School Distric		BEAM Professi 11 Greenway F Houston, Texas	Plaza, 2 s 77046	
TO:	[Street}	ctor Name] ate, Zip]			DATE:		
List all Subcor	ntractors	and manufact	urers proposed o docume		e Project as req	uired b	y the proposal
	(To b	e filled out by	the Contractor		ed to the Archite	ct.)	
Work/ Division F	Firm	Address	Phone	Fax	E-Mail		Representative

(Provide additional sheets as required.)

# END OF DOCUMENT AG

NOTE: THIS DOCUMENT MUST BE EXECUTED AND SUBMITTED WITH PROPOSAL

LIST OF SUBCONTRACTORS & MANUFACTURERS AG - 1

# DOCUMENT AH - AFFIDAVIT OF NON-ASBESTOS, LEAD, AND PCB USE IN PROJECT

Upon completion of this form, return to the Architect upon close-out of the project.

PROJECT:	Multi-Campus Building Envelope Repairs (Pkg	4) ARCHITECT:
	GISD Administration	BEAM Professionals
	3904 Avenue T	11 Greenway Plaza, 22 <sup>nd</sup> Floor
(Name, Address)	Galveston, Texas 77550	Houston, Texas 77046
	Galveston Independent School District	

ARCHITECT'S PROJECT NO. 18266

OWNER NAME:	Galveston Independent School District
(Address)	3904 Avenue T
	Galveston, Texas 77550

#### AFFIDAVIT

The undersigned affirms and certifies that "to the best of their knowledge and belief asbestos, lead, and PCB containing materials have not been used or incorporated into the Work and lead or lead bearing materials have not been incorporated into potable water systems", including, but not limited to those water systems for drinking fountains, all sinks, showers, bath tubs, residential and commercial kitchen equipment, ice machines, and hose bibs, as applicable to the project, and that lead sheet flashing used in through roof plumbing penetration applications is the only lead on the Project.

	Company	
	Printed Name	
	Signature	
STATE OF TEXAS		
) COUNTY OF )		
Sworn to and subscribed before me at, 20 .	, Texas, this the day	y of

NOTE: THIS DOCUMENT MUST BE EXECUTED AND SUBMITTED AT PROJECT CLOSE-OUT

# DOCUMENT AI – RELEASE OF LIEN FORMS

#### FORM 1: CONDITIONAL WAIVER FOR PROGRESS PAYMENTS

MULTI-CAMPUS BUILDING ENVELOPE REPAIRS (PACKAGE 4) GALVESTON INDEPENDENT SCHOOL DISTRICT

#### CONDITIONAL WAIVER AND RELEASE ON PROGRESS PAYMENT

On receipt by the signer of this document of a check from \_\_\_\_\_\_ (maker of check) in the sum of \$\_\_\_\_\_\_ payable to \_\_\_\_\_\_ (payee or payees of check) and when the check has been properly endorsed and has been paid by the bank on which it is drawn, this document becomes effective to release any mechanic's lien right, any right arising from a payment bond that complies with a state or federal statute, any common law payment bond right, any claim for payment, and any rights under any similar ordinance, rule, or statute related to claim or payment rights for persons in the signer's position that the signer has on the property of \_\_\_\_\_\_ (owner) located at

\_\_\_\_\_ (location) to the following extent:

\_\_\_\_\_ (job description).

This release covers a progress payment for all labor, services, equipment, or materials furnished to the property or to \_\_\_\_\_\_ (person with whom signer contracted) as indicated in the attached statement(s) or progress payment request(s), except for unpaid retention, pending modifications and changes, or other items furnished.

Before any recipient of this document relies on this document, the recipient should verify evidence of payment to the signer.

The signer warrants that the signer has already paid or will use the funds received from this progress payment to promptly pay in full all of the signer's laborers, subcontractors, material men, and suppliers for all work, materials, equipment, or services provided for or to the above referenced project in regard to the attached statement(s) or progress payment request(s).

Date		-		
		(Company name)		
Ву		(Signature)		
		(Printed/Typed name)		
		(Title)		
SWORN AND SUBSCRIBED before me at	, this _	day of	, 20	_ A.D.
Notary Public in and for the State of				

# **DOCUMENT AI – RELEASE OF LIEN FORMS**

#### FORM 2: UNCONDITIONAL WAIVER FOR PROGRESS PAYMENTS

#### MULTI-CAMPUS BUILDING ENVELOPE REPAIRS (PACKAGE 4) GALVESTON INDEPENDENT SCHOOL DISTRICT

NOTICE: THIS DOCUMENT WAIVES RIGHTS UNCONDITIONALLY AND STATES THAT YOU HAVE BEEN PAID FOR GIVING UP THOSE RIGHTS. IT IS PROHIBITED FOR A PERSON TO REQUIRE YOU TO SIGN THIS DOCUMENT IF YOU HAVE NOT BEEN PAID THE PAYMENT AMOUNT SET FORTH BELOW. IF YOU HAVE NOT BEEN PAID, USE A CONDITIONAL RELEASE FORM.

UNCONDITIONAL WAIVER AND RELEASE ON PROGRESS PAYMENT

The signer of this document has been paid and has received a progress payment in the sum of for all labor, services, equipment, or materials furnished to the property or to (person with whom signer contracted) on the property of (Owner) located at (location) to the following extent: (job description). The signer therefore waives and releases any mechanic's lien right, any right arising from a payment bond that complies with a state or federal statute, any common law payment bond right, any claim for payment, and any rights under any similar ordinance, rule, or statute related to claim or payment rights for persons in the signer's position that the signer has on the above referenced project to the following extent:

This release covers a progress payment for all labor, services, equipment, or materials furnished to the property or to \_\_\_\_\_\_ (person with whom signer contracted) as indicated in the attached statement(s) or progress payment request(s), except for unpaid retention, pending modifications and changes, or other items furnished.

The signer warrants that the signer has already paid or will use the funds received from this progress payment to promptly pay in full all of the signer's laborers, subcontractors, material men, and suppliers for all work, materials, equipment, or services provided for or to the above referenced project in regard to the attached statement(s) or progress payment request(s).

Date		
	(Company name)	
Ву	(Signature)	
	(Printed/Typed name)	
	(Title)	
SWORN AND SUBSCRIBED before me at	, this day of	. 20
A.D		
Notary Public in and for the State of		

#### DOCUMENT AI -RELEASE OF LIEN FORMS

#### FORM 3: CONDITIONAL WAIVER FOR FINAL PAYMENT

MULTI-CAMPUS BUILDING ENVELOPE REPAIRS (PACKAGE 4) GALVESTON INDEPENDENT SCHOOL DISTRICT

#### CONDITIONAL WAIVER AND RELEASE ON FINAL PAYMENT

Project \_\_\_\_\_ Job No. \_\_\_\_\_

On receipt by the signer of this document of a check from		check from (maker of check) in
the sum of \$	payable to	(payee or payees of check)
and when the check l	nas been properly endorsed and ha	as been paid by the bank on which it is drawn, this
document becomes e	effective to release any mechanic's	lien right, any right arising from a payment bond
that complies with a s	state or federal statute, any commo	on law payment bond right, any claim for payment,
and any rights under	any similar ordinance, rule, or statu	ute related to claim or payment rights for persons
in the signer's positio	n that the signer has on the propert	rty of (owner) located at

\_\_\_\_\_ (location) to the following extent:

\_\_\_\_\_ (job description).

This release covers the final payment to the signer for all labor, services, equipment, or materials furnished to the property or to \_\_\_\_\_\_ (person with whom signer contracted).

Before any recipient of this document relies on this document, the recipient should verify evidence of payment to the signer.

The signer warrants that the signer has already paid or will use the funds received from this final payment to promptly pay in full all of the signer's laborers, subcontractors, material men, and suppliers for all work, materials, equipment, or services provided for or to the above referenced project up to the date of this waiver and release.

Notary Public in and for the State of		
SWORN AND SUBSCRIBED before me at	,, this day of	. 20 A.D.
	(Printed/Typed name) (Title)	
Ву	(Signature)	
Date	(Company name)	

# **DOCUMENT AI – RELEASE OF LIEN FORMS**

#### FORM 4: UNCONDITIONAL WAIVER FOR FINAL PAYMENTS

# MULTI-CAMPUS BUILDING ENVELOPE REPAIRS (PACKAGE 4) GALVESTON INDEPENDENT SCHOOL DISTRICT

NOTICE: THIS DOCUMENT WAIVES RIGHTS UNCONDITIONALLY AND STATES THAT YOU HAVE BEEN PAID FOR GIVING UP THOSE RIGHTS. IT IS PROHIBITED FOR A PERSON TO REQUIRE YOU TO SIGN THIS DOCUMENT IF YOU HAVE NOT BEEN PAID THE PAYMENT AMOUNT SET FORTH BELOW. IF YOU HAVE NOT BEEN PAID, USE A CONDITIONAL RELEASE FORM.

# UNCONDITIONAL WAIVER AND RELEASE ON FINAL PAYMENT

Project \_\_\_\_\_\_ Job No. \_\_\_\_\_\_

The signer of this document has been paid in full for all labor, services, equipment, or materials furnished to the property or to \_\_\_\_\_\_\_ (person with whom signer contracted) on the property of \_\_\_\_\_\_\_ (owner) located at \_\_\_\_\_\_\_ (location) to the following extent: \_\_\_\_\_\_\_ (job description). The signer therefore waives and releases any mechanic's lien right, any right arising from a payment bond that complies with a state or federal statute, any common law payment bond right, any claim for payment, and any rights under any similar ordinance, rule, or statute related to claim or payment rights for persons in the signer's position.

The signer warrants that the signer has already paid or will use the funds received from this final payment to promptly pay in full all of the signer's laborers, subcontractors, material men, and suppliers for all work, materials, equipment, or services provided for or to the above referenced project in regard to the attached statement(s) or progress payment request(s).

Date	
	(Company name)
Ву	(Signature)
	(Printed/Typed name)
	(Title)
SWORN AND SUBSCRIBED before me at,	, this day of 20 A.D.

Notary Public in and for the State of \_\_\_\_\_.

#### DOCUMENT AJ - GALVESTON INDEPENDENT SCHOOL DISTRICT CONTRACTOR SITE RULES

- A. No foul language or spitting on floor.
- B. No tobacco products on school property. On new construction projects, tobacco products are prohibited after air conditioning systems are initially activated.
- C. The possession or use of alcohol or illegal drugs is strictly prohibited.
- D. No tank tops workers must be fully clothed.
- E. No workers with a history of felony convictions or warrants.
- F. No parking on grass, under shade trees, sidewalks or non-vehicular paved areas.
- G. Entry into any Galveston ISD facility must be cleared in advance with the campus office by signing in at time of arrival and signing out upon departure.
- H. Contractor's employees, Subcontractors and their agents and employees working on any District facility must wear picture identification with the company name. Any exceptions must be approved in advance with the designated District representative.
- I. Keep the premises free from accumulation of waste, materials or rubbish caused by the work under this contract at each site. Boxes must be broken down <u>prior</u> to removal from the building. Upon completion of the work, and prior to the final inspection, have the premises in a neat and clean condition.
- J. Take all precautions necessary for the safety of, and provide protection to prevent damage, injury or loss to:
  - All employees on the project and all other persons who may be affected thereby.
  - All the work and all materials to be incorporated therein, whether in storage on or off the site.
  - All property at the site and adjacent thereto including trees, shrubs, lawns, walks, pavements, roadways, structures, utilities and any other school property.
- K. A competent supervisor who understands the full scope of the work shall be on site at all times.
- L. School administrative services shall at all times have priority over the Contractor's use/service/etc.
- M. Any work that may interfere with school activities must be authorized in advance through administrative channels. A management plan will be devised to minimize the effect of the interference.
- N. The Contractor shall be responsible to Galveston ISD for acts and omissions of the Contractor's employees, Subcontractors and their agents and employees, and other persons performing portions of the work under the contract.
- O. No work within the confines of a secured building will be allowed without at least one District custodian present. The contractor must pay the Galveston ISD Custodial Department in advance for the cost of adding a custodian to a building for after-hours work.
- P. Doors must not be propped open when working after-hours.

- Q. Unless otherwise noted, any contractor working on District property must supply the Owner with proof of insurance naming the Districts co-insured on that policy for property and liability.
- R. Only the designated District representative who let the contract for services will be authorized to sign documents that require releases or acceptance of work by the District.

# END OF DOCUMENT AJ

# DOCUMENT AK - CERTIFICATE OF RESIDENCY

The State of TX has passed a law concerning non-resident contractors. This law can be found in Texas Education Code under Chapter 2252, Subchapter A. This law makes it necessary for the Galveston Independent School District to determine the residency of its bidders. In part, this law reads as follows:

"Section: 2252.001, (3)'Non-resident bidder' refers to a person is not a resident. (4)'Resident bidder' refers to a person whose principal place of business is in this state, including a contractor whose ultimate parent company or majority owner has its principal place of business in this state."

I certify that under the above my company is a

Resident Bidder		Non-resident Bidder
My or Our principal place of business is	in the city of	in the state of
<u>.</u>		
Authorized Signature (for above document)		
	Company	
	Printed Name	
	Signature	
STATE OF TEXAS ) ) COUNTY OF		)
Sworn to and subscribed before me at, 20	, Τε	exas, this the day of

Notary Public in and for \_\_\_\_\_ County, Texas

# **DOCUMENT AL - AFFIDAVIT OF NON-COLLUSION**

STATE OF TEXAS

COUNTY OF

#### AFFIDAVIT

By submission of this proposal, the undersigned certifies that:

)

)

- a. This proposal has been independently arrived at without collusion with any other offeror or with any other competitor;
- b. This proposal has not been knowingly disclosed and will not be knowingly disclosed, to any other offeror competitor or potential competitor, prior to the opening of proposals for this project;
- c. No attempt has been or will be made to induce any other person, partnership or corporation to submit or not submit a proposal;
- d. The undersigned certifies that he is fully informed regarding the accuracy of the statements contained in this certification, and that the penalties herein are applicable to the offeror as well as to any person signing in his behalf.

		Company				
		Printed Name				
		Signature				
STATE OF TEXAS	)					
COUNTY OF	)					
Sworn to and subscribed	before me at _ , 20	,	Texas, this	the	day	of
Notary Public in and for	 County, Texa	as				
	END O	F DOCUMENT AL				

# NOTE: THIS FORM MUST BE EXECUTED AND SUBMITTED WITH PROPOSAL

AFFIDAVIT OF NON-COLLUSION AL - 1

# DOCUMENT AM - NOTICE OF NO BID/PROPOSAL

#### GALVESTON INDEPENDENT SCHOOL DISTRICT

Use this form for No Bid/Proposal Notification. Return the completed form at time Proposal Opening date to the Purchasing Department, P.O. Box 660, Galveston, TX 77553 or Fax 409-762-8841.

If form does not apply, check below and submit with Proposal.

GISD RFCSP Number: 20-01

Vendors/Proposers who respond to this request with a formal response or this form will remain on our mailing list. Vendors/Proposers making no response at all are subject to removal from our Proposer listing.

Check:

- □ 1. Does not apply.
- □ 2. We have decided not to submit a proposal for the reason(s) listed below; however, we wish to remain on the Proposers list.
- □ 3. The product/service we represent should be listed in another category. (Specify and attach VENDOR/PROPOSER INFORMATION SHEET).
- □ 4. We wish to be removed from the Proposers list for the reason(s) listed below:

Explanation for Non-submission:

<u> </u>			
Company N	ame:	Phone Number:	
Address:		Fax Number:	
		Internet Number:	
Mail To:	"Notice of Non-Submission"		
	Purchasing Department GALVESTON INDEPENDENT SC P.O. BOX 660 GALVESTON, TEXAS 77553	HOOL DISTRICT	
Signature of	f Proposer's Representative:		
Title:		Date:	
	NOTE: THIS DOCUMENT MUST BE EX		

# DOCUMENT AN - CERTIFICATION OF CRIMINAL HISTORY RECORD INFORMATION REVIEW BY CONTRACTOR-EMPLOYER

# Certifying Affidavit Submitted to:

Name of School District:	Galveston ISD
Mailing Address:	<u>3904 Avenue T</u> Galveston, Texas 77550
Project:	Multi-Campus Building Envelope Repairs (Package 4)
STATE OF TEXAS	§
COUNTY OF	§

(I) The undersigned representative, on behalf of the contracting firm identified below, swears and affirms to Galveston Independent School District (the "District") that such firm has obtained, reviewed and verified, from a law enforcement or criminal justice agency or a private entity that is consumer reporting agency governed by the Fair Credit Reporting Act (15 U.S.C. §§ 1681 et seq.) the criminal history record information of all employees hired **before January 1**, 2008, who (a) have or will have continuing duties related to the contracted services, and (b) have or will have direct contact with students. Such employees are identified by name on Schedule **A** attached hereto. The undersigned further swears and affirms no employees who meet the requirements of (a) and (b) herein and/or identified on Schedule **A** have been convicted of any offense identified in Section 22.085 of the Texas Education Code.

(2) The undersigned representative, on behalf of the contracting firm identified below, swears and affirms to the District, that such firm has obtained, reviewed and verified, from the Texas Department of Public Safety criminal clearinghouse, the national criminal history record information of all employees hired **on or after January 1, 2008**, who (a) have or will have continuing duties related to the contracted services, and (b) have or will have direct contact with students. Such employees are identified by name on Schedule B attached hereto. The undersigned further swears and affirms no employees who meet the requirements of (a) and (b) herein and/or identified on Schedule B have been convicted of any offense identified in Section 22.085 of the Texas Education Code.

(3) The undersigned firm swears and covenants that no present or future employee will provide services to the Project that involve direct contact with students unless and until such employee's national criminal history record information has been reviewed and cleared as required by Paragraph (2) above, and an updated Certification has submitted by the contracting firm to the District with an updated Schedule B identifying such employees. In the event of an emergency, an employee who has not been previously certified may only provide services that involve direct contact with students if such employee is escorted by a District representative.

(4) The undersigned firm swears and covenants that, upon receipt of information, directly or indirectly, that any employee of the contracting firm has been convicted of an offense identified in Section 22.085 of the Texas Education Code, the contracting firm will immediately remove such employee from the Project and notify the District.

(5) Furthermore, if requested by the District, the name, driver's license number, and any other information required by the DPS will be submitted to the District for any person on either Schedule **A** or Schedule B.

, being duly sworn, affirms and ce	ertifies that he/she is the
------------------------------------	-----------------------------

\_\_\_\_\_ (position) of \_\_\_\_\_\_(contracting

firm), and that all statements and acknowledgements contained herein are true and correct, and that he/she has the authority to bind such firm to the covenants set out above.

SUBSCRIBED AND SWORN TO BEFORE ME this		day of _	 	,
Notary Public	State of		 	
My Commission Expires:				

**END OF SECTION AN** 

# **DOCUMENT AO - CONFLICT OF INTEREST QUESTIONNAIRE**

#### INSTRUCTIONS

According to Local Government Code, Chapter 176, a person or an agent of a person who contracts or seeks to contract for the sale or purchase of property, goods, or services with Dickinson Independent School District must file a completed Conflict of Interest Questionnaire with the District Legal Department not later than the seventh business day after the date that the person begins contract discussions or negotiations with the District or submits to the District an application, response to a request for proposals or bids, correspondence, or another writing related to a potential agreement with the District.

This Conflict of Interest Questionnaire must be filed annually by September 1 as long as the person or the agent of the person continues to contract or seek to contract for the sale or purchase of property, goods, or services with the District or not later than the 7th business day after the date the originally filed questionnaire becomes incomplete or inaccurate.

The completion of the Conflict of Interest Questionnaire is <u>not</u> needed if the person is an employee of a governmental entity and is acting in the employee's official capacity.

Explanation of the Conflict of Interest Questionnaire

- 1. Name of person doing business with the District.
- 2. Check the box if you are filing an update to a previously filed questionnaire.
- 3. Describe each affiliation or business relationship with an employee or contractor of the District who makes recommendations to a District officer with respect to expenditure of money. If no affiliation or business relationship exists, state "NONE."

Examples:

If your spouse, parent, or child is the District's Director of Purchasing and a bid is being submitted to the Purchasing Department, this relationship must be reported.

If your spouse, parent, or child is the Principal at a School and your business may sell items directly to that school, this relationship must be reported.

If you or your spouse, parent, or child is in business with a District employee that would be making a recommendation concerning a purchase or sales transaction involving you, the relationship must be reported.

If you employ or do business with a spouse, parent, or child of a District employee that would be making a recommendation concerning a purchase or sales transaction involving you, the relationship must be reported.

If you are a District employee and would be making a recommendation concerning a purchase or sales transaction involving you, the relationship must be reported.

If your spouse, parent, or child is a teacher that does not make recommendations concerning purchasing or sales transactions, this relationship should **<u>not</u>** be reported.

If your spouse, parent, or child is a Principal at a School and a bid is being considered by a separate department such as Facilities Planning (Construction Department), this relationship should **not** be reported.

4. Describe each affiliation or business relationship with a person who is a District officer and who appoints or employs a District officer that is the subject of this questionnaire. If no affiliation or business relationship exists, state "NONE."

Example:

If you or your spouse, parent, or child is related to, employs, or is in business with a District officer or their spouse, parent, or child, this relationship must be reported.

5. Name of District officer with whom you have an affiliation or business relationship.

For each person listed under question #4, complete page 2. If answers to A, B, and C are NO, indicate the name of the District officer, but do not complete section D.

6. Describe any other affiliation or business relationship that might cause a conflict of interest.

Example:

If your neighbor or friend is a District employee that would be making a recommendation concerning a purchase or sales transaction involving you and you feel that your relationship with this employee could affect their recommendation, this relationship must be reported.

If any other situation exists that would result in a conflict of interest, the relationship must be reported.

7. Sign and date this form.

Submit the completed form to the District. If any disclosures are indicated under questions #3 or #4, the form will be posted on the District's website.

# END OF SECTION AO

INSERT CONFLICT OF INTEREST QUESTIONNAIRE

# DOCUMENT BA - CONTRACT DOCUMENTS

#### I. CONSTRUCTION CONTRACT AGREEMENT

A. The contract for the construction of the project shall be executed by the successful Offeror on the 2007 Edition of AIA Document A101 "Standard Form of Agreement Between Owner and Contractor." Said contract, fully executed, shall be delivered to the Owner within ten (10) days of receipt of said contract.

# II. CONDITIONS OF THE CONTRACT

- A. The General Conditions of the Contract for Construction, AIA Document A201, 2007 Edition, is hereby specifically made a part of the Contract Documents, whether attached hereto or not; and as supplemented and amended herein, constitutes the General Conditions.
- B. Supplementary Conditions:
  - 1. The Supplementary Conditions contain modifications to the General Conditions of the Contract for Construction, AIA Document A201. Where any part of that document is modified by Supplementary Conditions, the unaltered provisions of the General Conditions shall remain in effect. Refer to Document CB for the Supplementary Conditions.

#### III. AVAILABILITY

- A. Printed copies of these documents may be examined in the Architect's office. AIA Documents may be obtained from the Houston Chapter of the American Institute of Architects, 315 Capitol, Suite 120, Houston, Texas 77002; (713) 520-0155. Copies may also be obtained from local architects' supplies stores.
- B. Failure to obtain and examine these documents in no way relieves the Contractor, Subcontractors, Sub-subcontractors, and material suppliers of responsibilities incorporated in the Agreement.

#### END OF DOCUMENT BA

Bond No.: \_\_\_\_\_

# DOCUMENT BB - TEXAS STATUTORY PERFORMANCE BOND

(Penalty of this bond must be 100% of contract amount)

(hereinafter called the Obligee) in the amount of \_\_\_\_\_

Dollars(\$\_\_\_\_\_) for the payment whereof, the said Principal and Surety bind themselves, and their heirs, administrators, executors, successors and assigns, jointly and severally, firmly by these presents.

WHEREAS, the Principal has entered into a certain written contract with the Obligee, dated the \_\_\_\_\_ day of \_\_\_\_\_, 20\_\_\_, for

#### MULTI-CAMPUS BUILDING ENVELOPE REPAIRS (PACKAGE 4) GALVESTON INDEPENDENT SCHOOL DISTRICT

which contract is hereby referred to and made a part hereof as fully and the same extent as if copied at length herein.

NOW, THEREFORE, THE CONDITION OF THIS OBLIGATION IS SUCH, that if the said Principal shall faithfully perform the work in accordance with the plans, specifications and contract documents, then this obligation shall be void; otherwise to remain in full force and effect.

PROVIDED, HOWEVER, that this bond is executed pursuant to the provisions of Chapter 2253 of the Texas Government Code and all liabilities on this bond shall be determined in accordance with the provisions of said Chapter to the same extent as if it were copied at length herein.

IN WITNESS WHEREOF, the said Principal and Surety have signed and sealed this Instrument this \_\_\_\_\_ Day of \_\_\_\_\_\_, 20\_\_\_\_.

		Principal	(Seal)
Surety Address	Ву:		
Questo Talanta na Number		Surety	(Seal)
Surety Telephone Number	Ву:	Attorney-in-Fact	
	END OF DOCUM	ENT BB	

#### DOCUMENT BC - TEXAS STATUTORY PAYMENT BOND

Bond No.: \_\_\_\_\_

(Penalty of this bond must be 100% of contract amount)

KNOW ALL MEN BY THESE PRESENTS, that: \_\_\_\_

(Hereinafter called the Principal), as principal, and \_\_\_\_\_

(Hereinafter called the Obligee) in the amount of \_\_\_\_\_

Dollars(\$\_\_\_\_\_) for the payment whereof, the said Principal and Surety bind themselves, and their heirs, administrators, executors, successors and assigns, jointly and severally, firmly by these presents.

WHEREAS, the Principal has entered into a certain written contract with the Obligee, dated the \_\_\_\_\_ Day of \_\_\_\_\_\_, 20\_\_\_, for

#### MULTI-CAMPUS BUILDING ENVELOPE REPAIRS (PACKAGE 4) GALVESTON INDEPENDENT SCHOOL DISTRICT

Which contract is hereby referred to and made a part hereof as fully and the same extent as if copied at length herein.

NOW, THEREFORE, THE CONDITION OF THIS OBLIGATION IS SUCH, that if the said Principal shall pay all claimants supplying labor and material to him or a Subcontractor in the prosecution of the work provided for in said contract, then this obligation shall be void; otherwise to remain in full force and effect.

PROVIDED, HOWEVER, that this bond is executed pursuant to the provisions of Chapter 2253 of the Texas Government Code and all liabilities on this bond to all such claimants shall be determined in accordance with the provisions of said Chapter to the same extent as if it were copied at length herein.

IN WITNESS WHEREOF, the said Principal and Surety have signed and sealed this Instrument this \_\_\_\_\_ Day of \_\_\_\_\_\_, 20\_\_\_\_.

		Principal (Seal)
Surety Address	Ву:	
		Surety (Seal)
Surety Telephone Number	By:At	torney-in-Fact
	END OF DOCUMENT BC	

#### DOCUMENT BD - INSURANCE AND BONDS REQUIREMENTS FOR CONTRACTORS AND FACILITY RENTERS - GALVESTON INDEPENDENT SCHOOL DISTRICT INSURANCE MANAGEMENT

#### 1.0 GENERAL

- A. The District shall require that the following insurance requirements be met on public works contracts:
  - 1. No Work will be commenced until all requirements of this Section have been approved by the District in writing.
  - 2. The District shall be furnished a Declaration of Insurance evidencing all policies and endorsements required by this Section prior to proceeding with any work.
  - 3. The insurance shall contain a provision that at least thirty days prior written notice shall be given to the District in the event of cancellation, material change, or non-renewal.
  - 4. Insurance shall be underwritten by a company rated not less than B+ VII in Best's latest published guide.
  - 5. There shall be a hold harmless agreement in which the Contractor assumes liability on the contract and holds the School District harmless.
  - 6. The Contractor shall purchase and maintain in force the following kinds of insurance and bonds for operations under construction contracts and as specified in each section.
  - 7. No deletions/exclusions from standard coverage form are allowed without the written consent of Galveston Independent School District.
  - 8. Furnish copies of subcontractors Certificates of Insurance to the Owner.
  - 9. Furnish copies of Worker Compensation Documents to the Owner.

#### 2.0 CASUALTY INSURANCE

A. Worker's Compensation Insurance Coverage

Definitions:

Certificate of coverage ("Certificate") - A copy of a certificate of insurance, a certificate of authority to self-insure issued by the commission, or a coverage agreement (TWCC-81, TWCC-82, TWCC-83, or TWCC-84), showing statutory workers' compensation insurance coverage for the person's or entity's employees providing services on a project, for the duration of the project.

Duration of the project - includes the time from the beginning of the work on the project until one (1) year after Substantial Completion of the project.

Persons providing services on the project ("subcontractor" in Texas Labor Code 406.096) - includes all persons or entities performing all or part of the services the contractor has undertaken to perform on the project, regardless of whether that person has employees. This includes, without limitation, independent contractors, subcontractors, leasing companies, motor carriers, owner-operators, employees of any such entity, or employees of any entity that furnishes persons to provide services on the project. "Services" shall include, without limitation, providing, hauling, or delivering equipment or materials, or providing labor, transportation, or other service related to a project. "Services" does not include activities unrelated to the project, such as food/beverage vendors, office supply deliveries, and delivery of portable toilets.

1. The contractor shall provide coverage, based on proper reporting of classification codes and payroll amounts and filing of any coverage agreements, which meets the statutory requirements of Texas Labor Code 401.011 (44) for all employees of the contractor providing services on the project for the duration of the project.

- 2. The contractor must provide a certificate of coverage to the governmental entity prior to being awarded the contract. The certificate shall show Galveston Independent School District as the certificate holder. The policy must be endorsed to provide a "waiver of subrogation in favor of Galveston Independent School District.
- 3. If the coverage period shown on the contractor's current certificate of coverage ends during the duration of the project, the contractor must, prior to the end of the coverage period, file a new certificate of coverage with the governmental entity showing the coverage has been extended.
- 4. The contractor shall obtain from each person providing services on a project, and provide to the governmental entity:
  - a. a certificate of coverage, prior to that person beginning work on the project, so the governmental entity will have on file certificates of coverage showing coverage for all persons providing services on a project; and
  - b. no later than seven (7) days after receipt by the contractor, a new certificate of coverage showing extension of coverage, if the coverage period shown on the current certificate of coverage ends during the duration of the project.
- 5. The contractor shall retain all required certificates of coverage for the duration of the project and two (2) years thereafter.
- 6. The contractor shall notify the governmental entity in writing by certified mail or personal delivery, within 10 days after the contractor knows or should know, of any change that materially affects the provision of coverage of any person providing services on the project.
- 7. The contractor shall post on each project site a notice, in the text, form and manner prescribed by the Texas Workers' Compensation Commission, informing all persons providing services on the project that they are required to be covered, and stating how a person may verify coverage and report lack of coverage.
- 8. The contractor shall contractually require each person with whom it contracts to provide services on a project, to:
  - a. Provide coverage, based on proper reporting of classification codes and payroll amounts and filing of any coverage agreements, which meets the statutory requirements of Texas Labor Code, Section 401.011 (44) for all of its employees providing services on the project, for the duration of the project. The policy must be endorsed to provide a "waiver of subrogation" in favor of Galveston Independent School District;
  - b. Provide to the contractor, prior to that person beginning work on the project, a certificate of coverage showing that coverage is being provided for all employees of the person providing services on the project, for the duration of the project. The certificate shall show Galveston Independent School District as the certificate holder;
  - c. provide the contractor, prior to the end of the coverage period, a new certificate of coverage showing extension of coverage, if the coverage period shown on the current certificate of coverage ends during the duration of the project;
  - d. obtain from each other person with whom it contracts, and provide to the contractor:
    - 1) A certificate of coverage, prior to the other person beginning work on the project; and
    - 2) A new certificate of coverage showing extension of coverage, prior to the end of the coverage period, if the coverage period shown on the current certificate of coverage ends during the duration of the project.

- e. retain all required certificates of coverage on file for the duration of the project and for one year thereafter;
- f. notify the governmental entity in writing by certified mail or personal delivery, within 10 days after the person knew or should have known, of any change that materially affects the provisions of coverage of any person providing services on the project; and
- g. Contractually require each person with whom it contracts to perform as required by paragraphs a g, with the certificates of coverage to be provided to the person for whom they are providing services.
- 9. By signing this contract or providing or causing to be provided a certificate of coverage, the contractor is representing to the governmental entity that all employees of the contractor who will provide services on the project will be covered by workers' compensation coverage for the duration of the project, that the coverage will be based on proper reporting of classification codes and payroll amounts, and that all coverage agreements will be filed with the appropriate insurance carrier or, in the case of self-insured, with the commission's Division of Self-Insurance Regulation. Providing false or misleading information may subject the contractor to administrative penalties, criminal penalties, civil penalties, or other civil actions.
- 10. The contractor's failure to comply with any of these provisions is a breach of contract by the contractor which entitles the governmental entity to declare the contract void if the contractor does not remedy the breach within ten days after receipt of notice of breach from the governmental entity.
- 11. The Contractor shall post the following language:

#### **REQUIRED WORKERS' COMPENSATION COVERAGE**

"The law requires that each person working on this site or providing services related to this construction project must be covered by workers' compensation insurance. This includes persons providing, hauling, or delivering equipment or materials, or providing labor or transportation or other service related to the project, regardless of the identity of their employer or status as an employee." Furnish copies of Workers' Compensation coverage for each person working on the project.

"Call the Texas Workers' Compensation Commission at (512) 440 - 3789 to receive information on the legal requirements for coverage, to verify whether your employer has provided the required coverage, or to report an employer's failure to provide coverage."

B. Commercial General Liability Insurance (Occurrence basis only):

Bodily Injury and Property Damage	Each Occurrence General Aggregate	\$300,000 \$600,000
Products/Completed Operations	Aggregate	\$600,000
Personal and Advertising Injury	Occurrence	\$300,000
Fire Damage, Legal Liability	Any one fire	\$50,000
Medical Expenses	Any one person	\$5,000

C. The Owner shall be named as an additional insured by endorsement on the Contractor's policy as to the subject job.

#### 2.1 AUTOMOBILE LIABILITY INSURANCE

- A. Business (Commercial) Automobile Liability Insurance
  - 1. Coverage for all owned, non-owned and hired vehicles:

Bodily Insurance	Each Person	\$100,000
	Each Accident	\$300,000
Property Damage	Each Occurrence	\$100,000

#### 2.2 UMBRELLA LIABILITY INSURANCE (EXCESS)

\$1,000,000

- A. The Owner shall be named as an additional insured on the Contractor's policy as to the subject job.
- B. This policy shall provide coverage over the Workmen's Compensation, Commercial General Liability and Business Automobile Liability policies.

#### 2.3 OWNERS AND CONTRACTORS PROTECTIVE LIABILITY INSURANCE

A. Naming the Owner as insured with the following primary limits, irrespective of whether occurrence consists of bodily injury, death, property damage, or combination thereof:

Each Occurrence \$1,000,000.00 Aggregate \$1,000,000.00

- B. Covering the work to be performed for Owner by the Contractor and its subcontractor, if any. It is required that the Contractor and the subcontractor, if any, be designated in the Declaration of the policy.
- C. The definition of insured in the policy shall be endorsed to include officers and employees of the Galveston Independent School District, with respect to the work performed by the Contractor.
- D. The policy shall be written with same company as CGL policy.
- E. The Architects and Consultants shall be additional insured but only will have excess coverage. The full policy limits will protect the School District if needed and only the excess will protect the Architects and Consultants.
- F. The original policy to be placed on file with the Galveston Independent School District.
- G. Coverage through one (1) year after Substantial Completion of the project.

#### 2.4 PROPERTY INSURANCE (BUILDER'S RISK/INSTALLATION FLOATER)

- A. The policy shall be written in the name of the Owner, Contractor, and subcontractors as their interest may appear.
- B. The policy shall be written on an all risk basis for physical loss or damage and include theft, vandalism, malicious mischief.
- C. The amount of coverage shall be for the full insurable value of work.
- D. The deductible shall not be over \$1,000.00 without the approval of the Owner. (Deductible losses shall be paid by the Contractor.)
- E. The policy shall include an endorsement allowing Owner occupancy, and the insurance shall not be canceled or altered on account of partial occupancy prior to completion.

- F. A subrogation clause shall waive subrogation as to the Contractor, subcontractor, subsubcontractors, the Owner and his employees and representatives.
- G. The original builders risk policy shall be furnished to the Owner prior to start of the job and maintained through Substantial Completion

### 3.0 BONDS

- A. Bonds are required for public works contracts under the following circumstances:
  - 1. Performance Bond and Labor and Material Payment Bond, each in a personal sum equal to 100% of contract sum if the formal contract is in excess of \$25.000.00.
  - 2. A Proposal Bond or Proposal Security in the amount of 10% of any proposal of \$25,000.00 or more must be submitted with formal proposals on public works contracts or as otherwise specified in each contract.
  - 3. Copies of the bonds shall be filed with the county clerk and the owner shall receive a file receipt.
  - 4. Performance and Payment Bonds shall remain in force for one (1) year after completion of the contract.
  - 5. The Work will not be started until the bonds and issuing companies have been accepted as satisfactory by the Owner.
  - 6. The original bonds will be delivered to the Owner with an attached authorized power of attorney.

#### END OF DOCUMENT BD

# **AIA** Document A201° – 2007

# General Conditions of the Contract for Construction

#### for the following PROJECT:

(Name and location or address) Multi-Campus Building Envelope Repairs (Phase 4) 3904 Avenue T. Galveston, TX 77550.

#### THE OWNER:

(Name, legal status and address) Galveston Independent School District 3904 Avenue T. Galveston, TX 77550.

#### THE ARCHITECT:

(Name, legal status and address) **BEAM Professionals** 11 Greenway Plaza, 22<sup>nd</sup> Floor Houston, TX 77046

#### TABLE OF ARTICLES

- **GENERAL PROVISIONS** 1
- 2 OWNER
- 3 CONTRACTOR
- 4 ARCHITECT
- 5 SUBCONTRACTORS
- CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS 6
- 7 **CHANGES IN THE WORK**
- 8 TIME
- 9 **PAYMENTS AND COMPLETION**
- 10 PROTECTION OF PERSONS AND PROPERTY
- 11 **INSURANCE AND BONDS**
- UNCOVERING AND CORRECTION OF WORK 12
- 13 MISCELLANEOUS PROVISIONS
- 14 TERMINATION OR SUSPENSION OF THE CONTRACT
- 15 **CLAIMS AND DISPUTES**

#### ADDITIONS AND DELETIONS:

The author of this document has added information needed for its completion. The author may also have revised the text of the original AIA standard form. An Additions and Deletions Report that notes added information as well as revisions to the standard form text is available from the author and should be reviewed. A vertical line in the left margin of this document indicates where the author has added necessary information and where the author has added to or deleted from the original AIA text.

This document has important legal consequences. Consultation with an attorney is encouraged with respect to its completion or modification.

1

1

AIA Document A201<sup>®</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

INDEX

(Topics and numbers in **bold** are section headings.)

Acceptance of Nonconforming Work 9.6.6, 9.9.3, 12.3 Acceptance of Work 9.6.6, 9.8.2, 9.9.3, 9.10.1, 9.10.3, 12.3 Access to Work 3.16, 6.2.1, 12.1 Accident Prevention 10 Acts and Omissions 3.2, 3.3.2, 3.12.8, 3.18, 4.2.3, 8.3.1, 9.5.1, 10.2.5, 10.2.8, 13.4.2, 13.7, 14.1, 15.2 Addenda 1.1.1.3.11 Additional Costs, Claims for 3.7.4, 3.7.5, 6.1.1, 7.3.7.5, 10.3, 15.1.4 **Additional Inspections and Testing** 9.4.2, 9.8.3, 12.2.1, 13.5 Additional Insured 11.1.4 Additional Time, Claims for 3.2.4, 3.7.4, 3.7.5, 3.10.2, 8.3.2, 15.1.5 Administration of the Contract 3.1.3, 4.2, 9.4, 9.5 Advertisement or Invitation to Bid 1.1.1 Aesthetic Effect 4.2.13 Allowances 3.8. 7.3.8 All-risk Insurance 11.3.1, 11.3.1.1 **Applications for Payment** 4.2.5, 7.3.9, 9.2, 9.3, 9.4, 9.5.1, 9.6.3, 9.7, 9.10, 11.1.3 Approvals 2.1.1, 2.2.2, 2.4, 3.1.3, 3.10.2, 3.12.8, 3.12.9, 3.12.10, 4.2.7, 9.3.2, 13.5.1 Arbitration 8.3.1, 11.3.10, 13.1, 15.3.2, 15.4 ARCHITECT 4 Architect, Definition of 4.1.1 Architect, Extent of Authority 2.4, 3.12.7, 4.1, 4.2, 5.2, 6.3, 7.1.2, 7.3.7, 7.4, 9.2, 9.3.1, 9.4, 9.5, 9.6.3, 9.8, 9.10.1, 9.10.3, 12.1, 12.2.1, 13.5.1, 13.5.2, 14.2.2, 14.2.4, 15.1.3, 15.2.1 Architect, Limitations of Authority and Responsibility 2.1.1, 3.12.4, 3.12.8, 3.12.10, 4.1.2, 4.2.1, 4.2.2, 4.2.3, 4.2.6, 4.2.7, 4.2.10, 4.2.12, 4.2.13, 5.2.1, 7.4, 9.4.2, 9.5.3, 9.6.4, 15.1.3, 15.2 Architect's Additional Services and Expenses 2.4, 11.3.1.1, 12.2.1, 13.5.2, 13.5.3, 14.2.4

Architect's Administration of the Contract 3.1.3, 4.2, 3.7.4, 15.2, 9.4.1, 9.5 Architect's Approvals 2.4, 3.1.3, 3.5, 3.10.2, 4.2.7 Architect's Authority to Reject Work 3.5, 4.2.6, 12.1.2, 12.2.1 Architect's Copyright 1.1.7, 1.5 Architect's Decisions 3.7.4, 4.2.6, 4.2.7, 4.2.11, 4.2.12, 4.2.13, 4.2.14, 6.3, 7.3.7, 7.3.9, 8.1.3, 8.3.1, 9.2, 9.4.1, 9.5, 9.8.4, 9.9.1, 13.5.2, 15.2, 15.3 Architect's Inspections 3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.8.3, 9.9.2, 9.10.1, 13.5 Architect's Instructions 3.2.4, 3.3.1, 4.2.6, 4.2.7, 13.5.2 Architect's Interpretations 4.2.11, 4.2.12 Architect's Project Representative 4.2.10 Architect's Relationship with Contractor 1.1.2, 1.5, 3.1.3, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2, 3.5, 3.7.4, 3.7.5, 3.9.2, 3.9.3, 3.10, 3.11, 3.12, 3.16, 3.18, 4.1.2, 4.1.3, 4.2, 5.2, 6.2.2, 7, 8.3.1, 9.2, 9.3, 9.4, 9.5, 9.7, 9.8, 9.9, 10.2.6, 10.3, 11.3.7, 12, 13.4.2, 13.5, 15.2 Architect's Relationship with Subcontractors 1.1.2, 4.2.3, 4.2.4, 4.2.6, 9.6.3, 9.6.4, 11.3.7 Architect's Representations 9.4.2, 9.5.1, 9.10.1 Architect's Site Visits 3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.5.1, 9.9.2, 9.10.1, 13.5 Asbestos 10.3.1 Attorneys' Fees 3.18.1, 9.10.2, 10.3.3 Award of Separate Contracts 6.1.1, 6.1.2 Award of Subcontracts and Other Contracts for **Portions of the Work** 5.2 **Basic Definitions** 1.1 **Bidding Requirements** 1.1.1, 5.2.1, 11.4.1 **Binding Dispute Resolution** 9.7, 11.3.9, 11.3.10, 13.1, 15.2.5, 15.2.6.1, 15.3.1, 15.3.2, 15.4.1 **Boiler and Machinery Insurance** 11.3.2 Bonds, Lien 7.3.7.4, 9.10.2, 9.10.3 **Bonds, Performance, and Payment** 7.3.7.4, 9.6.7, 9.10.3, 11.3.9, 11.4 **Building Permit** 3.7.1

Init. 1

Capitalization 1.3 Certificate of Substantial Completion 9.8.3, 9.8.4, 9.8.5 **Certificates for Payment** 4.2.1, 4.2.5, 4.2.9, 9.3.3, 9.4, 9.5, 9.6.1, 9.6.6, 9.7, 9.10.1, 9.10.3, 14.1.1.3, 14.2.4, 15.1.3 Certificates of Inspection, Testing or Approval 13.5.4 Certificates of Insurance 9.10.2, 11.1.3 **Change Orders** 1.1.1, 2.4, 3.4.2, 3.7.4, 3.8.2.3, 3.11, 3.12.8, 4.2.8, 5.2.3, 7.1.2, 7.1.3, **7.2**, 7.3.2, 7.3.6, 7.3.9, 7.3.10, 8.3.1, 9.3.1.1, 9.10.3, 10.3.2, 11.3.1.2, 11.3.4, 11.3.9, 12.1.2, 15.1.3 Change Orders, Definition of 7.2.1 **CHANGES IN THE WORK** 2.2.1, 3.11, 4.2.8, 7, 7.2.1, 7.3.1, 7.4, 8.3.1, 9.3.1.1, 11.3.9 Claims, Definition of 15.1.1 **CLAIMS AND DISPUTES** 3.2.4, 6.1.1, 6.3, 7.3.9, 9.3.3, 9.10.4, 10.3.3, 15, 15.4 Claims and Timely Assertion of Claims 15.4.1 **Claims for Additional Cost** 3.2.4, 3.7.4, 6.1.1, 7.3.9, 10.3.2, 15.1.4 **Claims for Additional Time** 3.2.4, 3.7.4, 6.1.1, 8.3.2, 10.3.2, 15.1.5 Concealed or Unknown Conditions, Claims for 3.7.4 Claims for Damages 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Claims Subject to Arbitration 15.3.1, 15.4.1 **Cleaning Up** 3.15, 6.3 Commencement of the Work, Conditions Relating to 2.2.1, 3.2.2, 3.4.1, 3.7.1, 3.10.1, 3.12.6, 5.2.1, 5.2.3, 6.2.2, 8.1.2, 8.2.2, 8.3.1, 11.1, 11.3.1, 11.3.6, 11.4.1, 15.1.4 Commencement of the Work, Definition of 8.1.2 **Communications Facilitating Contract** Administration 3.9.1. 4.2.4 Completion, Conditions Relating to 3.4.1, 3.11, 3.15, 4.2.2, 4.2.9, 8.2, 9.4.2, 9.8, 9.9.1, 9.10, 12.2, 13.7, 14.1.2 **COMPLETION, PAYMENTS AND** Completion, Substantial 4.2.9, 8.1.1, 8.1.3, 8.2.3, 9.4.2, 9.8, 9.9.1, 9.10.3, 12.2, 13.7

Compliance with Laws 1.6, 3.2.3, 3.6, 3.7, 3.12.10, 3.13, 4.1.1, 9.6.4, 10.2.2, 11.1, 11.3, 13.1, 13.4, 13.5.1, 13.5.2, 13.6, 14.1.1, 14.2.1.3, 15.2.8, 15.4.2, 15.4.3 Concealed or Unknown Conditions 3.7.4, 4.2.8, 8.3.1, 10.3 Conditions of the Contract 1.1.1, 6.1.1, 6.1.4 Consent, Written 3.4.2, 3.7.4, 3.12.8, 3.14.2, 4.1.2, 9.3.2, 9.8.5, 9.9.1, 9.10.2, 9.10.3, 11.3.1, 13.2, 13.4.2, 15.4.4.2 **Consolidation or Joinder** 15.4.4 **CONSTRUCTION BY OWNER OR BY** SEPARATE CONTRACTORS 1.1.4.6 Construction Change Directive, Definition of 7.3.1 **Construction Change Directives** 1.1.1, 3.4.2, 3.12.8, 4.2.8, 7.1.1, 7.1.2, 7.1.3, 7.3, 9.3.1.1 Construction Schedules, Contractor's 3.10, 3.12.1, 3.12.2, 6.1.3, 15.1.5.2 **Contingent Assignment of Subcontracts** 5.4, 14.2.2.2 **Continuing Contract Performance** 15.1.3 Contract, Definition of 1.1.2 CONTRACT, TERMINATION OR SUSPENSION OF THE 5.4.1.1, 11.3.9, 14 Contract Administration 3.1.3, 4, 9.4, 9.5 Contract Award and Execution, Conditions Relating to 3.7.1, 3.10, 5.2, 6.1, 11.1.3, 11.3.6, 11.4.1 Contract Documents, Copies Furnished and Use of 1.5.2, 2.2.5, 5.3 Contract Documents, Definition of 1.1.1 **Contract Sum** 3.7.4, 3.8, 5.2.3, 7.2, 7.3, 7.4, **9.1**, 9.4.2, 9.5.1.4, 9.6.7, 9.7, 10.3.2, 11.3.1, 14.2.4, 14.3.2, 15.1.4, 15.2.5 Contract Sum, Definition of 9.1 Contract Time 3.7.4, 3.7.5, 3.10.2, 5.2.3, 7.2.1.3, 7.3.1, 7.3.5, 7.4, 8.1.1, 8.2.1, 8.3.1, 9.5.1, 9.7, 10.3.2, 12.1.1, 14.3.2, 15.1.5.1, 15.2.5 Contract Time, Definition of 8.1.1 CONTRACTOR 3 Contractor, Definition of 3.1, 6.1.2

AIA Document A201<sup>®</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

**Contractor's Construction Schedules** 3.10, 3.12.1, 3.12.2, 6.1.3, 15.1.5.2 Contractor's Employees 3.3.2, 3.4.3, 3.8.1, 3.9, 3.18.2, 4.2.3, 4.2.6, 10.2, 10.3, 11.1.1, 11.3.7, 14.1, 14.2.1.1 **Contractor's Liability Insurance** 11.1 Contractor's Relationship with Separate Contractors and Owner's Forces 3.12.5, 3.14.2, 4.2.4, 6, 11.3.7, 12.1.2, 12.2.4 Contractor's Relationship with Subcontractors 1.2.2, 3.3.2, 3.18.1, 3.18.2, 5, 9.6.2, 9.6.7, 9.10.2, 11.3.1.2, 11.3.7, 11.3.8 Contractor's Relationship with the Architect 1.1.2, 1.5, 3.1.3, 3.2.2, 3.2.3, 3.2.4, 3.3.1, 3.4.2, 3.5, 3.7.4, 3.10, 3.11, 3.12, 3.16, 3.18, 4.1.3, 4.2, 5.2, 6.2.2, 7, 8.3.1, 9.2, 9.3, 9.4, 9.5, 9.7, 9.8, 9.9, 10.2.6, 10.3, 11.3.7, 12, 13.5, 15.1.2, 15.2.1 Contractor's Representations 3.2.1, 3.2.2, 3.5, 3.12.6, 6.2.2, 8.2.1, 9.3.3, 9.8.2 Contractor's Responsibility for Those Performing the Work 3.3.2, 3.18, 5.3, 6.1.3, 6.2, 9.5.1, 10.2.8 Contractor's Review of Contract Documents 3.2 Contractor's Right to Stop the Work 9.7 Contractor's Right to Terminate the Contract 14.1, 15.1.6 Contractor's Submittals 3.10, 3.11, 3.12.4, 4.2.7, 5.2.1, 5.2.3, 9.2, 9.3, 9.8.2, 9.8.3, 9.9.1, 9.10.2, 9.10.3, 11.1.3, 11.4.2 Contractor's Superintendent 3.9, 10.2.6 Contractor's Supervision and Construction Procedures 1.2.2, 3.3, 3.4, 3.12.10, 4.2.2, 4.2.7, 6.1.3, 6.2.4, 7.1.3, 7.3.5, 7.3.7, 8.2, 10, 12, 14, 15.1.3 Contractual Liability Insurance 11.1.1.8, 11.2 Coordination and Correlation 1.2, 3.2.1, 3.3.1, 3.10, 3.12.6, 6.1.3, 6.2.1 Copies Furnished of Drawings and Specifications 1.5, 2.2.5, 3.11 Copyrights 1.5, 3.17 Correction of Work 2.3, 2.4, 3.7.3, 9.4.2, 9.8.2, 9.8.3, 9.9.1, 12.1.2, 12.2 **Correlation and Intent of the Contract Documents** 1.2 Cost, Definition of 7.3.7 Costs 2.4, 3.2.4, 3.7.3, 3.8.2, 3.15.2, 5.4.2, 6.1.1, 6.2.3, 7.3.3.3, 7.3.7, 7.3.8, 7.3.9, 9.10.2, 10.3.2, 10.3.6, 11.3, 12.1.2, 12.2.1, 12.2.4, 13.5, 14

**Cutting and Patching** 3.14, 6.2.5 Damage to Construction of Owner or Separate Contractors 3.14.2, 6.2.4, 10.2.1.2, 10.2.5, 10.4, 11.1.1, 11.3, 12.2.4 Damage to the Work 3.14.2, 9.9.1, 10.2.1.2, 10.2.5, 10.4, 11.3.1, 12.2.4 Damages, Claims for 3.2.4, 3.18, 6.1.1, 8.3.3, 9.5.1, 9.6.7, 10.3.3, 11.1.1, 11.3.5, 11.3.7, 14.1.3, 14.2.4, 15.1.6 Damages for Delay 6.1.1, 8.3.3, 9.5.1.6, 9.7, 10.3.2 Date of Commencement of the Work, Definition of 8.1.2 Date of Substantial Completion, Definition of 8.1.3 Day. Definition of 8.1.4 Decisions of the Architect 3.7.4, 4.2.6, 4.2.7, 4.2.11, 4.2.12, 4.2.13, 15.2, 6.3, 7.3.7, 7.3.9, 8.1.3, 8.3.1, 9.2, 9.4, 9.5.1, 9.8.4, 9.9.1, 13.5.2, 14.2.2, 14.2.4, 15.1, 15.2 **Decisions to Withhold Certification** 9.4.1, 9.5, 9.7, 14.1.1.3 Defective or Nonconforming Work, Acceptance, Rejection and Correction of 2.3, 2.4, 3.5, 4.2.6, 6.2.5, 9.5.1, 9.5.2, 9.6.6, 9.8.2, 9.9.3, 9.10.4, 12.2.1 Definitions 1.1, 2.1.1, 3.1.1, 3.5, 3.12.1, 3.12.2, 3.12.3, 4.1.1, 15.1.1, 5.1, 6.1.2, 7.2.1, 7.3.1, 8.1, 9.1, 9.8.1 **Delays and Extensions of Time** 3.2, 3.7.4, 5.2.3, 7.2.1, 7.3.1, 7.4, **8.3**, 9.5.1, 9.7, 10.3.2, 10.4, 14.3.2, 15.1.5, 15.2.5 Disputes 6.3, 7.3.9, 15.1, 15.2 **Documents and Samples at the Site** 3.11 Drawings, Definition of 1.1.5 Drawings and Specifications, Use and Ownership of 3.11 Effective Date of Insurance 8.2.2, 11.1.2 Emergencies 10.4, 14.1.1.2, 15.1.4 Employees, Contractor's 3.3.2, 3.4.3, 3.8.1, 3.9, 3.18.2, 4.2.3, 4.2.6, 10.2, 10.3.3, 11.1.1, 11.3.7, 14.1, 14.2.1.1 Equipment, Labor, Materials or 1.1.3, 1.1.6, 3.4, 3.5, 3.8.2, 3.8.3, 3.12, 3.13, 3.15.1, 4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.7, 9.3.2, 9.3.3, 9.5.1.3, 9.10.2, 10.2.1, 10.2.4, 14.2.1.1, 14.2.1.2

AIA Document A201<sup>®</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

Execution and Progress of the Work 1.1.3, 1.2.1, 1.2.2, 2.2.3, 2.2.5, 3.1, 3.3.1, 3.4.1, 3.5, 3.7.1, 3.10.1, 3.12, 3.14, 4.2, 6.2.2, 7.1.3, 7.3.5, 8.2, 9.5.1, 9.9.1, 10.2, 10.3, 12.2, 14.2, 14.3.1, 15.1.3 Extensions of Time 3.2.4, 3.7.4, 5.2.3, 7.2.1, 7.3, 7.4, 9.5.1, 9.7, 10.3.2, 10.4, 14.3, 15.1.5, 15.2.5 **Failure of Payment** 9.5.1.3, 9.7, 9.10.2, 13.6, 14.1.1.3, 14.2.1.2 Faulty Work (See Defective or Nonconforming Work) **Final Completion and Final Payment** 4.2.1, 4.2.9, 9.8.2, 9.10, 11.1.2, 11.1.3, 11.3.1, 11.3.5, 12.3, 14.2.4, 14.4.3 Financial Arrangements, Owner's 2.2.1, 13.2.2, 14.1.1.4 Fire and Extended Coverage Insurance 11.3.1.1 **GENERAL PROVISIONS Governing Law** 13.1 Guarantees (See Warranty) **Hazardous Materials** 10.2.4, 10.3 Identification of Subcontractors and Suppliers 5.2.1 Indemnification 3.17, 3.18, 9.10.2, 10.3.3, 10.3.5, 10.3.6, 11.3.1.2, 11.3.7 Information and Services Required of the Owner 2.1.2, 2.2, 3.2.2, 3.12.4, 3.12.10, 6.1.3, 6.1.4, 6.2.5, 9.6.1, 9.6.4, 9.9.2, 9.10.3, 10.3.3, 11.2, 11.4, 13.5.1, 13.5.2, 14.1.1.4, 14.1.4, 15.1.3 **Initial Decision** 15.2 Initial Decision Maker, Definition of 1.1.8 Initial Decision Maker, Decisions 14.2.2, 14.2.4, 15.2.1, 15.2.2, 15.2.3, 15.2.4, 15.2.5 Initial Decision Maker, Extent of Authority 14.2.2, 14.2.4, 15.1.3, 15.2.1, 15.2.2, 15.2.3, 15.2.4, 15.2.5 **Injury or Damage to Person or Property** 10.2.8, 10.4 Inspections 3.1.3, 3.3.3, 3.7.1, 4.2.2, 4.2.6, 4.2.9, 9.4.2, 9.8.3, 9.9.2, 9.10.1, 12.2.1, 13.5 Instructions to Bidders 1.1.1 Instructions to the Contractor 3.2.4, 3.3.1, 3.8.1, 5.2.1, 7, 8.2.2, 12, 13.5.2 Instruments of Service, Definition of 1.1.7 Insurance 3.18.1, 6.1.1, 7.3.7, 9.3.2, 9.8.4, 9.9.1, 9.10.2, 11

**Insurance, Boiler and Machinery** 11.3.2 **Insurance, Contractor's Liability** 11.1 Insurance, Effective Date of 8.2.2. 11.1.2 Insurance, Loss of Use 11.3.3 **Insurance, Owner's Liability** 11.2 **Insurance, Property** 10.2.5, 11.3 Insurance, Stored Materials 9.3.2 **INSURANCE AND BONDS** 11 Insurance Companies, Consent to Partial Occupancy 9.9.1 Intent of the Contract Documents 1.2.1, 4.2.7, 4.2.12, 4.2.13, 7.4 Interest 13.6 Interpretation 1.2.3, 1.4, 4.1.1, 5.1, 6.1.2, 15.1.1 Interpretations, Written 4.2.11, 4.2.12, 15.1.4 Judgment on Final Award 15.4.2 Labor and Materials, Equipment 1.1.3, 1.1.6, **3.4**, 3.5, 3.8.2, 3.8.3, 3.12, 3.13, 3.15.1, 4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.7, 9.3.2, 9.3.3, 9.5.1.3, 9.10.2, 10.2.1, 10.2.4, 14.2.1.1, 14.2.1.2 Labor Disputes 8.3.1 Laws and Regulations 1.5, 3.2.3, 3.6, 3.7, 3.12.10, 3.13, 4.1.1, 9.6.4, 9.9.1, 10.2.2, 11.1.1, 11.3, 13.1, 13.4, 13.5.1, 13.5.2, 13.6, 14, 15.2.8, 15.4 Liens 2.1.2, 9.3.3, 9.10.2, 9.10.4, 15.2.8 Limitations, Statutes of 12.2.5, 13.7, 15.4.1.1 Limitations of Liability 2.3, 3.2.2, 3.5, 3.12.10, 3.17, 3.18.1, 4.2.6, 4.2.7, 4.2.12, 6.2.2, 9.4.2, 9.6.4, 9.6.7, 10.2.5, 10.3.3, 11.1.2, 11.2, 11.3.7, 12.2.5, 13.4.2 Limitations of Time 2.1.2, 2.2, 2.4, 3.2.2, 3.10, 3.11, 3.12.5, 3.15.1, 4.2.7, 5.2, 5.3, 5.4.1, 6.2.4, 7.3, 7.4, 8.2, 9.2, 9.3.1, 9.3.3, 9.4.1, 9.5, 9.6, 9.7, 9.8, 9.9, 9.10, 11.1.3, 11.3.1.5, 11.3.6, 11.3.10, 12.2, 13.5, 13.7, 14, 15 Loss of Use Insurance 11.3.3 Material Suppliers 1.5, 3.12.1, 4.2.4, 4.2.6, 5.2.1, 9.3, 9.4.2, 9.6, 9.10.5 Materials, Hazardous 10.2.4, 10.3

Init.

1

AIA Document A201<sup>®</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

Materials, Labor, Equipment and 1.1.3, 1.1.6, 1.5.1, 3.4.1, 3.5, 3.8.2, 3.8.3, 3.12, 3.13, 3.15.1, 4.2.6, 4.2.7, 5.2.1, 6.2.1, 7.3.7, 9.3.2, 9.3.3, 9.5.1.3, 9.10.2, 10.2.1.2, 10.2.4, 14.2.1.1, 14.2.1.2 Means, Methods, Techniques, Sequences and Procedures of Construction 3.3.1, 3.12.10, 4.2.2, 4.2.7, 9.4.2 Mechanic's Lien 2.1.2, 15.2.8 Mediation 8.3.1, 10.3.5, 10.3.6, 15.2.1, 15.2.5, 15.2.6, **15.3**, 15.4.1 **Minor Changes in the Work** 1.1.1, 3.12.8, 4.2.8, 7.1, 7.4 **MISCELLANEOUS PROVISIONS** 13 Modifications, Definition of 1.1.1 Modifications to the Contract 1.1.1, 1.1.2, 3.11, 4.1.2, 4.2.1, 5.2.3, 7, 8.3.1, 9.7, 10.3.2, 11.3.1 **Mutual Responsibility** 6.2 Nonconforming Work, Acceptance of 9.6.6, 9.9.3, 12.3 Nonconforming Work, Rejection and Correction of 2.3, 2.4, 3.5, 4.2.6, 6.2.4, 9.5.1, 9.8.2, 9.9.3, 9.10.4, 12.2.1 Notice 2.2.1, 2.3, 2.4, 3.2.4, 3.3.1, 3.7.2, 3.12.9, 5.2.1, 9.7, 9.10, 10.2.2, 11.1.3, 12.2.2.1, 13.3, 13.5.1, 13.5.2, 14.1, 14.2, 15.2.8, 15.4.1 Notice, Written 2.3, 2.4, 3.3.1, 3.9.2, 3.12.9, 3.12.10, 5.2.1, 9.7, 9.10, 10.2.2, 10.3, 11.1.3, 11.3.6, 12.2.2.1, 13.3, 14, 15.2.8, 15.4.1 Notice of Claims 3.7.4, 10.2.8, 15.1.2, 15.4 Notice of Testing and Inspections 13.5.1, 13.5.2 Observations, Contractor's 3.2, 3.7.4 Occupancy 2.2.2, 9.6.6, 9.8, 11.3.1.5 Orders, Written 1.1.1, 2.3, 3.9.2, 7, 8.2.2, 11.3.9, 12.1, 12.2.2.1, 13.5.2, 14.3.1 **OWNER** 2 **Owner**, Definition of 2.1.1 **Owner, Information and Services Required of the** 2.1.2, 2.2, 3.2.2, 3.12.10, 6.1.3, 6.1.4, 6.2.5, 9.3.2, 9.6.1, 9.6.4, 9.9.2, 9.10.3, 10.3.3, 11.2, 11.3, 13.5.1, 13.5.2, 14.1.1.4, 14.1.4, 15.1.3

**Owner's** Authority 1.5, 2.1.1, 2.3, 2.4, 3.4.2, 3.8.1, 3.12.10, 3.14.2, 4.1.2, 4.1.3, 4.2.4, 4.2.9, 5.2.1, 5.2.4, 5.4.1, 6.1, 6.3, 7.2.1, 7.3.1, 8.2.2, 8.3.1, 9.3.1, 9.3.2, 9.5.1, 9.6.4, 9.9.1, 9.10.2, 10.3.2, 11.1.3, 11.3.3, 11.3.10, 12.2.2, 12.3, 13.2.2, 14.3, 14.4, 15.2.7 **Owner's Financial Capability** 2.2.1, 13.2.2, 14.1.1.4 **Owner's Liability Insurance** 11.2 Owner's Relationship with Subcontractors 1.1.2, 5.2, 5.3, 5.4, 9.6.4, 9.10.2, 14.2.2 **Owner's Right to Carry Out the Work** 2.4, 14.2.2 **Owner's Right to Clean Up** 6.3 **Owner's Right to Perform Construction and to Award Separate Contracts** 6.1 **Owner's Right to Stop the Work** 2.3 Owner's Right to Suspend the Work 14.3 Owner's Right to Terminate the Contract 14.2 **Ownership and Use of Drawings, Specifications** and Other Instruments of Service 1.1.1, 1.1.6, 1.1.7, **1.5**, 2.2.5, 3.2.2, 3.11, 3.17, 4.2.12, 5.3 Partial Occupancy or Use 9.6.6, 9.9, 11.3.1.5 Patching, Cutting and 3.14, 6.2.5 Patents 3.17 Payment, Applications for 4.2.5, 7.3.9, 9.2, 9.3, 9.4, 9.5, 9.6.3, 9.7, 9.8.5, 9.10.1, 14.2.3, 14.2.4, 14.4.3 **Payment, Certificates for** 4.2.5, 4.2.9, 9.3.3, 9.4, 9.5, 9.6.1, 9.6.6, 9.7, 9.10.1, 9.10.3, 13.7, 14.1.1.3, 14.2.4 Payment, Failure of 9.5.1.3, 9.7, 9.10.2, 13.6, 14.1.1.3, 14.2.1.2 Payment, Final 4.2.1, 4.2.9, 9.8.2, 9.10, 11.1.2, 11.1.3, 11.4.1, 12.3, 13.7, 14.2.4, 14.4.3 Payment Bond, Performance Bond and 7.3.7.4, 9.6.7, 9.10.3, 11.4 **Payments**, **Progress** 9.3, 9.6, 9.8.5, 9.10.3, 13.6, 14.2.3, 15.1.3 PAYMENTS AND COMPLETION Payments to Subcontractors 5.4.2, 9.5.1.3, 9.6.2, 9.6.3, 9.6.4, 9.6.7, 14.2.1.2 PCB 10.3.1

Init. 1

AIA Document A201<sup>®</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

**Performance Bond and Payment Bond** 7.3.7.4, 9.6.7, 9.10.3, 11.4 Permits, Fees, Notices and Compliance with Laws 2.2.2, 3.7, 3.13, 7.3.7.4, 10.2.2 PERSONS AND PROPERTY, PROTECTION OF 10 Polychlorinated Biphenyl 10.3.1 Product Data, Definition of 3.12.2 **Product Data and Samples, Shop Drawings** 3.11, 3.12, 4.2.7 **Progress and Completion** 4.2.2, 8.2, 9.8, 9.9.1, 14.1.4, 15.1.3 **Progress Payments** 9.3, 9.6, 9.8.5, 9.10.3, 13.6, 14.2.3, 15.1.3 **Project**, Definition of 1.1.4 **Project Representatives** 4.2.10 **Property Insurance** 10.2.5, 11.3 **PROTECTION OF PERSONS AND PROPERTY** 10 **Regulations and Laws** 1.5, 3.2.3, 3.6, 3.7, 3.12.10, 3.13, 4.1.1, 9.6.4, 9.9.1, 10.2.2, 11.1, 11.4, 13.1, 13.4, 13.5.1, 13.5.2, 13.6, 14, 15.2.8, 15.4 Rejection of Work 3.5, 4.2.6, 12.2.1 Releases and Waivers of Liens 9.10.2 Representations 3.2.1, 3.5, 3.12.6, 6.2.2, 8.2.1, 9.3.3, 9.4.2, 9.5.1, 9.8.2, 9.10.1 Representatives 2.1.1, 3.1.1, 3.9, 4.1.1, 4.2.1, 4.2.2, 4.2.10, 5.1.1, 5.1.2, 13.2.1 Responsibility for Those Performing the Work 3.3.2, 3.18, 4.2.3, 5.3, 6.1.3, 6.2, 6.3, 9.5.1, 10 Retainage 9.3.1, 9.6.2, 9.8.5, 9.9.1, 9.10.2, 9.10.3 **Review of Contract Documents and Field Conditions by Contractor** 3.2, 3.12.7, 6.1.3 Review of Contractor's Submittals by Owner and Architect 3.10.1, 3.10.2, 3.11, 3.12, 4.2, 5.2, 6.1.3, 9.2, 9.8.2 Review of Shop Drawings, Product Data and Samples by Contractor 3.12 **Rights and Remedies** 1.1.2, 2.3, 2.4, 3.5, 3.7.4, 3.15.2, 4.2.6, 5.3, 5.4, 6.1, 6.3, 7.3.1, 8.3, 9.5.1, 9.7, 10.2.5, 10.3, 12.2.2, 12.2.4, 13.4, 14, 15.4 **Royalties, Patents and Copyrights** 

3.17

Init.

1

Rules and Notices for Arbitration 15.4.1 Safety of Persons and Property 10.2, 10.4 **Safety Precautions and Programs** 3.3.1, 4.2.2, 4.2.7, 5.3, 10.1, 10.2, 10.4 Samples, Definition of 3.12.3 Samples, Shop Drawings, Product Data and 3.11, 3.12, 4.2.7 Samples at the Site, Documents and 3.11 **Schedule of Values** 9.2, 9.3.1 Schedules, Construction 3.10, 3.12.1, 3.12.2, 6.1.3, 15.1.5.2 Separate Contracts and Contractors 1.1.4, 3.12.5, 3.14.2, 4.2.4, 4.2.7, 6, 8.3.1, 12.1.2 Shop Drawings, Definition of 3.12.1 Shop Drawings, Product Data and Samples 3.11, 3.12, 4.2.7 Site, Use of 3.13, 6.1.1, 6.2.1 Site Inspections 3.2.2, 3.3.3, 3.7.1, 3.7.4, 4.2, 9.4.2, 9.10.1, 13.5 Site Visits, Architect's 3.7.4, 4.2.2, 4.2.9, 9.4.2, 9.5.1, 9.9.2, 9.10.1, 13.5 Special Inspections and Testing 4.2.6, 12.2.1, 13.5 Specifications, Definition of 1.1.6 **Specifications** 1.1.1, 1.1.6, 1.2.2, 1.5, 3.11, 3.12.10, 3.17, 4.2.14 Statute of Limitations 13.7, 15.4.1.1 Stopping the Work 2.3, 9.7, 10.3, 14.1 Stored Materials 6.2.1, 9.3.2, 10.2.1.2, 10.2.4 Subcontractor, Definition of 5.1.1 **SUBCONTRACTORS** 5 Subcontractors, Work by 1.2.2, 3.3.2, 3.12.1, 4.2.3, 5.2.3, 5.3, 5.4, 9.3.1.2, 9.6.7 **Subcontractual Relations** 5.3, 5.4, 9.3.1.2, 9.6, 9.10, 10.2.1, 14.1, 14.2.1 Submittals 3.10, 3.11, 3.12, 4.2.7, 5.2.1, 5.2.3, 7.3.7, 9.2, 9.3, 9.8, 9.9.1, 9.10.2, 9.10.3, 11.1.3 Submittal Schedule 3.10.2, 3.12.5, 4.2.7 Subrogation, Waivers of 6.1.1, 11.3.7

AIA Document A201<sup>®</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

**Substantial Completion** 4.2.9, 8.1.1, 8.1.3, 8.2.3, 9.4.2, **9.8**, 9.9.1, 9.10.3, 12.2, 13.7 Substantial Completion, Definition of 9.8.1 Substitution of Subcontractors 5.2.3. 5.2.4 Substitution of Architect 4.1.3 Substitutions of Materials 3.4.2, 3.5, 7.3.8 Sub-subcontractor, Definition of 5.1.2 Subsurface Conditions 3.7.4 **Successors and Assigns** 13.2 Superintendent **3.9**, 10.2.6 **Supervision and Construction Procedures** 1.2.2, 3.3, 3.4, 3.12.10, 4.2.2, 4.2.7, 6.1.3, 6.2.4, 7.1.3, 7.3.7, 8.2, 8.3.1, 9.4.2, 10, 12, 14, 15.1.3 Surety 5.4.1.2, 9.8.5, 9.10.2, 9.10.3, 14.2.2, 15.2.7 Surety, Consent of 9.10.2, 9.10.3 Surveys 2.2.3 Suspension by the Owner for Convenience 14.3 Suspension of the Work 5.4.2, 14.3 Suspension or Termination of the Contract 5.4.1.1, 14 Taxes 3.6, 3.8.2.1, 7.3.7.4 **Termination by the Contractor** 14.1, 15.1.6 **Termination by the Owner for Cause** 5.4.1.1, 14.2, 15.1.6 Termination by the Owner for Convenience 14.4 Termination of the Architect 4.1.3 Termination of the Contractor 14.2.2 **TERMINATION OR SUSPENSION OF THE** CONTRACT 14 **Tests and Inspections** 3.1.3, 3.3.3, 4.2.2, 4.2.6, 4.2.9, 9.4.2, 9.8.3, 9.9.2, 9.10.1, 10.3.2, 11.4.1, 12.2.1, 13.5 TIME 8 Time, Delays and Extensions of 3.2.4, 3.7.4, 5.2.3, 7.2.1, 7.3.1, 7.4, 8.3, 9.5.1, 9.7, 10.3.2, 10.4, 14.3.2, 15.1.5, 15.2.5

**Time Limits** 2.1.2, 2.2, 2.4, 3.2.2, 3.10, 3.11, 3.12.5, 3.15.1, 4.2, 5.2, 5.3, 5.4, 6.2.4, 7.3, 7.4, 8.2, 9.2, 9.3.1, 9.3.3, 9.4.1, 9.5, 9.6, 9.7, 9.8, 9.9, 9.10, 11.1.3, 12.2, 13.5, 13.7, 14, 15.1.2, 15.4 **Time Limits on Claims** 3.7.4, 10.2.8, 13.7, 15.1.2 Title to Work 9.3.2, 9.3.3 **Transmission of Data in Digital Form** 1.6 **UNCOVERING AND CORRECTION OF WORK** 12 **Uncovering of Work** 12.1 Unforeseen Conditions, Concealed or Unknown 3.7.4. 8.3.1. 10.3 Unit Prices 7.3.3.2, 7.3.4 Use of Documents 1.1.1, 1.5, 2.2.5, 3.12.6, 5.3 Use of Site 3.13, 6.1.1, 6.2.1 Values, Schedule of 9.2, 9.3.1 Waiver of Claims by the Architect 13.4.2 Waiver of Claims by the Contractor 9.10.5, 13.4.2, 15.1.6 Waiver of Claims by the Owner 9.9.3, 9.10.3, 9.10.4, 12.2.2.1, 13.4.2, 14.2.4, 15.1.6 Waiver of Consequential Damages 14.2.4, 15.1.6 Waiver of Liens 9.10.2, 9.10.4 Waivers of Subrogation 6.1.1, 11.3.7 Warranty 3.5, 4.2.9, 9.3.3, 9.8.4, 9.9.1, 9.10.4, 12.2.2, 13.7 Weather Delays 15.1.5.2 Work, Definition of 1.1.3 Written Consent 1.5.2, 3.4.2, 3.7.4, 3.12.8, 3.14.2, 4.1.2, 9.3.2, 9.8.5, 9.9.1, 9.10.2, 9.10.3, 11.4.1, 13.2, 13.4.2, 15.4.4.2 Written Interpretations 4.2.11, 4.2.12 Written Notice 2.3, 2.4, 3.3.1, 3.9, 3.12.9, 3.12.10, 5.2.1, 8.2.2, 9.7, 9.10, 10.2.2, 10.3, 11.1.3, 12.2.2, 12.2.4, 13.3, 14, 15.4.1Written Orders 1.1.1, 2.3, 3.9, 7, 8.2.2, 12.1, 12.2, 13.5.2, 14.3.1, 15.1.2

AIA Document A201<sup>®</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

#### ARTICLE 1 **GENERAL PROVISIONS** § 1.1 BASIC DEFINITIONS

#### § 1.1.1 THE CONTRACT DOCUMENTS

The Contract Documents are enumerated in the Agreement between the Owner and Contractor (hereinafter the Agreement) and consist of the Agreement, Conditions of the Contract (General, Supplementary and other Conditions), Drawings, Specifications, Addenda issued prior to execution of the Contract, other documents listed in the Agreement and Modifications issued after execution of the Contract. A Modification is (1) a written amendment to the Contract signed by both parties, (2) a Change Order, (3) a Construction Change Directive or (4) a written order for a minor change in the Work issued by the Architect. Unless specifically enumerated in the Agreement, the Contract Documents do not include the advertisement or invitation to bid, Instructions to Bidders, sample forms, other information furnished by the Owner in anticipation of receiving bids or proposals, the Contractor's bid or proposal, or portions of Addenda relating to bidding requirements.

#### § 1.1.2 THE CONTRACT

The Contract Documents form the Contract for Construction. The Contract represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either written or oral. The Contract may be amended or modified only by a Modification. The Contract Documents shall not be construed to create a contractual relationship of any kind (1) between the Contractor and the Architect or the Architect's consultants, (2) between the Owner and a Subcontractor or a Sub-subcontractor, (3) between the Owner and the Architect or the Architect's consultants or (4) between any persons or entities other than the Owner and the Contractor. The Architect shall, however, be entitled to performance and enforcement of obligations under the Contract intended to facilitate performance of the Architect's duties.

#### § 1.1.3 THE WORK

The term "Work" means the construction and services required by the Contract Documents, whether completed or partially completed, and includes all other labor, materials, equipment and services provided or to be provided by the Contractor to fulfill the Contractor's obligations. The Work may constitute the whole or a part of the Project.

#### § 1.1.4 THE PROJECT

The Project is the total construction of which the Work performed under the Contract Documents may be the whole or a part and which may include construction by the Owner and by separate contractors.

#### § 1.1.5 THE DRAWINGS

The Drawings are the graphic and pictorial portions of the Contract Documents showing the design, location and dimensions of the Work, generally including plans, elevations, sections, details, schedules and diagrams.

#### § 1.1.6 THE SPECIFICATIONS

The Specifications are that portion of the Contract Documents consisting of the written requirements for materials, equipment, systems, standards and workmanship for the Work, and performance of related services.

#### § 1.1.7 INSTRUMENTS OF SERVICE

Instruments of Service are representations, in any medium of expression now known or later developed, of the tangible and intangible creative work performed by the Architect and the Architect's consultants under their respective professional services agreements. Instruments of Service may include, without limitation, studies, surveys, models, sketches, drawings, specifications, and other similar materials.

#### § 1.1.8 INITIAL DECISION MAKER

The Initial Decision Maker is the person identified in the Agreement to render initial decisions on Claims in accordance with Section 15.2 and certify termination of the Agreement under Section 14.2.2.

#### § 1.2 CORRELATION AND INTENT OF THE CONTRACT DOCUMENTS

§ 1.2.1 The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the Work by the Contractor. The Contract Documents are complementary, and what is required by one shall be as binding as if required by all; performance by the Contractor shall be required only to the extent consistent with the Contract Documents and reasonably inferable from them as being necessary to produce the indicated results.

AIA Document A201<sup>®</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

Init. 1

§ 1.2.2 Organization of the Specifications into divisions, sections and articles, and arrangement of Drawings shall not control the Contractor in dividing the Work among Subcontractors or in establishing the extent of Work to be performed by any trade.

§ 1.2.3 Unless otherwise stated in the Contract Documents, words that have well-known technical or construction industry meanings are used in the Contract Documents in accordance with such recognized meanings.

#### § 1.3 CAPITALIZATION

Terms capitalized in these General Conditions include those that are (1) specifically defined, (2) the titles of numbered articles or (3) the titles of other documents published by the American Institute of Architects.

#### § 1.4 INTERPRETATION

In the interest of brevity the Contract Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an," but the fact that a modifier or an article is absent from one statement and appears in another is not intended to affect the interpretation of either statement.

#### § 1.5 OWNERSHIP AND USE OF DRAWINGS, SPECIFICATIONS AND OTHER INSTRUMENTS OF SERVICE

§ 1.5.1 The Architect and the Architect's consultants shall be deemed the authors and owners of their respective Instruments of Service, including the Drawings and Specifications, and will retain all common law, statutory and other reserved rights, including copyrights. The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers shall not own or claim a copyright in the Instruments of Service. Submittal or distribution to meet official regulatory requirements or for other purposes in connection with this Project is not to be construed as publication in derogation of the Architect's or Architect's consultants' reserved rights.

§ 1.5.2 The Contractor, Subcontractors, Sub-subcontractors and material or equipment suppliers are authorized to use and reproduce the Instruments of Service provided to them solely and exclusively for execution of the Work. All copies made under this authorization shall bear the copyright notice, if any, shown on the Instruments of Service. The Contractor, Subcontractors, Sub-subcontractors, and material or equipment suppliers may not use the Instruments of Service on other projects or for additions to this Project outside the scope of the Work without the specific written consent of the Owner, Architect and the Architect's consultants.

#### § 1.6 TRANSMISSION OF DATA IN DIGITAL FORM

If the parties intend to transmit Instruments of Service or any other information or documentation in digital form, they shall endeavor to establish necessary protocols governing such transmissions, unless otherwise already provided in the Agreement or the Contract Documents.

#### ARTICLE 2 OWNER

#### § 2.1 GENERAL

§ 2.1.1 The Owner is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Owner shall designate in writing a representative who shall have express authority to bind the Owner with respect to all matters requiring the Owner's approval or authorization. Except as otherwise provided in Section 4.2.1, the Architect does not have such authority. The term "Owner" means the Owner or the Owner's authorized representative.

§ 2.1.2 The Owner shall furnish to the Contractor within fifteen days after receipt of a written request, information necessary and relevant for the Contractor to evaluate, give notice of or enforce mechanic's lien rights. Such information shall include a correct statement of the record legal title to the property on which the Project is located, usually referred to as the site, and the Owner's interest therein.

#### § 2.2 INFORMATION AND SERVICES REQUIRED OF THE OWNER

§ 2.2.1 Prior to commencement of the Work, the Contractor may request in writing that the Owner provide reasonable evidence that the Owner has made financial arrangements to fulfill the Owner's obligations under the Contract. Thereafter, the Contractor may only request such evidence if (1) the Owner fails to make payments to the Contractor as the Contract Documents require; (2) a change in the Work materially changes the Contract Sum; or (3) the Contractor identifies in writing a reasonable concern regarding the Owner's ability to make payment when due. The Owner shall furnish such evidence as a condition precedent to commencement or continuation of the Work or the

Init. 1

AIA Document A201® - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 10 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

portion of the Work affected by a material change. After the Owner furnishes the evidence, the Owner shall not materially vary such financial arrangements without prior notice to the Contractor.

§ 2.2.2 Except for permits and fees that are the responsibility of the Contractor under the Contract Documents, including those required under Section 3.7.1, the Owner shall secure and pay for necessary approvals, easements, assessments and charges required for construction, use or occupancy of permanent structures or for permanent changes in existing facilities.

§ 2.2.3 The Owner shall furnish surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site. The Contractor shall be entitled to rely on the accuracy of information furnished by the Owner but shall exercise proper precautions relating to the safe performance of the Work.

§ 2.2.4 The Owner shall furnish information or services required of the Owner by the Contract Documents with reasonable promptness. The Owner shall also furnish any other information or services under the Owner's control and relevant to the Contractor's performance of the Work with reasonable promptness after receiving the Contractor's written request for such information or services.

§ 2.2.5 Unless otherwise provided in the Contract Documents, the Owner shall furnish to the Contractor one copy of the Contract Documents for purposes of making reproductions pursuant to Section 1.5.2.

#### § 2.3 OWNER'S RIGHT TO STOP THE WORK

If the Contractor fails to correct Work that is not in accordance with the requirements of the Contract Documents as required by Section 12.2 or repeatedly fails to carry out Work in accordance with the Contract Documents, the Owner may issue a written order to the Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Contractor or any other person or entity, except to the extent required by Section 6.1.3.

#### § 2.4 OWNER'S RIGHT TO CARRY OUT THE WORK

If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and fails within a ten-day period after receipt of written notice from the Owner to commence and continue correction of such default or neglect with diligence and promptness, the Owner may, without prejudice to other remedies the Owner may have, correct such deficiencies. In such case an appropriate Change Order shall be issued deducting from payments then or thereafter due the Contractor the reasonable cost of correcting such deficiencies, including Owner's expenses and compensation for the Architect's additional services made necessary by such default, neglect or failure. Such action by the Owner and amounts charged to the Contractor are both subject to prior approval of the Architect. If payments then or thereafter due the Contractor are not sufficient to cover such amounts, the Contractor shall pay the difference to the Owner.

#### ARTICLE 3 CONTRACTOR

#### § 3.1 GENERAL

§ 3.1.1 The Contractor is the person or entity identified as such in the Agreement and is referred to throughout the Contract Documents as if singular in number. The Contractor shall be lawfully licensed, if required in the jurisdiction where the Project is located. The Contractor shall designate in writing a representative who shall have express authority to bind the Contractor with respect to all matters under this Contract. The term "Contractor" means the Contractor or the Contractor's authorized representative.

§ 3.1.2 The Contractor shall perform the Work in accordance with the Contract Documents.

§ 3.1.3 The Contractor shall not be relieved of obligations to perform the Work in accordance with the Contract Documents either by activities or duties of the Architect in the Architect's administration of the Contract, or by tests, inspections or approvals required or performed by persons or entities other than the Contractor.

Init. 1

AIA Document A201® - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 11 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

#### § 3.2 REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS BY CONTRACTOR

§ 3.2.1 Execution of the Contract by the Contractor is a representation that the Contractor has visited the site, become generally familiar with local conditions under which the Work is to be performed and correlated personal observations with requirements of the Contract Documents.

§ 3.2.2 Because the Contract Documents are complementary, the Contractor shall, before starting each portion of the Work, carefully study and compare the various Contract Documents relative to that portion of the Work, as well as the information furnished by the Owner pursuant to Section 2.2.3, shall take field measurements of any existing conditions related to that portion of the Work, and shall observe any conditions at the site affecting it. These obligations are for the purpose of facilitating coordination and construction by the Contractor and are not for the purpose of discovering errors, omissions, or inconsistencies in the Contract Documents; however, the Contractor shall promptly report to the Architect any errors, inconsistencies or omissions discovered by or made known to the Contractor as a request for information in such form as the Architect may require. It is recognized that the Contractor's review is made in the Contractor's capacity as a contractor and not as a licensed design professional, unless otherwise specifically provided in the Contract Documents.

§ 3.2.3 The Contractor is not required to ascertain that the Contract Documents are in accordance with applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, but the Contractor shall promptly report to the Architect any nonconformity discovered by or made known to the Contractor as a request for information in such form as the Architect may require.

§ 3.2.4 If the Contractor believes that additional cost or time is involved because of clarifications or instructions the Architect issues in response to the Contractor's notices or requests for information pursuant to Sections 3.2.2 or 3.2.3, the Contractor shall make Claims as provided in Article 15. If the Contractor fails to perform the obligations of Sections 3.2.2 or 3.2.3, the Contractor shall pay such costs and damages to the Owner as would have been avoided if the Contractor had performed such obligations. If the Contractor performs those obligations, the Contractor shall not be liable to the Owner or Architect for damages resulting from errors, inconsistencies or omissions in the Contract Documents, for differences between field measurements or conditions and the Contract Documents, or for nonconformities of the Contract Documents to applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities.

#### § 3.3 SUPERVISION AND CONSTRUCTION PROCEDURES

§ 3.3.1 The Contractor shall supervise and direct the Work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for, and have control over, construction means, methods, techniques, sequences and procedures and for coordinating all portions of the Work under the Contract, unless the Contract Documents give other specific instructions concerning these matters. If the Contract Documents give specific instructions concerning construction means, methods, techniques, sequences or procedures, the Contractor shall evaluate the jobsite safety thereof and, except as stated below, shall be fully and solely responsible for the jobsite safety of such means, methods, techniques, sequences or procedures. If the Contractor determines that such means, methods, techniques, sequences or procedures may not be safe, the Contractor shall give timely written notice to the Owner and Architect and shall not proceed with that portion of the Work without further written instructions from the Architect. If the Contractor is then instructed to proceed with the required means, methods, techniques, sequences or procedures without acceptance of changes proposed by the Contractor, the Owner shall be solely responsible for any loss or damage arising solely from those Owner-required means, methods, techniques, sequences or procedures.

§ 3.3.2 The Contractor shall be responsible to the Owner for acts and omissions of the Contractor's employees, Subcontractors and their agents and employees, and other persons or entities performing portions of the Work for, or on behalf of, the Contractor or any of its Subcontractors.

§ 3.3.3 The Contractor shall be responsible for inspection of portions of Work already performed to determine that such portions are in proper condition to receive subsequent Work.

#### § 3.4 LABOR AND MATERIALS

§ 3.4.1 Unless otherwise provided in the Contract Documents, the Contractor shall provide and pay for labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities and services necessary for proper execution and completion of the Work, whether temporary or permanent and whether or not incorporated or to be incorporated in the Work.

§ 3.4.2 Except in the case of minor changes in the Work authorized by the Architect in accordance with Sections 3.12.8 or 7.4, the Contractor may make substitutions only with the consent of the Owner, after evaluation by the Architect and in accordance with a Change Order or Construction Change Directive.

§ 3.4.3 The Contractor shall enforce strict discipline and good order among the Contractor's employees and other persons carrying out the Work. The Contractor shall not permit employment of unfit persons or persons not properly skilled in tasks assigned to them.

#### § 3.5 WARRANTY

The Contractor warrants to the Owner and Architect that materials and equipment furnished under the Contract will be of good quality and new unless the Contract Documents require or permit otherwise. The Contractor further warrants that the Work will conform to the requirements of the Contract Documents and will be free from defects, except for those inherent in the quality of the Work the Contract Documents require or permit. Work, materials, or equipment not conforming to these requirements may be considered defective. The Contractor's warranty excludes remedy for damage or defect caused by abuse, alterations to the Work not executed by the Contractor, improper or insufficient maintenance, improper operation, or normal wear and tear and normal usage. If required by the Architect, the Contractor shall furnish satisfactory evidence as to the kind and quality of materials and equipment.

#### § 3.6 TAXES

The Contractor shall pay sales, consumer, use and similar taxes for the Work provided by the Contractor that are legally enacted when bids are received or negotiations concluded, whether or not yet effective or merely scheduled to go into effect.

#### § 3.7 PERMITS, FEES, NOTICES AND COMPLIANCE WITH LAWS

§ 3.7.1 Unless otherwise provided in the Contract Documents, the Contractor shall secure and pay for the building permit as well as for other permits, fees, licenses, and inspections by government agencies necessary for proper execution and completion of the Work that are customarily secured after execution of the Contract and legally required at the time bids are received or negotiations concluded.

§ 3.7.2 The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities applicable to performance of the Work.

§ 3.7.3 If the Contractor performs Work knowing it to be contrary to applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of public authorities, the Contractor shall assume appropriate responsibility for such Work and shall bear the costs attributable to correction.

§ 3.7.4 Concealed or Unknown Conditions. If the Contractor encounters conditions at the site that are (1) subsurface or otherwise concealed physical conditions that differ materially from those indicated in the Contract Documents or (2) unknown physical conditions of an unusual nature, that differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Contract Documents, the Contractor shall promptly provide notice to the Owner and the Architect before conditions are disturbed and in no event later than 21 days after first observance of the conditions. The Architect will promptly investigate such conditions and, if the Architect determines that they differ materially and cause an increase or decrease in the Contractor's cost of, or time required for, performance of any part of the Work, will recommend an equitable adjustment in the Contract Sum or Contract Time, or both. If the Architect determines that the conditions at the site are not materially different from those indicated in the Contract Documents and that no change in the terms of the Contract is justified, the Architect shall promptly notify the Owner and Contractor in writing, stating the reasons. If either party disputes the Architect's determination or recommendation, that party may proceed as provided in Article 15.

§ 3.7.5 If, in the course of the Work, the Contractor encounters human remains or recognizes the existence of burial markers, archaeological sites or wetlands not indicated in the Contract Documents, the Contractor shall immediately suspend any operations that would affect them and shall notify the Owner and Architect. Upon receipt of such notice, the Owner shall promptly take any action necessary to obtain governmental authorization required to resume the operations. The Contractor shall continue to suspend such operations until otherwise instructed by the Owner but shall

Init. 1

AIA Document A201<sup>®</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 13 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

continue with all other operations that do not affect those remains or features. Requests for adjustments in the Contract Sum and Contract Time arising from the existence of such remains or features may be made as provided in Article 15.

#### § 3.8 ALLOWANCES

§ 3.8.1 The Contractor shall include in the Contract Sum all allowances stated in the Contract Documents. Items covered by allowances shall be supplied for such amounts and by such persons or entities as the Owner may direct, but the Contractor shall not be required to employ persons or entities to whom the Contractor has reasonable objection.

§ 3.8.2 Unless otherwise provided in the Contract Documents,

- Allowances shall cover the cost to the Contractor of materials and equipment delivered at the site and .1 all required taxes, less applicable trade discounts;
- .2 Contractor's costs for unloading and handling at the site, labor, installation costs, overhead, profit and other expenses contemplated for stated allowance amounts shall be included in the Contract Sum but not in the allowances; and
- .3 Whenever costs are more than or less than allowances, the Contract Sum shall be adjusted accordingly by Change Order. The amount of the Change Order shall reflect (1) the difference between actual costs and the allowances under Section 3.8.2.1 and (2) changes in Contractor's costs under Section 3.8.2.2.

§ 3.8.3 Materials and equipment under an allowance shall be selected by the Owner with reasonable promptness.

#### § 3.9 SUPERINTENDENT

§ 3.9.1 The Contractor shall employ a competent superintendent and necessary assistants who shall be in attendance at the Project site during performance of the Work. The superintendent shall represent the Contractor, and communications given to the superintendent shall be as binding as if given to the Contractor.

§ 3.9.2 The Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner through the Architect the name and qualifications of a proposed superintendent. The Architect may reply within 14 days to the Contractor in writing stating (1) whether the Owner or the Architect has reasonable objection to the proposed superintendent or (2) that the Architect requires additional time to review. Failure of the Architect to reply within the 14 day period shall constitute notice of no reasonable objection.

§ 3.9.3 The Contractor shall not employ a proposed superintendent to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not change the superintendent without the Owner's consent, which shall not unreasonably be withheld or delayed.

#### § 3.10 CONTRACTOR'S CONSTRUCTION SCHEDULES

§ 3.10.1 The Contractor, promptly after being awarded the Contract, shall prepare and submit for the Owner's and Architect's information a Contractor's construction schedule for the Work. The schedule shall not exceed time limits current under the Contract Documents, shall be revised at appropriate intervals as required by the conditions of the Work and Project, shall be related to the entire Project to the extent required by the Contract Documents, and shall provide for expeditious and practicable execution of the Work.

§ 3.10.2 The Contractor shall prepare a submittal schedule, promptly after being awarded the Contract and thereafter as necessary to maintain a current submittal schedule, and shall submit the schedule(s) for the Architect's approval. The Architect's approval shall not unreasonably be delayed or withheld. The submittal schedule shall (1) be coordinated with the Contractor's construction schedule, and (2) allow the Architect reasonable time to review submittals. If the Contractor fails to submit a submittal schedule, the Contractor shall not be entitled to any increase in Contract Sum or extension of Contract Time based on the time required for review of submittals.

§ 3.10.3 The Contractor shall perform the Work in general accordance with the most recent schedules submitted to the Owner and Architect.

#### § 3.11 DOCUMENTS AND SAMPLES AT THE SITE

Init.

1

The Contractor shall maintain at the site for the Owner one copy of the Drawings, Specifications, Addenda, Change Orders and other Modifications, in good order and marked currently to indicate field changes and selections made during construction, and one copy of approved Shop Drawings, Product Data, Samples and similar required

AIA Document A201® - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes: (1447905136)

submittals. These shall be available to the Architect and shall be delivered to the Architect for submittal to the Owner upon completion of the Work as a record of the Work as constructed.

#### § 3.12 SHOP DRAWINGS, PRODUCT DATA AND SAMPLES

Init.

1

§ 3.12.1 Shop Drawings are drawings, diagrams, schedules and other data specially prepared for the Work by the Contractor or a Subcontractor, Sub-subcontractor, manufacturer, supplier or distributor to illustrate some portion of the Work.

§ 3.12.2 Product Data are illustrations, standard schedules, performance charts, instructions, brochures, diagrams and other information furnished by the Contractor to illustrate materials or equipment for some portion of the Work.

§ 3.12.3 Samples are physical examples that illustrate materials, equipment or workmanship and establish standards by which the Work will be judged.

§ 3.12.4 Shop Drawings, Product Data, Samples and similar submittals are not Contract Documents. Their purpose is to demonstrate the way by which the Contractor proposes to conform to the information given and the design concept expressed in the Contract Documents for those portions of the Work for which the Contract Documents require submittals. Review by the Architect is subject to the limitations of Section 4.2.7. Informational submittals upon which the Architect is not expected to take responsive action may be so identified in the Contract Documents. Submittals that are not required by the Contract Documents may be returned by the Architect without action.

§ 3.12.5 The Contractor shall review for compliance with the Contract Documents, approve and submit to the Architect Shop Drawings, Product Data, Samples and similar submittals required by the Contract Documents in accordance with the submittal schedule approved by the Architect or, in the absence of an approved submittal schedule, with reasonable promptness and in such sequence as to cause no delay in the Work or in the activities of the Owner or of separate contractors.

§ 3.12.6 By submitting Shop Drawings, Product Data, Samples and similar submittals, the Contractor represents to the Owner and Architect that the Contractor has (1) reviewed and approved them, (2) determined and verified materials, field measurements and field construction criteria related thereto, or will do so and (3) checked and coordinated the information contained within such submittals with the requirements of the Work and of the Contract Documents.

§ 3.12.7 The Contractor shall perform no portion of the Work for which the Contract Documents require submittal and review of Shop Drawings, Product Data, Samples or similar submittals until the respective submittal has been approved by the Architect.

§ 3.12.8 The Work shall be in accordance with approved submittals except that the Contractor shall not be relieved of responsibility for deviations from requirements of the Contract Documents by the Architect's approval of Shop Drawings, Product Data, Samples or similar submittals unless the Contractor has specifically informed the Architect in writing of such deviation at the time of submittal and (1) the Architect has given written approval to the specific deviation as a minor change in the Work, or (2) a Change Order or Construction Change Directive has been issued authorizing the deviation. The Contractor shall not be relieved of responsibility for errors or omissions in Shop Drawings, Product Data, Samples or similar submittals by the Architect's approval thereof.

§ 3.12.9 The Contractor shall direct specific attention, in writing or on resubmitted Shop Drawings, Product Data, Samples or similar submittals, to revisions other than those requested by the Architect on previous submittals. In the absence of such written notice, the Architect's approval of a resubmission shall not apply to such revisions.

§ 3.12.10 The Contractor shall not be required to provide professional services that constitute the practice of architecture or engineering unless such services are specifically required by the Contract Documents for a portion of the Work or unless the Contractor needs to provide such services in order to carry out the Contractor's responsibilities for construction means, methods, techniques, sequences and procedures. The Contractor shall not be required to provide professional services in violation of applicable law. If professional design services or certifications by a design professional related to systems, materials or equipment are specifically required of the Contractor by the Contract Documents, the Owner and the Architect will specify all performance and design criteria that such services must satisfy. The Contractor shall cause such services or certifications to be provided by a properly licensed design professional, whose signature and seal shall appear on all drawings, calculations, specifications, certifications, Shop

AIA Document A201<sup>®</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 15 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

Drawings and other submittals prepared by such professional. Shop Drawings and other submittals related to the Work designed or certified by such professional, if prepared by others, shall bear such professional's written approval when submitted to the Architect. The Owner and the Architect shall be entitled to rely upon the adequacy, accuracy and completeness of the services, certifications and approvals performed or provided by such design professionals, provided the Owner and Architect have specified to the Contractor all performance and design criteria that such services must satisfy. Pursuant to this Section 3.12.10, the Architect will review, approve or take other appropriate action on submittals only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Contractor shall not be responsible for the adequacy of the performance and design criteria specified in the Contract Documents.

#### § 3.13 USE OF SITE

The Contractor shall confine operations at the site to areas permitted by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities and the Contract Documents and shall not unreasonably encumber the site with materials or equipment.

#### § 3.14 CUTTING AND PATCHING

§ 3.14.1 The Contractor shall be responsible for cutting, fitting or patching required to complete the Work or to make its parts fit together properly. All areas requiring cutting, fitting and patching shall be restored to the condition existing prior to the cutting, fitting and patching, unless otherwise required by the Contract Documents.

§ 3.14.2 The Contractor shall not damage or endanger a portion of the Work or fully or partially completed construction of the Owner or separate contractors by cutting, patching or otherwise altering such construction, or by excavation. The Contractor shall not cut or otherwise alter such construction by the Owner or a separate contractor except with written consent of the Owner and of such separate contractor; such consent shall not be unreasonably withheld. The Contractor shall not unreasonably withhold from the Owner or a separate contractor the Contractor's consent to cutting or otherwise altering the Work.

#### § 3.15 CLEANING UP

§ 3.15.1 The Contractor shall keep the premises and surrounding area free from accumulation of waste materials or rubbish caused by operations under the Contract. At completion of the Work, the Contractor shall remove waste materials, rubbish, the Contractor's tools, construction equipment, machinery and surplus materials from and about the Project.

§ 3.15.2 If the Contractor fails to clean up as provided in the Contract Documents, the Owner may do so and Owner shall be entitled to reimbursement from the Contractor.

#### § 3.16 ACCESS TO WORK

The Contractor shall provide the Owner and Architect access to the Work in preparation and progress wherever located.

#### § 3.17 ROYALTIES, PATENTS AND COPYRIGHTS

The Contractor shall pay all royalties and license fees. The Contractor shall defend suits or claims for infringement of copyrights and patent rights and shall hold the Owner and Architect harmless from loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process or product of a particular manufacturer or manufacturers is required by the Contract Documents, or where the copyright violations are contained in Drawings, Specifications or other documents prepared by the Owner or Architect. However, if the Contractor has reason to believe that the required design, process or product is an infringement of a copyright or a patent, the Contractor shall be responsible for such loss unless such information is promptly furnished to the Architect.

#### § 3.18 INDEMNIFICATION

Init.

1

§ 3.18.1 To the fullest extent permitted by law the Contractor shall indemnify and hold harmless the Owner, Architect, Architect's consultants, and agents and employees of any of them from and against claims, damages, losses and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work, provided that such claim, damage, loss or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the Work itself), but only to the extent caused by the negligent acts or omissions of the Contractor, a Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, regardless of whether or not such claim, damage, loss or expense is caused in part by a

AIA Document A201<sup>®</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 16 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

party indemnified hereunder. Such obligation shall not be construed to negate, abridge, or reduce other rights or obligations of indemnity that would otherwise exist as to a party or person described in this Section 3.18.

§ 3.18.2 In claims against any person or entity indemnified under this Section 3.18 by an employee of the Contractor, a Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, the indemnification obligation under Section 3.18.1 shall not be limited by a limitation on amount or type of damages, compensation or benefits payable by or for the Contractor or a Subcontractor under workers' compensation acts, disability benefit acts or other employee benefit acts.

#### ARTICLE 4 ARCHITECT

#### § 4.1 GENERAL

§ 4.1.1 The Owner shall retain an architect lawfully licensed to practice architecture or an entity lawfully practicing architecture in the jurisdiction where the Project is located. That person or entity is identified as the Architect in the Agreement and is referred to throughout the Contract Documents as if singular in number.

§ 4.1.2 Duties, responsibilities and limitations of authority of the Architect as set forth in the Contract Documents shall not be restricted, modified or extended without written consent of the Owner, Contractor and Architect. Consent shall not be unreasonably withheld.

§ 4.1.3 If the employment of the Architect is terminated, the Owner shall employ a successor architect as to whom the Contractor has no reasonable objection and whose status under the Contract Documents shall be that of the Architect.

#### § 4.2 ADMINISTRATION OF THE CONTRACT

§ 4.2.1 The Architect will provide administration of the Contract as described in the Contract Documents and will be an Owner's representative during construction until the date the Architect issues the final Certificate for Payment. The Architect will have authority to act on behalf of the Owner only to the extent provided in the Contract Documents.

§ 4.2.2 The Architect will visit the site at intervals appropriate to the stage of construction, or as otherwise agreed with the Owner, to become generally familiar with the progress and quality of the portion of the Work completed, and to determine in general if the Work observed is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect will not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work. The Architect will not have control over, charge of, or responsibility for, the construction means, methods, techniques, sequences or procedures, or for the safety precautions and programs in connection with the Work, since these are solely the Contractor's rights and responsibilities under the Contract Documents, except as provided in Section 3.3.1.

§ 4.2.3 On the basis of the site visits, the Architect will keep the Owner reasonably informed about the progress and quality of the portion of the Work completed, and report to the Owner (1) known deviations from the Contract Documents and from the most recent construction schedule submitted by the Contractor, and (2) defects and deficiencies observed in the Work. The Architect will not be responsible for the Contractor's failure to perform the Work in accordance with the requirements of the Contract Documents. The Architect will not have control over or charge of and will not be responsible for acts or omissions of the Contractor, Subcontractors, or their agents or employees, or any other persons or entities performing portions of the Work.

#### § 4.2.4 COMMUNICATIONS FACILITATING CONTRACT ADMINISTRATION

Except as otherwise provided in the Contract Documents or when direct communications have been specially authorized, the Owner and Contractor shall endeavor to communicate with each other through the Architect about matters arising out of or relating to the Contract. Communications by and with the Architect's consultants shall be through the Architect. Communications by and with Subcontractors and material suppliers shall be through the Contractor. Communications by and with separate contractors shall be through the Owner.

§ 4.2.5 Based on the Architect's evaluations of the Contractor's Applications for Payment, the Architect will review and certify the amounts due the Contractor and will issue Certificates for Payment in such amounts.

§ 4.2.6 The Architect has authority to reject Work that does not conform to the Contract Documents. Whenever the Architect considers it necessary or advisable, the Architect will have authority to require inspection or testing of the Work in accordance with Sections 13.5.2 and 13.5.3, whether or not such Work is fabricated, installed or completed.

Init.

However, neither this authority of the Architect nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Contractor, Subcontractors, material and equipment suppliers, their agents or employees, or other persons or entities performing portions of the Work.

**§ 4.2.7** The Architect will review and approve, or take other appropriate action upon, the Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the limited purpose of checking for conformance with information given and the design concept expressed in the Contract Documents. The Architect's action will be taken in accordance with the submittal schedule approved by the Architect or, in the absence of an approved submittal schedule, with reasonable promptness while allowing sufficient time in the Architect's professional judgment to permit adequate review. Review of such submittals is not conducted for the purpose of determining the accuracy and completeness of other details such as dimensions and quantities, or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Contractor as required by the Contract Documents. The Architect's review of the Contractor's submittals shall not relieve the Contractor of the obligations under Sections 3.3, 3.5 and 3.12. The Architect's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by the Architect, of any construction means, methods, techniques, sequences or procedures. The Architect's approval of a specific item shall not indicate approval of an assembly of which the item is a component.

**§ 4.2.8** The Architect will prepare Change Orders and Construction Change Directives, and may authorize minor changes in the Work as provided in Section 7.4. The Architect will investigate and make determinations and recommendations regarding concealed and unknown conditions as provided in Section 3.7.4.

**§ 4.2.9** The Architect will conduct inspections to determine the date or dates of Substantial Completion and the date of final completion; issue Certificates of Substantial Completion pursuant to Section 9.8; receive and forward to the Owner, for the Owner's review and records, written warranties and related documents required by the Contract and assembled by the Contractor pursuant to Section 9.10; and issue a final Certificate for Payment pursuant to Section 9.10.

§ 4.2.10 If the Owner and Architect agree, the Architect will provide one or more project representatives to assist in carrying out the Architect's responsibilities at the site. The duties, responsibilities and limitations of authority of such project representatives shall be as set forth in an exhibit to be incorporated in the Contract Documents.

§ 4.2.11 The Architect will interpret and decide matters concerning performance under, and requirements of, the Contract Documents on written request of either the Owner or Contractor. The Architect's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness.

**§ 4.2.12** Interpretations and decisions of the Architect will be consistent with the intent of, and reasonably inferable from, the Contract Documents and will be in writing or in the form of drawings. When making such interpretations and decisions, the Architect will endeavor to secure faithful performance by both Owner and Contractor, will not show partiality to either and will not be liable for results of interpretations or decisions rendered in good faith.

§ 4.2.13 The Architect's decisions on matters relating to aesthetic effect will be final if consistent with the intent expressed in the Contract Documents.

§ 4.2.14 The Architect will review and respond to requests for information about the Contract Documents. The Architect's response to such requests will be made in writing within any time limits agreed upon or otherwise with reasonable promptness. If appropriate, the Architect will prepare and issue supplemental Drawings and Specifications in response to the requests for information.

#### ARTICLE 5 SUBCONTRACTORS

#### § 5.1 DEFINITIONS

§ 5.1.1 A Subcontractor is a person or entity who has a direct contract with the Contractor to perform a portion of the Work at the site. The term "Subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Subcontractor or an authorized representative of the Subcontractor. The term "Subcontractor" does not include a separate contractor or subcontractors of a separate contractor.

§ 5.1.2 A Sub-subcontractor is a person or entity who has a direct or indirect contract with a Subcontractor to perform a portion of the Work at the site. The term "Sub-subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Sub-subcontractor or an authorized representative of the Sub-subcontractor.

#### § 5.2 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OF THE WORK

§ 5.2.1 Unless otherwise stated in the Contract Documents or the bidding requirements, the Contractor, as soon as practicable after award of the Contract, shall furnish in writing to the Owner through the Architect the names of persons or entities (including those who are to furnish materials or equipment fabricated to a special design) proposed for each principal portion of the Work. The Architect may reply within 14 days to the Contractor in writing stating (1) whether the Owner or the Architect has reasonable objection to any such proposed person or entity or (2) that the Architect requires additional time for review. Failure of the Owner or Architect to reply within the 14-day period shall constitute notice of no reasonable objection.

§ 5.2.2 The Contractor shall not contract with a proposed person or entity to whom the Owner or Architect has made reasonable and timely objection. The Contractor shall not be required to contract with anyone to whom the Contractor has made reasonable objection.

§ 5.2.3 If the Owner or Architect has reasonable objection to a person or entity proposed by the Contractor, the Contractor shall propose another to whom the Owner or Architect has no reasonable objection. If the proposed but rejected Subcontractor was reasonably capable of performing the Work, the Contract Sum and Contract Time shall be increased or decreased by the difference, if any, occasioned by such change, and an appropriate Change Order shall be issued before commencement of the substitute Subcontractor's Work. However, no increase in the Contract Sum or Contract Time shall be allowed for such change unless the Contractor has acted promptly and responsively in submitting names as required.

§ 5.2.4 The Contractor shall not substitute a Subcontractor, person or entity previously selected if the Owner or Architect makes reasonable objection to such substitution.

#### § 5.3 SUBCONTRACTUAL RELATIONS

By appropriate agreement, written where legally required for validity, the Contractor shall require each Subcontractor, to the extent of the Work to be performed by the Subcontractor, to be bound to the Contractor by terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities, including the responsibility for safety of the Subcontractor's Work, which the Contractor, by these Documents, assumes toward the Owner and Architect. Each subcontract agreement shall preserve and protect the rights of the Owner and Architect under the Contract Documents with respect to the Work to be performed by the Subcontractor so that subcontracting thereof will not prejudice such rights, and shall allow to the Subcontractor, unless specifically provided otherwise in the subcontract agreement, the benefit of all rights, remedies and redress against the Contractor that the Contractor, by the Contract Documents, has against the Owner. Where appropriate, the Contractor shall require each Subcontractor to enter into similar agreements with Sub-subcontractors. The Contractor shall make available to each proposed Subcontractor, prior to the execution of the subcontract agreement, copies of the Contract Documents to which the Subcontractor will be bound, and, upon written request of the Subcontractor, identify to the Subcontractor terms and conditions of the proposed subcontract agreement that may be at variance with the Contract Documents. Subcontractors will similarly make copies of applicable portions of such documents available to their respective proposed Sub-subcontractors.

#### § 5.4 CONTINGENT ASSIGNMENT OF SUBCONTRACTS

§ 5.4.1 Each subcontract agreement for a portion of the Work is assigned by the Contractor to the Owner, provided that

- .1 assignment is effective only after termination of the Contract by the Owner for cause pursuant to Section 14.2 and only for those subcontract agreements that the Owner accepts by notifying the Subcontractor and Contractor in writing; and
- .2 assignment is subject to the prior rights of the surety, if any, obligated under bond relating to the Contract.

When the Owner accepts the assignment of a subcontract agreement, the Owner assumes the Contractor's rights and obligations under the subcontract.

Init. 1

AIA Document A201® - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 19 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

§ 5.4.2 Upon such assignment, if the Work has been suspended for more than 30 days, the Subcontractor's compensation shall be equitably adjusted for increases in cost resulting from the suspension.

§ 5.4.3 Upon such assignment to the Owner under this Section 5.4, the Owner may further assign the subcontract to a successor contractor or other entity. If the Owner assigns the subcontract to a successor contractor or other entity, the Owner shall nevertheless remain legally responsible for all of the successor contractor's obligations under the subcontract.

#### ARTICLE 6 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS § 6.1 OWNER'S RIGHT TO PERFORM CONSTRUCTION AND TO AWARD SEPARATE CONTRACTS

§ 6.1.1 The Owner reserves the right to perform construction or operations related to the Project with the Owner's own forces, and to award separate contracts in connection with other portions of the Project or other construction or operations on the site under Conditions of the Contract identical or substantially similar to these including those portions related to insurance and waiver of subrogation. If the Contractor claims that delay or additional cost is involved because of such action by the Owner, the Contractor shall make such Claim as provided in Article 15.

§ 6.1.2 When separate contracts are awarded for different portions of the Project or other construction or operations on the site, the term "Contractor" in the Contract Documents in each case shall mean the Contractor who executes each separate Owner-Contractor Agreement.

§ 6.1.3 The Owner shall provide for coordination of the activities of the Owner's own forces and of each separate contractor with the Work of the Contractor, who shall cooperate with them. The Contractor shall participate with other separate contractors and the Owner in reviewing their construction schedules. The Contractor shall make any revisions to the construction schedule deemed necessary after a joint review and mutual agreement. The construction schedules shall then constitute the schedules to be used by the Contractor, separate contractors and the Owner until subsequently revised.

§ 6.1.4 Unless otherwise provided in the Contract Documents, when the Owner performs construction or operations related to the Project with the Owner's own forces, the Owner shall be deemed to be subject to the same obligations and to have the same rights that apply to the Contractor under the Conditions of the Contract, including, without excluding others, those stated in Article 3, this Article 6 and Articles 10, 11 and 12.

#### § 6.2 MUTUAL RESPONSIBILITY

§ 6.2.1 The Contractor shall afford the Owner and separate contractors reasonable opportunity for introduction and storage of their materials and equipment and performance of their activities, and shall connect and coordinate the Contractor's construction and operations with theirs as required by the Contract Documents.

§ 6.2.2 If part of the Contractor's Work depends for proper execution or results upon construction or operations by the Owner or a separate contractor, the Contractor shall, prior to proceeding with that portion of the Work, promptly report to the Architect apparent discrepancies or defects in such other construction that would render it unsuitable for such proper execution and results. Failure of the Contractor so to report shall constitute an acknowledgment that the Owner's or separate contractor's completed or partially completed construction is fit and proper to receive the Contractor's Work, except as to defects not then reasonably discoverable.

§ 6.2.3 The Contractor shall reimburse the Owner for costs the Owner incurs that are payable to a separate contractor because of the Contractor's delays, improperly timed activities or defective construction. The Owner shall be responsible to the Contractor for costs the Contractor incurs because of a separate contractor's delays, improperly timed activities, damage to the Work or defective construction.

§ 6.2.4 The Contractor shall promptly remedy damage the Contractor wrongfully causes to completed or partially completed construction or to property of the Owner or separate contractors as provided in Section 10.2.5.

§ 6.2.5 The Owner and each separate contractor shall have the same responsibilities for cutting and patching as are described for the Contractor in Section 3.14.

AIA Document A201<sup>®</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 20 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

### § 6.3 OWNER'S RIGHT TO CLEAN UP

If a dispute arises among the Contractor, separate contractors and the Owner as to the responsibility under their respective contracts for maintaining the premises and surrounding area free from waste materials and rubbish, the Owner may clean up and the Architect will allocate the cost among those responsible.

#### ARTICLE 7 CHANGES IN THE WORK § 7.1 GENERAL

§ 7.1.1 Changes in the Work may be accomplished after execution of the Contract, and without invalidating the Contract, by Change Order, Construction Change Directive or order for a minor change in the Work, subject to the limitations stated in this Article 7 and elsewhere in the Contract Documents.

§ 7.1.2 A Change Order shall be based upon agreement among the Owner, Contractor and Architect; a Construction Change Directive requires agreement by the Owner and Architect and may or may not be agreed to by the Contractor; an order for a minor change in the Work may be issued by the Architect alone.

§ 7.1.3 Changes in the Work shall be performed under applicable provisions of the Contract Documents, and the Contractor shall proceed promptly, unless otherwise provided in the Change Order, Construction Change Directive or order for a minor change in the Work.

#### § 7.2 CHANGE ORDERS

§ 7.2.1 A Change Order is a written instrument prepared by the Architect and signed by the Owner, Contractor and Architect stating their agreement upon all of the following:

- .1 The change in the Work;
- .2 The amount of the adjustment, if any, in the Contract Sum; and
- .3 The extent of the adjustment, if any, in the Contract Time.

#### § 7.3 CONSTRUCTION CHANGE DIRECTIVES

§ 7.3.1 A Construction Change Directive is a written order prepared by the Architect and signed by the Owner and Architect, directing a change in the Work prior to agreement on adjustment, if any, in the Contract Sum or Contract Time, or both. The Owner may by Construction Change Directive, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions, the Contract Sum and Contract Time being adjusted accordingly.

§ 7.3.2 A Construction Change Directive shall be used in the absence of total agreement on the terms of a Change Order.

§ 7.3.3 If the Construction Change Directive provides for an adjustment to the Contract Sum, the adjustment shall be based on one of the following methods:

- Mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to .1 permit evaluation;
- .2 Unit prices stated in the Contract Documents or subsequently agreed upon;
- .3 Cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee; or
- As provided in Section 7.3.7. .4

§ 7.3.4 If unit prices are stated in the Contract Documents or subsequently agreed upon, and if quantities originally contemplated are materially changed in a proposed Change Order or Construction Change Directive so that application of such unit prices to quantities of Work proposed will cause substantial inequity to the Owner or Contractor, the applicable unit prices shall be equitably adjusted.

§ 7.3.5 Upon receipt of a Construction Change Directive, the Contractor shall promptly proceed with the change in the Work involved and advise the Architect of the Contractor's agreement or disagreement with the method, if any, provided in the Construction Change Directive for determining the proposed adjustment in the Contract Sum or Contract Time.

AIA Document A201<sup>®</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 21 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

§ 7.3.6 A Construction Change Directive signed by the Contractor indicates the Contractor's agreement therewith, including adjustment in Contract Sum and Contract Time or the method for determining them. Such agreement shall be effective immediately and shall be recorded as a Change Order.

§ 7.3.7 If the Contractor does not respond promptly or disagrees with the method for adjustment in the Contract Sum, the Architect shall determine the method and the adjustment on the basis of reasonable expenditures and savings of those performing the Work attributable to the change, including, in case of an increase in the Contract Sum, an amount for overhead and profit as set forth in the Agreement, or if no such amount is set forth in the Agreement, a reasonable amount. In such case, and also under Section 7.3.3.3, the Contractor shall keep and present, in such form as the Architect may prescribe, an itemized accounting together with appropriate supporting data. Unless otherwise provided in the Contract Documents, costs for the purposes of this Section 7.3.7 shall be limited to the following:

- .1 Costs of labor, including social security, old age and unemployment insurance, fringe benefits required by agreement or custom, and workers' compensation insurance;
- .2 Costs of materials, supplies and equipment, including cost of transportation, whether incorporated or consumed;
- .3 Rental costs of machinery and equipment, exclusive of hand tools, whether rented from the Contractor or others:
- .4 Costs of premiums for all bonds and insurance, permit fees, and sales, use or similar taxes related to the Work; and
- .5 Additional costs of supervision and field office personnel directly attributable to the change.

§ 7.3.8 The amount of credit to be allowed by the Contractor to the Owner for a deletion or change that results in a net decrease in the Contract Sum shall be actual net cost as confirmed by the Architect. When both additions and credits covering related Work or substitutions are involved in a change, the allowance for overhead and profit shall be figured on the basis of net increase, if any, with respect to that change.

§ 7.3.9 Pending final determination of the total cost of a Construction Change Directive to the Owner, the Contractor may request payment for Work completed under the Construction Change Directive in Applications for Payment. The Architect will make an interim determination for purposes of monthly certification for payment for those costs and certify for payment the amount that the Architect determines, in the Architect's professional judgment, to be reasonably justified. The Architect's interim determination of cost shall adjust the Contract Sum on the same basis as a Change Order, subject to the right of either party to disagree and assert a Claim in accordance with Article 15.

§ 7.3.10 When the Owner and Contractor agree with a determination made by the Architect concerning the adjustments in the Contract Sum and Contract Time, or otherwise reach agreement upon the adjustments, such agreement shall be effective immediately and the Architect will prepare a Change Order. Change Orders may be issued for all or any part of a Construction Change Directive.

#### § 7.4 MINOR CHANGES IN THE WORK

The Architect has authority to order minor changes in the Work not involving adjustment in the Contract Sum or extension of the Contract Time and not inconsistent with the intent of the Contract Documents. Such changes will be effected by written order signed by the Architect and shall be binding on the Owner and Contractor.

# ARTICLE 8 TIME

## § 8.1 DEFINITIONS

§ 8.1.1 Unless otherwise provided, Contract Time is the period of time, including authorized adjustments, allotted in the Contract Documents for Substantial Completion of the Work.

§ 8.1.2 The date of commencement of the Work is the date established in the Agreement.

§ 8.1.3 The date of Substantial Completion is the date certified by the Architect in accordance with Section 9.8.

§ 8.1.4 The term "day" as used in the Contract Documents shall mean calendar day unless otherwise specifically defined.

#### § 8.2 PROGRESS AND COMPLETION

§ 8.2.1 Time limits stated in the Contract Documents are of the essence of the Contract. By executing the Agreement the Contractor confirms that the Contract Time is a reasonable period for performing the Work.

§ 8.2.2 The Contractor shall not knowingly, except by agreement or instruction of the Owner in writing, prematurely commence operations on the site or elsewhere prior to the effective date of insurance required by Article 11 to be furnished by the Contractor and Owner. The date of commencement of the Work shall not be changed by the effective date of such insurance.

§ 8.2.3 The Contractor shall proceed expeditiously with adequate forces and shall achieve Substantial Completion within the Contract Time.

#### § 8.3 DELAYS AND EXTENSIONS OF TIME

§ 8.3.1 If the Contractor is delayed at any time in the commencement or progress of the Work by an act or neglect of the Owner or Architect, or of an employee of either, or of a separate contractor employed by the Owner; or by changes ordered in the Work; or by labor disputes, fire, unusual delay in deliveries, unavoidable casualties or other causes beyond the Contractor's control; or by delay authorized by the Owner pending mediation and arbitration; or by other causes that the Architect determines may justify delay, then the Contract Time shall be extended by Change Order for such reasonable time as the Architect may determine.

§ 8.3.2 Claims relating to time shall be made in accordance with applicable provisions of Article 15.

§ 8.3.3 This Section 8.3 does not preclude recovery of damages for delay by either party under other provisions of the Contract Documents.

## ARTICLE 9 PAYMENTS AND COMPLETION

### § 9.1 CONTRACT SUM

Init.

1

The Contract Sum is stated in the Agreement and, including authorized adjustments, is the total amount payable by the Owner to the Contractor for performance of the Work under the Contract Documents.

#### § 9.2 SCHEDULE OF VALUES

Where the Contract is based on a stipulated sum or Guaranteed Maximum Price, the Contractor shall submit to the Architect, before the first Application for Payment, a schedule of values allocating the entire Contract Sum to the various portions of the Work and prepared in such form and supported by such data to substantiate its accuracy as the Architect may require. This schedule, unless objected to by the Architect, shall be used as a basis for reviewing the Contractor's Applications for Payment.

#### § 9.3 APPLICATIONS FOR PAYMENT

§ 9.3.1 At least ten days before the date established for each progress payment, the Contractor shall submit to the Architect an itemized Application for Payment prepared in accordance with the schedule of values, if required under Section 9.2, for completed portions of the Work. Such application shall be notarized, if required, and supported by such data substantiating the Contractor's right to payment as the Owner or Architect may require, such as copies of requisitions from Subcontractors and material suppliers, and shall reflect retainage if provided for in the Contract Documents.

§ 9.3.1.1 As provided in Section 7.3.9, such applications may include requests for payment on account of changes in the Work that have been properly authorized by Construction Change Directives, or by interim determinations of the Architect, but not yet included in Change Orders.

§ 9.3.1.2 Applications for Payment shall not include requests for payment for portions of the Work for which the Contractor does not intend to pay a Subcontractor or material supplier, unless such Work has been performed by others whom the Contractor intends to pay.

§ 9.3.2 Unless otherwise provided in the Contract Documents, payments shall be made on account of materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work. If approved in advance by the Owner, payment may similarly be made for materials and equipment suitably stored off the site at a location agreed upon in writing. Payment for materials and equipment stored on or off the site shall be conditioned upon

compliance by the Contractor with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect the Owner's interest, and shall include the costs of applicable insurance, storage and transportation to the site for such materials and equipment stored off the site.

§ 9.3.3 The Contractor warrants that title to all Work covered by an Application for Payment will pass to the Owner no later than the time of payment. The Contractor further warrants that upon submittal of an Application for Payment all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, to the best of the Contractor's knowledge, information and belief, be free and clear of liens, claims, security interests or encumbrances in favor of the Contractor, Subcontractors, material suppliers, or other persons or entities making a claim by reason of having provided labor, materials and equipment relating to the Work.

#### § 9.4 CERTIFICATES FOR PAYMENT

§ 9.4.1 The Architect will, within seven days after receipt of the Contractor's Application for Payment, either issue to the Owner a Certificate for Payment, with a copy to the Contractor, for such amount as the Architect determines is properly due, or notify the Contractor and Owner in writing of the Architect's reasons for withholding certification in whole or in part as provided in Section 9.5.1.

§ 9.4.2 The issuance of a Certificate for Payment will constitute a representation by the Architect to the Owner, based on the Architect's evaluation of the Work and the data comprising the Application for Payment, that, to the best of the Architect's knowledge, information and belief, the Work has progressed to the point indicated and that the quality of the Work is in accordance with the Contract Documents. The foregoing representations are subject to an evaluation of the Work for conformance with the Contract Documents upon Substantial Completion, to results of subsequent tests and inspections, to correction of minor deviations from the Contract Documents prior to completion and to specific qualifications expressed by the Architect. The issuance of a Certificate for Payment will further constitute a representation that the Contractor is entitled to payment in the amount certified. However, the issuance of a Certificate for Payment will not be a representation that the Architect has (1) made exhaustive or continuous on-site inspections to check the quality or quantity of the Work, (2) reviewed construction means, methods, techniques, sequences or procedures, (3) reviewed copies of requisitions received from Subcontractors and material suppliers and other data requested by the Owner to substantiate the Contractor's right to payment, or (4) made examination to ascertain how or for what purpose the Contractor has used money previously paid on account of the Contract Sum.

#### § 9.5 DECISIONS TO WITHHOLD CERTIFICATION

§ 9.5.1 The Architect may withhold a Certificate for Payment in whole or in part, to the extent reasonably necessary to protect the Owner, if in the Architect's opinion the representations to the Owner required by Section 9.4.2 cannot be made. If the Architect is unable to certify payment in the amount of the Application, the Architect will notify the Contractor and Owner as provided in Section 9.4.1. If the Contractor and Architect cannot agree on a revised amount, the Architect will promptly issue a Certificate for Payment for the amount for which the Architect is able to make such representations to the Owner. The Architect may also withhold a Certificate for Payment or, because of subsequently discovered evidence, may nullify the whole or a part of a Certificate for Payment previously issued, to such extent as may be necessary in the Architect's opinion to protect the Owner from loss for which the Contractor is responsible, including loss resulting from acts and omissions described in Section 3.3.2, because of

- .1 defective Work not remedied;
- .2 third party claims filed or reasonable evidence indicating probable filing of such claims unless security acceptable to the Owner is provided by the Contractor;
- .3 failure of the Contractor to make payments properly to Subcontractors or for labor, materials or equipment;
- .4 reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum;
- .5 damage to the Owner or a separate contractor;
- .6 reasonable evidence that the Work will not be completed within the Contract Time, and that the unpaid balance would not be adequate to cover actual or liquidated damages for the anticipated delay; or
- .7 repeated failure to carry out the Work in accordance with the Contract Documents.

§ 9.5.2 When the above reasons for withholding certification are removed, certification will be made for amounts previously withheld.

§ 9.5.3 If the Architect withholds certification for payment under Section 9.5.1.3, the Owner may, at its sole option, issue joint checks to the Contractor and to any Subcontractor or material or equipment suppliers to whom the

Contractor failed to make payment for Work properly performed or material or equipment suitably delivered. If the Owner makes payments by joint check, the Owner shall notify the Architect and the Architect will reflect such payment on the next Certificate for Payment.

#### § 9.6 PROGRESS PAYMENTS

§ 9.6.1 After the Architect has issued a Certificate for Payment, the Owner shall make payment in the manner and within the time provided in the Contract Documents, and shall so notify the Architect.

§ 9.6.2 The Contractor shall pay each Subcontractor no later than seven days after receipt of payment from the Owner the amount to which the Subcontractor is entitled, reflecting percentages actually retained from payments to the Contractor on account of the Subcontractor's portion of the Work. The Contractor shall, by appropriate agreement with each Subcontractor, require each Subcontractor to make payments to Sub-subcontractors in a similar manner.

§ 9.6.3 The Architect will, on request, furnish to a Subcontractor, if practicable, information regarding percentages of completion or amounts applied for by the Contractor and action taken thereon by the Architect and Owner on account of portions of the Work done by such Subcontractor.

§ 9.6.4 The Owner has the right to request written evidence from the Contractor that the Contractor has properly paid Subcontractors and material and equipment suppliers amounts paid by the Owner to the Contractor for subcontracted Work. If the Contractor fails to furnish such evidence within seven days, the Owner shall have the right to contact Subcontractors to ascertain whether they have been properly paid. Neither the Owner nor Architect shall have an obligation to pay or to see to the payment of money to a Subcontractor, except as may otherwise be required by law.

§ 9.6.5 Contractor payments to material and equipment suppliers shall be treated in a manner similar to that provided in Sections 9.6.2, 9.6.3 and 9.6.4.

§ 9.6.6 A Certificate for Payment, a progress payment, or partial or entire use or occupancy of the Project by the Owner shall not constitute acceptance of Work not in accordance with the Contract Documents.

§ 9.6.7 Unless the Contractor provides the Owner with a payment bond in the full penal sum of the Contract Sum, payments received by the Contractor for Work properly performed by Subcontractors and suppliers shall be held by the Contractor for those Subcontractors or suppliers who performed Work or furnished materials, or both, under contract with the Contractor for which payment was made by the Owner. Nothing contained herein shall require money to be placed in a separate account and not commingled with money of the Contractor, shall create any fiduciary liability or tort liability on the part of the Contractor for breach of trust or shall entitle any person or entity to an award of punitive damages against the Contractor for breach of the requirements of this provision.

#### § 9.7 FAILURE OF PAYMENT

If the Architect does not issue a Certificate for Payment, through no fault of the Contractor, within seven days after receipt of the Contractor's Application for Payment, or if the Owner does not pay the Contractor within seven days after the date established in the Contract Documents the amount certified by the Architect or awarded by binding dispute resolution, then the Contractor may, upon seven additional days' written notice to the Owner and Architect, stop the Work until payment of the amount owing has been received. The Contract Time shall be extended appropriately and the Contract Sum shall be increased by the amount of the Contractor's reasonable costs of shut-down, delay and start-up, plus interest as provided for in the Contract Documents.

#### § 9.8 SUBSTANTIAL COMPLETION

§ 9.8.1 Substantial Completion is the stage in the progress of the Work when the Work or designated portion thereof is sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work for its intended use.

§ 9.8.2 When the Contractor considers that the Work, or a portion thereof which the Owner agrees to accept separately, is substantially complete, the Contractor shall prepare and submit to the Architect a comprehensive list of items to be completed or corrected prior to final payment. Failure to include an item on such list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents.

AIA Document A201<sup>®</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 25 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

§ 9.8.3 Upon receipt of the Contractor's list, the Architect will make an inspection to determine whether the Work or designated portion thereof is substantially complete. If the Architect's inspection discloses any item, whether or not included on the Contractor's list, which is not sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work or designated portion thereof for its intended use, the Contractor shall, before issuance of the Certificate of Substantial Completion, complete or correct such item upon notification by the Architect. In such case, the Contractor shall then submit a request for another inspection by the Architect to determine Substantial Completion.

§ 9.8.4 When the Work or designated portion thereof is substantially complete, the Architect will prepare a Certificate of Substantial Completion that shall establish the date of Substantial Completion, shall establish responsibilities of the Owner and Contractor for security, maintenance, heat, utilities, damage to the Work and insurance, and shall fix the time within which the Contractor shall finish all items on the list accompanying the Certificate. Warranties required by the Contract Documents shall commence on the date of Substantial Completion of the Work or designated portion thereof unless otherwise provided in the Certificate of Substantial Completion.

§ 9.8.5 The Certificate of Substantial Completion shall be submitted to the Owner and Contractor for their written acceptance of responsibilities assigned to them in such Certificate. Upon such acceptance and consent of surety, if any, the Owner shall make payment of retainage applying to such Work or designated portion thereof. Such payment shall be adjusted for Work that is incomplete or not in accordance with the requirements of the Contract Documents.

#### § 9.9 PARTIAL OCCUPANCY OR USE

§ 9.9.1 The Owner may occupy or use any completed or partially completed portion of the Work at any stage when such portion is designated by separate agreement with the Contractor, provided such occupancy or use is consented to by the insurer as required under Section 11.3.1.5 and authorized by public authorities having jurisdiction over the Project. Such partial occupancy or use may commence whether or not the portion is substantially complete, provided the Owner and Contractor have accepted in writing the responsibilities assigned to each of them for payments, retainage, if any, security, maintenance, heat, utilities, damage to the Work and insurance, and have agreed in writing concerning the period for correction of the Work and commencement of warranties required by the Contract Documents. When the Contractor considers a portion substantially complete, the Contractor shall prepare and submit a list to the Architect as provided under Section 9.8.2. Consent of the Contractor to partial occupancy or use shall not be unreasonably withheld. The stage of the progress of the Work shall be determined by written agreement between the Owner and Contractor or, if no agreement is reached, by decision of the Architect.

§ 9.9.2 Immediately prior to such partial occupancy or use, the Owner, Contractor and Architect shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work.

§ 9.9.3 Unless otherwise agreed upon, partial occupancy or use of a portion or portions of the Work shall not constitute acceptance of Work not complying with the requirements of the Contract Documents.

#### § 9.10 FINAL COMPLETION AND FINAL PAYMENT

§ 9.10.1 Upon receipt of the Contractor's written notice that the Work is ready for final inspection and acceptance and upon receipt of a final Application for Payment, the Architect will promptly make such inspection and, when the Architect finds the Work acceptable under the Contract Documents and the Contract fully performed, the Architect will promptly issue a final Certificate for Payment stating that to the best of the Architect's knowledge, information and belief, and on the basis of the Architect's on-site visits and inspections, the Work has been completed in accordance with terms and conditions of the Contract Documents and that the entire balance found to be due the Contractor and noted in the final Certificate is due and payable. The Architect's final Certificate for Payment will constitute a further representation that conditions listed in Section 9.10.2 as precedent to the Contractor's being entitled to final payment have been fulfilled.

§ 9.10.2 Neither final payment nor any remaining retained percentage shall become due until the Contractor submits to the Architect (1) an affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the Owner or the Owner's property might be responsible or encumbered (less amounts withheld by Owner) have been paid or otherwise satisfied, (2) a certificate evidencing that insurance required by the Contract Documents to remain in force after final payment is currently in effect and will not be canceled or allowed to expire until at least 30 days' prior written notice has been given to the Owner, (3) a written statement that the Contractor knows of no substantial reason that the insurance will not be renewable to cover the period required by the Contract

Init. 1

AIA Document A201<sup>®</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 26 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

Documents, (4) consent of surety, if any, to final payment and (5), if required by the Owner, other data establishing payment or satisfaction of obligations, such as receipts, releases and waivers of liens, claims, security interests or encumbrances arising out of the Contract, to the extent and in such form as may be designated by the Owner. If a Subcontractor refuses to furnish a release or waiver required by the Owner, the Contractor may furnish a bond satisfactory to the Owner to indemnify the Owner against such lien. If such lien remains unsatisfied after payments are made, the Contractor shall refund to the Owner all money that the Owner may be compelled to pay in discharging such lien, including all costs and reasonable attorneys' fees.

§ 9.10.3 If, after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Contractor or by issuance of Change Orders affecting final completion, and the Architect so confirms, the Owner shall, upon application by the Contractor and certification by the Architect, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed and accepted. If the remaining balance for Work not fully completed or corrected is less than retainage stipulated in the Contract Documents, and if bonds have been furnished, the written consent of surety to payment of the balance due for that portion of the Work fully completed and accepted shall be submitted by the Contractor to the Architect prior to certification of such payment. Such payment shall be made under terms and conditions governing final payment, except that it shall not constitute a waiver of claims.

§ 9.10.4 The making of final payment shall constitute a waiver of Claims by the Owner except those arising from

- .1 liens, Claims, security interests or encumbrances arising out of the Contract and unsettled;
- .2 failure of the Work to comply with the requirements of the Contract Documents; or
- .3 terms of special warranties required by the Contract Documents.

§ 9.10.5 Acceptance of final payment by the Contractor, a Subcontractor or material supplier shall constitute a waiver of claims by that payee except those previously made in writing and identified by that payee as unsettled at the time of final Application for Payment.

#### PROTECTION OF PERSONS AND PROPERTY **ARTICLE 10** § 10.1 SAFETY PRECAUTIONS AND PROGRAMS

The Contractor shall be responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the performance of the Contract.

#### § 10.2 SAFETY OF PERSONS AND PROPERTY

§ 10.2.1 The Contractor shall take reasonable precautions for safety of, and shall provide reasonable protection to prevent damage, injury or loss to

- employees on the Work and other persons who may be affected thereby; .1
- .2 the Work and materials and equipment to be incorporated therein, whether in storage on or off the site, under care, custody or control of the Contractor or the Contractor's Subcontractors or Sub-subcontractors; and
- .3 other property at the site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, structures and utilities not designated for removal, relocation or replacement in the course of construction.

§ 10.2.2 The Contractor shall comply with and give notices required by applicable laws, statutes, ordinances, codes, rules and regulations, and lawful orders of public authorities bearing on safety of persons or property or their protection from damage, injury or loss.

§ 10.2.3 The Contractor shall erect and maintain, as required by existing conditions and performance of the Contract, reasonable safeguards for safety and protection, including posting danger signs and other warnings against hazards, promulgating safety regulations and notifying owners and users of adjacent sites and utilities.

§ 10.2.4 When use or storage of explosives or other hazardous materials or equipment or unusual methods are necessary for execution of the Work, the Contractor shall exercise utmost care and carry on such activities under supervision of properly qualified personnel.

§ 10.2.5 The Contractor shall promptly remedy damage and loss (other than damage or loss insured under property insurance required by the Contract Documents) to property referred to in Sections 10.2.1.2 and 10.2.1.3 caused in

whole or in part by the Contractor, a Sub-subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under Sections 10.2.1.2 and 10.2.1.3, except damage or loss attributable to acts or omissions of the Owner or Architect or anyone directly or indirectly employed by either of them, or by anyone for whose acts either of them may be liable, and not attributable to the fault or negligence of the Contractor. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under Section 3.18.

§ 10.2.6 The Contractor shall designate a responsible member of the Contractor's organization at the site whose duty shall be the prevention of accidents. This person shall be the Contractor's superintendent unless otherwise designated by the Contractor in writing to the Owner and Architect.

§ 10.2.7 The Contractor shall not permit any part of the construction or site to be loaded so as to cause damage or create an unsafe condition.

#### § 10.2.8 INJURY OR DAMAGE TO PERSON OR PROPERTY

If either party suffers injury or damage to person or property because of an act or omission of the other party, or of others for whose acts such party is legally responsible, written notice of such injury or damage, whether or not insured, shall be given to the other party within a reasonable time not exceeding 21 days after discovery. The notice shall provide sufficient detail to enable the other party to investigate the matter.

#### § 10.3 HAZARDOUS MATERIALS

§ 10.3.1 The Contractor is responsible for compliance with any requirements included in the Contract Documents regarding hazardous materials. If the Contractor encounters a hazardous material or substance not addressed in the Contract Documents and if reasonable precautions will be inadequate to prevent foreseeable bodily injury or death to persons resulting from a material or substance, including but not limited to asbestos or polychlorinated biphenyl (PCB), encountered on the site by the Contractor, the Contractor shall, upon recognizing the condition, immediately stop Work in the affected area and report the condition to the Owner and Architect in writing.

§ 10.3.2 Upon receipt of the Contractor's written notice, the Owner shall obtain the services of a licensed laboratory to verify the presence or absence of the material or substance reported by the Contractor and, in the event such material or substance is found to be present, to cause it to be rendered harmless. Unless otherwise required by the Contract Documents, the Owner shall furnish in writing to the Contractor and Architect the names and qualifications of persons or entities who are to perform tests verifying the presence or absence of such material or substance or who are to perform the task of removal or safe containment of such material or substance. The Contractor and the Architect will promptly reply to the Owner in writing stating whether or not either has reasonable objection to the persons or entities proposed by the Owner. If either the Contractor or Architect has an objection to a person or entity proposed by the Owner, the Owner shall propose another to whom the Contractor and the Architect have no reasonable objection. When the material or substance has been rendered harmless, Work in the affected area shall resume upon written agreement of the Owner and Contractor. By Change Order, the Contract Time shall be extended appropriately and the Contract Sum shall be increased in the amount of the Contractor's reasonable additional costs of shut-down, delay and start-up.

§ 10.3.3 To the fullest extent permitted by law, the Owner shall indemnify and hold harmless the Contractor, Subcontractors, Architect, Architect's consultants and agents and employees of any of them from and against claims, damages, losses and expenses, including but not limited to attorneys' fees, arising out of or resulting from performance of the Work in the affected area if in fact the material or substance presents the risk of bodily injury or death as described in Section 10.3.1 and has not been rendered harmless, provided that such claim, damage, loss or expense is attributable to bodily injury, sickness, disease or death, or to injury to or destruction of tangible property (other than the Work itself), except to the extent that such damage, loss or expense is due to the fault or negligence of the party seeking indemnity.

§ 10.3.4 The Owner shall not be responsible under this Section 10.3 for materials or substances the Contractor brings to the site unless such materials or substances are required by the Contract Documents. The Owner shall be responsible for materials or substances required by the Contract Documents, except to the extent of the Contractor's fault or negligence in the use and handling of such materials or substances.

AIA Document A201® - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 28 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

§ 10.3.5 The Contractor shall indemnify the Owner for the cost and expense the Owner incurs (1) for remediation of a material or substance the Contractor brings to the site and negligently handles, or (2) where the Contractor fails to perform its obligations under Section 10.3.1, except to the extent that the cost and expense are due to the Owner's fault or negligence.

§ 10.3.6 If, without negligence on the part of the Contractor, the Contractor is held liable by a government agency for the cost of remediation of a hazardous material or substance solely by reason of performing Work as required by the Contract Documents, the Owner shall indemnify the Contractor for all cost and expense thereby incurred.

#### § 10.4 EMERGENCIES

Init.

1

In an emergency affecting safety of persons or property, the Contractor shall act, at the Contractor's discretion, to prevent threatened damage, injury or loss. Additional compensation or extension of time claimed by the Contractor on account of an emergency shall be determined as provided in Article 15 and Article 7.

## ARTICLE 11 INSURANCE AND BONDS

#### § 11.1 CONTRACTOR'S LIABILITY INSURANCE

§ 11.1.1 The Contractor shall purchase from and maintain in a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located such insurance as will protect the Contractor from claims set forth below which may arise out of or result from the Contractor's operations and completed operations under the Contract and for which the Contractor may be legally liable, whether such operations be by the Contractor or by a Subcontractor or by anyone directly or indirectly employed by any of them, or by anyone for whose acts any of them may be liable:

- .1 Claims under workers' compensation, disability benefit and other similar employee benefit acts that are applicable to the Work to be performed;
- .2 Claims for damages because of bodily injury, occupational sickness or disease, or death of the Contractor's employees;
- Claims for damages because of bodily injury, sickness or disease, or death of any person other than the .3 Contractor's employees;
- .4 Claims for damages insured by usual personal injury liability coverage;
- .5 Claims for damages, other than to the Work itself, because of injury to or destruction of tangible property, including loss of use resulting therefrom;
- .6 Claims for damages because of bodily injury, death of a person or property damage arising out of ownership, maintenance or use of a motor vehicle;
- .7 Claims for bodily injury or property damage arising out of completed operations; and
- .8 Claims involving contractual liability insurance applicable to the Contractor's obligations under Section 3.18.

§ 11.1.2 The insurance required by Section 11.1.1 shall be written for not less than limits of liability specified in the Contract Documents or required by law, whichever coverage is greater. Coverages, whether written on an occurrence or claims-made basis, shall be maintained without interruption from the date of commencement of the Work until the date of final payment and termination of any coverage required to be maintained after final payment, and, with respect to the Contractor's completed operations coverage, until the expiration of the period for correction of Work or for such other period for maintenance of completed operations coverage as specified in the Contract Documents.

§ 11.1.3 Certificates of insurance acceptable to the Owner shall be filed with the Owner prior to commencement of the Work and thereafter upon renewal or replacement of each required policy of insurance. These certificates and the insurance policies required by this Section 11.1 shall contain a provision that coverages afforded under the policies will not be canceled or allowed to expire until at least 30 days' prior written notice has been given to the Owner. An additional certificate evidencing continuation of liability coverage, including coverage for completed operations, shall be submitted with the final Application for Payment as required by Section 9.10.2 and thereafter upon renewal or replacement of such coverage until the expiration of the time required by Section 11.1.2. Information concerning reduction of coverage on account of revised limits or claims paid under the General Aggregate, or both, shall be furnished by the Contractor with reasonable promptness.

§ 11.1.4 The Contractor shall cause the commercial liability coverage required by the Contract Documents to include (1) the Owner, the Architect and the Architect's consultants as additional insureds for claims caused in whole or in part by the Contractor's negligent acts or omissions during the Contractor's operations; and (2) the Owner as an additional

insured for claims caused in whole or in part by the Contractor's negligent acts or omissions during the Contractor's completed operations.

#### § 11.2 OWNER'S LIABILITY INSURANCE

The Owner shall be responsible for purchasing and maintaining the Owner's usual liability insurance.

#### § 11.3 PROPERTY INSURANCE

§ 11.3.1 Unless otherwise provided, the Owner shall purchase and maintain, in a company or companies lawfully authorized to do business in the jurisdiction in which the Project is located, property insurance written on a builder's risk "all-risk" or equivalent policy form in the amount of the initial Contract Sum, plus value of subsequent Contract Modifications and cost of materials supplied or installed by others, comprising total value for the entire Project at the site on a replacement cost basis without optional deductibles. Such property insurance shall be maintained, unless otherwise provided in the Contract Documents or otherwise agreed in writing by all persons and entities who are beneficiaries of such insurance, until final payment has been made as provided in Section 9.10 or until no person or entity other than the Owner has an insurable interest in the property required by this Section 11.3 to be covered, whichever is later. This insurance shall include interests of the Owner, the Contractor, Subcontractors and Sub-subcontractors in the Project.

§ 11.3.1.1 Property insurance shall be on an "all-risk" or equivalent policy form and shall include, without limitation, insurance against the perils of fire (with extended coverage) and physical loss or damage including, without duplication of coverage, theft, vandalism, malicious mischief, collapse, earthquake, flood, windstorm, falsework, testing and startup, temporary buildings and debris removal including demolition occasioned by enforcement of any applicable legal requirements, and shall cover reasonable compensation for Architect's and Contractor's services and expenses required as a result of such insured loss.

§ 11.3.1.2 If the Owner does not intend to purchase such property insurance required by the Contract and with all of the coverages in the amount described above, the Owner shall so inform the Contractor in writing prior to commencement of the Work. The Contractor may then effect insurance that will protect the interests of the Contractor, Subcontractors and Sub-subcontractors in the Work, and by appropriate Change Order the cost thereof shall be charged to the Owner. If the Contractor is damaged by the failure or neglect of the Owner to purchase or maintain insurance as described above, without so notifying the Contractor in writing, then the Owner shall bear all reasonable costs properly attributable thereto.

§ 11.3.1.3 If the property insurance requires deductibles, the Owner shall pay costs not covered because of such deductibles.

§ 11.3.1.4 This property insurance shall cover portions of the Work stored off the site, and also portions of the Work in transit.

§ 11.3.1.5 Partial occupancy or use in accordance with Section 9.9 shall not commence until the insurance company or companies providing property insurance have consented to such partial occupancy or use by endorsement or otherwise. The Owner and the Contractor shall take reasonable steps to obtain consent of the insurance company or companies and shall, without mutual written consent, take no action with respect to partial occupancy or use that would cause cancellation, lapse or reduction of insurance.

#### § 11.3.2 BOILER AND MACHINERY INSURANCE

The Owner shall purchase and maintain boiler and machinery insurance required by the Contract Documents or by law, which shall specifically cover such insured objects during installation and until final acceptance by the Owner; this insurance shall include interests of the Owner, Contractor, Subcontractors and Sub-subcontractors in the Work, and the Owner and Contractor shall be named insureds.

#### § 11.3.3 LOSS OF USE INSURANCE

The Owner, at the Owner's option, may purchase and maintain such insurance as will insure the Owner against loss of use of the Owner's property due to fire or other hazards, however caused. The Owner waives all rights of action against the Contractor for loss of use of the Owner's property, including consequential losses due to fire or other hazards however caused.

AIA Document A201<sup>®</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 30 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

§ 11.3.4 If the Contractor requests in writing that insurance for risks other than those described herein or other special causes of loss be included in the property insurance policy, the Owner shall, if possible, include such insurance, and the cost thereof shall be charged to the Contractor by appropriate Change Order.

§ 11.3.5 If during the Project construction period the Owner insures properties, real or personal or both, at or adjacent to the site by property insurance under policies separate from those insuring the Project, or if after final payment property insurance is to be provided on the completed Project through a policy or policies other than those insuring the Project during the construction period, the Owner shall waive all rights in accordance with the terms of Section 11.3.7 for damages caused by fire or other causes of loss covered by this separate property insurance. All separate policies shall provide this waiver of subrogation by endorsement or otherwise.

§ 11.3.6 Before an exposure to loss may occur, the Owner shall file with the Contractor a copy of each policy that includes insurance coverages required by this Section 11.3. Each policy shall contain all generally applicable conditions, definitions, exclusions and endorsements related to this Project. Each policy shall contain a provision that the policy will not be canceled or allowed to expire, and that its limits will not be reduced, until at least 30 days' prior written notice has been given to the Contractor.

# § 11.3.7 WAIVERS OF SUBROGATION

The Owner and Contractor waive all rights against (1) each other and any of their subcontractors, sub-subcontractors, agents and employees, each of the other, and (2) the Architect, Architect's consultants, separate contractors described in Article 6, if any, and any of their subcontractors, sub-subcontractors, agents and employees, for damages caused by fire or other causes of loss to the extent covered by property insurance obtained pursuant to this Section 11.3 or other property insurance applicable to the Work, except such rights as they have to proceeds of such insurance held by the Owner as fiduciary. The Owner or Contractor, as appropriate, shall require of the Architect, Architect's consultants, separate contractors described in Article 6, if any, and the subcontractors, sub-subcontractors, agents and employees of any of them, by appropriate agreements, written where legally required for validity, similar waivers each in favor of other parties enumerated herein. The policies shall provide such waivers of subrogation by endorsement or otherwise. A waiver of subrogation shall be effective as to a person or entity even though that person or entity would otherwise have a duty of indemnification, contractual or otherwise, did not pay the insurance premium directly or indirectly, and whether or not the person or entity had an insurable interest in the property damaged.

§ 11.3.8 A loss insured under the Owner's property insurance shall be adjusted by the Owner as fiduciary and made payable to the Owner as fiduciary for the insureds, as their interests may appear, subject to requirements of any applicable mortgagee clause and of Section 11.3.10. The Contractor shall pay Subcontractors their just shares of insurance proceeds received by the Contractor, and by appropriate agreements, written where legally required for validity, shall require Subcontractors to make payments to their Sub-subcontractors in similar manner.

§ 11.3.9 If required in writing by a party in interest, the Owner as fiduciary shall, upon occurrence of an insured loss, give bond for proper performance of the Owner's duties. The cost of required bonds shall be charged against proceeds received as fiduciary. The Owner shall deposit in a separate account proceeds so received, which the Owner shall distribute in accordance with such agreement as the parties in interest may reach, or as determined in accordance with the method of binding dispute resolution selected in the Agreement between the Owner and Contractor. If after such loss no other special agreement is made and unless the Owner terminates the Contract for convenience, replacement of damaged property shall be performed by the Contractor after notification of a Change in the Work in accordance with Article 7.

§ 11.3.10 The Owner as fiduciary shall have power to adjust and settle a loss with insurers unless one of the parties in interest shall object in writing within five days after occurrence of loss to the Owner's exercise of this power; if such objection is made, the dispute shall be resolved in the manner selected by the Owner and Contractor as the method of binding dispute resolution in the Agreement. If the Owner and Contractor have selected arbitration as the method of binding dispute resolution, the Owner as fiduciary shall make settlement with insurers or, in the case of a dispute over distribution of insurance proceeds, in accordance with the directions of the arbitrators.

# § 11.4 PERFORMANCE BOND AND PAYMENT BOND

§ 11.4.1 The Owner shall have the right to require the Contractor to furnish bonds covering faithful performance of the Contract and payment of obligations arising thereunder as stipulated in bidding requirements or specifically required in the Contract Documents on the date of execution of the Contract.

Init. 1

AIA Document A201<sup>®</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 31 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

§ 11.4.2 Upon the request of any person or entity appearing to be a potential beneficiary of bonds covering payment of obligations arising under the Contract, the Contractor shall promptly furnish a copy of the bonds or shall authorize a copy to be furnished.

#### ARTICLE 12 UNCOVERING AND CORRECTION OF WORK § 12.1 UNCOVERING OF WORK

§ 12.1.1 If a portion of the Work is covered contrary to the Architect's request or to requirements specifically expressed in the Contract Documents, it must, if requested in writing by the Architect, be uncovered for the Architect's examination and be replaced at the Contractor's expense without change in the Contract Time.

§ 12.1.2 If a portion of the Work has been covered that the Architect has not specifically requested to examine prior to its being covered, the Architect may request to see such Work and it shall be uncovered by the Contractor. If such Work is in accordance with the Contract Documents, costs of uncovering and replacement shall, by appropriate Change Order, be at the Owner's expense. If such Work is not in accordance with the Contract Documents, such costs and the cost of correction shall be at the Contractor's expense unless the condition was caused by the Owner or a separate contractor in which event the Owner shall be responsible for payment of such costs.

# § 12.2 CORRECTION OF WORK

# § 12.2.1 BEFORE OR AFTER SUBSTANTIAL COMPLETION

The Contractor shall promptly correct Work rejected by the Architect or failing to conform to the requirements of the Contract Documents, whether discovered before or after Substantial Completion and whether or not fabricated, installed or completed. Costs of correcting such rejected Work, including additional testing and inspections, the cost of uncovering and replacement, and compensation for the Architect's services and expenses made necessary thereby, shall be at the Contractor's expense.

# § 12.2.2 AFTER SUBSTANTIAL COMPLETION

§ 12.2.2.1 In addition to the Contractor's obligations under Section 3.5, if, within one year after the date of Substantial Completion of the Work or designated portion thereof or after the date for commencement of warranties established under Section 9.9.1, or by terms of an applicable special warranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, the Contractor shall correct it promptly after receipt of written notice from the Owner to do so unless the Owner has previously given the Contractor a written acceptance of such condition. The Owner shall give such notice promptly after discovery of the condition. During the one-year period for correction of Work, if the Owner fails to notify the Contractor and give the Contractor an opportunity to make the correction, the Owner waives the rights to require correction by the Contractor and to make a claim for breach of warranty. If the Contractor fails to correct nonconforming Work within a reasonable time during that period after receipt of notice from the Owner or Architect, the Owner may correct it in accordance with Section 2.4.

§ 12.2.2.2 The one-year period for correction of Work shall be extended with respect to portions of Work first performed after Substantial Completion by the period of time between Substantial Completion and the actual completion of that portion of the Work.

§ 12.2.2.3 The one-year period for correction of Work shall not be extended by corrective Work performed by the Contractor pursuant to this Section 12.2.

§ 12.2.3 The Contractor shall remove from the site portions of the Work that are not in accordance with the requirements of the Contract Documents and are neither corrected by the Contractor nor accepted by the Owner.

§ 12.2.4 The Contractor shall bear the cost of correcting destroyed or damaged construction, whether completed or partially completed, of the Owner or separate contractors caused by the Contractor's correction or removal of Work that is not in accordance with the requirements of the Contract Documents.

§ 12.2.5 Nothing contained in this Section 12.2 shall be construed to establish a period of limitation with respect to other obligations the Contractor has under the Contract Documents. Establishment of the one-year period for correction of Work as described in Section 12.2.2 relates only to the specific obligation of the Contractor to correct the Work, and has no relationship to the time within which the obligation to comply with the Contract Documents may be

sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to the Contractor's obligations other than specifically to correct the Work.

# § 12.3 ACCEPTANCE OF NONCONFORMING WORK

If the Owner prefers to accept Work that is not in accordance with the requirements of the Contract Documents, the Owner may do so instead of requiring its removal and correction, in which case the Contract Sum will be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made.

#### ARTICLE 13 MISCELLANEOUS PROVISIONS § 13.1 GOVERNING LAW

The Contract shall be governed by the law of the place where the Project is located except that, if the parties have selected arbitration as the method of binding dispute resolution, the Federal Arbitration Act shall govern Section 15.4.

# § 13.2 SUCCESSORS AND ASSIGNS

§ 13.2.1 The Owner and Contractor respectively bind themselves, their partners, successors, assigns and legal representatives to covenants, agreements and obligations contained in the Contract Documents. Except as provided in Section 13.2.2, neither party to the Contract shall assign the Contract as a whole without written consent of the other. If either party attempts to make such an assignment without such consent, that party shall nevertheless remain legally responsible for all obligations under the Contract.

§ 13.2.2 The Owner may, without consent of the Contractor, assign the Contract to a lender providing construction financing for the Project, if the lender assumes the Owner's rights and obligations under the Contract Documents. The Contractor shall execute all consents reasonably required to facilitate such assignment.

# § 13.3 WRITTEN NOTICE

Written notice shall be deemed to have been duly served if delivered in person to the individual, to a member of the firm or entity, or to an officer of the corporation for which it was intended; or if delivered at, or sent by registered or certified mail or by courier service providing proof of delivery to, the last business address known to the party giving notice.

# § 13.4 RIGHTS AND REMEDIES

§ 13.4.1 Duties and obligations imposed by the Contract Documents and rights and remedies available thereunder shall be in addition to and not a limitation of duties, obligations, rights and remedies otherwise imposed or available by law.

§ 13.4.2 No action or failure to act by the Owner, Architect or Contractor shall constitute a waiver of a right or duty afforded them under the Contract, nor shall such action or failure to act constitute approval of or acquiescence in a breach there under, except as may be specifically agreed in writing.

# § 13.5 TESTS AND INSPECTIONS

§ 13.5.1 Tests, inspections and approvals of portions of the Work shall be made as required by the Contract Documents and by applicable laws, statutes, ordinances, codes, rules and regulations or lawful orders of public authorities. Unless otherwise provided, the Contractor shall make arrangements for such tests, inspections and approvals with an independent testing laboratory or entity acceptable to the Owner, or with the appropriate public authority, and shall bear all related costs of tests, inspections and approvals. The Contractor shall give the Architect timely notice of when and where tests and inspections are to be made so that the Architect may be present for such procedures. The Owner shall bear costs of (1) tests, inspections or approvals that do not become requirements until after bids are received or negotiations concluded, and (2) tests, inspections or approvals where building codes or applicable laws or regulations prohibit the Owner from delegating their cost to the Contractor.

§ 13.5.2 If the Architect, Owner or public authorities having jurisdiction determine that portions of the Work require additional testing, inspection or approval not included under Section 13.5.1, the Architect will, upon written authorization from the Owner, instruct the Contractor to make arrangements for such additional testing, inspection or approval by an entity acceptable to the Owner, and the Contractor shall give timely notice to the Architect of when and where tests and inspections are to be made so that the Architect may be present for such procedures. Such costs, except as provided in Section 13.5.3, shall be at the Owner's expense.

Init. 1

AIA Document A201<sup>®</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 33 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

§ 13.5.3 If such procedures for testing, inspection or approval under Sections 13.5.1 and 13.5.2 reveal failure of the portions of the Work to comply with requirements established by the Contract Documents, all costs made necessary by such failure including those of repeated procedures and compensation for the Architect's services and expenses shall be at the Contractor's expense.

§ 13.5.4 Required certificates of testing, inspection or approval shall, unless otherwise required by the Contract Documents, be secured by the Contractor and promptly delivered to the Architect.

§ 13.5.5 If the Architect is to observe tests, inspections or approvals required by the Contract Documents, the Architect will do so promptly and, where practicable, at the normal place of testing.

§ 13.5.6 Tests or inspections conducted pursuant to the Contract Documents shall be made promptly to avoid unreasonable delay in the Work.

# § 13.6 INTEREST

Init.

1

Payments due and unpaid under the Contract Documents shall bear interest from the date payment is due at such rate as the parties may agree upon in writing or, in the absence thereof, at the legal rate prevailing from time to time at the place where the Project is located.

#### § 13.7 TIME LIMITS ON CLAIMS

The Owner and Contractor shall commence all claims and causes of action, whether in contract, tort, breach of warranty or otherwise, against the other arising out of or related to the Contract in accordance with the requirements of the final dispute resolution method selected in the Agreement within the time period specified by applicable law, but in any case not more than 10 years after the date of Substantial Completion of the Work. The Owner and Contractor waive all claims and causes of action not commenced in accordance with this Section 13.7.

#### ARTICLE 14 TERMINATION OR SUSPENSION OF THE CONTRACT § 14.1 TERMINATION BY THE CONTRACTOR

§ 14.1.1 The Contractor may terminate the Contract if the Work is stopped for a period of 30 consecutive days through no act or fault of the Contractor or a Subcontractor, Sub-subcontractor or their agents or employees or any other persons or entities performing portions of the Work under direct or indirect contract with the Contractor, for any of the following reasons:

- .1 Issuance of an order of a court or other public authority having jurisdiction that requires all Work to be stopped;
- .2 An act of government, such as a declaration of national emergency that requires all Work to be stopped;
- .3 Because the Architect has not issued a Certificate for Payment and has not notified the Contractor of the reason for withholding certification as provided in Section 9.4.1, or because the Owner has not made payment on a Certificate for Payment within the time stated in the Contract Documents; or
- .4 The Owner has failed to furnish to the Contractor promptly, upon the Contractor's request, reasonable evidence as required by Section 2.2.1.

§ 14.1.2 The Contractor may terminate the Contract if, through no act or fault of the Contractor or a Subcontractor, Sub-subcontractor or their agents or employees or any other persons or entities performing portions of the Work under direct or indirect contract with the Contractor, repeated suspensions, delays or interruptions of the entire Work by the Owner as described in Section 14.3 constitute in the aggregate more than 100 percent of the total number of days scheduled for completion, or 120 days in any 365-day period, whichever is less.

§ 14.1.3 If one of the reasons described in Section 14.1.1 or 14.1.2 exists, the Contractor may, upon seven days' written notice to the Owner and Architect, terminate the Contract and recover from the Owner payment for Work executed, including reasonable overhead and profit, costs incurred by reason of such termination, and damages.

§ 14.1.4 If the Work is stopped for a period of 60 consecutive days through no act or fault of the Contractor or a Subcontractor or their agents or employees or any other persons performing portions of the Work under contract with the Contractor because the Owner has repeatedly failed to fulfill the Owner's obligations under the Contract Documents with respect to matters important to the progress of the Work, the Contractor may, upon seven additional days' written notice to the Owner and the Architect, terminate the Contract and recover from the Owner as provided in Section 14.1.3.

AIA Document A201<sup>®</sup> - 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 34 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

# § 14.2 TERMINATION BY THE OWNER FOR CAUSE

§ 14.2.1 The Owner may terminate the Contract if the Contractor

- .1 repeatedly refuses or fails to supply enough properly skilled workers or proper materials;
- .2 fails to make payment to Subcontractors for materials or labor in accordance with the respective agreements between the Contractor and the Subcontractors;
- .3 repeatedly disregards applicable laws, statutes, ordinances, codes, rules and regulations, or lawful orders of a public authority; or
- .4 otherwise is guilty of substantial breach of a provision of the Contract Documents.

§ 14.2.2 When any of the above reasons exist, the Owner, upon certification by the Initial Decision Maker that sufficient cause exists to justify such action, may without prejudice to any other rights or remedies of the Owner and after giving the Contractor and the Contractor's surety, if any, seven days' written notice, terminate employment of the Contractor and may, subject to any prior rights of the surety:

- Exclude the Contractor from the site and take possession of all materials, equipment, tools, and .1 construction equipment and machinery thereon owned by the Contractor;
- .2 Accept assignment of subcontracts pursuant to Section 5.4; and
- .3 Finish the Work by whatever reasonable method the Owner may deem expedient. Upon written request of the Contractor, the Owner shall furnish to the Contractor a detailed accounting of the costs incurred by the Owner in finishing the Work.

§ 14.2.3 When the Owner terminates the Contract for one of the reasons stated in Section 14.2.1, the Contractor shall not be entitled to receive further payment until the Work is finished.

§ 14.2.4 If the unpaid balance of the Contract Sum exceeds costs of finishing the Work, including compensation for the Architect's services and expenses made necessary thereby, and other damages incurred by the Owner and not expressly waived, such excess shall be paid to the Contractor. If such costs and damages exceed the unpaid balance, the Contractor shall pay the difference to the Owner. The amount to be paid to the Contractor or Owner, as the case may be, shall be certified by the Initial Decision Maker, upon application, and this obligation for payment shall survive termination of the Contract.

# § 14.3 SUSPENSION BY THE OWNER FOR CONVENIENCE

§ 14.3.1 The Owner may, without cause, order the Contractor in writing to suspend, delay or interrupt the Work in whole or in part for such period of time as the Owner may determine.

§ 14.3.2 The Contract Sum and Contract Time shall be adjusted for increases in the cost and time caused by suspension, delay or interruption as described in Section 14.3.1. Adjustment of the Contract Sum shall include profit. No adjustment shall be made to the extent

- .1 that performance is, was or would have been so suspended, delayed or interrupted by another cause for which the Contractor is responsible; or
- .2 that an equitable adjustment is made or denied under another provision of the Contract.

# § 14.4 TERMINATION BY THE OWNER FOR CONVENIENCE

§ 14.4.1 The Owner may, at any time, terminate the Contract for the Owner's convenience and without cause.

§ 14.4.2 Upon receipt of written notice from the Owner of such termination for the Owner's convenience, the Contractor shall

- .1 cease operations as directed by the Owner in the notice;
- .2 take actions necessary, or that the Owner may direct, for the protection and preservation of the Work; and
- .3 except for Work directed to be performed prior to the effective date of termination stated in the notice, terminate all existing subcontracts and purchase orders and enter into no further subcontracts and purchase orders.

§ 14.4.3 In case of such termination for the Owner's convenience, the Contractor shall be entitled to receive payment for Work executed, and costs incurred by reason of such termination, along with reasonable overhead and profit on the Work not executed.

AIA Document A201<sup>®</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 35 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

Init. 1

# ARTICLE 15 CLAIMS AND DISPUTES

# § 15.1 CLAIMS

# § 15.1.1 DEFINITION

A Claim is a demand or assertion by one of the parties seeking, as a matter of right, payment of money, or other relief with respect to the terms of the Contract. The term "Claim" also includes other disputes and matters in question between the Owner and Contractor arising out of or relating to the Contract. The responsibility to substantiate Claims shall rest with the party making the Claim.

# § 15.1.2 NOTICE OF CLAIMS

Claims by either the Owner or Contractor must be initiated by written notice to the other party and to the Initial Decision Maker with a copy sent to the Architect, if the Architect is not serving as the Initial Decision Maker. Claims by either party must be initiated within 21 days after occurrence of the event giving rise to such Claim or within 21 days after the claimant first recognizes the condition giving rise to the Claim, whichever is later.

# § 15.1.3 CONTINUING CONTRACT PERFORMANCE

Pending final resolution of a Claim, except as otherwise agreed in writing or as provided in Section 9.7 and Article 14, the Contractor shall proceed diligently with performance of the Contract and the Owner shall continue to make payments in accordance with the Contract Documents. The Architect will prepare Change Orders and issue Certificates for Payment in accordance with the decisions of the Initial Decision Maker.

# § 15.1.4 CLAIMS FOR ADDITIONAL COST

If the Contractor wishes to make a Claim for an increase in the Contract Sum, written notice as provided herein shall be given before proceeding to execute the Work. Prior notice is not required for Claims relating to an emergency endangering life or property arising under Section 10.4.

# § 15.1.5 CLAIMS FOR ADDITIONAL TIME

§ 15.1.5.1 If the Contractor wishes to make a Claim for an increase in the Contract Time, written notice as provided herein shall be given. The Contractor's Claim shall include an estimate of cost and of probable effect of delay on progress of the Work. In the case of a continuing delay, only one Claim is necessary.

§ 15.1.5.2 If adverse weather conditions are the basis for a Claim for additional time, such Claim shall be documented by data substantiating that weather conditions were abnormal for the period of time, could not have been reasonably anticipated and had an adverse effect on the scheduled construction.

# § 15.1.6 CLAIMS FOR CONSEQUENTIAL DAMAGES

The Contractor and Owner waive Claims against each other for consequential damages arising out of or relating to this Contract. This mutual waiver includes

- damages incurred by the Owner for rental expenses, for losses of use, income, profit, financing, .1 business and reputation, and for loss of management or employee productivity or of the services of such persons; and
- .2 damages incurred by the Contractor for principal office expenses including the compensation of personnel stationed there, for losses of financing, business and reputation, and for loss of profit except anticipated profit arising directly from the Work.

This mutual waiver is applicable, without limitation, to all consequential damages due to either party's termination in accordance with Article 14. Nothing contained in this Section 15.1.6 shall be deemed to preclude an award of liquidated damages, when applicable, in accordance with the requirements of the Contract Documents.

# § 15.2 INITIAL DECISION

Init.

1

§ 15.2.1 Claims, excluding those arising under Sections 10.3, 10.4, 11.3.9, and 11.3.10, shall be referred to the Initial Decision Maker for initial decision. The Architect will serve as the Initial Decision Maker, unless otherwise indicated in the Agreement. Except for those Claims excluded by this Section 15.2.1, an initial decision shall be required as a condition precedent to mediation of any Claim arising prior to the date final payment is due, unless 30 days have passed after the Claim has been referred to the Initial Decision Maker with no decision having been rendered. Unless the Initial Decision Maker and all affected parties agree, the Initial Decision Maker will not decide disputes between the Contractor and persons or entities other than the Owner.

§ 15.2.2 The Initial Decision Maker will review Claims and within ten days of the receipt of a Claim take one or more of the following actions: (1) request additional supporting data from the claimant or a response with supporting data from the other party, (2) reject the Claim in whole or in part, (3) approve the Claim, (4) suggest a compromise, or (5) advise the parties that the Initial Decision Maker is unable to resolve the Claim if the Initial Decision Maker lacks sufficient information to evaluate the merits of the Claim or if the Initial Decision Maker concludes that, in the Initial Decision Maker's sole discretion, it would be inappropriate for the Initial Decision Maker to resolve the Claim.

§ 15.2.3 In evaluating Claims, the Initial Decision Maker may, but shall not be obligated to, consult with or seek information from either party or from persons with special knowledge or expertise who may assist the Initial Decision Maker in rendering a decision. The Initial Decision Maker may request the Owner to authorize retention of such persons at the Owner's expense.

§ 15.2.4 If the Initial Decision Maker requests a party to provide a response to a Claim or to furnish additional supporting data, such party shall respond, within ten days after receipt of such request, and shall either (1) provide a response on the requested supporting data, (2) advise the Initial Decision Maker when the response or supporting data will be furnished or (3) advise the Initial Decision Maker that no supporting data will be furnished. Upon receipt of the response or supporting data, if any, the Initial Decision Maker will either reject or approve the Claim in whole or in part.

§ 15.2.5 The Initial Decision Maker will render an initial decision approving or rejecting the Claim, or indicating that the Initial Decision Maker is unable to resolve the Claim. This initial decision shall (1) be in writing; (2) state the reasons therefor; and (3) notify the parties and the Architect, if the Architect is not serving as the Initial Decision Maker, of any change in the Contract Sum or Contract Time or both. The initial decision shall be final and binding on the parties but subject to mediation and, if the parties fail to resolve their dispute through mediation, to binding dispute resolution.

§ 15.2.6 Either party may file for mediation of an initial decision at any time, subject to the terms of Section 15.2.6.1.

§ 15.2.6.1 Either party may, within 30 days from the date of an initial decision, demand in writing that the other party file for mediation within 60 days of the initial decision. If such a demand is made and the party receiving the demand fails to file for mediation within the time required, then both parties waive their rights to mediate or pursue binding dispute resolution proceedings with respect to the initial decision.

§ 15.2.7 In the event of a Claim against the Contractor, the Owner may, but is not obligated to, notify the surety, if any, of the nature and amount of the Claim. If the Claim relates to a possibility of a Contractor's default, the Owner may, but is not obligated to, notify the surety and request the surety's assistance in resolving the controversy.

§ 15.2.8 If a Claim relates to or is the subject of a mechanic's lien, the party asserting such Claim may proceed in accordance with applicable law to comply with the lien notice or filing deadlines.

# § 15.3 MEDIATION

§ 15.3.1 Claims, disputes, or other matters in controversy arising out of or related to the Contract except those waived as provided for in Sections 9.10.4, 9.10.5, and 15.1.6 shall be subject to mediation as a condition precedent to binding dispute resolution.

§ 15.3.2 The parties shall endeavor to resolve their Claims by mediation which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Mediation Procedures in effect on the date of the Agreement. A request for mediation shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the mediation. The request may be made concurrently with the filing of binding dispute resolution proceedings but, in such event, mediation shall proceed in advance of binding dispute resolution proceedings, which shall be stayed pending mediation for a period of 60 days from the date of filing, unless stayed for a longer period by agreement of the parties or court order. If an arbitration is stayed pursuant to this Section 15.3.2, the parties may nonetheless proceed to the selection of the arbitrator(s) and agree upon a schedule for later proceedings.

Init. 1

AIA Document A201<sup>®</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered 37 trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

§ 15.3.3 The parties shall share the mediator's fee and any filing fees equally. The mediation shall be held in the place where the Project is located, unless another location is mutually agreed upon. Agreements reached in mediation shall be enforceable as settlement agreements in any court having jurisdiction thereof.

# § 15.4 ARBITRATION

§ 15.4.1 If the parties have selected arbitration as the method for binding dispute resolution in the Agreement, any Claim subject to, but not resolved by, mediation shall be subject to arbitration which, unless the parties mutually agree otherwise, shall be administered by the American Arbitration Association in accordance with its Construction Industry Arbitration Rules in effect on the date of the Agreement. A demand for arbitration shall be made in writing, delivered to the other party to the Contract, and filed with the person or entity administering the arbitration. The party filing a notice of demand for arbitration must assert in the demand all Claims then known to that party on which arbitration is permitted to be demanded.

§ 15.4.1.1 A demand for arbitration shall be made no earlier than concurrently with the filing of a request for mediation, but in no event shall it be made after the date when the institution of legal or equitable proceedings based on the Claim would be barred by the applicable statute of limitations. For statute of limitations purposes, receipt of a written demand for arbitration by the person or entity administering the arbitration shall constitute the institution of legal or equitable proceedings based on the Claim.

§ 15.4.2 The award rendered by the arbitrator or arbitrators shall be final, and judgment may be entered upon it in accordance with applicable law in any court having jurisdiction thereof.

§ 15.4.3 The foregoing agreement to arbitrate and other agreements to arbitrate with an additional person or entity duly consented to by parties to the Agreement shall be specifically enforceable under applicable law in any court having jurisdiction thereof.

# § 15.4.4 CONSOLIDATION OR JOINDER

§ 15.4.4.1 Either party, at its sole discretion, may consolidate an arbitration conducted under this Agreement with any other arbitration to which it is a party provided that (1) the arbitration agreement governing the other arbitration permits consolidation, (2) the arbitrations to be consolidated substantially involve common questions of law or fact, and (3) the arbitrations employ materially similar procedural rules and methods for selecting arbitrator(s).

§ 15.4.4.2 Either party, at its sole discretion, may include by joinder persons or entities substantially involved in a common question of law or fact whose presence is required if complete relief is to be accorded in arbitration, provided that the party sought to be joined consents in writing to such joinder. Consent to arbitration involving an additional person or entity shall not constitute consent to arbitration of any claim, dispute or other matter in question not described in the written consent.

§ 15.4.4.3 The Owner and Contractor grant to any person or entity made a party to an arbitration conducted under this Section 15.4, whether by joinder or consolidation, the same rights of joinder and consolidation as the Owner and Contractor under this Agreement.

# Additions and Deletions Report for

AIA<sup>®</sup> Document A201<sup>®</sup> – 2007

This Additions and Deletions Report, as defined on page 1 of the associated document, reproduces below all text the author has added to the standard form AIA document in order to complete it, as well as any text the author may have added to or deleted from the original AIA text. Added text is shown underlined. Deleted text is indicated with a horizontal line through the original AIA text.

Note: This Additions and Deletions Report is provided for information purposes only and is not incorporated into or constitute any part of the associated AIA document. This Additions and Deletions Report and its associated document were generated simultaneously by AIA software at 13:03:09 ET on 09/02/2020.

# PAGE 1

Multi-Campus Building Envelope Repairs (Phase 4) 3904 Avenue T. Galveston, TX 77550.

. . .

Galveston Independent School District 3904 Avenue T. Galveston, TX 77550.

...

**BEAM Professionals** 11 Greenway Plaza, 22nd Floor Houston, TX 77046

Additions and Deletions Report for AIA Document A201<sup>®</sup> – 2007. Copyright © 1911, 1915, 1918, 1925, 1937, 1951, 1958, 1961, 1963, 1966, 1970, 1976, 1987, 1997 and 2007 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, "A201," and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under 1 Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes:

# **Certification of Document's Authenticity**

*AIA*<sup>®</sup> *Document D*401<sup>™</sup> – 2003

I, , hereby certify, to the best of my knowledge, information and belief, that I created the attached final document simultaneously with its associated Additions and Deletions Report and this certification at 13:03:09 ET on 09/02/2020 under Order No. 8348483959 from AIA Contract Documents software and that in preparing the attached final document I made no changes to the original text of AIA<sup>®</sup> Document A201<sup>TM</sup> – 2007, General Conditions of the Contract for Construction, as published by the AIA in its software, other than those additions and deletions shown in the associated Additions and Deletions Report.

(Signed)			
(Title)			
(Dated)			

AIA Document D401<sup>™</sup> – 2003. Copyright © 1992 and 2003 by The American Institute of Architects. All rights reserved. The "American Institute of Architects," "AIA," the AIA Logo, and "AIA Contract Documents" are registered trademarks and may not be used without permission. This document was produced by AIA software at 13:03:09 ET on 09/02/2020 under Order No.8348483959 which expires on 09/12/2020, is not for resale, is licensed for one-time use only, and may only be used in accordance with the AIA Contract Documents® Terms of Service. To report copyright violations, e-mail copyright@aia.org. User Notes: (1447905136)

# DOCUMENT CB - SUPPLEMENTARY CONDITIONS TO THE GENERAL CONDITIONS OF THE CONTRACT FOR CONSTRUCTION

Supplement AIA Document A201, 2007 Edition as follows:

#### **ARTICLE 1 – GENERAL PROVISION**

1.1 BASIC DEFINITIONS

Revise the first sentence in Subparagraph 1.1.1 as follows:

#### 1.1.1 THE CONTRACT DOCUMENTS

The Contract Documents are enumerated in the Agreement between the Owner and Contractor (hereinafter the Agreement) and consist of the Agreement, the Conditions of the Contract (General, Supplementary and other Conditions), Performance Bond, Labor and Material Payment Bond, the Drawings, the Specifications, all Addenda issued prior to execution of the Agreement and all Modifications thereto.

Add the following text to Subparagraph 1.1.3, THE WORK:

- 1.1.3 It also includes all supplies, skill, supervision, transportation services and other facilities and things necessary, proper or incidental to the carrying out and completion of the terms of the contract and all other items of cost or value needed to produce, construct and fully complete the public work identified by the Contract Documents.
- 1.1.7 INSTRUMENTS OF SERVICE: Delete this paragraph in its entirety.

Add the following Subparagraphs:

# 1.1.9 DESCRIPTION OF PARTIES

The following definitions apply to parties named in the Contract Documents.

1.	Owner:	Galveston Independent School District Administration Building 3904 Ave. T Galveston, Texas 77550 Phone: (409) 766-5158 Representative: Mr. Paul Byers, Director of Facilities
2.	Architect:	BEAM Professionals 11 Greenway Plaza 22 <sup>nd</sup> Floor Houston, Texas 77046 Phone: (713) 965-0608; Fax: (713) 961-4571
3.	MEP/Tech Consultant:	PBK <i>Leaf Engineers</i> 11 Greenway Plaza, 22 <sup>nd</sup> Floor Houston, Texas 77046 Phone: (713) 965-0608; Fax (713) 961-4571

# 1.1.10 ADDENDA

Addenda are written or graphic instruments issued prior to the execution of the Contract, which modify or interpret the proposal documents, including Drawings and Specifications, by additions, deletions, clarifications or corrections. Addenda will become part of the Contract Documents when the Construction Agreement is executed.

# 1.1.11 APPROVED, APPROVED EQUAL, APPROVED EQUIVALENT, OR EQUAL

The terms Approved and Approved Equal relate to the substitution of materials, equipment or procedure approved in writing by the Architect prior to receipt of proposals.

#### 1.1.12 ABBREVIATIONS

N.I.C. By Others; By Owner; Existing	Not in contract. Indicating work not to be done by this Contractor under this Agreement.
AIA	American Institute of Architects
ACI	American Concrete Institute
AISC	American Institute of Steel Construction
AISI	American Iron and Steel Institute
ASA	American Standards Association
ASTM	American Society for Testing Materials
AWSC	American Welding Society Code
FS	Federal Specification
NES	National Electrical Code
SPR	Simplified Practice Recommendation
UL	Underwriters Laboratories, Inc.

#### 1.1.13 PROPOSAL DOCUMENTS

Proposal Documents consist of all documents bound into or referenced in the Project Manual, the Drawings, and Addenda related thereto. The Project Manual contains the Proposal Requirements, Sample Forms, Conditions of the Contract, the Specifications, and a list of Drawings, and Schedules, some of which are bound into the Project Manual (Other Drawings and Schedules are bound separately).

# 1.1.14 MISCELLANEOUS OTHER WORDS

<u>Provide</u>: Whenever the word "provide" is used in these documents, it shall mean the same as "furnish and install".

# 1.2 CORRELATION AND INTENT OF THE CONTRACT DOCUMENTS

Add the following Subparagraphs:

#### 1.2.4 PRECEDENCE OF THE CONTRACT DOCUMENTS

The most recent issued Document takes precedence over previous issued forms of the same Document. The order of precedence is as follows with the highest authority listed first.

- .1 The Agreement
- .2 The Addenda
- .3 Conditions of the Contract, Drawings, and Specifications shall have equal authority. Should these documents disagree in themselves, the Architect will select the appropriate method for performing the work at no additional increase in the Contract Cost.

# 1.2.5 RELATION OF SPECIFICATIONS AND DRAWINGS

The Drawings and Specifications are correlative and have equal authority and priority. Should they disagree in themselves, or with each other, base the proposals on the most expensive combination of quality and quantity of work indicated. The appropriate method of performing the Work, in the event of the above mentioned disagreements, will be made by the Architect.

#### 1.2.6 OPTIONAL MATERIALS, BRANDS AND PROCESSES

When more than one is specified for a particular item of Work, the choice shall be the Contractor's. The final selection of color and pattern will be made from the range available within the option selected by the Contractor, unless the item is specified to match a specific color or sample furnished. Where particular items are specified only products of those named manufacturers are acceptable. Certain specified construction and equipment details may not be regularly included as part of the named manufacturer's standard catalog equipment but shall be provided by the manufacturer as required for the proper functioning of the equipment. Reasonable minor variations in equipment are expected and will be acceptable; however, indicated and specified performance and material requirements are minimum, and will be required in addition to standard accessories. The Architect reserves the right to determine the equality of equipment and materials that deviate from any of the indicated and specified requirements.

Add Paragraph 1.7 and following Subparagraphs:

# 1.7 MISCELLANEOUS OTHER DEFINITIONS

#### 1.7.1 ADDENDA, ADDENDUM

Documents issued by the Architect prior to execution of the Owner Contractor Agreement that modify or clarify the Proposal Documents. The addenda become a part of the Contract Documents

#### 1.7.2 ALTERNATE PROPOSAL(S)

A separate amount stated on the Proposal Form which, if accepted by the Owner, will be added to or deducted from the Base Proposal. If accepted, the work that corresponds to the alternate proposal will become part of the Agreement between Owner and Contractor. Alternative proposals shall remain valid for a period of 30 days after receipt of proposals, regardless if an Owner Contractor Agreement has been executed, unless indicated otherwise herein.

#### 1.7.3 BASE PROPOSAL

The Contractor's proposal for the Work, not including any Alternatives.

# 1.7.4 CONTRACT TIME

The period of time which is established in the Contract Documents for Substantial completion (Delivery of Equipment) of the Work.

#### 1.7.5 DATE OF AGREEMENT

The date the Owner formally awards a Contract for Construction of the Work. This date will be inserted on the first page of the Agreement Between Owner and Contractor and shall be referenced in Performance Bond and Payment Bond forms. See also Date of Commencement of the Work.

#### 1.7.6 DATE OF COMMENCEMENT OF THE WORK

The date of either (1) the fully executed Agreement Between Owner and Contractor, or (2) a written Notice to Proceed is delivered to the Contractor or (3) **November 30, 2020** whichever comes first. This date constitutes day zero ("0") of the stated Contract Time.

1.7.7 DATE OF FINAL COMPLETION

The end of construction. Refer Paragraph 9.10.

1.7.8 DATE OF SUBSTANTIAL COMPLETION (DELIVERY OF EQUIPMENT)

Refer Subparagraph 8.1.3 and Paragraph 9.8. Contractor shall be Substantially Complete by date stated on the Proposal Form, agreed to by the Owner and incorporated into the Agreement Between Owner and Contractor.

#### 1.7.9 DAY

The following days are referenced in the documents:

- .1 Calendar Days: The days of the Gregorian Calendar. The Contract Time is established in Calendar Days and extensions of time granted for Regular Work Days lost, if any, will be converted to Calendar Days.
- .2 Holidays: The days officially recognized by the construction industry in this area as a holiday; normally limited to the observance days of New Year's Day, Memorial Day, Fourth of July, Labor Day, Thanksgiving Day and the day after, and Christmas Day.
- .3 Regular Work Days: All calendar days except holidays, Saturdays, and Sundays. Requests for extensions of time shall be requested on the basis of Regular Work Days, and those days, if approved; will be converted to calendar days by multiplying by a factor of one and four-tenths (1.4).
- .4 Anticipated Weather Days: An allowance of Regular Work Days, established as probable days lost due to weather delays; said allowance to be included in the Contractor's proposed Completion Time on his Proposal Form.
- .5 Weather Days: Regular Work Days when rain, flooding, snow, unusually high winds, excessively wet grounds, or similar circumstances prevent progress on major portions of the Work. The Contractor will be entitled to an extension of the Contract Time for the net additional time, if any, which results from deducting the amount of Anticipated Weather Days from the total amount of Weather Days.
- .6 Net Weather Days: The difference in working days between Anticipated Weather Days and Weather Days.

# 1.7.10 NOTICE TO PROCEED

A notice that may be given by the Owner to the Contractor that directs the Contractor to start the Work. It may also establish the Date of Commencement of the Work.

# 1.7.11 PUNCH LIST

A comprehensive list prepared by the Contractor prior to Substantial completion (Delivery of Equipment) to establish all items to be completed or corrected; this list may be supplemented by the Architect or Owner. Refer to Subparagraph 9.8.2.

# **ARTICLE 2 - OWNER**

# 2.2 INFORMATION AND SERVICES REQUIRED OF THE OWNER

Delete the text of Subparagraph 2.2.5 in its entirety and substitute the following:

- 2.2.5 The Contractor will be furnished free of charge, 25 copies of the Drawings and Specifications for the execution of the work. The Contractor shall pay actual reproduction costs of any additional copies required.
- 2.4 OWNER'S RIGHT TO CARRY OUT THE WORK

Delete text of Subparagraph 2.4.1 in its entirety and substitute the following:

- 2.4.1 If the Contractor defaults or neglects to carry out the work in accordance with the Contract Documents and fails, after receipt of written notice from the Owner, to commence and continue correction of such default or neglect with diligence and promptness, the Owner may, without prejudice to other remedies the Owner may have, correct such deficiencies. In such case, an appropriate Change Order shall be issued deducting from payments then or thereafter due the contractor the cost of correcting such deficiencies, including compensation for the Architect's additional services and expenses made necessary by such default, neglect or failure. Such action by the Owner and amounts charged to the Contractor are both subject to prior approval of the Architect. If payments then or thereafter due the Contractor are not sufficient to cover such amounts, the Contractor shall pay the difference to the Owner.
- 2.5 OWNER'S RIGHT TO OCCUPY THE PROJECT

Add the following Subparagraphs:

- 2.5.1 The Owner shall have the right to occupy or use without prejudice to the right of either party, any completed or largely completed portions of the project, notwithstanding the time for completing the entire work or such portions may not have expired. Such occupancy and use shall not constitute acceptance of any work not in accordance with the Contract Documents.
- 2.5.2 If such prior use delays the completion of the project, the Contractor shall be entitled to extension of time, which claim shall be in writing with supporting data attached.
- 2.5.3 Refer to Article 11 Insurance and Bonds regarding property insurance requirements in the event of such occupancy.

# **ARTICLE 3 - CONTRACTOR**

3.2 REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS BY CONTRACTOR

Delete text of Subparagraph 3.2.2 and substitute the following:

3.2.2 The Contractor shall carefully study and compare the Agreement, Conditions of the Contract, Drawings, Specifications, Addenda, and Modifications and shall at once report to the Architect any error, inconsistency, or omission he may discover. Contractor shall be liable for any damage to Owner for failure to report any error, inconsistency or omission he may discover or should have

discovered, but he shall not be liable to Owner or Architect for any damage resulting from such error, inconsistency or omission which he should not have discovered or which he did discover and at once so reported. Contractor shall do no work without approved Drawings and Specifications.

Add the following Subparagraphs:

- 3.2.5 The Contractor shall not be entitled to additional compensation for the "rework portion" of any additional work caused by his failure to carefully study and compare the contract documents prior to execution of the Work.
- 3.2.6 The Contractor shall make a reasonable attempt to interpret the Contract Documents before asking the Architect for assistance in interpretation. The Contractor shall not ask the Architect for observation of work prior to the Contractor's field superintendent's personal inspection of the work and his determination that the work complies with the Contract Documents. The Contractor shall arrange meetings prior to commencement of the work of all major subcontractors to allow the subcontractor to demonstrate his understanding of the documents to the Architect and to allow the subcontractor to ask for any interpretation he may require.
- 3.2.7 If, in the opinion of the Architect, the Contractor does not make a reasonable effort to comply with the above requirements of the Contract Documents and this causes the Architect or his Consultants to expend an unreasonable amount of time in the discharge of the duties imposed on him by the contract Documents, then the Contractor shall bear the cost of compensation for the Architect's additional services made necessary by such failure. The Architect will give the Contractor prior notice of intent to bill for additional services related to above requirements before additional services are performed.
- 3.2.8 If the Contractor has knowledge that any of the products or systems specified will perform in a manner that will limit the Contractor's ability to satisfactorily perform the work or to honor his Warranty, he shall promptly notify the Architect in writing, providing substantiation for his position. Any necessary changes, including substitution of materials, shall be accomplished by appropriate Modification.
- 3.20 PREVAILING WAGE RATES
- 3.20.1 No employee used in this construction may be paid less than the minimum wage rate provided herein in Article 16. The Contractor must comply with all Davis-Bacon and Related Acts.
- 3.3 SUPERVISION AND CONSTRUCTION PROCEDURES

Add the following Subparagraph:

- 3.3.4 The Contractor shall provide a full time superintendent to remain on site throughout duration of Project dedicated strictly to the Project.
- 3.5 WARRANTY

Add the following Subparagraphs:

3.5.2 In the event of failure of materials, products, or workmanship, either during construction or the warranty period (which shall be one (1) year from the Date of Substantial completion (Delivery of Equipment), except where a longer period is specified), the Contractor shall take appropriate measures to assure correction or replacement of the defective items, whether notified by the Owner or Architect. Items of work first performed after Substantial Completion (Delivery of Equipment) shall have their warranties extended by the period of time between Substantial completion (Delivery of Equipment) and the actual performance of the Work.

- 3.5.3 Refer to warranty forms included under Section 01710 Close-Out Procedures, which will be required prior to final payment.
- 3.5.4 Appropriately 11 months after Substantial Completion, the Contractor shall accompany the Owner and Architect on a complete reinspection of the Project and be responsible for correcting of any additional deficiencies observed or reported.
- 3.6 TAXES

Delete text of Subparagraph 3.6.1, and substitute the following.

- 3.6.1 The Owner is exempt from the Texas Sales Tax on any purchase of tangible personal property and will issue Certificates of Exemption from the Texas Sales Tax on materials furnished by Contractors on School Construction projects. The Contractor shall give a written statement to the Owner (with a copy to the Architect) as to the proration of costs of skilled crafts, labor and materials for the project prior to awarding of a Construction Contract. The contractors shall obtain Certificates of Resale from their suppliers in order to avoid payment of the State Sales Tax on materials incorporated in School jobs. Failure of the Contractor to obtain Certificates of Resale from their suppliers shall make the Contractor responsible for absorbing the tax.
- 3.7 PERMITS, FEES, NOTICES AND COMPLIANCES WITH LAWS
- 3.7.1 Supplement Subparagraph 3.7.1, as follows:

Unless otherwise provided in the Contract Documents, the Contractor shall secure and pay for the building permit and other permits and governmental fees, licenses and inspections necessary for proper execution and completion of the Work which are customarily secured after execution of the Contract and which are legally required when bids are received or negotiations concluded.

Delete text of Subparagraph 3.7.3 in its entirety and substitute the following:

3.7.3 It is not the Contractor's responsibility to ascertain that the Contract Documents are in accordance with Applicable laws, statutes ordinances, building codes, and rules and regulations. However, if the Contractor observes that portions of the Contract Documents are to variance therewith, the Contractor shall promptly notify the Architect and Owner in writing and necessary changes shall be accomplished by appropriate Modification.

Add the following subparagraph:

- 3.7.3.1 If the Contractor performs Work which he knew or should have known to be contrary to applicable laws, statues, ordinances, building codes, local rules or regulations, without such notice to the Architect and Owner, the Contractor shall assume full responsibility for such Work and shall bear there attributable costs.
- 3.8 ALLOWANCES

Delete text of Subparagraph 3.8.1, and substitute the following:

3.8.1 The General Contractor shall include in his proposal the allowances stated in the Specifications. These stated allowances represent the cost estimate of the materials and equipment delivered and unloaded at the site. The Contractor's handling costs on site, overhead, profit, and other expenses contemplated for the allowance material and equipment shall be included in allowance only where called for in the various sections of these specifications.

The Contractor shall purchase the allowance materials and equipment as directed by the Architect on the basis of the lowest responsible proposal of at least three (3) competitive proposals. If the

actual cost of the materials and equipment delivered and unloaded at the site is more or less than all the allowance estimates, the Contract Sum will be adjusted accordingly by Change Order.

#### 3.9 SUPERINTENDENT

Delete Subparagraph 3.9.1, in its entirety and substitute the following:

3.9.1 Prime Contractor shall employ competent superintendent and necessary assistants who shall be in attendance at the Project site during the progress of the work. The Superintendent shall be satisfactory to the Owner and shall not be changed except with the consent of the Architect, unless the Superintendent leaves the employment of the Contractor. No increase in Contract Time or Contract Sum shall be allowed in the event the Owner or Architects objects to any nominated superintendent.

#### 3.10 CONTRACTOR'S CONSTRUCTION SCHEDULES

Add the following Subparagraph:

- 3.10.4 The Contractor shall submit to the Architect with each monthly Application for Payment, a copy of the progress schedule showing all modifications required to have the schedule reflect appropriate revisions and shall take whatever action is necessary to assure that the project completion schedule is met.
- 3.12 SHOP DRAWINGS, PRODUCT DATA AND SAMPLES

Add the following Subparagraphs:

- 3.12.11 The Contractor shall submit complete drawings, data and samples to the Architect at least 30 days prior to the date the Contractor needs the reviewed submittals returned. The Contractor shall be prepared to submit color samples on any key items (such as quarry tile, vinyl wall covering, etc.) within 30 days of the award of Contract. Once samples of all key items are received, the Architect will finalize color selections.
- 3.12.12 The Contractor shall submit the number of copies of product data and samples which the Contractor and his subcontractors need for their use PLUS two (2) additional sets for the Architect, one (1) additional set for the Owner and one (1) additional set for each of the Architect's consultants involved with the particular Section of Work. Where shop drawings are involved, submit one (1) high quality reproducible transparency and one (1) opaque print of the shop drawing for the Architect plus one (1) additional opaque print for each of the Architect's consultants involved with the particular Section of Work. The reproducible transparency will be marked by the Architect and/or his consultants. After final review and correction of the submittal, the Contractor shall send one (1) corrected set to the Architect and one (1) to each of the Architect's consultants involved with the particular Section of Work.
- 3.12.13 The Contractor shall provide composite drawings within three (3) months of contract signing showing how all piping, ductwork, lights, conduit, equipment, etc. will fit into the ceiling space alotted, including clearances required by the manufacturer, by Code, or in keeping with good construction practice. Space for all trade elements must be considered on the same drawing. Drawings shall be at 1/4 inch per foot minimum scale and shall include invert elevations and sections required to meet intended purpose.
- 3.15 CLEANING UP

Add the following Subparagraph:

3.15.3 Prior to the Architect's inspection for Substantial completion (Delivery of Equipment), the Contractor shall clean exterior and interior surfaces exposed to view; remove temporary labels, stains, and foreign substances; polish transparent and glossy surfaces; clean equipment and fixtures to a sanitary condition; clean site; remove trash and surplus materials from the site.

Add following Paragraphs in their entirety:

# 3.19 ANTITRUST VIOLATIONS

3.19.1 To permit the Owner to recover damages suffered; in antitrust violations, the Owner/Contractor Agreement shall include the following, "Contractor hereby assigns to Owner any and all claims for overcharges associated with this contract which are under the antitrust laws of the United States, 15 U.S.C.A., Sec. 1 et.seq. (1973)". The Contractor shall include this provision in his agreements with each subcontractor and supplier. Each subcontractor shall include such provisions in agreements with sub-subcontractors and suppliers.

# **ARTICLE 4 - ADMINISTRATION OF THE CONTRACT**

4.2 ADMINISTRATION OF THE CONTRACT

Delete text of Subparagraphs 4.2.2 and substitute the following:

The Architect, as a representative of the Owner, will visit the site at intervals appropriate to the stage of the Contractor's operations (1) to become generally familiar with and (2) to endeavor to guard the Owner against defects and deficiencies in the Work, and (3) to determine in general if the Work is being performed in a manner indicating that the Work, when fully completed, will be in accordance with the Contract Documents. However, the Architect will not be required to make exhaustive or continuous on-site inspections to check the quality or quantity of the Work or for safety precautions and programs in connection with the Work.

Add the following text to Subparagraph 4.2.3:

4.2.3 The Architect shall endeavor to guard the Owner against defects and deficiencies in the Work.

Delete text of following Subparagraphs and substitute the following:

- 4.2.6 The Architect will have authority to reject Work which does not conform to the Contract Documents. Whenever the Architect considers it necessary or advisable for implementation of the intent of the Contract Documents, the Architect will have authority to require additional inspection or testing of the Work in accordance with Subparagraphs 13.5.2 and 13.5.3, whether or not such Work is fabricated, installed or completed. However, neither this authority of the Architect nor a decision made reasonably and in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the Architect to the Contractor, Subcontractors, material an equipment suppliers, their agents or employees, or other persons performing portions of the Work.
- 4.2.12 Interpretations and decisions of the Architect will be consistent with the intent of and reasonably inferable from the Contract Documents and will be in writing or in the form of drawings. When making such interpretations and decisions, the Architect will endeavor to secure faithful performance by both Owner and Contractor, will not show partiality to either and will not be liable for results of interpretations or decisions so rendered reasonably and in good faith.

# **ARTICLE 5 - SUBCONTRACTORS**

5.2 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OF THE WORK

Delete text of Subparagraphs 5.2.1, 5.2.2, 5.2.3, and 5.2.4 in their entirety and substitute the following:

- 5.2.1 As soon as practicable after Award of the Contract but no later than 10 days prior to the submittal date for the Contractor's first Application for Payment, Contractor shall furnish to the Owner and Architect in writing the names of the persons or entities (including those who are to furnish materials or equipment fabricated to a special design) proposed for each of the principal portions of the Work. Where Subcontractors or Sub-subcontractors have been listed in the Specifications or in an Addendum as a Listed Subcontractor the proposed entity shall be one of those firms listed, unless agreement has been reached to accept a proposed Substitute Subcontractor as listed on the Proposal Form. Regarding proposed persons or entities to perform portions of the Work where no Listed Subcontractors have been listed or approved by Addendum, the Architect will promptly reply to the Contractor in writing stating whether or not the Owner or the Architect, after due investigation, has objection to any such proposed person or entity. Failure of the Owner or Architect to reply promptly shall constitute notice of no objection. Failure of the Contractor to submit the subject names in a timely manner will delay processing of the Contractor's Application for Payment.
- 5.2.2 The Contractor shall not contract with a proposed person or entity to whom the Owner or Architect has made a timely objection. The Contractor shall not be required to contract with anyone to whom the Contractor has made an objection under the provisions of Subparagraph 5.2.1.
- 5.2.3 If the Owner or Architect has objection to a person or entity proposed by the Contractor, the Contractor shall propose another to whom the Owner or Architect has no objection. The Contract Sum shall be increased or decreased by the difference in cost occasioned by such change and an appropriate Change Order shall be issued. However, no increase in the Contract Sum shall be allowed for such change unless the Contractor has acted promptly and responsively in submitting names as required in Subparagraph 5.2.1.
- 5.2.4 Prior to such substitution the Contractor shall notify the Architect of his intent and reasons for such proposed substitutions. The Contractor shall not change a Subcontractor, person or entity previously selected if the Owner or Architect makes objection to such change.

Add Subparagraphs 5.2.5 and 5.2.6 as follows:

- 5.2.5 The Contractor shall submit the list of proposed Subcontractors on AIA Document G805. The Contractor may obtain blank copies from the Architect.
- 5.2.6 The Contractor is required to visit the site and completely familiarize himself with the existing conditions prior to the proposal. No additional increase in the Contract amount will be provided when existing or known conditions require a certain amount of work to comply with the intent of the Contract Documents.

# **ARTICLE 7 - CHANGES IN THE WORK**

7.2 CHANGE ORDERS

Add Subparagraph 7.2.2:

7.2.2 The cost or credit to the Owner resulting from a change in the work shall be determined in one or more of the ways listed below. The first method listed shall be used unless the Architect determines that the method is inappropriate, in which case another method shall be selected:

- A. By mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation. Where additional work is involved, the lump sum shall represent the estimated cost of labor and materials plus markups to cover overhead and profit:
  - 01 To compensate the contractor or subcontractor actually performing a part of the work for the combined cost of overhead and profit, the performing party shall be entitled to a single markup not to exceed 10% of the estimated cost of that part of the work.
  - 02 To compensate the contractor for the combined cost of overhead and profit on work performed by subcontractors, the Contractor shall be entitled to a single markup not to exceed 10% of the subcontract amount.
  - 03 When a subcontractor performs the work of a change, the 10% markup for combined overhead and profit shall be used only by the sub-subcontractor. The Contractor and Subcontractor would each be entitled to a single markup not to exceed 10% of the cost to them for the Subcontractor and sub-subcontractor, respectively.
- B. By unit prices stated in the Contract Documents or subsequently agreed upon. Additional markups for overhead and profit will not be allowed in Unit Price work.
- C. By cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee.

# 7.3 CONSTRUCTION CHANGE DIRECTIVES

Delete text of Subparagraph 7.3.3 in its entirety and substitute the following:

- 7.3.3 The cost or credit to the Owner resulting from a change in the Work shall be determined in one or more ways listed below. The first method listed shall be used unless the Architect determines that the method is inappropriate, in which case another method shall be selected.
  - .1 By mutual acceptance of a lump sum properly itemized and supported by sufficient substantiating data to permit evaluation. Where additional Work is involved, the lump sum shall represent the estimated cost of labor and materials plus markups to cover overhead and profit:

To compensate the Contractor or Subcontractor actually performing a part of the Work for the combined cost of overhead and profit, the performing party shall be entitled to a single markup not to exceed 10% of the estimated cost of that part of the Work.

To compensate the Contractor for the combined cost of overhead and profit on work performed by Subcontractors, the Contractor shall be entitled to a single markup not to exceed 10% of the subcontract amount.

When a Sub-subcontract performs the Work of a change, the 10% markup for combined overhead and profit shall be used only by the Sub-subcontractor. The Contractor and Subcontractor would each be entitled to a single markup not to exceed 10% of the cost to them from the Subcontractor and Sub-subcontractor respectively.

- .2 By unit prices stated in the Contract Documents or subsequently agreed upon. Additional markups for overhead and profit will not be allowed in Unit Price Work.
- .3 By cost to be determined in a manner agreed upon by the parties and a mutually acceptable fixed or percentage fee.

Add the following Subparagraph:

#### 7.5 CHANGES FUNDED BY ALLOWANCES

7.5.1 Allowances balances may be used to fund changes in the work.

The Contractor will not be allowed an overhead and profit mark-up when changes in the work are funded by one of the Allowances. For Subcontractor and Sub-Subcontractor overhead and profit mark-up when changes in the work are funded by one of the Allowances, refer to Paragraph 7.2.2.

#### **ARTICLE 8 - TIME**

#### 8.1 DEFINITIONS

Delete Subparagraph 8.1.2 in its entirety and substitute the following:

- 8.1.2 Unless agreed otherwise, the date inserted on the Agreement form and the Date of Commencement of the Work shall be as follows:
  - .1 The date inserted on the first page of the Agreement form will be the date the Owner formally awards the Contract. As soon as feasible after receipt of Proposals, the Architect will present Agreement forms to the Contractor for his review and signature; the Contractor will be allowed a maximum of five (5) days from the date the prepared Agreements are presented to him to 1) obtain the required bond forms and insurance certificates and 2) return the executed Agreement and supporting documents to the Architect for transmittal to the Owner for his final review and execution.
  - .2 The Date of Commencement of the Work shall be as defined in paragraph 1.7.6 above.

#### 8.3 DELAYS AND EXTENSIONS OF TIME

Add the following Subparagraph:

8.3.4 The following is a requirement of the Contract and will be included in the Agreement Between Owner and Contractor under Time of Completion and the blank spaces will be completed indicating the completion date as stated on the Proposal Form or agreed to by the Owner.

"The Work to be performed under this Contract shall be commenced upon Notice to Proceed and shall be Substantially Completed by **February 15, 2019** or by such dates thereafter as may be established in any written extensions granted under Article 8 of the General Conditions. The parties hereto agree that time is of the essence of this contract and that the pecuniary damages which would be suffered by the Owner, if the Contractor does not complete all work called for in the contract documents by the specified date, are in their very nature difficult of ascertainment.

It is therefore expressly agreed as a part of the consideration inducing the Owner to execute this contract that the Owner may deduct from the final payment made to the Contractor a sum equal to **\$500.00** per calendar day for each and every calendar day beyond the agreed date which the Contractor shall require for Substantial Completion of the work included in this contract. It is expressly understood that the said sum per day is agreed upon as a fair estimate of the pecuniary damages which will be sustained by the Owner in the event that the work is not completed within the agreed time, or within the legally extended time, if any, otherwise provided for herein. Said sum shall be considered as liquidated damages only and in no sense shall be considered a penalty, said damage being caused by additional compensation to personnel, for loss of interest on money and other miscellaneous increased costs, all of which are difficult of exact ascertainment."

Failure to complete and close-out project **60 days** after Substantial Completion will result in liquidated damages being assessed in the amount of **\$500.00** per calendar day until close-out occurs.

# **ARTICLE 9 - PAYMENTS AND COMPLETION**

9.2 SCHEDULE OF VALUES

Add Subparagraph 9.2.2 as follows:

- 9.2.2 In order to facilitate the review of Applications for Payment, the Schedule of Values shall be submitted on AIA Documents G702 and G703 or other similar forms approved by the Owner, and shall include the following:
  - .1 General Contractor's costs for Contractor's fee, bonds and insurance, mobilization, etc., shall be listed as individual line items.
  - .2 Contractor's costs for various construction items shall be detailed. For example, concrete Work shall be subdivided into footings, grade beams, floor slabs, paving, etc. These subdivisions shall appear as individual line items.
  - .3 On major subcontracts, such as mechanical, electrical and plumbing, the schedule shall indicate line items and amounts in detail (for example; underground, major equipment, fixtures, installation of fixtures, start up, etc.)
  - .4 Costs for subcontract Work shall be listed without any additional of General Contractor's costs for overhead, profit or supervisions.
  - .5 Where payment for stored materials may be requested prior to installation, material and labor shall be listed as separate line items.
  - .6 Sample pages from an approved schedule of values are included following this document.

# 9.3 APPLICATIONS FOR PAYMENT

Delete Subparagraph 9.3.1 in its entirety and substitute the following:

9.3.1 At least ten (10) days before the date established for each progress payment, the Contractor shall submit to the Architect an itemized Application for Payment for operations completed in accordance with the schedule of values. Prior to this submittal, the Contractor shall contact the Architect's Field Department for on-site review of the proposed application. Upon approval by the Architect's Field Department, the Application for Payment shall be notarized and submitted to the Architect.

Included shall be data required to support the Contractor's right to payment as may be required by the Owner or Architect, such as copies of requisitions from subcontractors and material suppliers, and reflecting retainage, if provided for elsewhere in the contract documents.

Delete Subparagraph 9.3.2 in its entirety and substitute the following:

- 9.3.2 Payments will be made on account of materials or equipment 1) incorporated in the Work and 2) suitably stored at the site or 3) suitably stored at some off-site location provided the following conditions are met for off-site storage:
  - .1 The location must be agreed to, in writing, by the Owner and Surety.
  - .2 The location must be a bonded warehouse.
  - .3 Surety must agree, in writing, to each request for payment.
  - .4 The Contractor must bear the cost of the Owner's and Architect's expenses related to visiting the off-site storage area.

Payments for materials or equipment stored on or off the site shall be conditioned upon submission by the Contractor of bills of sale or such other procedures satisfactory to the Owner to establish the Owner's title to such materials or equipment or otherwise protect the Owner's interest, including applicable insurance (naming the Owner as insured) and transportation to the site for those materials and equipment stored off the site. Under no circumstances will the Owner reimburse the Contractor for down payments, deposits, or other advance payments for materials or equipment.

The Contractor acknowledges that the review of materials stored off site is an additional service of the Architect and shall be charged for that service. The cost for that service will be established by the Architect and is not subject to appeal.

Add the following Subparagraph:

- 9.3.4 Contractors shall submit application in quadruplicate using AIA Document G702 and G703, Application and Certificate for Payment, 1992 Edition. All blanks in the form must be completed and signatures of Contractor and Notary Public must be original on each form.
- 9.4 CERTIFICATES FOR PAYMENT

Add the following Subparagraph:

- 9.4.3 The Architect will affix his signature to the same form described in Paragraph 9.3.4 to signify his certification of payment provided the application is otherwise satisfactory.
- 9.6 PROGRESS PAYMENTS

Add the following Subparagraph:

9.6.8 Based upon Applications for Payment submitted to the Architect by the Contractor and Certificates for Payment issued by the Architect, the Owner shall make progress payments on account of the Contract Sum to the Contractor as provided in the conditions of the Contract as follows:

On or about the 15th day of each month 95% of the proportion of the Contract Sum properly allocable to labor, materials and equipment incorporated in the Work and 95% of the portion of the Contract Sum properly allocable to materials and equipment suitably stored at the site or at some other location agreed upon in writing by the parties, up to the first (1st) day of that month; less the aggregate of previous payments in each case; and upon Substantial Completion (Delivery of Equipment) of the entire work, a sum sufficient to increase the total payments to 95% of the Contract Sum less such retainages as the Architect shall determine for all incomplete work and unsettled claims.

- 9.7 FAILURE OF PAYMENT
- 9.7.1 Delete the phrase "or awarded binding dispute resolution".
- 9.8 SUBSTANTIAL COMPLETION
- 9.8.1 Add the following:

The following items are a partial list of requirements, as applicable to the Project, which must be completed <u>prior</u> to establishment of Substantial Completion.

- 01 All exterior clean-up must be complete.
- 02 All operation and maintenance manuals are delivered and approved ("D-slant" ring binders in triplicate).
- 03 Requirements of Specification Section 01710, Closeout Procedures.
- 9.10 FINAL COMPLETION AND FINAL PAYMENT

At Subparagraph 9.10.2, modify as follows:

- .1 On line 8, delete the phrase "if any".
- .2 On line 8, delete the phrase "If required by the Owner".

Add the following to Subparagraph 9.10.2:

- Prior to final payment, the Contractor shall submit in triplicate to the Architect the following completed forms:
- 01 Contractor's Affidavit of Payment of Debts and Claims, AIA Document G706.
- 02 Contractor's Affidavit of Release of Liens, AIA Document G706A.
- 03 Consent of Surety to Final Payment, AIA Document G707.
- 04 General Contractor's Guarantee notarized
- 05 Subcontractor's Guarantees notarized
- 06 Subcontractor's Affidavit of Release of Liens with subcontractor's supplier's separate release with signature and notary seal on same page.
- 07 Each Offeror (and subcontractor and supplier submitting a proposal to an Offeror) shall submit a notarized affidavit stating that no asbestos, PCB or lead (except for flashing in roofing) containing building materials were installed in this project. (Document AN)
- 08 Each subcontractor and supplier submitting a proposal to an Offeror shall submit a notarized affidavit stating that no asbestos, PCB or lead (except for flashing in roofing) containing building materials were installed in this project. (attached to Section 01710)
- 09 Maintenance and inspection manuals. Three (3) sets of each bound in a 3 inch "D-slant" ring binder.
- 10 Final list of subcontractors (Document AG).

Documents identified as affidavit must be notarized. All manuals will contain an index listing the information submitted. The index sections will be divided and identified by tabbing each section as listed in the index.

Upon request, the Architect will furnish the Contractor with blank copies of the forms listed above. Final payment, constituting the entire unpaid balance of the Contract Sum shall be paid by the Owner to the Contractor 31 days after substantial completion (delivery of equipment) unless otherwise stipulated in the Certificate of Completion, the contract fully performed, and Final Certificate of Payment has been issued by the Architect.

The Owner may accept certain portions of the work as being complete prior to the acceptance of the entire project. If certain areas are accepted by the Owner as being complete, and if the Contractor has completed all of the requirements for final payment of that portion of work, then the Owner may release retainage for that area/portion of work. Amounts of retainage shall be agreed upon by both Owner and Contractor prior to final acceptance of these areas.

Refer to Section 01 77 00, Close-out Procedures for additional requirements.

# **ARTICLE 10 - PROTECTION OF PERSONS AND PROPERTY**

# 10.3 HAZARDOUS MATERIALS

Delete text of Subparagraph 10.3.2 in its entirety and substitute the following:

10.3.2 If requested in writing by the Contractor, the Owner shall obtain the services of a licensed laboratory to verify a presence or absence of the material or substance reported by the Contractor and, in the even such material or substance is found to be present, to verify that is has been rendered harmless. If requested in writing by the Contractor or Architect, the Owner shall furnish in writing to the Contractor and Architect, the names and qualifications of persons or entities who are to perform tests verifying the presence or absence of such material or substance. The Contractor and the Architect will promptly reply to the Owner in writing stating whether or not either has reasonable objection to the persons or entities proposed by the Owner. If either the Contractor or Architect has an objection to a person or entity proposed by the Owner, the Owner shall propose another to whom the Contractor and the Architect have no reasonable objection.

Add the following Subparagraphs:

# 10.5 ASBESTOS, LEAD OR PCBs CONTAINING MATERIALS

10.5.1 The contractor and each subcontractor, prior to final payment, shall submit a notarized statement on their letterhead certifying "to the best of their information, knowledge, and belief asbestos, asbestos containing materials, and PCBs have not been used or incorporated into the Work and lead or lead bearing materials have not been incorporated into potable water systems." For the purpose of definition as used in this statement, the term "potable water systems" includes, but is not limited to, those water systems for drinking fountains, all sinks, showers, bath tubs, residential and commercial kitchen equipment, ice machines, and hose bibbs, as applicable to the project. The Contractor shall also obtain such statements from Subcontractors and all such statements shall be notarized.

# **ARTICLE 11 - INSURANCE AND BONDS**

11.1 CONTRACTOR'S LIABILITY INSURANCE

Add the following to Paragraph 11.1.3.

11.1.3.1 Refer to Document CE - Insurance for minimum Contractor's Insurance required. The Contractor is urged to carry such additional insurance as he may deem appropriate to provide protection from risks assumed under this contract. The Contractor shall fill in and submit one (1) copy of the completed Certificate of Insurance forms to the Owner prior to commencement of the Work. The required insurance must be written by a Company licensed to do business in the State of Texas and be acceptable to the Owner.

Comply with minimum insurance requirements as specified in Insurance under the Owner's Terms & Conditions. If anything in this document is contrary to what is stated in the Owner's document, the Owner's documents will take precedence, unless directed otherwise by the Owner.

A. Worker's Compensation Insurance Coverage

# Definitions:

Certificate of coverage ("Certificate") - A copy of a certificate of insurance, a certificate of authority to self-insure issued by the commission, or a coverage agreement (TWCC-81, TWCC-82, TWCC-83, or TWCC-84), showing statutory workers' compensation insurance coverage for the person's or entity's employees providing services on a project, for the duration of the project.

Duration of the project - includes the time from the beginning of the work on the project until the contractor's/person's work on the project has been completed and accepted by the governmental entity.

Persons providing services on the project ("subcontractor" in Texas Labor Code 406.096) - includes all persons or entities performing all or part of the services the contractor has undertaken to perform on the project, regardless of whether that person has employees. This includes, without limitation, independent contractors, subcontractors, leasing companies, motor carriers, owner-operators, employees of any such entity, or employees of any entity that furnishes persons to provide services on the project. "Services" shall include, without limitation, providing, hauling, or delivering equipment or materials, or providing labor, transportation, or other service related to a project. "Services" does not include activities unrelated to the project, such as food/beverage vendors, office supply deliveries, and delivery of portable toilets.

- 01 The contractor shall provide coverage, based on proper reporting of classification codes and payroll amounts and filing of any coverage agreements, which meets the statutory requirements of Texas Labor Code 401.011 (44) for all employees of the contractor providing services on the project for the duration of the project.
- 02 The contractor must provide a certificate of coverage to the governmental entity prior to being awarded the contract.
- 03 If the coverage period shown on the contractor's current certificate of coverage ends during the duration of the project, the contractor must, prior to the end of the coverage period, file a new certificate of coverage with the governmental entity showing the coverage has been extended.
- 04 The contractor shall obtain from each person providing services on a project, and provide to the governmental entity:
  - a. a certificate of coverage, prior to that person beginning work on the project, so the governmental entity will have on file certificates of coverage showing coverage for all persons providing services on a project; and
  - b. no later than seven (7) days after receipt by the contractor, a new certificate of coverage showing extension of coverage, if the coverage period shown on the current certificate of coverage ends during the duration of the project.
- 05 The contractor shall retain all required certificates of coverage for the duration of the project and one year thereafter.
- 06 The contractor shall notify the governmental entity in writing by certified mail or personal delivery, within 10 days after the contractor knew or should have known, of any change that materially affects the provision of coverage of any person providing services on the project.
- 07 The contractor shall post on each project site a notice, in the text, form and manner prescribed by the Texas Workers' Compensation Commission, informing all persons providing services on the project that they are required to be covered, and stating how a person may verify coverage and report lack of coverage.
- 08 The contractor shall contractually require each person with whom it contracts to provide services on a project, to:
  - provide coverage, based on proper reporting of classification codes and payroll amounts and filing of any coverage agreements, which meets the statutory requirements of Texas Labor Code, Section 401.011 (44) for all of its employees providing services on the project, for the duration of the project;
  - b. provide to the contractor, prior to that person beginning work on the project, a certificate of coverage showing that coverage is being provided for all employees of the person providing services on the project, for the duration of the project;
  - c. provide the contractor, prior to the end of the coverage period, a new certificate of coverage showing extension of coverage, if the coverage period shown on the current certificate of coverage ends during the duration of the project;

- d. obtain from each other person with whom it contracts, and provide to the contractor:
  - 1) a certificate of coverage, prior to the other person beginning work on the project; and
  - a new certificate of coverage showing extension of coverage, prior to the end of the coverage period, if the coverage period shown on the current certificate of coverage ends during the duration of the project.
- e. retain all required certificates of coverage on file for the duration of the project and for one year thereafter;
- f. notify the governmental entity in writing by certified mail or personal delivery, within 10 days after the person knew or should have known, of any change that materially affects the provisions of coverage of any person providing services on the project; and
- g. contractually require each person with whom it contracts to perform as required by paragraphs a g, with the certificates of coverage to be provided to the person for whom they are providing services.
- 09 By signing this contract or providing or causing to be provided a certificate of coverage, the contractor is representing to the governmental entity that all employees of the contractor who will provide services on the project will be covered by workers' compensation coverage for the duration of the project, that the coverage will be based on proper reporting of classification codes and payroll amounts, and that all coverage agreements will be filed with the appropriate insurance carrier or, in the case of self-insured, with the commission's Division of Self-Insurance Regulation. Providing false or misleading information may subject the contractor to administrative penalties, criminal penalties, civil penalties, or other civil actions.
- 10 The contractor's failure to comply with any of these provisions is a breach of contract by the contractor which entitles the governmental entity to declare the contract void if the contractor does not remedy the breach within ten days after receipt of notice of breach from the governmental entity.
- 11 The Contractor shall post the following language:

# REQUIRED WORKERS' COMPENSATION COVERAGE

"The law requires that each person working on this site or providing services related to this construction project must be covered by workers' compensation insurance. This includes persons providing, hauling, or delivering equipment or materials, or providing labor or transportation or other service related to the project, regardless of the identity of their employer or status as an employee."

"Call the Texas Workers' Compensation Commission at 512 - 440 - 3789 to receive information on the legal requirements for coverage, to verify whether your employer has provided the required coverage, or to report an employer's failure to provide coverage."

# 11.5 PERFORMANCE BOND AND LABOR AND MATERIAL PAYMENT BOND

Supplement to Article 11.

11.5.1 The Performance Bond Form and Labor and Material Payment Bond - Substitute the following for Subparagraph 11.4.1 as set forth below:

The Contractor shall furnish a Statutory Performance Bond in an amount equal to One Hundred Percent (100%) of the Contract Sum as security for the faithful performance of this Contract and also a Statutory Labor and Material payment Bond in an amount not less than One Hundred Percent (100%) of the Contract Sum as security for the payment for all persons performing labor on the project under this Contract and furnishing materials in connection with this Contract. The Performance Bond and the Labor and Material Payment Bond may be in one or in separate instruments in accordance with local law and shall be delivered to the Owner not later than the date of execution of the Contract.

- A. The Contractor shall provide an affidavit showing proof that his bonding company meets the following criteria:
  - 01 The Bonding Company must be domiciled in the United States.
  - 02 The Bonding Company must be licensed in the State of Texas.
  - 03 The Bonding Company must be acceptable to the Owner.
- B. Bond forms shall be subject to the Owner's approval. File copies of the bond with the County Clerk and furnish the Owner a file receipt.
- C. Performance and payment bonds shall remain in force throughout the warranty period of the contract.
- D. The work will not be started until the bonds and issuing companies have been accepted as satisfactory by the Owner.
- E. The original bonds will be delivered to the Owner with an attached authorization power of attorney.

# **ARTICLE 13 - MISCELLANEOUS PROVISIONS**

13.6 INTEREST

Delete paragraph in its entirety.

# **ARTICLE 14 - TERMINATION OR SUSPENSION OF THE CONTRACT**

Add the following Subparagraph:

- 14.2.5 Contractor hereby assigns the Owner any and all claims for overcharges associated with this Contract which arise under the antitrust laws of the United States, 15 U.S.C.A. Section 1 ET.SEQ. (1973).
- 14.4 TERMINATION BY THE OWNER FOR CONVENIENCE

Delete Subparagraph 14.4.3 in its entirety and replace with the following:

14.4.3 In case of such termination for the Owner's convenience, the Contractor shall be entitled to receive payment from the Owner on the same basis provided in Subparagraph 14.1.3.

# **ARTICLE 15 - LABOR STANDARDS**

15.1 CLAIMS

Delete the text of Subparagraphs 15.1.1 and 15.1.3 in their entirety and substitute the following:

- 15.1.1 Definition. A Claim is a demand or assertion by one of the parties seeking, as a matter of right, adjustment or interpretation of Contract terms, payment of money, extension of time or other relief with respect to the terms of the Contract. The term "Claim" also includes other disputes and matters in question between the Owner, Architect and Contractor arising out of or relating to the Contract.
- 15.1.3 Continuing Contract Performance. Pending final resolution of a Claim, unless otherwise agreed in writing, the Contractor shall proceed diligently with performance of the Contract.
- 15.2 INITIAL DECISION

Delete the text of Subparagraph 15.2.1 and 15.2.2 in their entirety and substitute the following:

15.2.1 Decisions of Architect. Claims including those alleging an error or omission by the Architect, shall be referred initially to the Architect for action. If the parties are unable to agree, appeal shall be as stated at ARTICLE 4A.

ARTICLE 15A: Any claim, disputes or matters arising out of this contract between the Architect, Owner and Contractor or any combination of those parties shall be submitted to a court of appropriate jurisdiction.

15.2.2 The Architect will review Claims and taken one or more of the following preliminary actions within ten (10) days of receipt of a Claim: (1) request Additional supporting data from the claimant; (2) submit a schedule to the parties indicating when the Architect expects to take action; (3) reject the Claim in whole or in part, stating reasons for rejection,; (4) recommend approval of the Claim by the other party or (5) suggest a compromise. The Architect shall notify the surety, if any, of the nature and amount of the Claim.

Add the following Subparagraph:

15.2.9 If a Claim has not been resolved after consideration of the foregoing and of further evidence presented by the parties or requested by the Architect, the Architect will notify the parties in writing that the Architect's decision will be made within seven (7) days.

Upon expiration of such time period, the Architect will render to the parties the Architect's written decision relative to the Claim, including and change in the Contract Sum or Contract Time or both. If there is a surety and there appears to be possibility of a Contractor's default, the Architect shall notify the surety and request the surety's assistance in resolving the controversy.

15.4 ARBITRATION

Delete Subparagraphs 15.4.1 thru 15.4.4.3 in their entirety and all other references to arbitration.

Add the following Subparagraph:

15.4.1 Any claims, disputes, or matters arising out of this contract between the Contractor and the Owner or the Architect not settled by mediation, shall be submitted to a court of appropriate jurisdiction. It is understood and agreed that, in the event that any dispute, controversy, or conflict arises during the design and construction of the Project or following its completion, the parties hereto will cooperate in good faith, if possible, to resolve the issues without resorting to litigation. Should the parties be unable to reach agreement, an independent mediator may be selected by mutual consent of the parties to assist in a further effort to resolve the dispute. Furthermore, if the parties mutually agree to mediation, each party included in the mediation will bear an equal share of all costs related to the mediation.

"Any claims, disputes or matters arising out of the contract will be submitted to mediation only upon the mutual consent of the parties. In the event that mutual consent is not achieved, the parties are free to pursue any claims, disputes or matters in any manner allowed by law."

Add the following Article:

# **ARTICLE 16 – LABOR STANDARDS**

- 16.1 PREVAILING WAGE RATES
- 16.1.1 Contractor and each Subcontractor shall pay to all laborers, workmen, and mechanics employed in execution of this Contract not less than rates set forth by law and as noted in the following Wage Rate Scale, for each craft or type of workman or mechanic needed to execute Contract. Strict conformance to all Davis-Bacon and Related Acts are the sole responsibility of the Contractor.
- 16.1.2 Determination of prevailing wages shall not be construed to prohibit payment of more than rates named.

# **DOCUMENT CC - TERMS AND CONDITIONS**

ITEMS BELOW APPLY TO AND BECOME A PART OF CONTRACT <u>UNLESS</u> SUPERSEDED BY ANY ATTACHED SUPPLEMENTAL SPECIAL REQUIREMENTS/CONDITIONS ON SPECIFICATIONS IN WHICH CASE ATTACHED CONDITIONS WILL PREVAIL.

# ANY EXCEPTION REQUEST MUST BE IN WRITING ON THE DEVIATION SHEET PROVIDED

# GOVERNING LAW

This invitation to propose is governed by the Texas Education School Law Bulletin, Texas Local Government Code, Local School Policy, and any law prevailing which hold to school districts as a whole. Offeror is further advised that these requirements shall be fully governed by the laws of the State of Texas and that Galveston Independent School District may request and rely on advice, decisions and opinions of the Attorney General of Texas and the School Attorney concerning any portion of these requirements. On any contract involving construction, repair, or renovation, successful Offeror must furnish performance and payment bonds as prescribed by Texas State Law. All bonds must be issued by a surety approved by the State of Texas.

# PROPOSAL FORM COMPLETION

Proposal should be submitted on attached form <u>ONLY</u>. Fill out and return to the Purchasing Department one (1) complete proposal form. Each proposal shall be placed in a separate envelope, sealed, and properly identified with the <u>proposal title and date to be opened</u>. An authorized representative should sign the Cover Sheet.

# PROPOSAL RETURNS

Proposals must be received in the Purchasing Office <u>before</u> the hour and date specified. Late proposals will not be considered under any circumstances. FAXED PROPOSALS WILL NOT BE ACCEPTED.

# ADDENDA

When specifications are revised, the Galveston Independent School District Director of Purchasing will issue an addendum addressing the nature of the change. Changes will become part of the original packet. Offerors must sign the addendum and include it in the returned proposal packet.

# HOLD HARMLESS AGREEMENT

Contractor, the successful Offeror, shall indemnify and hold Galveston ISD harmless from all claims for personal injury, death and/or property damage resulting directly or indirectly from contractor's performance. Contractor shall procure and maintain, with respect to the subject matter, appropriate insurance coverage including, as a minimum, public liability and property damage with adequate limits to cover contractor's liability as may arise directly or indirectly from work performed under terms of this proposal. Certification of such coverage must be provided to the school district upon request.

# **SEVERABILITY**

If any section, subsection, paragraph, sentence, clause, phrase or word of these requirements or the specifications shall be held invalid, such holding shall not affect the remaining portions of these requirements and the specifications. It is hereby declared that such remaining portions would have been included in these requirements and the specifications as though the invalid portion had been omitted.

# <u>TAXES</u>

GISD is exempt from Federal Excise Tax, State Tax, and Local Tax. Do not include tax in proposal. If it is determined that tax was included in the proposal, it will not be included in the tabulation or any awards. Texas Sales Tax Exemption Certificates will be furnished upon written request to the Galveston ISD Director of Purchasing.

# FISCAL FUNDING

A multi-year contract or lease/purchase arrangement (if requested by the Special Requirements/Instructions, or any contract continuing as a result of an extension option, must include

BEAM Professionals Project No. 18267R

fiscal funding out. If, for any reason, funds are not appropriated to continue the lease or contract, said lease or contract shall become null and void on the last day of the current appropriation of funds. No contract or agreement may exceed a period of one (1) year from the date of contract; however, it may include options to renew at one (1) year with increments not to exceed two (2) years extension (three [3] years total). Should funding for the continuance of this contract be withdrawn by the Board, GISD reserves the right to cancel without pecuniary risk or penalty upon thirty (30) days written notice of intent.

# PRICING

Prices for all goods and/or services shall be firm for the duration of this said contract and shall be stated on the Pricing/Delivery Information form. Proposal prices must be firm for <u>Acceptance by Board of</u> <u>Trustees</u> 120 days from proposal opening date, <u>UNLESS</u> specified otherwise in the proposal. **Prices shall be inclusive**: All prices must be written in ink or typewritten. Proposal unit price on quantity specified, extend, and show total. In case of errors in extension, unit prices shall govern. Pricing on all transportation, freight, and other charges are to be prepaid by the proposer and included in the proposal prices. If there are any additional charges of any kind, other than those mentioned above, specified or unspecified, Offeror MUST indicate the items required and state cost or forfeit the right to payment for such items.

#### SILENCE OF SPECIFICATIONS

The items furnished under this contract shall be new, unused of the latest product in production to commercial trade and shall be of the highest quality as to materials used and workmanship. Manufacturer furnishing these items shall be experienced in design and construction of such items and shall be an established supplier of the item proposed. All interpretation of specifications shall be based on the following statement: The apparent silence of specifications as to any detail, or the apparent omission from it of a detailed description concerning any point, shall be regarded as meaning that only the best commercial practice is to prevail and that only material and workmanship of the finest quality are to be used.

#### SUPPLEMENTAL MATERIALS

Offerors are responsible for including all pertinent product data in the returned proposal package. Literature, brochures, data sheets, specification information, completed forms requested as part of the proposal package and any other facts, which may affect the evaluation, and subsequent contract award should be included. Failure to include all necessary and proper supplemental material may be cause to reject the entire proposal.

# MATERIAL SAFETY DATA SHEETS

Under the "Hazardous Communication Act", commonly known as the "Texas Right To Know Act", an Offeror must provide MSDS sheets <u>with each delivery</u>. Material Safety Data Sheets, which are applicable to hazardous substances, defined in the Act should be mailed to the Purchasing Department, 3904 Ave T, Galveston, TX 77550. Failure of the Offeror to furnish this documentation will be cause to reject any proposal applying thereto.

# NAME BRANDS

Specifications <u>may</u> reference name brands and model numbers. It is not the intent of Galveston ISD to restrict these proposals in such cases, but to establish a desired quality level of merchandise or to meet a pre-established standard due to like existing items. Offerors may offer items of equal stature and the burden of proof of such stature rests with the offeror. GISD shall act as sole judge in determining equality and acceptability of products offered. GISD may require a sample of material or to visually inspect a site that has used requested material. If this is to be requested, the offeror has three (3) working days to respond to the request.

# COLOR SELECTION

Determination of colors of materials is a right reserved by the using department unless otherwise specified in the proposal. Unspecified colors shall be quoted as standard colors, NOT colors which require upcharges or special handling. Unspecified fabrics or vinyl should be construed as medium grade. If offeror fails to get color/material approvals prior to delivery of merchandise, the using department may refuse to accept the items and demand correct shipment without penalty, subject to other legal remedies.

#### **EVALUATION**

Evaluation shall be used as a determinant as to which proposed items or services are the most efficient and/or most economical for Galveston ISD. It shall be based on all factors, which have a bearing on price and performance of the items in the user environment. All proposals are subject to tabulation by the GISD Purchasing Department and recommendation to the Board of Trustees. Compliance with all proposal requirements, delivery and needs of the using department are considerations in evaluating proposals. **Pricing is NOT the only criteria for making a recommendation**. The Galveston Director of Purchasing reserves the rights to contact any Offeror, at any time, to clarify, verify or request information with regard to any proposal.

#### INSPECTIONS

Galveston ISD reserves the right to inspect any item(s) or service location for compliance with specifications and requirements and needs of the using department. If an Offeror cannot furnish a sample of a proposed item, where applicable, for review, or fails to satisfactorily show an ability to perform, GISD can reject the proposal as inadequate. All items proposed must be new, in first class condition, included containers suitable for shipment and storage, unless otherwise indicated on the proposal.

#### TESTING

Galveston ISD reserves the right to test equipment, supplies, material and goods proposed for quality, compliance with specifications and ability to meet the needs of the user. Demonstration units must be available for review. Should the goods or services fail to meet requirements and/or be unavailable for evaluation, the proposal is subject to rejection.

#### **DISQUALIFICATION OF OFFEROR**

Upon signing this document, an Offeror offering to sell supplies, materials, services, or equipment to Galveston ISD certifies that the Offeror has not violated the antitrust laws of this state (codified in Texas Business and Commerce Code §15.01, et seq., as amended) or the federal antitrust laws. The Offeror also certifies that it has not communicated directly or indirectly the proposal made to any competitor or any other person engaged in such line of business. Any or all proposals may be rejected if GISD believes that collusion exists among the Offerors. Proposals in which the prices are obviously unbalanced may be rejected. If multiple proposals are submitted by an Offeror and after the proposals are opened, one of the proposals is withdrawn, the result will be that all of the proposals submitted by that Offeror will be withdrawn; however, nothing herein prohibits a vendor from submitting multiple proposals for different products or services.

#### AWARD

Galveston ISD reserves the right to award this contract on the basis of LOWEST AND BEST PROPOSAL in accordance with the laws of the State of Texas, to waive any formality or irregularity, to make awards to more than one Offeror, to reject any or all proposals. In determining to whom to award a contract, the district may consider: purchase price, reputation of vendor's goods, quality of vendor's goods and services, extent to which goods or services meet the districts needs, vendor's past relationship with the district, the impact on the ability of the district to comply with laws and rules relating to historically underutilized businesses, total long-term cost to the district to acquire the vendor's goods and services, and any other relevant factor that a private business entity would consider in selecting a vendor.

#### ASSIGNMENT

The successful Offeror may not assign, sell or otherwise transfer this contract without written permission of Galveston ISD.

# TERM CONTRACTS

The term of this contract shall be for, or approximately for, one year with an option to extend for two additional years should both parties agree.

#### MAINTENANCE

Maintenance required for equipment proposals should be available by a manufacturer authorized maintenance facility. Costs for this service shall be shown on the Pricing/Delivery Information form. If Galveston ISD opts to include maintenance, it shall be so stated in the purchase order and said cost will be included. Service will commence only upon expiration of applicable warranties and should be priced accordingly.

#### CONTRACT OBLIGATION

Galveston ISD Board of Trustees must award the contract and the Director of Purchasing or other person authorized by the Galveston ISD must sign the contract before it becomes binding on Galveston ISD or the Offerors. Department heads are NOT authorized to sign agreements for Galveston ISD. Binding agreements shall remain in effect until all products and/or services covered by this purchase have been satisfactorily delivered and accepted.

#### TITLE TRANSFER

Title and Risk of Loss of goods shall not pass to Galveston ISD until Galveston ISD actually receives and takes possession of the goods at the point or points of delivery. Generally, deliveries may be made between 8:00 a.m. and 3:30 p.m., Monday through Friday. The place of delivery shall be shown under the "Deliver To:" address.

#### WARRANTIES

Offerors shall furnish all data pertinent to warranties or guarantees, which may apply to items in the proposal. Offerors may not limit or exclude any implied warranties. Offeror warrants that product sold to Galveston ISD shall conform to the standards established by the U.S. Department of Labor under the Occupational Safety and Health Act of 1970. In the event product does not conform to OSHA Standards, where applicable, Galveston ISD may return the product for correction or replacement at the Offeror's expense. If Offeror fails to make the appropriate correction within a reasonable time, Galveston ISD may correct at the Offeror's expense.

# PURCHASE ORDER AND DELIVERY

The successful Offeror shall not deliver products or provide services without a Galveston ISD Purchase Order, signed by an authorized representative of the GISD Purchasing Department, Director of Purchasing. The Offeror in the proper place on the Pricing/Delivery Information form shall indicate the fastest, most reasonable delivery time. Any special information concerning delivery should also be included, on a separate sheet, if necessary. All items shall be shipped **F.O.B. Destination/Freight Prepaid** unless otherwise stated in the specifications. Every tender or delivery of goods must fully comply with all provisions of these requirements and the specifications including time, delivery and quality. Nonconformance shall constitute a breach that must be rectified prior to expiration of the time for performance. Failure to rectify within the performance period will be considered cause to reject future deliveries and cancellation of the contract by Galveston ISD without prejudice to other remedies provided by law. Where delivery times are critical, GISD reserves the right to award accordingly. Delivery: Proposal should show the number of days required delivering the articles to the specified location under normal conditions after receipt of an order. Failure to state the delivery time obligates the Offeror to make delivery in 14 calendar days. Delivery shall be made during normal working hours of Warehouse 8:00 am till 3:30 p.m., unless prior approval has been obtained from the Galveston Independent School District (GISD).

#### **CONTRACT RENEWALS**

Renewals may be made ONLY by written agreement between Galveston ISD Purchasing Department and the Offeror.

# INVOICES AND PAYMENTS

Offerors shall submit invoices in duplicate on each purchase order or purchase release after each delivery, indicating the purchase order number. Invoices must be itemized. Any invoice, which cannot be verified by

BEAM Professionals Project No. 18267R

the contract price and/or is otherwise incorrect, will be returned to the Offeror for correction. Under term contracts, when multiple deliveries and/or services are required, the Offeror may invoice following each delivery and GISD will pay on invoice. Contracts providing for a monthly charge will be billed and paid on a monthly basis only. All billing statements shall be mailed to: Galveston Independent School District Accounts Payable, P.O. Box 660, Galveston, Texas 77553-0660.

### **TERMINATION**

Galveston ISD reserves the right to terminate the contract for default if Offeror breaches any of the terms therein, including warranties of Offeror or if the Offeror becomes insolvent or commits acts of bankruptcy. Such right of termination is in addition to and not in lieu of any other remedies which Galveston ISD may have in law or equity. Default may be construed as, but not limited to, failure to deliver the proper goods and/or services within the proper amount of time, and/or to properly perform any and all services required to GISD's satisfaction and/or to meet all other obligations and requirements. Contracts may be terminated without cause upon thirty (30) days written notice to either party unless otherwise specified.

#### PAYMENT/PERFORMANCE BONDS

If a construction contract is over \$25,000 a payment bond and possibly a performance bond will be required. Any construction contract valued at/over \$100,000 will require a performance bond.

# END OF DOCUMENT CC

# **DOCUMENT CD - SPECIAL TERMS AND CONDITIONS**

#### The terms of this agreement will be mutually agreed upon.

If at any time the Offeror's service is not in accordance with the instructions, conditions, and specifications set forth by GISD, such act will constitute grounds for the cancellation of the contractual agreement. A thirty-day (30) notice will be given upon cancellation of contract agreement. Agreement can be cancelled by GISD for non-performance.

Agreement will include completion of all requested contractual work requested with final approval by the GISD.

Offeror may not assign or transfer this agreement nor its rights and privileges granted under this agreement, either in whole or part, without written consent of GISD. If any portion of project is sub-let with GISD approval, GISD must be provided with documentation that said contractor has all applicable license and insurance as stated for original contractor.

GISD will require a payment bond and possibly a performance bond if contract is valued at over \$25,000.00, this cost to contractor must be included as part of original cost, not to be invoiced separate as to GISD paying cost for said documents.

Any changes in the scope of work must be approved in writing by GISD and attached to original proposal document in the purchasing department. Amendments will then be made to the Purchase Order and contractor will be provided a revised cop of addition to the contract.

GISD and contractor will perform a walk-thru when contractor stated 50% of stated contract terms have been fulfilled. A progression payment will be made at the approval of both parties at 50%, and 75% of completion of contract. There will be a 5% retainage fee of total project until final walk-thru and completion of project. Upon approval of GISD that the contract has been fulfilled, final payment will be initiated.

GISD reserves the right to accept or reject all or any part of any proposal, waive minor technicalities, and award the proposal to serve the best interest of GISD.

All items and services must be in full compliance to all allocable Federal, State, and Local laws, ordinances, codes and regulations.

GISD reserves the right to cancel this contract without pecuniary risk or penalty upon written notice of the intent. The board reserves the right to cancel this contract upon written notice of the intent.

If at any time the Offeror fails to fulfill or abide by the terms, conditions, or specifications of this contract, GISD reserves the right to cancel upon thirty days written notification of the intent.

# END OF DOCUMENT CD

# DOCUMENT CE - INSURANCE

Offeror shall provide evidence of liability insurance. GISD retains the right of approval for insurance coverage.

Copies of the successful Offeror's liability insurance and workman's compensation certificates will be required. This certificate does not amend, extend or alter the coverage afforded by the policies below:

Insurance Requirement	Limits:			
General Liability : Gene Products – Cor Commercial General Li	\$1,000,000 \$1,000,000			
Personal & Adv. Injury	-	\$1,000,000		
Claims Made Occurren	\$1,000,000			
Fire Damage (Any one	\$50,000			
Medical Expense (any o	\$1,000			
Automobile Liability Any Auto Bodily Injury	Combined Single Limit	\$1,000,000		
All Owned Autos	(Per Person)			
Scheduled Autos	Bodily Injury			
Hired Autos				
Non-Owned Autos	Property Damage			
Garage Liability				

Umbrella Form - Excess Liability (Offeror to specify limits of coverage).

Worker's Comp. and Employer's Liability Statutory Each Accident \$500,000 Disease – Policy

The insurance requirements as listed above also apply to any sub-contractor(s) in the event that any work is sublet. The contractor is responsible to insure that the sub-contractor(s) meet(s) the minimum insurance requirements as by law.

Should any of the above-described policies be canceled before the expiration date, the issuing company will mail a thirty (30) days written notice to the certificate holder, GISD.

The contractor shall agree to waive all rights of subrogation against GISD, its officials, employees and volunteers for losses arising from work performed by the contractor for GISD.

Vendor shall indemnify and hold harmless the GISD and its Board of Trustees, officers agents, employees from all suits, actions, losses, damages, claims or liability of any character, type, or description, including but not limited to, all expenses of litigation, court cost, penalties, and attorney's fees the GISD incurs defending any action, suit or claim from any source whatsoever and of any kind or nature arising directly or indirectly on the part of Offeror, it agents, servants, employees, contractors, and suppliers, out of the operation under this agreement.

Refer to Document CB, Supplementary Conditions for additional insurance requirements.

Offerors shall submit a certificate of insurance for the above insurance requirements with their proposal.

# END OF DOCUMENT CE

# **DOCUMENT CF - GENERAL REQUIREMENTS**

# **RFP** Document

The Offeror shall return the entire RFP document with the proposal, with all signature and information lines filled in. Signatures shall be legible and signed in black ink.

<u>Detailed Description of Deliverables</u> - The Offeror's response to the rates constitutes the description of the deliverable services. The response must indicate the manner in which the proposed deliverables will fulfill the requirements of the specifications. As detailed an explanation as possible must be included.

<u>Submission of Proposals</u> - All proposals must be received at the address shown on the Cover Sheet of this RFCSP no later according to the GISD's Purchasing Office Date/Time stamp on **April 4, 2019 at 2:00 p.m.** One original and two (2) separately bound copies (total of 3) of the proposal shall be delivered to GISD on or before the due date for the proposal specified above. The proposal must follow the format indicated in this section. Unnecessarily elaborate proposals, beyond that sufficient to present a complete effective responsive proposal, are not desired. Repetitious references to a catalog, brochure, or standard contract, which are not relevant to the questions posed in this RFP, may be considered non-responsive.

All proposal packages must be clearly marked with the RFP title, due date, and time as shown on the RFP Cover Sheet.

Proposals received by GISD after the due date and time will not be considered for award. Any late proposals will be returned unopened to the offeror upon request only.

<u>Proposal Inquiries</u> – All inquiries and requests for information affecting this RFP must be submitted in writing to the GISD Purchasing Department, at the issuing office address, to the attention of the Purchasing Coordinator. GISD reserves the right to determine whether questions merit response. If questions are answered, written copies of the questions and answers will be shared with all prospective Offerors who received a copy of the original RFP. Only answers signed by the Purchasing Coordinator of GISD will be considered as official by GISD. No negotiations, decisions, or actions shall be initiated by any Offeror or as a result of any verbal discussion with any GISD employee.

<u>Post-Proposal Conference</u> – If questions arise as to the content of the proposal during evaluation, GISD may request a conference with any or all Offerors. Such conferences will not be for Offeror presentation, but rather, to allow GISD to ask questions regarding proposal content.

<u>Correct Errors of Omissions</u> – GISD may, at its option, allow all Offerors, responding to the RFP, a three-(3) calendar day period to correct errors of omissions to their response to the RFP. Should this necessity arise, GISD will contact each Offeror affected. Each Offeror must submit written corrections to it's proposal within three (3) calendar days of notification. The intent of this option is to allow proposals with only technical or minor errors or omissions to be corrected in lieu of the GISD being forced to reject all proposals and reissue the RFP. GISD reserves the sole right to determine whether an error is minor or major. Major errors or omissions, such as not including prices, will not be considered by GISD as a minor or technical error or omission and will result in disgualifying the proposal from further evaluation.

<u>Withdrawal of Offers</u> - Responses to the RFP may be withdrawn by written notice received prior to the due date and time specified for receipt of the proposals.

#### Proprietary/Confidential Information

Offerors to the RFP are advised that all materials submitted to the GISD as a part of their proposal, become and remain the property of the GISD, and are subject to the Public Information Act, and therefore, may be viewed and/or copied by any member of the public, including news agencies and competitors. <u>Therefore, all materials deemed "Proprietary/Confidential" by the Offeror must be</u> conspicuously marked as such. Preferably, each document/page deemed as such, should exhibit the word or words "PROPRIETARY/CONFIDENTIAL", stamped in red ink, in large letters. Materials so marked will remain the property of GISD, will not be made available to the public, and

consequently cannot be returned to the Offeror. Failure on part of the Offeror to indicate the proprietary/confidential status for the applicable documents as recommended will release the GISD from any action or actions should said documents be made public. Offerors should note that entire proposals cannot be considered "Proprietary/Confidential", and any proposals so marked will be considered "non-responsive".

<u>Term of the Proposal</u> – All proposals made in response to the RFCSP must remain open and in effect for a period of not less than one (1) month after the due date for proposals. Any proposal selected by GISD for contract negotiations shall remain open and in effect beyond six (6) months until superseded by a fully executed contract or until GISD notifies the Offeror in writing that contract negotiations have ended, and the proposal has been rejected.

<u>Best and Final Offers (BAFO)</u> – GISD reserves the right to conduct discussions, either oral or written, with those Offerors determined by GISD to be reasonably susceptible to being selected for award. If discussions are held, GISD may request best and final offers. BAFO shall include:

Notice that discussions are concluded; Notice that this is the opportunity to submit written best and final offers; A common cut-off date and time for submission of the best and final offer; Notice of any changes in GISD's requirements; Notice that if any modification is submitted, it must be received by the cut-off time or it will not be considered.

<u>Rejection of Proposals</u> – GISD reserves the right to reject any or all proposals received, or to award a contract on the basis of initial offers received without discussions or clarifications. Therefore, the proposal should contain the Offeror's best price and technical response based on the RFP. GISD reserves the right to re-open discussions after receipt of best and final offers if it is clearly in GISD's best interest to do so. If discussions are re-opened, GISD will issue an additional request for best and final offers from all Offerors determined by GISD to be reasonably susceptible to being selected for award.

<u>Negotiations</u> - Following an evaluation of the best and final offers, GISD will select for negotiations the offer that is most advantageous to GISD, considering price or cost and the evaluation factors in the RFP.

<u>Clarification of Issues</u> – GISD also reserves the right to conduct clarifications to resolve minor issues. GISD retains sole discretion to determine whether contact with the Offerors is for clarification or discussion.

<u>Contract Components</u> – Any or all portions of this RFP and all portions of the Offerors proposal shall be incorporated by reference as part of the final contractual agreement.

<u>GISD Obligations</u> – GISD accepts no obligations for costs incurred by Offerors responding to this RFP or on being awarded the contract. GISD reserves the right to select a proposal without discussion with the Offerors. It is understood that proposals shall become part of GISD's official files. Retention of these proposals does not obligate GISD to any action. GISD reserves the right to reject any and all proposals received.

<u>Taxes</u> – GISD is exempt from state, federal, and local taxes, and will not be responsible for any taxes levied on the Contractor as a result of the contract resulting from this RFP.

<u>Equal Opportunity</u> – It is the public policy of GISD, at all levels of procurement, to promote equal opportunity in employment and in contracting opportunities, and to promote and encourage the participation of minority, small and other disadvantaged business entities in employment and contracting opportunities involving the State as fully as possible. GISD, therefore, committed to pursue such avenues in its employment and contracting activities which will further the goals of this policy.

Similarly, a demonstrated commitment consistent with the goals of this policy by those with whom the GISD does business, including those Offerors responding to the solicitation is highly desirable by GISD.

# END OF DOCUMENT CF

# For other counties or areas, go to <u>http://www.wdol.gov/dba.aspx</u>.

# DOCUMENT CG - PREVAILING WAGE RATE DETERMINATION INFORMATION

The following information is from Chapter 2258 Texas Government Code: Effective 1 March 2016

# 2258.021. Right to be Paid Prevailing Wage Rates.

- (a) A worker employed on a public work by or on behalf of the state or a political subdivision of the state shall be paid:
  - (1) not less than the general prevailing rate of per diem wages for work of a similar character in the locality in which the work is performed; and
  - (2) not less than the general prevailing rate of per diem wages for legal holiday and overtime work.
- (b) Subsection (a) does not apply to maintenance work.
- (c) A worker is employed on a public work for the purposes of this section if the worker is employed by a contractor or subcontractor in the execution of a contract for the public work with the state, a political subdivision of the state, or any officer or public body of the state or a political subdivision of the state.

#### 2258.023. Prevailing Wage Rates to be Paid by Contractor and Subcontractor; Penalty.

- (a) The contractor who is awarded a contract by a public body or a subcontractor of the contractor shall pay not less than the rates determined under Section 2258.022 to a worker employed by it in the execution of the contract.
- (b) A contractor or subcontractor who violates this section shall pay to the state or a political subdivision of the state on whose behalf the contract is made, \$60 for each worker employed for each calendar day or part of the day that the worker is paid less than the wage rates stipulated in the contract. A public body awarding a contract shall specify this penalty in the contract.
- (c) A contractor or subcontractor does not violate this section if a public body awarding a contract does not determine the prevailing wage rates and specify the rates in the contract as provided by Section 2258.022.
- (d) The public body shall use any money collected under this section to offset the costs incurred in the administration of this chapter.
- (e) A municipality is entitled to collect a penalty under this section only if the municipality has a population of more than 10,000.

#### 2258.051. Duty of Public Body to Hear Complaints and Withhold Payment.

A public body awarding a contract, and an agent or officer of the public body, shall:

- (1) take cognizance of complaints of all violations of this chapter committed in the execution of the contract; and
- (2) withhold money forfeited or required to be withheld under this chapter from the payments to the contractor under the contract, except that the public body may not withhold money from other than the final payment without a determination by the public body that there is good cause to believe that the contractor has violated this chapter.

Texas Gulf Coast Area

CLASSIFICATION	HOURLY RATE
Asbestos Worker	\$15.42
Bricklayers; Masons	\$18.34
Carpenters/Caseworker	\$21.50
Carpet Layers/Floor Installers	\$20.03
Concrete Finishers	\$16.13
Data Comm / Telecom Installer	\$23.50
Drywall Installers; Ceiling Installers	\$16.69
Electricians	\$22.44
Elevator Mechanics	\$30.00
Fire Proofing Installer	\$19.13
Glaziers	\$19.87
Heavy Equipment Operators	\$18.18
Insulators	\$16.16
Ironworkers	\$18.14
Laborers	\$11.81
Lather / Plasterer	\$18.03
Light Equipment Operators	\$15.21
Metal Building Assemblers	\$17.53
Millwrights	\$20.69
Painters/Wall Covering Installers	\$15.75
Pipefitters	\$25.70
Plumbers	\$26.50
Roofers	\$18.80
Sheet Metal Workers	\$20.46
Sprinkler Fitters	\$25.10
Steel Erector	\$19.33
Terrazzo Workers	\$19.67
Tile Setters	\$19.83
Waterproofers/Caulkers	\$19.00

This document was developed by PBK Architects, Inc. in strict accordance with the Texas Government Code Chapter 2258.

# Prevailing Wage Rates Worker Classification Definition Sheet

Asbestos Worker	Worker who removes & disposes of asbestos materials.				
Bricklayers/Masons	Craftsman who works with masonry products, stone, brick, block or any material substituting for those materials & accessories.				
Carpenter / Caseworker	Worker who builds wood structures or structures of any material which has replaced wood. Includes rough & finish carpentry, hardware and trim.				
Carpet Layer / Floor Installer	Worker who installs carpets and/or floor coverings-vinyl tile.				
Concrete Finisher	Worker who floats, trowels and finishes concrete.				
Data Communications/ Telecom Installer	Worker who installs data/telephone & television cable and associated equipment and accessories.				
Drywall / Ceiling Installer	Worker who installs metal framed walls & ceilings, drywall coverings, ceiling grids				
Electrician	Skilled craftsman who installs or repairs electrical wiring & devices. Includes fire alarm systems				
Elevator Mechanic	Craftsman skilled in the installation & maintenance of elevators.				
Fire Proofing Installer	Worker who sprays or applies fire proofing materials.				
Glazier	Worker who installs glass, glazing and glass framing.				
Heavy Equipment Operator	Includes, but not limited to, all Cat tractors, all derrick-powered, all power operated cranes, back-hoe, back-filler, power operated shovel, winch				
Insulator	Worker who applies, sprays or installs insulation.				
Iron Worker	Skilled craftsman who erects structural steel framing & installs structural concrete				
Laborer / Helper	Worker qualified for only unskilled or semi-skilled work. Lifting, carrying materials & tools, hauling, digging, clean-up.				
Lather / Plasterer	Worker who installs metal framing & lath. Worker who applies plaster to lathing and installs associated accessories.				
Light Equipment Operator	Includes, but not limited to, air compressors, truck crane driver, flex plane, building elevator, form grader, concrete mixer (less than 14cf), conveyer.				
Metal Building Assembler	Worker who assembles pre-made metal buildings.				
Millwright	Mechanic specializing in the installation of heavy machinery, conveyance, wrenches, dock levelers, hydraulic lifts & align pumps.				
Painter / Wall Covering Installer	Worker who prepares wall surfaces & applies paint and/or wall coverings, tape				
Pipefitter	Trained worker who installs piping systems, chilled water piping & hot water (boiler) piping, pneumatic tubing controls, chillers, boilers & associated				
Plumber	Skilled craftsman who installs domestic hot & cold water piping, waste piping, storm system piping, water closets, sinks, urinals, and related work.				
Roofer	Worker who installs roofing materials, Bitumen (asphalt & coal tar) felts, flashings, all types roofing membranes & associated products.				
Sheet Metal Worker	Worker who installs sheet metal products. Roof metal, flashings & curbs ductwork, mechanical equipment and associated metals.				
Sprinkler Fitter	Worker who installs fire sprinkler systems & fire protection equipment.				
Steel Erector	Worker who erects and dismantles structural steel frames of buildings and othe				
Terrazzo Worker	Craftsman who places & finishes Terrazzo.				
Tile Setter	Worker who prepares wall and/or floor surfaces & applies ceramic tiles to these				
Waterproofer / Caulker	Worker who applies water proofing material to buildings. Products include sealant, caulk, sheet membrane, liquid membranes, sprayed, rolled or brushed.				

# END OF DOCUMENT CG

# SECTION 01 11 00 - SUMMARY OF WORK

PART 1 - GENERAL

### 1.1 **PROJECT DESCRIPTION**

### A. The Project, MULTI-CAMPUS BUILDING ENVELOPE REPAIRS (PACKAGE 4) Galveston Independent School District GISD RFCSP #20-01

#### 1.2 SCOPE OF WORK

- A. The Work shall consist of roof replacement to the existing facility as described in the Contract Documents.
- B. Base Proposal No.1: Work to include building envelope repairs at:

#### Annex: 3904 Avenue T, Galveston, TX 77550

Remove existing gutters, downspouts, metal flashing beneath gutters and replace with new perimeter edge metal, gutters and downspouts per specifications. Flash new edge metal into existing single ply roof. Coat single ply roof with acrylic coating. Power wash, prep, prime and paint metal panels and canopies per specifications.

Waterproofing scope: Remove, grind, clean, and replace all soft sealants around window and door openings, at vertical control joints, expansion joints, and at all penetrations.

Alternate #1: Powerwash, prep, prime, and paint brick.

<u>Administration Building: 3904 Avenue T, Galveston, TX 77550</u> Waterproofing scope: Remove, grind, clean, and replace all soft sealants around window and door openings, at vertical control joints, expansion joints, and at all penetrations.

La Morgan Elementary: 1410 37th St, Galveston, TX 77550 Alternate #2: Clear Elastomeric Sealant To Be Applied To Entire Campus.

Waterproofing Scope: Apply soft sealants at all soffit perimeters on campus.

<u>Parker Elementary: 6802 Jones Dr, Galveston, TX 77551</u> Waterproofing scope: Remove, grind, clean, and replace all soft sealants around window and door openings, at vertical control joints, expansion joints, and at all penetrations. See specifications for joints larger than 2".

Wetglazing scope: Wetglaze all existing windows and doors with high quality silicone sealant per specifications. Include glazing of all metal joints with wetglazing. Contractor to ensure not to block weeps.

Waterproofing scope: Apply soft sealants at all soffit perimeters on campus.

Alternate #3: Clear elastomeric sealant to be applied to entire campus.

<u>Weis Collegiate Academy:7100 Stewart Rd, Galveston, TX 77551</u> Wetglazing Scope: Wetglaze all existing windows and doors with high quality silicone sealant

Per specifications. Include glazing of all metal joints with wetglazing. Contractor to ensure not to block weeps.

Waterproofing scope: Apply soft sealants at all soffit perimeters on campus.

Alternate Proposal No.1: POWER WASH, PREP, PRIME, AND PAINT BRICK AT GISD ANNEX.

1. This Alternate shall provide pricing to add the scope of work outlined in the drawings.

Alternate Proposal No.2: CLEAR ELASTOMERIC SEALANT TO BE APPLIED TO ENTIRE CAMPUS AT LA MORGAN ELEMENTARY.

1. This Alternate shall provide pricing to add the scope of work outlined in the drawings.

Alternate Proposal No.3: : CLEAR ELASTOMERIC SEALANT TO BE APPLIED TO ENTIRE CAMPUS AT PARKER ELEMENTARY.

1. This Alternate shall provide pricing to add the scope of work outlined in the drawings.

# 1.3 BUDGET

A. TBD

# 1.4 SALVAGED MATERIALS

- A. Owner may salvage all items deemed reusable or necessary to keep from facilities to be demolished prior to the start of demolition.
- B. Contractor shall remove and turn additional items over to the Owner, as directed.
- C. Contractor shall demolish, remove and salvage all other items of demolished work.

# 1.5 CONTRACTS AND USE OF SITE

- A. Contractor Use of Premises:
  - 1. Confine operations at site to areas permitted by:
    - a. Law
    - b. Ordinances
    - c. Permits
    - d. Contract Documents
  - 2. Do not unreasonably encumber site with materials or equipment.
  - 3. Assume full responsibility for protection and safekeeping of products stored on premises.
  - 4. Obtain and pay for use of additional storage or work areas as needed for operations.
  - 5. Contractor shall establish secured staging area for work and coordinate and provide for safe passage and exit from building areas during construction, as determined by City and District officials.
  - 6. Contractor shall coordinate all construction activities with Owner.
  - 7. Owner reserves the right to perform construction operations with its own forces or to employ separate contractors on portions of the Project. Contractor shall coordinate with this work in terms of providing site access, work space, and storage space, cooperation of work forces, scheduling, and technical requirements.
  - 8. Coordinate all utility shutdowns with Owner and, as required, with local utility companies, prior to commencement of shutdown.

# B. Owner Occupancy:

- 1. Partial Owner Occupancy: The Owner reserves the right to place and install equipment in completed areas of the building, prior to Substantial Completion provided that such occupancy does not interfere with completion of the Work. Such placing of equipment and partial occupancy shall not constitute acceptance of the total Work.
- 2. A Certificate of Substantial Completion will be executed in accordance with conditions of the Contract.
- 3. Contractor shall obtain a Certificate of Occupancy from local building officials prior to Owner occupancy.
- 4. Prior to partial Owner occupancy, mechanical and electrical systems shall be fully operational. Required inspections and tests shall have been successfully completed. Upon occupancy the Owner will provide operation and maintenance of mechanical and electrical systems in occupied portions of the building.
- 5. Prior to partial Owner occupancy, emergency and life safety systems shall be fully operational. Emergency and life safety systems include, but are not limited to, fire sprinkler systems, fire alarm systems, and emergency egress devices. For emergency exiting purposes, the path of travel shall be clearly delineated and functional. If required, temporary barricades shall separate on-going construction from occupied spaces as allowed by the governing agency holding jurisdiction over the Project. Required inspections and tests shall have been successfully completed. Upon occupancy the Owner will provide operation and maintenance of emergency and life safety systems in occupied portions of the building.
- C. Owner-Furnished Items:
  - 1. The Owner may provide items to the Contractor for installation in accordance with manufacturer's recommendations and instructions.
  - 2. The Owner will arrange and pay for delivery of Owner-furnished items in accordance with the Contractor's Construction Schedule, and will inspect deliveries for damage.
  - 3. If Owner-furnished items are damaged, defective or missing, through no fault of the Contractor, the Owner will arrange for replacement.
  - 4. The Contractor is responsible for designating the delivery dates of Owner-furnished items in the Contractor's Construction Schedule and for receiving, unloading and handling Owner-furnished items at the site. The Contractor is responsible for protecting Owner-furnished items from damage, including damage from exposure to elements, and to repair or replace items damaged as a result of his operations.
- D. Coordination with Owner's Forces or Owner's Contractors:
  - 1. Provide site access, space allocation, scheduling, scheduling coordination, coordination of work forces and coordination of technical requirements with contractors that may be selected and employed by Owner to perform work simultaneously and in conjunction with the Work, which may include, but shall not be limited to the following, as applicable to the Project:
    - a. Materials Inspection and Testing Agency
    - b. HVAC Testing, Adjusting, Balancing Agency
    - c. Energy Management System Contractor
    - d. Data and Cabling System Contractor
    - e. Telephone System Contractor
    - f. Modular Furniture Installer
    - g. Lighting and Sound
    - h. Surveying

# 1.6 **PROTECTION OF EXISTING PROPERTY**

- A. Contractor shall provide and maintain adequate protection of all Owner's existing property during duration of Project.
- B. Contractor shall verify location of all existing underground pipelines on site with the owner of such pipelines and authorities having jurisdiction and shall provide and maintain adequate protection of all such pipelines during duration of Project.
- C. Protection of Trees:
  - 1. Provide wood barricades around trees and shrubs at their drip line in traffic areas to protect them from construction operations until Substantial Completion, or until barricade removal is directed by Architect.

# 1.7 USE OF ASBESTOS FREE MATERIALS, PRODUCTS AND SYSTEMS

A. The Contractor is reminded to refer to Document AB, Instructions to Offerors for requirements regarding asbestos containing materials (ACM).

# PART 2 - PRODUCTS

# 2.1 MATERIALS

A. Refer to Specification Sections.

# PART 3 - EXECUTION

# 3.1 CONSTRUCTION SCHEDULE

- A. The Owner has a critical need for the work to begin on November 30, 2020 and shall have the entire work Substantially Complete by February 22, 2020.
- B. Refer to Section 01 32 16 for other scheduling requirements, and to Document CB Supplementary Conditions for information concerning liquidated damages.

# END OF SECTION

# SECTION 01 21 00 - ALLOWANCES

#### **PART 1 - GENERAL**

#### 1.1 CONDITIONS

- A. Cash allowances are hereby established for Owner's Contingency, Scope of Work items and materials in the amounts listed below and shall be included in the Contract Sum. These sums shall be reconciled in accordance with Article 3.8 of the General Conditions.
- B. Allowances for materials, such as brick, tile, etc., shall be for the net cost of materials only, without sales tax, delivered and unloaded at the jobsite. The party who makes the purchase (Contractor or subcontractor) shall include handling costs on site, labor, overhead, profit and other expenses contemplated for each allowance in the Contractor's Sum and not in the allowance. Include labor under allowance, only when labor is specified to be included.
- C. Allowances for Scope of Work, such as Owner's contingency, graphics, technology, etc., will be adjusted, as necessary, to reflect the difference between the allowance amount stated and Contractor's handling costs, cost of materials, without sales tax, plus labor, subcontract costs, with overhead and profit markup, and any other reasonable costs, except the Contractor's overhead and profit, which is not allowed.
- D. Contractor shall cause the work covered by these allowances to be performed for such amounts and by such persons as the Architect may direct or by persons selected by competitive sealed proposals, but he will not be required to employ persons against whom he makes reasonable objection. If any items cost less than the amount listed, the Owner shall be given a credit in the amount of the difference. If the Owner so desires, credits in one allowance category may be transferred to any other allowance category. If any items cost more than the amount listed, such adjustment will include additional handling costs on the site, labor, installation costs, overhead, profit and other expenses resulting to the Contractor or subcontractor from any increase over the original allowance, unless such increase is funded by a transfer of funds from other allowances in which case no overhead and profit will be allowed. If the final cost of all allowances, when determined, is more or less than the sum of the allowances, the Contract Sum will be adjusted accordingly by Change Order.
- E. Contractor shall proceed with the work in question only after receiving written directions executed by the Owner and the Architect. Such direction will be provided by an Allowance Expenditure Authorization prepared by the Architect and executed by Owner, Architect and Contractor. Owner will not be obligated to pay the cost of any work completed without prior authorization.
- F. Unexpended balance of allowance sums shall revert to the Owner in the final settlement of the Contract.

# PART 2 - PRODUCTS

Not Used

# PART 3 - EXECUTION

# 3.1 ALLOWANCES

- A. Contingency Allowance for Base Proposal: \$50,000.00
  - 1. Contractor shall include the amount indicated above in his Base Proposal as a contingency to cover the cost of hidden, concealed or otherwise unforeseen conditions which develop during completion of the work. Contractor shall be allowed to recover all costs associated with the completion of work under this contingency, however, no overhead or profit will be allowed.
- B. Unit Price Allowance No.1 (to include 400 LF in each dimension in base bid): \$TBD
  - 1. Contractor shall include the amount indicated above in his Base Proposal as a contingency to cover the cost of hidden, concealed or otherwise unforeseen conditions which develop during completion of the tuck-pointing work. Contractor shall be allowed to recover all costs associated with the completion of work under this contingency, however, no overhead or profit will be allowed.

# END OF SECTION 01 21 00

# SECTION 01 22 00 - MEASUREMENT AND PAYMENT (UNIT PRICES)

### PART 1 - GENERAL

#### 1.1 SECTION INCLUDES

A. Measurement and payment criteria applicable to portions of the Work performed under a unit price payment method.

B. Defect assessment and non-payment for rejected work.

# 1.2 AUTHORITY

A. Measurement methods delineated in the individual specification sections complement the criteria of this Section. In the event of conflict, the requirements of the individual specification section govern.

B. Take all measurements and compute quantities. The Architect will verify measurements and quantities.

### 1.3 UNIT QUANTITIES SPECIFIED

- A. Quantities indicated in the Contract Documents are for bidding and contract purposes only. Quantities and measurements supplied or placed in the Work and verified by the Architect determine payment.
- B. If the actual Work requires more or fewer quantities than those quantities indicated, provide the required quantities at the unit sum/prices contracted.

# 1.4 MEASUREMENT OF QUANTITIES

- A. Measurement Devices:
  - 1. Weigh Scales: Inspected, tested and certified by the applicable State Weights and Measures Department within the past year.
  - 2. Platform Scales: Of sufficient size and capacity to accommodate the conveying vehicle.
  - 3. Metering Devices: Inspected, tested and certified by the applicable State department within the past year.
- B. Measurement by Weight: Concrete reinforcing steel, rolled or formed steel or other metal shapes will be measured by handbook weights. Welded assemblies will be measured by handbook or scale weight.
- C. Measurement by Volume: Measured by cubic dimension using mean length, width and height or thickness.
- D. Measurement by Area: Measured by square dimension using mean length and width or radius.
- E. Linear Measurement: Measured by linear dimension, at the item centerline or mean chord.
- F. Stipulated Sum/Price Measurement: Items measured by weight, volume, area, or linear means or combination, as appropriate, as a completed item or unit of the Work.

# 1.5 PAYMENT

- A. Payment Includes: Full compensation for all required labor, Products, tools, equipment, plant, transportation, services and incidentals; erection, application or installation of an item of the Work; overhead and profit.
- B. Final payment for Work governed by unit prices will be made on the basis of the actual measurements and quantities accepted by the Architect multiplied by the unit/sum price for Work which is incorporated in or made necessary by the Work.

# 1.6 DEFECT ASSESSMENT

- A. Replace the Work, or portions of the Work, not conforming to specified requirements.
- B. If, in the opinion of the Architect, it is not practical to remove and replace the Work, the Architect will direct one (1) of the following remedies:
  - 1. The defective Work may remain, but the unit sum/price will be adjusted to a new sum/price or reduced 50 percent at the discretion of the Architect.
  - 2. The defective Work will be partially repaired to the instructions of the Architect, and the unit sum/price will be adjusted to a new sum/price or reduced 50 percent at the discretion of the Architect.
- C. The individual specification sections may modify these options or may identify a specific formula or percentage sum/price reduction.
- D. The authority of the Architect to assess the defect and identify payment adjustment is final.

# 1.7 NON-PAYMENT FOR REJECTED PRODUCTS

- A. Payment will not be made for any of:
  - 1. Products wasted or disposed of in a manner that is not acceptable.
  - 2. Products determined as unacceptable before or after placement.
  - 3. Products not completely unloaded from the transporting vehicle.
  - 4. Products placed beyond the lines and levels of the required Work.
  - 5. Products remaining on hand after completion of the Work.
  - 6. Loading, hauling and disposing of rejected Products.

# PART 2 - PRODUCTS

Not Used

# PART 3 - EXECUTION

# 3.1 SCHEDULE OF UNIT PRICES

- A. Unit Price No. 1: Removal and replacement of treated wood: (Include 400 linear feet of each dimension in Base Proposal)
  - 1. Unit Price shall be for entire unit cost including overhead and profit to provide removal and replacement of treated wood nailers which are not shown on the drawings, but which are found to be required.
    - 2X4 2X6
    - 2X8
    - 2X12

- 2. The units to be used shall be dollars per linear foot.
- 3. Refer to Specification Sections for materials.
- 4. Enter unit price on Proposal Form, include 400 LF of each dimension in base bid as unit price contingency.

# END OF SECTION 01 22 00

# SECTION 01 23 00 - ALTERNATES

### PART 1 - GENERAL

# 1.1 ALTERNATE PRICES

A. State, in the spaces provided in the proposal form, Alternate Prices for the work described below. The responsibility of determining quantity of Alternates rests with the Contractor. Base Proposal and Alternates shall include cost of all supporting elements required, so that no matter what combination of Base Proposal and Alternates are accepted, that portion shall be a complete entity in itself. Work for all Alternates shall be in strict accordance with the specification sections noted and applicable to the specific work.

### **PART 2 - PRODUCTS**

Not Used

# PART 3 - EXECUTION

# 3.1 ALTERNATES

- A. Alternate Proposal No.1: POWER WASH, PREP, PRIME, AND PAINT BRICK AT GISD ANNEX.
  - 1. This Alternate shall provide pricing to add the scope of work outlined in the drawings.
- B. Alternate Proposal No.2: CLEAR ELASTOMERIC SEALANT TO BE APPLIED TO ENTIRE CAMPUS AT LA MORGAN ELEMENTARY.
  - 1. This Alternate shall provide pricing to add the scope of work outlined in the drawings.
- C. Alternate Proposal No.3: : CLEAR ELASTOMERIC SEALANT TO BE APPLIED TO ENTIRE CAMPUS AT PARKER ELEMENTARY.
  - 1. This Alternate shall provide pricing to add the scope of work outlined in the drawings.

# END OF SECTION 01 23 00

# SECTION 01 25 13 - PRODUCT SUBSTITUTION PROCEDURES

# PART 1 - GENERAL

#### 1.1 SECTION INCLUDES

- A. Specified product compliance, and product quality assurance
- B. Specific administrative and procedural requirements for handling requests for substitutions made prior to award of Contract.
- C. Requirements for product delivery, storage and handling.

# 1.2 RELATED REQUIREMENTS

A. Instructions to Offerors: Product options and procedures for submittal of requests for substitutions during the Proposal period.

# 1.3 DESCRIPTION OF REQUIREMENTS

- A. Definitions: Definitions used in this Section are not intended to negate the meaning of other terms used in the Contract Documents, including such terms as "specialties", "systems", "structure", "finishes", "accessories", "furnishings", "special construction", and similar terms. Such terms are self-explanatory and have recognized meanings in the construction industry.
  - 1. Products: Shall mean items purchased for incorporation in the Work, regardless of whether they were specifically purchased for the project or taken from the Contractor's previously purchased stock. The term "product" as used herein includes the terms "material", "equipment", "system", and other terms of similar intent.
    - a. Named Products: Are those identified by the use of the manufacturer's name for a product, including such items as a make or model designation, as recorded in published product literature, of the latest issue as of the date of the Contract Documents.
    - b. Specified Products: same as Named Products.
  - 2. Materials: Shall mean products that must be substantially cut, shaped, worked, mixed, finished, refined, or otherwise fabricated, processed, or installed to form units of work.
  - 3. Equipment: Is defined as a product with operational parts, regardless of whether motorized or manually operated, and in particular, a product that requires service connections such as wiring or piping.

# 1.4 **PRODUCT QUALITY ASSURANCE**

- A. Source Limitations: To the fullest extent possible, provide products of the same generic kind, from a single source, for each unit of work.
  - 1. When it is discovered that specific products are available only from sources that do not or cannot produce an adequate quantity to complete project requirements in a timely manner, consult with the Architect/Engineer for a determination of what product quantities are most important before proceeding. The Architect/Engineer will designate those qualities, such as visual, structural, durability, or compatibility, that are most important. When the Architect/Engineer's determination has been made, select products from those sources that produce products that possess the most important qualities, to the fullest extent possible.

- B. Compatibility of Options: Compatibility of products is a basic requirement of product selection. When the Contractor is given the option of selecting between two (2) or more products for use on the project, the product selected must be compatible with other products previously selected, even if the products previously selected were also Contractor options. The complete compatibility between the various choices available to the Contractor is not assured by the various requirements of the Contract Documents, but must be provided by the Contractor.
- C. Or Equal:
  - 1. Where the phrase "or equal", "or equivalent", "or Architects approved equal", or similar phrasing, occurs in the Proposal Documents, do not assume that materials, equipment, or methods of construction will be approved by the Architect unless the item has been specifically approved for this Work by the Architect.
  - 2. The decision of the Architect shall be final.
- D. Where a proposed substitution involves the work of more than one (1) contractor, each contractor involved shall cooperate and coordinate the work with each other contractor involved, so as to provide uniformity and consistency and to assure the compatibility of products.
- E. Foreign Product Limitations: "Foreign products" as distinguished from "domestic products" are defined as products that are either manufactured substantially (50 percent or more of value) outside of the United States and its possessions, or produced or supplied by entities known to be substantially owned (more than 50 percent) by persons who are not citizens of nor living within the United States and its possessions.
  - 1. Except under one (1) of the following conditions, select and provide domestic, not foreign, products for inclusion in the Work.
    - a. There is no domestic product available that complies with the requirements of the Contract Documents.
    - b. Available domestic products that comply with the requirements of the Contract Documents are available only at prices or other procurement terms that are substantially higher (25 percent or more) than for available foreign products that comply with the requirements of the Contract Documents.
    - c. At the discretion of the Architect or Owner.
  - 2. Final determination and acceptance will be the responsibility of the Architect.
- F. Standards: Refer to Section 01 41 00, Regulatory Requirements for the applicability of industry standards to the products specified for the Project, and for the acronyms used in the text of the Specification Sections.

# 1.5 SUBSTITUTIONS OF PRODUCTS

- A. The products described in the Proposal Documents establish a standard of required function, dimension, appearance and quality to be met by any proposed substitution. The materials and equipment named in, and the procedures covered by these specifications have been selected as a standard because of quality, particular suitability or record of satisfactory performance. It is not intended to preclude the use of equal or better materials or equipment provided that same meets the requirements of the particular project and is approved in an Addendum as a substitution prior to the submission of proposals.
- B. No substitution will be considered prior to receipt of proposals unless written request for approval has been received by the Architect at least seven (7) days prior to the date for receipt of proposals. Each such request shall include the name of the material or equipment for which it is to be substituted and a complete description of the proposed

substitute including drawings, cuts, performance and test data and any other information necessary for an evaluation. The Architect's decision of approval or disapproval of a proposed substitution shall be final.

- C. If the Architect approves any proposed substitution prior to receipt of proposals, such approval will be set forth in an Addendum. Offerors shall not rely upon approvals made in any other manner.
- D. The Architect and Owner reserve the right to disapprove the use of any manufacturer who in their judgment is unsuitable for use on the Project and that decision will be final
- E. The following are not considered as substitutions:
  - 1. Revisions to the Contract Documents, when requested by the Owner, Architect, or any of their consultants are considered as "changes" not substitutions.
  - 2. Specified Contractor options on products and construction methods included in Contract Documents are choices made available to the Contractor and are not subject to the requirements specified in this Section for substitutions.
  - 3. Except as otherwise provided in the Contract Documents, the Contractor's determination of and compliance with governing authorities do not constitute "substitutions" and do not constitute a basis for change orders.
- F. The following may be considered as a reason for a request for substitution:
  - 1. The request is directly related to an "or approved equal" clause or similar language in the Contract Documents.
  - 2. The specified product or method of construction cannot be provided within the Contract Time in accordance with paragraph below concerning availability of specified items.
  - 3. The specified product or method of construction cannot receive necessary approval by a governing authority, and the requested substitution can be approved.
  - 4. A substantial advantage is offered the Owner, in terms of cost, time, energy conservation or other consideration of merit, after deducting offsetting responsibilities the Owner may be required to bear. These additional responsibilities may include such considerations as additional compensation to the Architect/Engineer for redesign and evaluation services, the increased cost of other work by the Owner or separate contractors, and similar considerations.
  - 5. The specified product or method of construction cannot be provided in a manner that is compatible with other materials, and where the Contractor certifies that the substitution will overcome the incompatibility.
  - 6. The specified product or method of construction cannot be coordinated with other materials, and where the Contractor certifies that the proposed substitution can be coordinated.
  - 7. The specified product or method of construction cannot provide a warranty required by the Contract Documents and where the Contractor certifies that the proposed substitution provides the required warranty.
- G. Availability of specified items:
  - 1. Verify prior to submittal of Proposal that all specified items will be available in time for installation during orderly and timely progress of the work.
  - 2. In the event specified items will not be so available, notify the Architect prior to receipt of Proposals. Submit Request for Substitutions in accordance with this section.
  - 3. The request will not be considered if the product or method cannot be provided as a result of the Contractor's failure to pursue the work promptly or coordinate activities properly.

- 4. Costs of delays because of non-availability of specified items, when such delays could have been avoided by the Contractor, will be back-charged as necessary and shall not be borne by the Owner.
- H. A request constitutes a representation that Offeror:
  - 1. Has investigated proposed product and determined that it meets or exceeds quality level of specified product.
  - 2. Will provide same warranty for Substitution as for specified product, except when inability to provide specified Warranty is reason for request for substitution as described above.
  - 3. Will coordinate installation and make changes to other Work which may be required for the Work to be complete with no additional cost to Owner.
  - 4. Waives claims for additional costs or time extension which may subsequently become apparent.
  - 5. Will reimburse the Owner and pay for all costs, including Architect/Engineer's redesign and evaluation costs resulting from the use of the proposed substitution, or for review or redesign services associated with re-approval by authorities having jurisdiction.

# I. No substitutions will be considered after the Award of Contract.

# 1.6 SUBSTITUTION REQUEST SUBMITTAL

- A. Requests for Substitutions: Submit three (3) copies of each request for substitution. In each request identify the product or fabrication or installation method to be replaced by the substitution; include related Specifications Section and Drawing numbers, and complete documentation showing compliance with the requirements for substitutions. Include, as appropriate, with each request, the following information:
  - 1. Product data, drawings and descriptions of products, fabrication and installation procedures.
  - 2. Samples, where applicable or requested.
  - 3. A detailed comparison of the significant qualities of the proposed substitution with those of the work originally specified. Significant qualities may include elements such as size, weight, durability, performance and visual effect, where applicable.
  - 4. Coordination information, including a list of changes or modifications needed by other parts of the work and to construction performed by the Owner and separate Contractors that will become necessary to accommodate the proposed substitution.
  - 5. A statement indicating the effect the substitution will have on the Contractor's Construction Schedule compared to the schedule without approval of the substitution. Indicate the effect of the proposed substitution on overall Contract Time.
  - 6. Cost information, including a proposal of the net change, if any in the Contract Sum.
  - 7. Certification by the Contractor to the effect that, in the Contractor's opinion, after thorough evaluation, the proposed substitution will result in work that in every significant respect is equal-to or better than the work required by the Contract Documents, and that it will perform adequately in the application indicated. Include the Contractor's waiver of rights to additional payment or time that may subsequently become necessary because of the failure of the substitution to perform adequately.
  - 8. A statement indicating the Contractor will reimburse the Owner and pay for all costs, including Architect/Engineer's re-design and evaluation costs resulting from the use of the proposed substitution.

B. Work-Related Submittals: The Contractor's submittal of, and the Architect/Engineer's acceptance of, Shop Drawings, Product Data, or Samples which are related to work not complying with the Contract Documents, does not constitute an acceptance or valid request for a substitution, nor approval thereof.

# 1.7 PRODUCT DELIVERY, STORAGE, AND HANDLING

- A. General: Deliver, store, and handle products in accordance with manufacturer's recommendations, using means and methods that will prevent damage, deterioration and loss, including theft. Control to prevent overcrowding of construction spaces or overloading of structure. In particular, coordinate delivery and installation to ensure minimum holding or storage times for items known or recognized to be flammable, hazardous, easily damaged, or sensitive to deterioration, theft and other sources of loss.
  - 1. Deliver products to the site in the manufacturer's sealed containers or other packaging system, complete with labels intact, and instructions for handling, storage, unpacking, installing, cleaning and protecting.
  - 2. Cover products subject to deterioration with impervious sheet covering. Provide ventilation to avoid condensation or potential degradation of product.
  - 3. Store loose granular materials on solid flat surfaces in a well-drained area. Prevent mixing with foreign matter.
  - 4. Store products at the site or in a bonded and insured off-site storage facility or warehouse in a manner that will facilitate inspection and measurement of quantity or counting of units. Periodically inspect to verify products are undamaged and are maintained in acceptable condition.
  - 5. Store heavy materials away from the project structure or in a manner that will not endanger the supporting construction.

# PART 2 - PRODUCTS

# 2.1 GENERAL PRODUCT COMPLIANCE

- A. General: Requirements for individual products are indicated in the Contract Documents; compliance with these requirements is in itself a contract requirement. These requirements may be specified in any one (1) of several different specifying methods, or in any combination of these methods. These methods include the following:
  - 1. Proprietary
  - 2. Descriptive
  - 3. Performance
  - 4. Compliance with Reference Standards

Compliance with codes, compliance with graphic details, allowances, and similar provisions of the Contract Documents also have a bearing on the selection process.

- B. Procedures for Selecting Products: The Contractor's options in selecting products are limited by requirements of the Contract Documents and governing regulations. They are not controlled by industry traditions or procedures experienced by the Contractor on previous construction projects. Required procedures include, but are not limited to the following for the various indicated methods of specifying:
  - 1. Proprietary and Semi-Proprietary Specification Requirements:
    - a. Single Product Name: Where only a single product or manufacturer is named, provide the product indicated, unless the specification indicates the possible consideration of other products. Advise the Architect/Engineer before proceeding, when it is discovered that the named product is not a reasonable or feasible solution.
    - b. Two (2) or More Product Names: Where two (2) or more products or manufacturers are named, provide one (1) of the products named, at the

Contractor's option. Exclude products that do not comply with specification requirements. Do not provide or offer to provide an unnamed product, unless the specification indicates the possible consideration of other products. Advise the Architect/Engineer before proceeding where none of the named products comply with specification requirements, or are not feasible for use. Where products or manufacturers are specified by name, accompanied by the term "or approved equal" or similar language, comply with this Section regarding "substitutions" to obtain approval from the Architect/Engineer for the use of an unnamed product.

- 2. Non-Proprietary Specification Requirements: Where the specifications name products or manufacturers that are available and may be incorporated in the Work, but do not restrict the Contractor to the use of these products only, the Contractor may, at his option, use any available product that complies with the Contract requirements.
- 3. Descriptive Specification Requirements: Where the specifications describe a product or assembly generically, in detail, listing the exact characteristics required, but without use of a brand name, provide products or assemblies that provide the characteristics indicated and otherwise comply with Contract requirements.
- 4. Performance Specification Requirements: Where the specifications require compliance with indicated performance requirements, provide products that comply with the specific performance requirements indicated, and that are recommended by the manufacturer for the application indicated. The manufacturer's recommendations may be contained in published product literature, or by the manufacturer's individual certification of performance. General overall performance of a product is implied where the product is specified for specific performances.
- 5. Compliance with Standards, Codes, and Regulations: Where the specifications require only compliance with an imposed standard, code or regulation, the Contractor has the option of selecting a product that complies with specification requirements, including standards, codes, and regulations.
- 6. Visual Matching: Where matching an established sample is required, the final judgment of whether a product proposed by the Contractor matches the sample satisfactorily will be determined by the Architect. Where there is no product available within the specified product category that matches the sample satisfactorily and also complies with other specified requirements, comply with the provisions of this Section regarding "substitutions" and other Contract Documents for "change orders" for the selection of a matching product in another product category, or for non-compliance with specified requirements.
- 7. Visual Selection: Except as otherwise indicated, where specified product requirements include the phrase "...as selected from the manufacturer's standard colors, patterns, textures..." or similar phrases, the Contractor has the option of selecting the product and manufacturer, provided the selection complies with other specified requirements. The Architect is subsequently responsible for selecting the color, pattern and texture from the product line selected by the Contractor.
- 8. Allowances: Refer to individual sections of the specifications and Section 01 21 00, Allowances for an indication of product selections that are controlled by established allowances, and for the procedures required for processing such selections.
- C. Producer's Statement of Applicability: Where individual specification sections indicate products that require a "Statement of Applicability" from the manufacturer or other producer, submit a written-certified statement from the producer stating that the producer has reviewed the proposed application of the product on the project. This statement shall

state that the producer agrees with or does not object to the Architect/Engineer's specification and the Contractor's selection of the product on the project is suitable and proper.

# 2.2 SUBSTITUTIONS

A. Condition: The Contractor's request for substitution will be received and considered when extensive revisions to Contract Documents are not required, when the proposed changes are in keeping with the general intent of the Contract Documents, when the request is timely, fully documented and properly submitted, and when one (1) or more of the above conditions are satisfied, all as judged and determined by the Architect/Engineer; otherwise the requests will be returned without action except to record non-compliance with these requirements.

# PART 3 - EXECUTION

# 3.1 INSTALLATION OF PRODUCTS

- A. General: Except as otherwise indicated in individual sections of these specifications, comply with the manufacturer's instructions and recommendations for installation of the products in the applications indicated.
- B. Anchor each product securely in place, accurately located and aligned with other work.
- C. Clean exposed surfaces and protect surfaces as necessary to ensure freedom from damage and deterioration at time of acceptance.
- D. Products and assemblies shall be installed complete, in-place, watertight and structurally sound.

# 3.2 INSTALLATION OF APPROVED SUBSTITUTIONS

- A. Coordinate all approved substitutions with adjacent work.
- B. Comply with the manufacturer's and/or supplier's instructions and recommendations for installation of the products in the applications indicated.
- C. Provide all items required by manufacturer and/or supplier regarding installation, i.e. supplemental supports, anchors, fasteners, painting, etc. whether or not indicated or specified.

# END OF SECTION 01 25 13

# SECTION 01 26 00 - CONTRACT MODIFICATION PROCEDURES

### PART 1 - GENERAL

#### 1.1 SECTION INCLUDES

- A. Change procedures.
- B. Defect assessment.

#### 1.2 GENERAL

A. Coordinate requirements of this Section with the requirements of the General and Supplementary Conditions of the Contract concerning change procedures.

### 1.3 CHANGE PROCEDURES

- A. Submittals: Submit name of individual authorized to receive change documents, and be responsible for informing others in Contractor's employ or Subcontractors of changes to the Work.
- B. Minor Changes: The Architect/Engineer may advise of minor changes in the Work not involving adjustment to Contract Sum/Price or Contract Time by issuing supplemental instructions on Minor Change form or by other similar documents in the form issued by the Architect.
- C. Change Proposal Request: The Architect may issue a Change Proposal Request (CPR) or other similar request for proposal in the form issued by the Architect, including a detailed description of proposed change with supplementary or revised Drawings and Specifications, a change in Contract Time for executing the change and the period of time during which the requested price will be considered valid. Contractor will prepare and submit estimate in the form of a Change Proposal so as to not cause delays in the Project.
- D. Use of allowances must be approved by issuance of Allowance Expenditure Authorization (AEA) by Architect prior to modification of the schedule of values. The AEA may be comprised of a single executed Change Proposal, an accumulation of executed Change Proposals, or other similar documentation in the form allowed by the Architect in accordance with Document CB, Supplementary Conditions of the Contract.
- E. Contractor may propose changes which, in his opinion, will provide value to the Owner, by submitting a request for change to Architect, describing proposed change and its full effect on the Work. Include a statement describing reason for the change, and effect on Contract Sum/Price and Contract Time with full documentation and a statement describing effect on Work by separate or other Contractors. If accepted by Architect and approved by Owner, submit a Change Order in accordance with the requirements of this Section. This request will not be considered a substitution except as defined by Section 01 25 13, Product Substitution Procedures. Owner is not obligated to accept this request.
- F. Construction Change Directive: Architect/Engineer may issue directive, on AIA Form G713 Construction Change Directive or other similar document in the form issued by the Architect, and signed by Owner, instructing Contractor to proceed with change in the Work, for subsequent inclusion in a Change Order. Document will describe changes in

the Work, and designate method of determining any change in Contract Sum/Price or Contract Time. Promptly execute change.

- G. Document each quotation for change in cost or time with sufficient data to allow evaluation of quotation.
- H. Change Order Forms: AIA G701 Change Order.
- I. Execution of Change Orders: The Architect will prepare and sign the Change Order, the contractor shall sign the Change Order indicating acceptance of the change, and then the Owner will execute the Change Order.
- J. Correlation Of Contractor Submittals:
  - 1. Promptly revise Schedule of Values and Application for Payment forms to record each authorized Change Order as separate line item and adjust Contract Sum/Price.
  - 2. Promptly revise progress schedules to reflect change in Contract Time, revise sub-schedules to adjust times for other items of work affected by the change, and resubmit.
  - 3. Promptly enter changes in Project Record Documents.

# 1.4 DEFECT ASSESSMENT

- A. Replace the Work, or portions of the Work, not conforming to specified requirements at no additional cost to the Owner.
- B. If, in the opinion of the Architect/Engineer or Owner, it is not practical to remove and replace the Work, the Architect will direct appropriate remedy or adjust payment.
- C. The defective Work may remain, but sum/price will be adjusted to new sum/price at the discretion of Architect or Owner.
- D. Individual specification sections may modify these options or may identify specific formula or percentage sum/price reduction.
- E. Authority of Architect/Engineer, or other appropriate agent identified to perform assessment by the Architect/Engineer or Owner, to assess defects and identify payment adjustments, is final.
- F. Non-Payment For Rejected Products: In addition to replacement of rejected Work, payment will not be made for rejected products for any of the following:
  - 1. Products wasted or disposed of in a manner that is not acceptable.
  - 2. Products determined as unacceptable before or after placement.
  - 3. Products not completely unloaded from transporting vehicle.
  - 4. Products placed beyond lines and levels of required Work.
  - 5. Products remaining on hand after completion of the Work.
  - 6. Loading, hauling, and disposing of rejected products.

PART 2 - PRODUCTS Not Used. PART 3 - EXECUTION Not Used.

# END OF SECTION 01 26 00

CONTRACT MODIFICATION PROCEDURES 01 26 00 - 2

# **SECTION 01 29 00 - PAYMENT PROCEDURES**

# PART 1 - GENERAL

# 1.1 SECTION INCLUDES

A. Procedures for submitting Applications for Payment.

# 1.2 GENERAL

A. Coordinate requirements of this Section with the requirements of the General and Supplementary Conditions of the Contract concerning payment procedures.

# 1.3 SCHEDULE OF VALUES

A. Submit printed schedule on AIA Form G703 - Continuation Sheet for G702 in accordance with Section 01 29 73, Schedule of Values. Contractor's standard form or electronic media printout will be considered but must be approved by the Owner.

# 1.4 APPLICATIONS FOR PAYMENT

- A. Submit four (4) notarized originals of each application on AIA Form G702 Application and Certificate for Payment and AIA G703 Continuation Sheet for G702 or other similar form approved by the Owner.
- B. Content and Format: Utilize Schedule of Values for listing items in Application for Payment.
- C. Submit updated construction schedule with each Application for Payment.
- D. Payment Period: Submit at intervals stipulated in the Agreement in accordance with Document CB, Supplementary Conditions of the Contract.
- E. Only materials stored on the project site shall be paid for unless the materials are stored in a bonded warehouse.
- F. Substantiating Data: When Architect/Engineer requires substantiating information, submit data justifying dollar amounts in question. Items which may be requested by the Architect or Owner to substantiate costs include, but are not limited to the following:
  - 1. Current Record Documents as specified in Section 01 77 00, Closeout Procedures, for review by Owner which will be returned to Contractor.
  - 2. Labor time sheets, purchase orders, or similar documentation.
  - 3. Affidavits attesting to off-site stored products.

# PART 2 - PRODUCTS

Not Used.

# PART 3 - EXECUTION

Not Used.

# END OF SECTION 01 29 00

# SECTION 01 29 73 - SCHEDULE OF VALUES

# PART 1 - GENERAL

#### 1.1 DESCRIPTION

- A. Work Included: Provide a detailed breakdown of the agreed Contract Sum showing values allocated to each of the various parts of the work, as specified herein and in other provisions of the Contract Documents.
- B. Coordinate requirements of this Section with the requirements of the General and Supplementary Conditions of the Contract concerning Schedule of Values.

# 1.2 QUALITY ASSURANCE

- A. Use required means to assure arithmetical accuracy of the sums described.
- B. When so required by the Owner, provide copies of the subcontracts or other data acceptable to the Owner, substantiating the sums described.

# 1.3 SUBMITTALS

- A. Prior to the first Application for Payment, submit a proposed schedule of values to the Owner, as outlined below:
  - 1. Meet with the Owner and determine additional data, if any, required to be submitted.
  - 2. Secure the Owner's approval of the schedule of values prior to submitting first Application for Payment.

# 1.4 SCHEDULE OF VALUES

- A. <u>The Schedule of Values shall be broken down into item costs for each specification section</u> <u>as a minimum</u>. After review by the Owner, the Schedule of Values shall be broken down into further items as required. (See following list and refer to the enclosed sample.). <u>In</u> <u>addition, total each Specification Division separately</u>.
- B. Schedule of Values Items in addition to Specification Sections.
  - 1. Mobilization
  - 2. Clean Up
  - 3. Building Permit
  - 4. Bonds, Insurance
  - 5. Misc. Mechanical Accessories
  - 6. Demolition
  - 7. Rough-In Labor (Roofing Tear off)
  - 8. Rough-In Material (Roofing Tear off)
  - 9. Finish Labor (Roofing cap ply)
  - 10. Finish Material (Roofing cap ply)
  - 11. Allowances (listed separately)
  - 12. Sealants
  - 13. Coatings
  - 14. Record drawings and close-out documents
  - 15. Submittals listed separately per mechanical, electrical and plumbing
  - 16. Roof warranty as a line item
  - 17. Donated items individually itemized at \$0.00 (zero dollars).

# PART 2 - PRODUCTS

Not Used

# PART 3 - EXECUTION

# 3.1 SCHEDULE OF VALUES

A. Refer to following sample.

# END OF SECTION 01 29 73

### SECTION 01 29 73

# SCHEDULE OF VALUES - SAMPLE

ltem No.	Description of Work	Scheduled	Work Completed		Stored	Total	%	Balance	Retainage
		Value	Previous	This	Materials	Completed		To Finish	
			App.	App.					
	Div. 1 - General Reqs.								
	Suoervision								
	Mobilization								
	Bonds & Insurance								
	Permits								
	Contractor's Fee								
	Close-Out Documents								
	Div. 1 - Total								
	Div. 6 - Wood & Plastics								
	Rough Carpentry - Labor								
	Rough Carpentry - Materials								
	Div. 6 - Total								
	Div. 7 - Thermal and								
	Moisture Protection								
	Waterpfng / Dampprfng-Matls								
	Waterpfng / Dampprfng-Labor								
	Mod Bit Roofing-Labor								
	Mod Bit Roofing-Materials								
	Mod Bit Roofing Guarantee								
	Roof Accessories								
	Building Sheet Metal - Labor								
	Building Sheet Metal - Matls								
	Bldg. Sheet Metal Guarantee								
	Roof Curbs								
	Roof Hatches								
	Sealants								
	Close-Out Documents								

## SECTION 01 31 13 - PROJECT COORDINATION

# PART 1 - GENERAL

#### 1.1 **REQUIREMENTS**

- A. General: notify the Architect whenever there is need of clarification or interpretation of the Contract Documents prior to commencement of work.
- B. Commencement of work without Architect's prior notification means Contractor's acceptance of responsibility.
- C. Commencement of work without Architect's prior notification implies Contractor's understanding of conditions, assemblies, methods, or procedures.
- D. The project superintendent shall notify the Owner on an ongoing basis of ongoing work.

#### 1.2 PRE-INSTALLATION CONFERENCE

- A. General: Notify the Architect 48 hours in advance of certain stages of construction, and, as required by the Architect, organize a pre-installation meeting with each trade individually prior to commencement of their portion of the Work. At a minimum, representatives of the Architect, the General Contractor's project superintendent, and the Sub-contractor's Foreman and Project Manager shall be present at each meeting. The Engineer shall be notified as applicable.
- B. As indicated in each specific section of this Project Manual, or as required by the Architect, these stages generally include, but are not necessarily limited to the following:
  - 1. Division 6 Rough Carpentry
  - 2. Division 7 installation of waterproofing, vapor barriers, flashing and sheet metal.
  - 3. Division 7 Installation of roofing system(s) and associated work.
- C. In addition to notifying the Architect, notify the Structural Engineer (48 hours) prior to the following stages:
  - 1. Drilling, reinforcing, and placing of first piers and footings.
  - 2. Placing first reinforcing and grade beams.
  - 3. Erecting structural steel elements.

# PART 2 - PRODUCTS

Not Used

#### PART 3 - EXECUTION

# 3.1 PRE-CONSTRUCTION CONFERENCE

A. The Contractor shall contact Architect at least ten (10) days prior to commencing construction in order for Architect to schedule a pre-construction meeting with Contractor, Architect, and Owner. This meeting must occur prior to commencement of any construction.

# 3.2 CONFERENCES AND MEETINGS

A. Refer to Section 01 31 19, Project Meetings for requirements pertaining to Preconstruction Conference, Progress Meetings, and Pre-installation Conferences.

# END OF SECTION 01 31 13

# SECTION 01 31 19 - PROJECT MEETINGS

# PART 1 - GENERAL

# 1.1 REQUIREMENTS INCLUDE

- A. The Architect's:
  - 1. Scheduling of each meeting (pre-construction meeting, periodic project meetings, and specialty called meetings throughout the progress of the Work).
  - 2. Preparation of agenda for meetings.
  - 3. Presiding at minutes, including all significant proceedings and decisions.
  - 4. Recording, reproducing, and distributing copies of meeting minutes within two (2) working days, excluding weekends and holidays, after each meeting to:
    - a. All participants in the meeting.
    - b. All parties affected by decisions made at the meeting.
  - 5. Providing status report of allowance funds.
- B. The Contractor's:
  - 1. Making physical arrangement for meetings.
  - 2. Participation in all meetings and conferences.
  - 3. Scheduling attendance of Job Superintendent, Project Coordinator, and other parties affecting or affected by decisions made at meetings and conferences as their interests require.
  - 4. Scheduling Pre-installation conferences.
  - 5. Scheduling Pre-Closeout Meeting
  - 6. Providing updated schedules.
  - 7. Providing status reports/logs of CPRs, MCs, and shop drawings/submittals.

# PART 2 - PRODUCTS

Not Used

# **PART 3 - EXECUTION**

# 3.1 PRE-CONSTRUCTION CONFERENCE

A. Contractor shall contact Architect at least ten (10) days prior to commencing construction in order for Architect to schedule a pre-construction meeting with Contractor, Architect, and Owner. This meeting must occur prior to commencement of any construction.

# B. Architect will:

- 1. Administer pre-construction conference for the establishment of communication methods, procedures and Owner requirements.
- 2. Administer site mobilization conference for clarification of Owner and Contractor.
- C. Location: At Project site as designated by the Architect.
- D. Attendance:
  - 1. Contractor or Contractor's Representative
  - 2. Job Superintendent
  - 3. Project Coordinator (Manager)
  - 4. Owner or Owner's Representative
  - 5. Major subcontractors
  - 6. Major suppliers

- 7. Architect's Representative
- 8. Architect's Field Representative
- 9. Consultants as needed
- 10. Others as appropriate
- E. Meeting Agenda, may include, but is not limited to:
  - 1. Discussion on major subcontracts and suppliers and projected construction schedules.
  - 2. Critical work sequencing.
  - 3. Major equipment deliveries and priorities. Discussion of long lead time items.
  - 4. Project coordination and designation of responsible personnel.
  - 5. Procedures and processing of field decisions, proposal requests, submittals, minor changes, change orders and applications for payment.
  - 6. Method of distribution of Contract Documents.
  - 7. Procedures for maintaining Record Documents.
  - 8. Use of premises, office work and storage areas, on-site parking, and Owner's requirements.
  - 9. Construction facilities and temporary utilities.
  - 10. Housekeeping procedures.

# 3.2 PROGRESS MEETINGS

- A. Architect will:
  - 1. Schedule project meetings throughout progress of the work at weekly intervals, and specially called meetings.
  - 2. Set agenda and administer said meetings.
  - 3. Preside at meetings.
  - 4. Record meeting minutes, including all significant proceedings and decisions.
  - 5. Reproduce and distribute copies of meeting minutes within two (2) working days, excluding weekends and holidays, after each meeting to:
    - a. All participants in the meeting.
    - b. All parties affected by decisions made at the meeting.
- B. Contractor shall:
  - 1. Make physical arrangements for meetings.
- C. Attendance:
  - 1. Contractor or Contractor's Representative
  - 2. Job Superintendent
  - 3. Project Coordinator (Manager)
  - 4. Owner or Owner's Representative
  - 5. Major subcontractors
  - 6. Major suppliers
  - 7. Architect's Field Representative
  - 8. Consultants as needed
  - 9. Others as appropriate
- D. Meeting Agenda, may include, but is not limited to:
  - 1. Review and approval of minutes of previous meeting.
  - 2. Review of Work progress since previous meeting.
  - 3. Field observations, problems, and conflicts.
  - 4. Review of off-site fabrication and delivery schedules.
  - 5. Corrective measures and procedures to regain projected schedule.
  - 6. Revisions to Construction Schedule.
  - 7. Plan progress and schedule during succeeding work period.
  - 8. Coordination of schedules.

- 9. Review submittal schedules and expedite as required.
- 10. Maintenance of quality standards.
- 11. Allowance balances.
- 12. Review of proposed changes and substitutions for:
  - a. Effect on Construction Schedule and on completion date.
    - b. Effect on other contracts of the Project.
- 13. Status of Allowance Expenditure Authorizations (AEAs).
- 14. Status of Change Proposal Requests (CPRs).
- 15. Status of Minor Changes (MCs).
- 16. Status of submittals, review of submittal log.
- 17. Other items and critical issues affecting Work.

# 3.3 PRE-INSTALLATION CONFERENCES

- A. In accordance with the requirements of Section 01 11 00, Notification of Architect Requirements, the Contractor will convene pre-installation conferences when required by individual specification Sections or as required by the Architect, prior to the Contractor commencing Work of the Section.
- B. Attendance, optional:
  - 1. General Contractor or Contractor's Representative
  - 2. Project Coordinator (Manager)
  - 3. Owner or Owner's Representative
  - 4. Architect's Project Manager (Project Executive)
- C. Attendance, required:
  - 1. Project Superintendent
  - 2. Architect's Field Representative
  - 3. Sub-contractor's Project Manager
  - 4. Sub-contractor's Foreman
  - 5. Engineer's Representative, as needed.
  - 6. Manufacturer's Representative, as needed.
  - 7. Governing Agency Official, as required
  - 8. Inspection Agency Representative, as required.
  - 9. Others affecting or affected by Work.
- D. Meeting Agenda, may include, but is not limited to:
  - 1. Review of conditions of installation.
  - 2. Preparation and installation procedures.
  - 3. Coordination with related work
  - 4. Review of the contract document requirements.
  - 5. Review of code enforcement or testing requirements.
  - 6. Questions related to work required.

# 3.4 PRE-CLOSEOUT MEETING

- A. In accordance with the requirements of Section 01 77 00, Closeout Procedures, the Contractor will convene a pre-closeout meeting when he considers the Work or designated portion of the Work is sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the work for its intended use.
- B. Attendance, required:
  - 1. Owner or Owner's Representative
  - 2. Project Coordinator (Manager)
  - 3. General Contractor or Contractor's Representative
  - 4. Project Superintendent

- 5. Architect's Project Manager (Project Executive)
- 6. Architect's Field Representative
- 7. Engineer's Representative, as needed.
- C. Meeting Agenda, may include, but is not limited to:
  - 1. Review of the contract document requirements for Substantial Completion and Project Closeout
  - 2. Review of Work which remains to be completed or corrected.
  - 3. Closeout Document review schedule and log
  - 4. Review of closeout procedures including, but not limited to Record Drawings, Warrantees, Operation and Maintenance Manuals, and Owner Demonstrations and Start-up.
  - 5. Review of code enforcement or testing requirements.
  - 6. Questions related to work required.

### END OF SECTION 01 31 19

# SECTION 01 32 16 - CONSTRUCTION PROGRESS SCHEDULE

### PART 1 GENERAL

### 1.1 SUBMITTALS

- A. Schedules:
  - 1. Preliminary Analysis: Within 10 (ten) days after receipt of Award of Contract, submit a preliminary construction schedule for review by Owner and Architect.
  - 2. Construction Schedule: Within 14 (fourteen) days after receipt of Notice to Proceed, submit one (1) reproducible and four (4) prints of the approved construction schedule.

### 1.2 RELIANCE UPON SCHEDULE

A. The construction schedule as approved by the Architect will be an integral part of the contract and will establish conditions for various activities and phases of constructions.

# 1.3 CONSTRUCTION SCHEDULE

- A. Diagram: Graphically show the order of all activities necessary to complete the work and the sequence in which each activity is to be accomplished.
- B. Activities shown on the diagram shall include but not necessarily be limited to:
  - 1. Project mobilization
  - 2. Submittals and approvals of shop drawings and samples
  - 3. Phasing of construction
  - 4. Procurement of equipment and critical materials
  - 5. Fabrication and installation of special material and equipment
  - 6. Final clean-up
  - 7. Final inspection and testing
- C. The construction schedule shall be updated and submitted with each Application for Payment.

### 1.4 CONSTRUCTION SCHEDULE LIMITATIONS

- A. Work performed under this Contract shall be done in accordance with the following paragraphs:
  - 1. All work may proceed immediately upon Notice to Proceed and continue uninterrupted.
  - 2. The Owner has a critical need for the work to begin upon Notice to Proceed and shall be Substantially Complete by January 11, 2019.
  - 3. Under the Base Proposal only, the successful Offeror will be 1) entitled to certain extensions of time and 2) subject to liquidated damages for work not completed beyond the agreed date which the Contractor shall require for Substantial Completion of the work included in this contract. Refer to Supplementary Conditions for additional requirements and liquidated damages.
  - 4. Failure to complete and close-out project after substantial completion may result in liquidated damages. Refer to Supplementary Conditions for additional requirements and liquidated damages.
  - 5. The Owner may at his discretion approve changes recommended by the successful Offeror to the above-mentioned schedule provided that the Owner's use of newly completed areas are not disrupted.

CONSTRUCTION PROGRESS SCHEDULE 01 32 16 - 1

# PART 2 - PRODUCTS

Not Used

# PART 3 - EXECUTION

Not Used

# END OF SECTION 01 32 16

# SECTION 01 33 00 - SUBMITTAL PROCEDURES

### PART 1 - GENERAL

# 1.1 SUBMITTAL PROCEDURES

- A. Transmit to the Architect/Engineer each item indicated in individual specification sections with approved form identifying:
  - 1. Date of submission and dates of any previous submissions.
  - 2. Project title and number
  - 3. Contract identification
  - 4. Names of Contractor, Supplier, Manufacturer
  - 5. Pertinent drawing sheet and detail number, and specification section number, as appropriate
  - 6. Deviations from Contract Documents.
- B. Contractor shall be responsible for initial review prior to submittal to Architect/Engineer to verify adequacy and conformance to contract requirements. Lack of review by Contractor may be grounds for rejection.
- C. Apply Contractor's stamp, signed, to each item submitted, certifying that review and verification of products, field dimensions, adjacent construction work and coordination of information is in accordance with the requirements of the work and contract documents.
- D. Transmit each item in accordance with approved schedule, and in such sequence as to cause no delay in the work or in the work of any other Contractor. Allow minimum of ten (10) days for adequate Architect/Engineer review of each submittal. Time may vary according to scope and complexity of item under review. Allow adequate time in schedule for revisions and resubmittal as deemed necessary.
- E. Submit one (1) opaque print or copy of the submittal to the Architect plus one (1) electronic original. Transmit the printed copy of consultant and engineering submittals directly to respective consultants with a transmittal and the electronic original to the Architect. The Architect and Consultant will make up the printed copy and return to the Contractor upon completion of review. It will be the Contractors responsibility to scan and distribute the necessary quantity of copies of the reviewed submittal to all concerned parties.
- F. Submit each item according to individual specification sections and identified by Division, Section, and individual submittal number. Maintain log according to each Division.
- G. Revise and resubmit submittal as required; identify all changes made since previous submittal.
  - 1. Make any corrections or changes in the submittals required by the Architect/Engineer and resubmit until approved.
  - 2. Submit new submittal as required for initial submittal.

# 1.2 PROPOSED PRODUCTS LIST

- A. Within 30 days after date of Notice to Proceed, submit list of major products proposed for use, with name of manufacturer, trade name, and model number of each product.
- B. For products specified only by reference standards, give manufacturer, trade name, model or catalog designation, and reference standards.

### 1.3 **PRODUCT DATA**

- A. Submit to Architect for review for limited purpose of checking for conformance with information given and design concept expressed in Contract Documents.
- B. Submit the number of copies of product data and samples which the Contractor and his subcontractors need for their use PLUS two (2) additional sets for the Architect, one (1) additional set for the Owner and one (1) additional set for each of the Architect's consultants involved with the particular Section of Work.
- C. Mark each copy to identify applicable products, models, options, and other data. Supplement manufacturers' standard data to provide information specific to this Project

### 1.4 MSDS SHEETS

- A. The Texas Asbestos Health Protection Rules (Title 25. Health Services, Part I. Texas Department of Health Chapter 295 - Occupational Health, Subchapter C - Texas Asbestos Health Protection) were approved and became effective on October 20, 1992, and amended March 27, 2003. The Rules established the procedures and means to implement the provisions of Senate Bill 1341 and House Bill 79.
- B. Pursuant to the above referenced Rules, submit MSDS Sheets showing that materials used in the Project, contain 1.0 percent or less asbestos. This requirement pertains to every material in every Section of the Specifications, as applicable to the Project, whether written therein, or not. Submit MSDS Sheets for materials, as applicable to the Project.

# 1.5 SHOP DRAWINGS

- A. Submit to Architect/Engineer for review for limited purpose of checking for conformance with information given and design concept expressed in Contract Documents.
- B. Indicate special utility and electrical characteristics, utility connection requirements, and location of utility outlets for service for functional equipment and appliances.
- C. All dimensions indicated on the drawings are based on the specific models and manufacturers of products, equipment, fixtures and miscellaneous items specified. If the Contractor uses an approved product by another listed manufacturer which is different than the specific model and manufacturer listed in these specifications, then the Contractor shall be solely responsible for the coordination of any dimensional changes required, including structural, relocation of walls, equipment, fixtures, ceilings and miscellaneous items. When dimensional changes are required in these situations, the Contractor shall submit a proposed modification drawing to the Architect for approval prior to proceeding with the work. All causes and effects of the dimensional change shall be indicated on the Contractor's drawing submittal.

### 1.6 SAMPLES

- A. Submit for review for limited purpose of checking for conformance with information given and design concept expressed in Contract Documents.
- B. Submit for aesthetic, color, or finish selection. Submit full range of manufacture's standard colors, textures, and patterns for Architect's selection.
- C. Submit samples to illustrate functional characteristics of the Product, with integral parts and attachment devices. Coordinate submittal of different categories for interfacing work.

- D. Submit the number specified in respective Specification Section; minimum of two (2), of which one (1) will be retained by Architect.
- E. Reviewed samples which may be used in the Work are indicated in individual specification sections.
- F. Samples will not be used for testing purposes unless specifically stated in specification section.

# 1.7 DESIGN DATA

- A. When required, submit for Architect/Engineer's knowledge as contract administrator or for Owner.
- B. Submit design data for information for limited purpose of assessing conformance with information given and design concept expressed in Contract Documents.

# 1.8 TEST REPORTS

- A. In accordance with Section 01 45 23, Inspection and Testing Laboratory Services, submit test reports for Architect/Engineer's knowledge as contract administrator or for Owner. Architect will determine whether corrective action is required.
- B. Submit test reports for information for limited purpose of assessing conformance with information given and design concept expressed in Contract Documents.

# 1.9 CERTIFICATES

- A. When specified in individual specification sections, submit certification by manufacturer, installation/application subcontractor, or Contractor to Architect, in quantities specified.
- B. Indicate material or product conforms to or exceeds specified requirements. Submit supporting reference data, affidavits, and certifications as appropriate.
- C. Certificates may be recent or previous test results on material or product, but must be acceptable to Architect and Owner.
- D. Submit required certificates in duplicate.

### 1.10 GUARANTEES

- A. When specified in individual specification sections, submit warranties by manufacturer, installation/application subcontractor, fabricator, or Contractor to Architect, in quantities specified.
- B. Submit warranties in accordance with Section 01 77 00, Closeout Procedures.

### 1.11 MANUFACTURER'S INSTRUCTIONS

- A. When specified in individual specification sections, submit printed instructions for delivery, storage, assembly, installation, start-up, adjusting, and finishing, to Architect for delivery to Owner in quantities specified.
- B. Indicate special procedures, perimeter conditions requiring special attention, and special environmental criteria required for application or installation.
- C. Submit required instructions in duplicate.

### 1.12 MANUFACTURER'S FIELD REPORTS

- A. Submit reports for Architect/Engineer's benefit as contract administrator or for Owner.
- B. Submit report in quantity specified or required within ten (10) days of observation to Architect for information. Architect will determine whether corrective action is required.
- C. Submit for information for limited purpose of assessing conformance with information given and design concept expressed in Contract Documents.

### 1.13 ERECTION DRAWINGS

- A. When required, submit drawings for Architect/Engineer's benefit or for Owner.
- B. Submit for information for limited purpose of assessing conformance with information given and design concept expressed in Contract Documents.
- C. Data indicating inappropriate or unacceptable Work may be subject to action by Architect/Engineer or Owner. Architect will determine whether corrective action is required.

### 1.14 CONSTRUCTION PHOTOGRAPHS

- A. Provide photographs monthly of site and construction throughout progress of Work produced by an experienced photographer, acceptable to Architect/Engineer.
- B. Photographs: digital; sent to Architect via email, or provide on non-rewritable compact disk. Along with Application for Payment, include one (1) reproducible copy of contact sheet of all photographs taken during that period indicating Work completed and identified as stated below.
- C. Photograph project conditions five (5) days maximum prior to submitting indicating relative progress of the Work. Do not photograph conditions previously photographed if no work has proceeded. As able, take photographs from same position indicating same view in successive installments.
- D. Take photographs as evidence of existing project conditions as follows:
  - 1. Site: Take four (4) site photographs at project corners
  - 2. Interior views: Take two (2) minimum interior photographs of each space under construction from differing directions or as required.
  - 3. Exterior views: Take two (2) photographs of each elevation.
  - 4. Details: Take as required to document concealed conditions, including, but not limited to, underground construction, utility penetrations and installation, steel erection, concrete and masonry reinforcing, waterproofing and flashing, and roofing installation.
  - 5. Cavity wall: Provide photographic evidence that cavity wall was maintained clean and free of debris and excess mortar.
- E. Identify each photograph with name of Project, room or view, and date.

# PART 2 - PRODUCTS

Not Used

PART 3 - EXECUTION Not Used

### END OF SECTION 01 33 00

SUBMITTAL PROCEDURES 01 33 00 - 4

# SECTION 01 35 16 - ALTERATION PROJECT PROCEDURES

#### PART 1 - GENERAL

#### 1.1 REQUIREMENTS INCLUDED

- A. This Section contains general provisions and requirements pertaining to all remodeling, removal and relocation of Work in the existing building and becomes a part of each Section and Division performing remodeling, removal and relocation Work for this Project with the same force and effect as if written in full therein.
- B. Take all necessary precautions to keep trespassers out of the Work areas. Secure Work areas from entry when Work is not in progress.
- C. Perform all alterations, remodeling, demolition, removal and relocation of Work in strict accordance with Owner's instructions and applicable Federal, State and local health and safety standards, codes and ordinances. Where conflicts occur, the more restrictive requirement shall govern.

### 1.2 RELATED WORK

A. Section 02 41 16 - Selective Demolition

#### 1.3 EXISTING CONDITIONS

- A. Obvious existing conditions, installations and obstructions affecting the Work shall be taken into consideration as necessary Work to be done, the same as though they were completely shown or described.
- B. Items of existing construction indicated to remain upon completion of the Contract, but which require removal to complete the Work, shall be carefully removed and replaced as required. The replaced Work shall match its condition at the start of the Work unless otherwise required.
- C. Visit the site to determine by inspection all existing conditions, including access to the site, the nature of structures, objects and materials to be encountered, and all other facts concerning or affecting the Work. Information on the Drawings showing existing conditions does not constitute a guarantee that other items may not be found or encountered.
- D. Utilities: Do not interrupt existing utilities serving occupied or used facilities, except when authorized by the Architect in writing two (2) weeks in advance. Provide temporary services during interruptions to existing utilities.

### PART 2 - PRODUCTS

### 2.1 SALVAGED MATERIALS

- A. The Owner reserves the right of first refusal on all salvage items. Remove remaining items from the site as Work progresses. Storage or sale of items on site is not permitted. Burning or burying of removed materials on site is not permitted.
- B. Store salvaged items in a dry, secure place on site.
- C. Salvaged items not required for use in repair of existing Work shall remain the property of the Owner.

D. Do not incorporate salvaged or used material in new construction except with permission of the Architect.

# 2.2 PRODUCTS FOR PATCHING, EXTENDING AND MATCHING

- A. Contract Documents do not define products or standards of workmanship present in existing construction. Determine products by inspection and by use of the existing. Provide same or similar quality products or types of construction as that in existing structure when needed to patch or extend existing Work.
- B. If reasonably matching products are not obtainable, improve appearance by minor relocating of some existing products and grouping new ones in some pattern arranged by the Architect. Do not replace products scheduled for retaining because matching ones are not obtainable, except as directed by Change Order.

### PART 3 - EXECUTION

### 3.1 PROTECTION OF WORK TO REMAIN

- A. Protect existing Work from damage. Use barricades, tarpaulins, temporary walls, plywood, planking, masking, or other suitable means and methods as approved by the Architect.
- B. If Work to remain in place is damaged, restore to original condition at no additional cost to the Owner.
- C. Concealed Conditions: If conditions cause changes in the Work from requirements of the Contract Documents, the Contract Sum will be adjusted in accordance with the General Conditions.

# 3.2 EXAMINATION

Β.

- A. Verify that areas are ready for alteration and remodeling.
  - Discrepancies: Verify dimensions and elevations indicated in layout of existing work.
    - 1. Prior to commencing work, carefully compare and check Contract Documents for discrepancies in locations or elevations of work to be executed.
    - 2. Refer discrepancies among Drawings and existing conditions to Architect for adjustment before work affected is performed.

# 3.3 PREPARATION

- Construct temporary fire-rated partitions to separate existing occupied areas from construction and alteration areas. Comply with provisions of Division 01 Section "Temporary Facilities and Controls."
- B. Cut, move, or remove items as necessary for access to alteration and renovation Work.
  - 1. Remove unsuitable material not marked for salvage, such as rotted wood, corroded metals, deteriorated masonry and concrete, and other deteriorated materials. Replace materials as specified for finished Work.
  - 2. Remove debris and abandoned items from area and from concealed spaces.
- C. Cutting and Removal: Perform cutting and removal work to remove minimum necessary, and in manner to avoid damage to adjacent work. Cut finish surfaces such as masonry, tile, plaster, or metals by methods to terminate surfaces in straight line at natural point of division.

- D. Prepare surface and remove surface finishes as necessary to provide for proper installation of new materials and finishes.
- E. Close openings in exterior surfaces to protect existing Work from weather and extremes of temperature and humidity. Insulate ductwork and piping to prevent condensation in exposed areas.
- F. Provide temporary barriers and closures to control operations to prevent spread of dust to occupied portions of building.

# 3.4 SECURITY AND PROTECTION FACILITIES INSTALLATION

- A. Temporary Egress: Maintain temporary egress from existing occupied facilities as indicated and as required by authorities having jurisdiction.
- B. Temporary Enclosures: Provide temporary enclosures for protection of construction, in progress and completed, from exposure, foul weather, other construction operations, and similar activities. Provide temporary weather tight enclosure for building exterior.
  - 1. Where heating or cooling is needed and permanent enclosure is not complete, provide insulated temporary enclosures. Coordinate enclosure with ventilating and material drying or curing requirements to avoid dangerous conditions and effects.
  - 2. Vertical Openings: Close openings of 25 sq. ft. or less with plywood or similar materials.
  - 3. Horizontal Openings: Close openings in floor or roof decks and horizontal surfaces with load-bearing, wood-framed construction.
  - 4. Install tarpaulins securely using fire-retardant-treated wood framing and other materials.
  - 5. Where temporary wood or plywood enclosure exceeds 100 sq. ft. in area, use fire-retardant-treated material for framing and main sheathing.
- C. Temporary Partitions: Erect and maintain dustproof partitions and temporary enclosures to limit dust and dirt migration and to separate areas from fumes and noise.
  - 1. Construct non-fire-rated dustproof partitions of not less than nominal 4-inch studs, 1/2-inch gypsum wallboard with joints taped on occupied side, and 1/2-inch fire-retardant plywood on construction side.
  - 2. Insulate partitions to provide noise protection to occupied areas.
  - 3. Seal joints and perimeter.
  - 4. Equip partitions with dustproof doors and security locks.
  - 5. Protect air-handling equipment.
  - 6. Weather-strip openings.
- D. Temporary Fire-Rated Partitions: Erect and maintain dustproof fire-rated partitions and temporary enclosures to limit dust and dirt migration and to separate occupied areas from construction, fumes, and noise. Fire-rated partitions shall be provided to separate existing occupied areas from construction areas in accordance with NFPA 241.
  - 1. Construct fire-rated dustproof partitions of not less than nominal 4-inch studs, 1/2-inch or 5/8-inch Type X gypsum wallboard on both sides, with joints taped.
  - 2. Extend partitions up to underside of existing structure to the greatest extent possible.
  - 3. Insulate partitions to provide noise protection to occupied areas.
  - 4. Seal joints and perimeter with fire-resistant joint sealant.
  - 5. Equip partitions with dustproof doors and security locks.
    - a. Protect openings in 1-hour fire-rated partitions with 45-minute hollow metal or solid core wood doors.
  - 6. Protect air-handling equipment.

- 7. Weather-strip openings.
- E. Temporary Fire Protection: Until fire-protection needs are supplied by permanent facilities, install and maintain temporary fire-protection facilities of types needed to protect against reasonably predictable and controllable fire losses. Comply with NFPA 241.
  - 1. Provide fire extinguishers, installed on walls on mounting brackets, visible and accessible from space being served, with sign mounted above.
    - a. Field Offices: Class A stored-pressure water-type extinguishers.
    - b. Other Locations: Class ABC dry-chemical extinguishers or a combination of extinguishers of NFPA-recommended classes for exposures.
    - c. Locate fire extinguishers where convenient and effective for their intended purpose; provide not less than one extinguisher on each floor at or near each usable stairwell.
  - 2. Store combustible materials in containers in fire-safe locations.
  - 3. Maintain unobstructed access to fire extinguishers, fire hydrants, temporary fireprotection facilities, stairways, and other access routes for firefighting. Prohibit smoking in hazardous fire-exposure areas.
  - 4. Supervise welding operations, combustion-type temporary heating units, and similar sources of fire ignition.
  - 5. Permanent Fire Protection: At earliest feasible date in each area of Project, complete installation of permanent fire-protection facility, including connected services, and place into operation and use. Instruct key personnel on use of facilities.
  - 6. Develop and supervise an overall fire-prevention and first-aid fire-protection program for personnel at Project site. Review needs with local fire department and establish procedures to be followed. Instruct personnel in methods and procedures. Post warnings and information.

# 3.5 MOISTURE AND MOLD CONTROL

- A. Contractor's Moisture-Protection Plan: Avoid trapping water in finished work. Document visible signs of mold that may appear during construction. Remove and replace materials with mold.
- B. Exposed Construction Phase: Before installation of weather barriers, when materials are subject to wetting and exposure and to airborne mold spores, protect as follows:
  - 1. Protect porous materials from water damage.
  - 2. Protect stored and installed material from flowing or standing water.
  - 3. Keep porous and organic materials from coming into prolonged contact with concrete.
  - 4. Remove standing water from decks.
  - 5. Keep deck openings covered or dammed.
- C. Partially Enclosed Construction Phase: After installation of weather barriers but before full enclosure and conditioning of building, when installed materials are still subject to infiltration of moisture and ambient mold spores, protect as follows:
  - 1. Do not load or install drywall or other porous materials or components, or items with high organic content, into partially enclosed building.
  - 2. Keep interior spaces reasonably clean and protected from water damage.
  - 3. Periodically collect and remove waste containing cellulose or other organic matter.
  - 4. Discard or replace water-damaged material.
  - 5. Do not install material that is wet.
  - 6. Discard, replace or clean stored or installed material that begins to grow mold.
  - 7. Perform work in a sequence that allows any wet materials adequate time to dry before enclosing the material in drywall or other interior finishes.

- D. Controlled Construction Phase of Construction: After completing and sealing of the building enclosure but prior to the full operation of permanent HVAC systems, maintain as follows:
  - 1. Control moisture and humidity inside building by maintaining effective dry-in conditions.
  - 2. Use permanent HVAC system to control humidity.
  - 3. Comply with manufacturer's written instructions for temperature, relative humidity, and exposure to water limits.
- E. Wet and Water-Damaged Materials:
  - 1. Hygroscopic materials that may support mold growth, including wood and gypsum-based products, that become wet during the course of construction and remain wet for 24 hours are considered defective.
  - 2. Measure moisture content of materials that have been exposed to moisture during construction operations or after installation. Record daily readings over a forty-eight hour period. Identify materials containing moisture levels higher than allowed. Report findings in writing to Architect.
  - 3. Remove materials that can not be completely restored to their manufactured moisture level within 48 hours.

# 3.6 **PROCEDURES**

- A. Refinishing At Removed Work: Cut below surface of substrate materials and patch over area of removal with finish materials so removal is not apparent.
- B. Remove and replace existing ceilings, and cut, patch, or replace existing walls, partitions and floors as may be necessary for access to valves, piping, conduit and tubing by mechanical and electrical trades as directed and approved by the Architect, and performed by the appropriate subcontractor for the Work involved, or by other properly qualified subcontractors.
- C. Patch and extend existing Work using skilled mechanics who are capable of matching existing quality and workmanship. Quality of patched or extended Work shall be not less than that specified for new Work.
- D. Cutting:
  - 1. Concrete and Masonry: Saw cut where feasible.
  - 2. Plaster: Cut back to sound plaster on straight lines, and back-bevel edges of remaining plaster. Trim and prepare existing lath for tying of new lath.
  - 3. Woodwork: Cut back to a joint or panel line. Undamaged removed materials may be reused.
  - 4. Resilient Tiles: Remove in whole units to natural breaking points or straight joint lines with no damaged or defective existing tiles remaining where joining new construction.
  - 5. Salvaged Materials: Carefully remove to avoid damage, thoroughly clean and reinstall as indicated, or as directed.
  - 6. Doors: Remove in such a manner as to facilitate filling in of openings or installation of new Work, as required by Drawings.
  - 7. Structural Elements: Remove only as shown on the Structural Drawings. If not specifically shown, but removal is required, perform such removal or alteration only upon written approval of the Architect. Do not damage or alter any structural element of the existing building.
- E. Patching:

- 1. Match existing Work where possible; if unavailable, use salvage material for patching and provide totally new material in areas where salvage has been removed; consult with the Architect concerning locations for salvaging materials.
- 2. Repairs or continuations of existing Work shall be relatively imperceptible in the finished Work when viewed under finished lighting conditions from a distance of six (6) feet.
- 3. Patching, Repairing and Finishing of Existing Work: Perform in compliance with the applicable requirements of the Specification Section covering the Work to be performed and the requirement of this Section.
- F. Erect scaffolding as necessary to gain access to the various parts of the Work. Provide structurally sound, rigidly braced and properly constructed scaffolding, shoring and bracing as necessary to positively protect the affected elements and building, and to support the activities or workmen and loads. Design and construction of scaffolds and supports shall be in accordance with applicable safety regulations. Material used shall be adequate to support anticipated loads with a properly calculated margin of safety.
- G. Noise Producing Equipment: Minimize use of noise producing equipment. Limit excessive noise to periods of vacancy or provide sound control. Arrange schedules in advance with the Architect.

# 3.7 EXISTING FURNITURE AND EQUIPMENT

- A. Owner Salvaged Items: Personal items in areas subject to remodeling will be removed before construction in those areas commences.
- B. Furniture Items: Before remodeling commences, Owner will remove all furniture and equipment from each space, store items as necessary. Owner will replace these items to the same locations after each remodeling phase is complete. Contractor to coordinate activities with Owner.

### 3.8 PAINTING

- A. Preparation: Prepare patched areas as required for new Work. Wash existing painted surfaces with neutral soap or detergent, thoroughly rinse, and sand when dry.
- B. Painting and Finishing: Conform to the applicable provisions of the Painting Section. Prepare bare areas and patches in existing painted surfaces with specified primer and intermediate coats, sanded smooth and flush with adjoining surfaces.

# 3.9 DISPOSAL OF DEBRIS

- A. Remove material, debris and rubbish resulting from Work of this Section from the building and site as it accumulates. Keep all areas of Work in "broom clean" condition as the Work progresses.
- B. At completion of renovation and remodeling Work in each area, provide final cleaning and return space to a condition suitable for use by the Owner.

# END OF SECTION 01 35 16

# SECTION 01 41 00 - REGULATORY REQUIREMENTS

# PART 1 - GENERAL

### 1.1 SECTION INCLUDES

- A. Quality Assurance.
- B. References Standards.
- C. Definitions.
- D. Abbreviations.
- E. Format and Specification Context Explanations.
- F. Drawing Symbols.
- G. General Requirements.

# 1.2 QUALITY ASSURANCE

- A. General:
  - 1. For products or workmanship specified by a standard of an association, trade, or Federal standards, comply with requirements of the standard, except when more rigid requirements are specified or are required by applicable code authorities having jurisdiction.
  - 2. The contractual relationship of the parties to the Contract should not be altered from the Contract Documents by mention or inference otherwise in any reference standard.
  - 3. Obtain copies of standards when required by Contract Documents.
  - 4. Maintain copy of standards at jobsite during submittals, planning, and progress of the specific work for which the standards pertain, until the date of Substantial Completion.
  - 5. In the absence of specific instructions in the specifications, materials, products, equipment and their installation shall conform to the applicable codes, regulations and standards specified therein. When a conflict exists between the applicable code, regulation and standard and that specified, the more stringent code regulation or standard shall prevail, except as authorized by applicable authorities having jurisdiction.
- B. Specifications and Drawings: The Drawings and Specifications are correlative and have equal authority and priority. Base disagreements in themselves or in each other on the most expensive combination of quantity and quality of work indicated. In the event of such disagreement bring it to the attention of the Architect, who will determine the appropriate method to perform the work.
- C. Industry Standards: Where compliance with two (2) or more industry standards or sets of requirements are specified, and overlapping of those different standards or requirements establishes different or conflicting minimums or levels of quality, the most stringent requirement is intended and will be enforced, unless specifically detailed language written into Contract Documents clearly indicates that a less stringent requirement is to be fulfilled. Refer apparently equal-but-different requirements, and uncertainties as to which level of quality is more stringent, to the Architect for a decision before proceeding.

- D. Contractor's Option: Except for overlapping or conflicting requirements, where more than one (1) set of requirements are specified for a particular unit of work, option is intended to be Contractor's regardless of whether or not it is specifically indicated as such.
- E. Minimum Quality/Quantity: In every instance, the quality level or quantity shown or specified is intended to be the minimum for the work to be performed or provided. Except as otherwise specifically indicated, the actual work may either comply exactly with the minimum (within specified tolerances), or may exceed that minimum within reasonable limits. In complying with requirements, indicated numeric values are either minimums or maximums as noted or as appropriate for context of requirements. Refer instances of uncertainty to Architect for decision before proceeding.
- F. Specialists' Assignments: In certain instances, specification text requires (or implies) that specific work is to be assigned to specialists, who are engaged for performance of work. Such assignments shall be recognized as special requirements over which the Contractor has no choice or option. These requirements should not be interpreted so as to conflict with applicable regulations, union jurisdiction settlements and similar conventions. Such assignments are intended to establish which party or entity involved in a specific unit of work is recognized as "expert" for the indicated construction processes or operations. Nevertheless, the final responsibility for fulfillment of the entire set of requirements remains with the Contractor.

# 1.3 **REFERENCE STANDARDS**

- A. Dates of codes, regulations and standards specified shall be the latest date of issue of that code, regulation or standard prior to the date of issue of this Project Manual or Document, except as modified or otherwise directed by the applicable codes and their supplements and amendments adopted by the code authorities having jurisdiction.
  - 1. Date of Issue The "date of issue" as it appears in the statement above, means the date which appears on the cover of the Project Manual or Document corresponding to the date of issue of the Contract Documents.
  - 2. Code Authorities: The "code authorities" as it appears in the statement above, means the authorities responsible for code enforcement.

### 1.4 **DEFINITIONS**

- General Explanation: A substantial amount of specification language consists of Α. definitions for terms found in other Contract Documents, including those in the AIA A201 General Conditions of the Contract for Construction, Supplementary Conditions, the Drawings, and the Specifications. Drawings must be recognized as being diagrammatic in nature and not completely descriptive of requirements indicated thereon. Certain terms used in the Contract Documents are defined in the General Conditions, Supplementary Conditions, and in this Section. Definitions and explanations contained in this Section are not necessarily either complete or exclusive, but are general for this Work to the extent that they are not stated more explicitly in another element of the Contract Documents. In the event of a conflict in definitions or explanations within the Contract Documents or whenever there is need of clarification or interpretation of definitions within or between the Contract Documents, notify the Architect immediately and proceed as directed. Except in cases where definitions are determined by code authorities having jurisdiction, the Architect's interpretation of all definitions will take precedence.
- B. General Requirements: The provisions or requirements of Division 1 Sections apply to entire Work of Contract and, where indicated, to other elements which are included in the Project.

- C. Special Conditions: Wherever the term "Special Conditions", appears in the Contract Documents, it refers collectively to all requirements of the Owner in addition to the sections in Division 1, General Requirements, and to Articles contained in the General Conditions and Supplementary Conditions.
- D. Architect: Wherever the term "Architect" or any derivative thereof appears in the Contract Documents, it means PBK, 11 Greenway Plaza, 22<sup>nd</sup> Floor, Houston, Texas 77046, (713) 965-0608, or their authorized representative(s).
- E. Bid, Competitive Sealed Proposal (CSP), Response, Offer, etc.: Wherever the term "Bid", "Competitive Sealed Proposal (CSP)", "Response", "Offer", "Proposal", or any derivative thereof, or similar term appears in the Contract Documents, they mean one and the same, and shall mean Competitive Sealed Proposal, which by definition allows the Owner to accept the "best value" for the school district based on factors other than cost in selecting the Contractor.
- F. Contractor, General Contractor, Construction Manager, etc: Wherever the term "Contractor", "General Contractor", "Construction Manager" or any derivative thereof, or similar term appear in the Contract Documents, they mean one and the same.
- G. Subcontractor, Sub-subcontractor, Bidder, etc.: Wherever the term "Subcontractor", Sub-subcontractor", "Bidder", "Bidder/Vendor", "Vendor", "Installer", "Integrator", "Respondent", "Offeror", or any derivative thereof, or similar term appears in the Contract Documents, they mean one and the same, and shall refer to the entity (person or firm) licensed and meeting all applicable regulations of the State of Texas and Department of Labor to perform the Work, or their authorized representative(s).
  - 1. Responsibilities: To avoid any misunderstanding or lack of interpretation, the responsibility for performing the Work is totally that of the entity defined above, and the resolutions proposed in his shop drawings and related documentation shall be demonstrated throughout the Work and specified warranty period.
  - 2. In the event of a controversy involving the Contract Documents or interpretation of Project requirements, the decision of the Architect will take precedence.
- H. District, School District, Owner, etc.: Wherever the term "District", "School District", "Owner", "Galveston ISD", GISD, or similar such term appears in the Contract Documents, it means the Galveston Independent School District, 3904 Avenue T Galveston, Texas 77550, or its authorized representative(s).
- I. Consultants: Wherever the term "Consultant", or any derivative thereof appears in the Contract Documents, it means the following to whom that portion of the work applies.
  - 1. Architect's Consultants:
    - a. Facilities Consultant: PBK, *Facilities Group,* 11 Greenway Plaza, 22<sup>nd</sup> Floor, Houston, Texas 77046 (713) 965-0608, or their authorized representative(s).
- J. Indicated: Wherever the term "indicated", or any derivative thereof appears in the Contract Documents, it means a cross-reference to graphic representations, notes, or schedules on Drawings, to other paragraphs or schedules in the Specifications, and to similar means of recording requirements in the Contract Documents. Where terms such as "shown", "noted", "scheduled", and "specified" are used in lieu of "indicated", it is for the purpose of helping reader locate cross-reference, and no limitation of location is intended except as specifically noted.
- K. Directed, Requested, Etc: Where not otherwise explained, terms such as "directed", "requested", "authorized", "selected", "approved", "required", "accepted", and "permitted" or any derivative thereof appears in the Contract Documents, it means as "directed by the

Architect", "requested by the Architect", and similar phrases with actions taken by the Architect. However, no meaning or otherwise shall be interpreted to extend the Architect's responsibility into Contractor's area of construction supervision.

- L. Approve: Wherever the term "Approve", or any derivative thereof appears in the Contract Documents, it means only the Architect, or an individual designated by him as his representative, can approve or disapprove contract actions. Even if the specifications indicate that an individual other than the Architect, such as the "Engineer" or "Consultant" will approve or disapprove an action, it is understood that only the Architect has this authority unless the individual is so designated by him in writing. Even when an individual is so designated, the Contractor may appeal the action to the Architect and the Architect's decision will be final. In no case will "approval" by the Architect be interpreted as a release of the Contractor from responsibility to fulfill requirements of the Contract Documents.
- M. Furnish: Wherever the term "Furnish", or any derivative thereof appears in the Contract Documents, it means supply or deliver to Project site, ready for unloading, unpacking, assembly, erection, placing, installing, anchoring, applying, curing, finishing, protecting, cleaning and similar operations, as applicable in each instance.
- N. Install: Wherever the term "Install", or any derivative thereof appears in the Contract Documents, it means performing the operations at the Project site, of unloading, unpacking, assembly, erection, placing, installing, anchoring, applying, curing, finishing, protecting, cleaning and similar operations, as applicable in each instance.
- O. Provide: Wherever the term "Provide", or any derivative thereof appears in the Contract Documents, it means furnish and install at the Project site, complete and ready for intended use, as applicable in each instance.
- P. Project, Site: Wherever the term "Project", "Site", or similar such term appears in the Contract Documents, it means the space available to the Contractor for performance of the Work, either exclusively or in conjunction with others performing work as part of the Project. The extent of project or site is shown on the Drawings, and may or may not be identical with description of land upon which Project is to be built.
- Q. Installer: Wherever the term "Installer", or any derivative thereof appears in the Contract Documents, it means the entity (person or firm) engaged by the Contractor or its subcontractor or sub-subcontractor for performance of a particular unit of work at the Project, including installation, erection, application and similar required operations. It is a general requirement that such entities (Installers) be expert in operations they are engaged to perform.
- R. Specialist: Wherever the term "Specialist", or any derivative thereof appears in the Contract Documents, it means an individual or firm of established reputation (or if newly organized, whose personnel have previously established a reputation in the same field), which is regularly engaged in, and which maintains a regular force of workmen skilled in either (as applicable) manufacturing or fabricating items required by the Contract, installing items required by the Contract, or otherwise performing work required by the Contract. Where the Contract Specification requires installation by a specialist, that term shall also be deemed to mean either the manufacturer of the item or firm who will perform the work under the manufacturer's direct supervision.
- S. Testing Laboratory: Wherever the term "Testing Laboratory", or any derivative thereof appears in the Contract Documents, it means an independent entity engaged to perform specific inspections or tests of the work, either at the Project site or elsewhere; and to report and (if required) interpret results of those inspections or tests.

### 1.5 FORMAT AND SPECIFICATION CONTEXT EXPLANATIONS

- A. Underscoring: Is used strictly to assist reader of specification text in scanning text for key words (for quick recall). No emphasis on or relative importance is intended where underscoring is used.
- B. Capitalization: Except for manufacturer, product, or trademark names, capitalization is used strictly to assist reader of specification text in scanning text for key words (for quick recall). No emphasis on or relative importance is intended where capitalization is used.
- C. Imperative language: Is used generally in specifications. Except as otherwise indicated, requirements expressed imperatively are to be performed by Contractor. For clarity of reading at certain locations, contrasting subjective language is used to describe responsibilities which must be fulfilled indirectly by the Contractor, or when so noted, by others.
- D. Section Numbering: Is used to facilitate cross-reference in Contract Documents. Sections are placed in Project Manual in numeric sequence; however, numbering sequence is not complete, and listing of sections at beginning of Project Manual must be consulted to determine numbers and names of specification sections in Contract Documents.
- E. Page Numbering: Pages are numbered independently for each section. The section number is shown preceded by the project number and followed by the page number at the bottom of each page, to facilitate the location of text. The project number is given to identify the project, for which specification was written, should the section become separated from the Project Manual.
- F. Specifying Methods: The techniques or methods of specifying to record requirements varies throughout text, and may include "prescriptive, "open-generic descriptive", "compliance with standards", "performance", or a combination of these. The method used for specifying one unit of work has no bearing on requirements for another unit of work.
- G. Abbreviations: The language of Specifications and other Contract Documents is of the abbreviated type in certain instances, and implies words and meanings which will be appropriately interpreted. Actual work abbreviations of a self-explanatory nature have been included in texts. Specific abbreviations have been established, principally for lengthy technical terminology and primarily in conjunction with coordination of specification requirements with notations on drawings and in schedules. These are frequently defined in section at first instance of use. Trade association names and titles of general standards are frequently abbreviated. Singular words will be interpreted as plural and plural words will be interpreted as singular where applicable and where full context of the Contract Documents so indicates. A list of typical abbreviations, includes, but is not limited to the following trade associations and organizations. Refer to Drawings and other Contract Documents for other abbreviations.

AA	Aluminum Association
AAMA	Architectural Aluminum Manufacturer's Assn.
AASHTO	American Association of State Highway and Transportation Officials
ACI	American Concrete Institute
ACIL	American Council of Independent Laboratories
AGA	American Gas Association
AGC	Associated General Contractors of America
AHA	American Hardboard Association
AHGA	American Hotdip Galvanizers Association
Al	Asphalt Institute
AIA	American Institute of Architects

AISC	American Institute of Steel Construction
AISI	American Iron & Steel Institute
AITC	American Institute of Timber Construction
ANSI	American National Standards Institute
APA	American Plywood Association
ARI	Air Conditioning & Refrigeration Institute
ASA	Acoustical Society of America
ASA	American Subcontractors Association
ASAHC	American Society of Architectural Hardware Consultants
ASC	•
	Adhesive & Sealant Council, Inc.
ASCE	American Society of Civil Engineers
ASHRAE	American Society of Heating, Refrigeration, and Air Conditioning Engineers
ASME	American Society of Mechanical Engineers
ASPE	American Society of Professional Engineers
ASPI	American Wood Preserver's Institute
ASTM	ASTM International
AWI	Architectural Woodwork Institute
AWS	American Welding Society
BIA	Brick Institute of America
BRI	Building Research Institute
CRA	California Redwood Association
CLFMI	Chain Link Fence Manufacturers Institute
CRSI	Concrete Reinforcing Steel Institute
CSI	Construction Specifications Institute
DHI	Door and Hardware Institute
EPA	Environmental Protection Agency
FTI	Facing Tile Institute
FGMA	Flat Glass Marketing Association
GA	Gypsum Association
HPMA	Hardwood Plywood Manufacturers Association
IBC	International Building Code
ICBO	International Conference of Building Officials
ICC	International Code Council
IEEE	Institute of Electrical and Electronic Engineers
JSMA	Joint Sealer Manufacturers Association
MFMA	Maple Flooring Manufacturers Association
ML/SFA	Metal Lath/Steel Framing Association
NAAMM	National Association of Architectural Metal Manufacturers
NAMM	National Association of Mirror Manufacturers
NBLP	National Bureau of Lathing & Plastering
NCPI	National Clay Pipe Institute
NCMA	National Concrete Masonry Association
NEMA	National Electrical Manufacturers Assn.
NESC	National Environmental Systems Contractors
NFPA	National Fire Protection Association
NFPA	National Forest Products Association
NHLA	National Hardwood Lumber Association
NOMMA	National Ornamental Metal Manufacturers Assn
NPVLA	National Paint, Varnish and Lacquer Assn.
NRMCA	National Ready Mixed Concrete Assn.
NRIMCA	
NSPE	National Roofing Contractors Association
NWMA	National Society of Professional Engineers National Woodwork Manufacturers Assn., Inc.
OSHA	
	Occupational Safety and Health Administration
PDCA	Painting and Decorating Contractors of America
PI	Perlite Institute, Inc.

PCA RFCI RVFC SBCCI SFPA	Portland Cement Association Resilient Floor Covering Institute Rubber and Vinyl Floor Council Southern Building Code Congress International, Inc. Southern Forest Products Association
SHLMA	Southern Hardwood Lumber Manufacturing Assn.
SDI	Steel Deck Institute
SDI	Steel Door Institute
SJI	Steel Joist Institute
SSPC	Steel Structures Painting Council
TCA	Tile Council of America, Inc.
UBC	Uniform Building Code
UL	Underwriter's Laboratories, Inc.
VBI	Venetian Blind Institute
VFI	Vinyl Fabrics Institute
WCLIB	West Coast Lumber Inspection Bureau
WRCLA	Western Red Cedar Lumber Association
WWPA	Western Wood Products Association

# 1.6 DRAWING SYMBOLS

- A. General: Except as otherwise indicated, graphic symbols used on drawings are those symbols recognized in the construction industry for purposes indicated. Where not otherwise noted, symbols defined by "Architectural Graphic Standards", published by the American Institute of Architects (AIA) and John Wiley & Sons, Inc., latest edition. Refer instances of uncertainty to Architect for clarification before proceeding.
- B. Mechanical/Electrical Drawings: Graphic symbols used in Mechanical/Electrical Drawings are generally aligned with symbols recommended by American Society of Heating, Refrigeration, and Air Conditioning Engineers (ASHRAE). Where appropriate, those symbols are supplemented by more specific symbols as recommended by other recognized technical organizations, including, but not limited to American Society of Mechanical Engineers (ASME), American Society of Professional Engineers (ASPE), Institute of Electrical and Electronic Engineers (IEEE) and similar organizations. Refer instances of uncertainty to Architect for clarification before proceeding.

### 1.7 GENERAL REQUIREMENTS

- A. Color, Texture, or Pattern Requirements:
  - 1. When color, texture, or pattern is specified, the item, product, or material shall be furnished in the specified color, texture, or pattern, as applicable.
  - 2. When more than one (1) approved manufacturer is named in the Specifications, Contractor may select any of the approved manufacturers and submit the full range of colors, textures, and patterns (standard and special) available of that manufacturer for the Architect's review and selection.
  - 3. When the term "match existing", or any derivative thereof appears in the Contract Documents, it means that the sample must match the Owner's existing work in every respect as to color, texture, and pattern, as applicable.
  - 4. When the term "match Architect's approved sample", or any derivative thereof appears in the Contract Documents, it means that the Architect has selected a sample which must be matched in every respect as to color, texture, and pattern, as applicable.
  - 5. When an item or product is specified of a manufacturer for which only one (1) color, texture, or pattern is available, and a color, texture, or pattern other than that one is specified, Contractor shall bring it to the attention of the Architect for a

decision prior to proceeding with the work. Do not proceed with the work until Architect has approved the color, texture, and pattern, as applicable.

- 6. When an item or product is specified of a manufacturer for which no color, texture, or pattern is specified, and colors, textures, and patterns are available, Contractor shall bring it to the attention of the Architect and submit the full range of colors, textures, and patterns (standard and special) available of that manufacturer for the Architect's review and selection. Do not proceed with the work until Architect has selected and approved the color, texture, and pattern, as applicable.
- 7. When due to the nature of the item, product, or material, i.e. face brick, tile pavers, natural stone, etc, Contractor shall submit sample or samples which exhibits the full range of characteristics (colors, i.e. lights and darks, as well as textures, and patterns) for which the item, product, or material is available. The Architect will select the color, texture, and pattern, as applicable, from those available and request a sample panel exhibiting the approved characteristics. The approved color range, texture, and pattern, as applicable will then become the standard for which all work on the project will be judged. Architect will be final judge as to having performed work in conformance with approved characteristics.
- 8. Under no circumstances are colors, textures, patterns, or any other characteristics for which an item, product, or material are available to be selected by anyone other than the Architect or his authorized representative.
- 9. Non-conforming work shall be removed from the site and replaced with new conforming work at no additional expense to Owner.
- B. Continuity of Building Envelope, Full Height Partitions, and Fire Rated Construction:
  - 1. Continuity of Building Envelope:
    - a. All materials such as exterior sheathing, membrane flashings, vapor barriers, insulations, dampproofing, waterproofing, roofing, flashings, etc. and all penetrations, holes, gaps, joints, and openings through such materials shall be sealed to ensure continuity of building envelope, whether indicated or not.
    - b. Refer instances of uncertainty to Architect for clarification before proceeding with work.
  - 2. Full Height Partitions:
    - a. All full height partitions shall be from floor to bottom of deck structure and shall be made to fit around steel joists, beams, etc., whether indicated or not.
    - b. Seal joints at top of partitions, in flutes of steel deck, and around structural elements with a compressible filler and/or sealant to accommodate movement due to expansion, contraction, and deflection, whether indicated or not. Treat seals in joints of fire rated partitions as specified below for fire rated construction, whether indicated or not.
    - c. Refer instances of uncertainty to Architect for clarification before proceeding with work.
  - 3. Fire Rated Construction:
    - a. All seals in fire rated construction, whether at top, bottom, or penetrations through fire rated construction, shall be made with firestopping and fire safing materials to maintain fire rating integrity of construction and satisfy authorities having jurisdiction, whether indicated or not.
    - b. Refer instances of uncertainty to Architect for clarification before proceeding with work.
- C. Plumbing Line Protection:
  - 1. Placing or washing materials, including, but not limited to the following, down any plumbing line or fixture is strictly forbidden.

- a. Concrete, cement, sludge, mortar, grout, plaster, or any other cementitious material
- b. Paint, paint thinner, turpentine, kerosene, gasoline, oil, or any other petroleum or hazardous products.
- 2. Cleaning painting equipment, including brushes in new or existing plumbing fixtures is strictly prohibited.
- 3. If requested, Contractor shall certify that all affected plumbing lines and fixtures are clean, free flowing and running. Plumbing lines and fixtures damaged as a result of any of the above shall be repaired or replaced at no expense to Owner. Contractor shall bear responsibility and all costs of fines, penalties, and legal fees attributed to violations as levied by authorities having jurisdiction.
- D. Hanging Items from Deck and Structure: Ducts, pipes, conduits, equipment, and other items indicated to be supported from the structure shall be accomplished using approved hangwires, hangers, or devices of type, size and material recommended to suit the application and installed in accordance with recommendations of the hanger or device manufacturer, Architect and/or Structural Engineer, or code authorities having jurisdiction, whichever is the more stringent requirement. Nothing shall be hung from the deck and structure unless directed to do so by the Architect and/or Structural Engineer. Powder activated devices in metal deck are not permitted.
- E. Ducts, Pipes, Conduits, and Wires: Shall be concealed in walls, chases, and enclosed areas out of view, unless specifically indicated as exposed or where exposure is required for proper function of item, such as air registers, air returns, louvers, grilles, vents, thermostats, electrical receptacles, telephone/data terminals and jacks, light switches, etc. Refer instances of uncertainty to Architect for clarification before proceeding.
- F. Fasteners:
  - 1. Unless specifically indicated or directed otherwise, all fasteners in work exposed to view, shall be concealed in the finished work.
  - 2. No fasteners shall show through or telegraph through exposed face of finished work and all finished surfaces shall be free of all evidence of the existence of fasteners.
  - 3. Fasteners shall be spaced to accurately and rigidly secure work in place.
  - 4. If not shown or otherwise required or recommended by manufacturer, standard, or code authorities having jurisdiction, fastener spacing shall not exceed 12 inches on center.
  - 5. Non-conforming work shall be removed from the site and replaced with new conforming work at no additional expense to Owner.
- G. Exposed Metal Work:
  - 1. Unless specifically indicated or directed otherwise, all exposed metal work shall be flat with all surfaces free of distortions, oil canning, waves, dents, scratches, weld marks, and other surface defects detrimental to good appearance or function.
  - 2. All steel exposed to exterior weather or moisture, either exposed or concealed in work, shall be hot-dip galvanized, phosphate treated for paint retention and shop prime painted.
  - 3. Non-conforming work shall be removed from the site and replaced with new conforming work at no additional expense to Owner.
- H. Continuous Date and Time Code Operated Devices:
  - 1. Devices used in the construction of this Project which use continuous date and time codes in their operation, whether software or hardware, and whether upgradable or not, including, but not limited to air handling, lighting, alarm, communication, security, and instrumentation systems, elevators, escalators and

other conveying systems, shall be Y2K compliant. In addition, such devices shall remain compliant for 100 years or the life of the device, whichever comes first.

# PART 2 - PRODUCTS

Not Used

#### **PART 3 - EXECUTION**

Not Used

# END OF SECTION 01 41 00

### SECTION 01 45 00 - QUALITY CONTROL

### PART 1 – GENERAL

### 1.1 SECTION INCLUDES

- A. Quality Assurance: Requirements for material and product quality and control of installation.
- B. Tolerances
- C. References and Standards
- D. Mock-ups
- E. Testing Laboratory Services
- F. Inspection Services
- G. Manufacturers' field services

# 1.2 RELATED SECTIONS

- A. Section 01 41 00 Regulatory Requirements
- B. Section 01 45 23 Testing and Inspecting Services
- C. Section 01 33 00 Submittal Procedures
- D. The Work of this Section shall be included as a part of all Sections of Work, whether referenced therein or not.

### 1.3 DESCRIPTION OF REQUIREMENTS

- A. Unless specifically noted otherwise, perform all Work shown, mentioned, or reasonably inferred and comply with all work restrictions.
- B. Many of the requirements specified elsewhere are included herein for reference and convenience. Where a conflict occurs between the Contract Documents, either within themselves or each other, the more stringent requirement or the most expensive combination of materials and workmanship shall prevail.
- C. Contractor shall:
  - 1. perform Work in accordance with the General Conditions, as specified herein, and with the quality control requirements of each Specification Section;
  - 2. perform Work in the highest quality workmanship, unless specified otherwise;
  - 3. join materials with a uniform and accurate fit so they meet with neat straight lines, free of smears, overlaps or irregularities, as applicable to the work;
  - 4. install all exposed materials appropriately level, plumb, and at accurate angles as shown and flush with adjoining materials;
  - 5. attach materials with sufficient strength, and with number and spacing of fasteners and attachments that will not fail until materials joined are broken or permanently deformed;
  - 6. use concealed fasteners, unless shown or directed otherwise.

# 1.4 QUALITY ASSURANCE AND CONTROL OF INSTALLATION

- A. Monitor quality control over suppliers, manufacturers, Products, services, site conditions, and workmanship, to produce Work of specified quality.
- B. Comply with manufacturers' instructions, including each step in sequence.
- C. Should manufacturer's instructions conflict with Contract Documents, request clarification from Architect/Engineer before proceeding.
- D. Comply with specified standards as minimum quality for the Work except where more stringent tolerances, codes, or specified requirements indicate higher standards or more precise workmanship.
- E. Perform Work by persons qualified to produce required and specified quality.
- F. Verify that field measurements are as indicated on shop drawings or as instructed by the manufacturer.
- G. Secure Products in place with positive anchorage devices designed and sized to withstand stresses, vibration, physical distortion, or disfigurement.

# 1.5 TOLERANCES

- A. Monitor fabrication and installation tolerance control of Products to produce acceptable Work. Do not permit tolerances to accumulate.
- B. Comply with manufacturers' tolerances. Should manufacturers' tolerances conflict with Contract Documents, request clarification from Architect/Engineer before proceeding.
- C. Adjust Products to appropriate dimensions; position before securing Products in place.

# 1.6 **REFERENCES AND STANDARDS**

- A. For Products or workmanship specified by association, trade, or other consensus standards, comply with requirements of the standard, except when more rigid requirements are specified or are required by applicable codes.
- B. Conform to reference standard by date of issue current on date of Owner-Contractor Agreement except where specific date is established by code.
- C. Obtain copies of standards where required by product specification sections.
- D. When specified reference standards conflict with Contract Documents, request clarification from Architect/Engineer before proceeding.
- E. Neither contractual relationships, duties, responsibilities of parties in Contract nor those of Architect/Engineer shall be altered from Contract Documents by mention or inference otherwise in reference documents.
- F. Refer to Section 01 41 00, Codes, Regulations and Standards, for additional information concerning applicable reference and standards requirements.

### 1.7 MOCK-UP REQUIREMENTS

- A. Tests will be performed under provisions identified in this section and identified in respective product specification sections.
- B. Assemble and erect specified items with specified attachment and anchorage devices, flashings, seals, and finishes.
- C. Accepted mock-ups shall be the comparison standard for remaining Work.
- D. Where mock-up has been accepted by Architect/Engineer and is specified in product specification sections to be removed; remove mock-up and clear area when directed to do so by Architect.
- E. Mock-up may be approved in phases as portions are completed.
- F. Project Mock-up Requirements: Provide an actual sample panel with the following properties:
  - 1. Size: Minimum 6 feet wide by 8 feet tall. Size may vary according to specific project requirements. Brace and support as required to withstand structural windloads.
  - 2. Materials: actual exterior finishes including, but not limited to face brick, cast stone, and plaster, actual building materials and assemblies indicating brick patterns on masonry and stud back-up as occurs with dampproofing and flashing as detailed, actual portion of aluminum storefront indicating jam, sill and head attachment and flashing details, and where appropriate, provide mock-up of special finish details, insets and reliefs, reveals, expansion and control joints, brick ledges, brick head and sills, pipe penetrations and waterproofing materials. Provide roof edge flashing and gutter section (as applicable) in pre-finished color as selected by Architect to cap the mock-up panel. Include a sealant joint at least 16 inches long.
  - 3. Drawing: Refer to mock-up diagram on Drawings for minimum project requirements. Mock-up drawing is for reference only. Actual mock-up drawing will be submitted by the Architect after submittals have been approved.

### 1.8 TESTING SERVICES

- A. Owner will appoint, employ, and pay for specified services of an independent firm to perform testing.
- B. The independent firm will perform tests and other services specified in individual specification sections and as required by the Architect/Engineer, Owner, or authority having jurisdiction.
- C. Testing and source quality control may occur on or off the project site. Perform off-site testing as required by the Architect/Engineer or the Owner.
- D. Reports will be submitted by the independent firm to the Owner, Architect/Engineer, and Contractor, indicating observations and results of tests and indicating compliance or non-compliance with Contract Documents.
- E. Cooperate with independent firm; furnish samples of materials, design mix, equipment, tools, storage, safe access, and assistance by incidental labor as requested.
  - 1. Notify Architect/Engineer and independent firm 48 hours prior to expected time for operations requiring services, or as specified in individual specification sections.

QUALITY CONTROL 01 45 00 - 3

- 2. Make arrangements with independent firm and pay for additional samples and tests required.
- F. Testing does not relieve Contractor to perform Work to contract requirements.
- G. Re-testing required because of non-conformance to specified requirements shall be performed by the same independent firm on instructions by the Architect/Engineer. Payment for re-testing will be charged to the Contractor by deducting testing charges from the Contract Sum/Price.
- H. Refer to Section 01 45 23, Inspection and Testing Laboratory Services, for additional information concerning testing, and submittal procedures and requirements for Testing Reports.

# 1.9 INSPECTION SERVICES

- A. Owner will appoint, employ, and pay for specified services of an independent firm to perform inspection.
- B. The independent firm will perform inspections and other services specified in individual specification sections and as required by the Architect/Engineer, Owner, or authority having jurisdiction.
- C. Inspecting may occur on or off the project site. Perform off-site inspecting as required by the Architect/Engineer or the Owner.
- D. Reports will be submitted by the independent firm to the Owner, Architect/Engineer, and Contractor, indicating inspection observations and indicating compliance or non-compliance with Contract Documents.
- E. Cooperate with independent firm; furnish safe access and assistance by incidental labor as requested.
  - 1. Notify Architect/Engineer and independent firm 48 hours prior to expected time for operations requiring services, or as specified in individual specification sections.
- G. Inspecting does not relieve Contractor to perform Work to contract requirements.
- H. Refer to Section 01 45 23, Inspection and Testing Laboratory Services, for additional information concerning inspections, and submittal procedures and requirements for Inspection Reports.

# 1.10 MANUFACTURERS' FIELD SERVICES

- A. When specified in individual specification sections, require material or product suppliers or manufacturers to provide qualified staff personnel to observe site conditions, conditions of surfaces and installation, quality of workmanship, start-up of equipment, test, adjust and balance of equipment as required, and to initiate instructions when necessary.
- B. Submit qualifications of observer to Architect/Engineer within ten (10) days after receipt of Notice to Proceed, in advance of required observations. Observer subject to approval of Architect/Engineer and Owner.
- C. Report observations and site decisions or instructions given to applicators or installers that are supplemental or contrary to manufacturers' written instructions.

D. Refer to Section 01 33 00, Submittal Procedures, for additional information concerning submittal procedures and requirements for Manufacturers Field Reports.

### PART 2 - PRODUCTS

Not Used.

#### **PART 3 - EXECUTION**

#### 3.1 EXAMINATION

- A. Verify existing site conditions and substrate surfaces are acceptable for subsequent Work. Beginning new Work means acceptance of existing conditions.
- B. Verify existing substrate is capable of structural support or attachment of new Work being applied or attached.
- C. Examine and verify specific conditions described in individual specification sections.
- D. Verify utility services are available, of correct characteristics, and in correct locations.

### 3.2 **PREPARATION**

- A. Clean substrate surfaces prior to applying next material or substance.
- B. Seal cracks or openings of substrate prior to applying next material or substance.
- C. Apply manufacturer required or recommended substrate primer, sealer, or conditioner prior to applying new material or substance in contact or bond.

# END OF SECTION 01 45 00

# SECTION 01 50 00 - TEMPORARY FACILITIES AND CONTROLS

### PART 1 - GENERAL

### 1.1 DESCRIPTION OF REQUIREMENTS

- A. Specific administrative and procedural minimum actions are specified in this Section, as extensions of provisions in other Contract Documents. These requirements have been included for special purposes as indicated. Nothing in this Section is intended to limit types and amounts of temporary work required, and no omission from this Section will be recognized as an indication that such temporary activity is not required for successful completion of the Work and compliance with requirements of the Contract Documents. Provisions of this Section are applicable to, but are not limited to the temporary power, temporary water, field office, sanitary facilities, storage facilities, signs, barriers, security, cleaning, first aid facilities, fire protection, construction aids, and parking facilities as further expanded in this Section.
- B. This Section covers the temporary facilities required by the Construction Manager for the Construction Manager's use only. Subcontractor's and Suppliers are responsible for their own temporary facilities as required by their scope of work. The Construction Manager will provide limited temporary facilities for the use of the subcontractor supplier as they become available. Subcontractors and Suppliers are referred to Section AC for further details.

### 1.2 JOB CONDITIONS

- A. General: Establish and initiate use of each temporary facility at time first reasonably required for proper performance of the Work. Terminate use and remove facilities at earliest reasonable time, when no longer required or when permanent facilities have, with authorized use, replaced their need.
- B. Conditions of Use:
  - 01 Install, operate, maintain and protect temporary facilities in a manner and at locations which will be safe, non-hazardous, sanitary, and protective of persons and property, and free of deleterious effects.
  - 02 Contractor shall be responsible for overloading or excess use of or damage resulting from the overloading or excess use of existing utilities.

### **PART 2 - PRODUCTS**

#### 2.1 MATERIALS

- A. Materials, not specifically described herein, but required for proper completion of Work of this Section, may be new or used as selected by the Contractor, but shall be of design, type, size, and strength recommended to suit intended purpose.
- B. Items required to protect the tenants, workmen, and public from danger, shall be sufficiently designed to protect them. Where required, exclude the public from all hazards.

# PART 3 - EXECUTION

### 3.1 UTILITIES

- A. Temporary Power: Construction Manager/Contractor shall use temporary power from existing utilities, but must provide all wiring, lamps, distribution of power, and equipment required for construction, inspection and testing of Work.
- B. Temporary Water: Construction Manager/Contractor shall use temporary water from existing utilities, but must provide all hoses and equipment required for construction, inspection and testing of Work.

# 3.2 FIELD OFFICE

A. Construction Manager/Contractor does not need to provide a job trailer. Job trailers may be provided if contractor deems them necessary.

#### 3.3 SANITARY FACILITIES

A. Construction Manager/Contractor shall furnish and maintain temporary sanitary facilities. Comply with regulations of State Department of Health and other authorities having jurisdiction. Construction Manager/Contractor may not use the Owner's facilities.

### 3.4 STORAGE FACILITIES

- A. Construction Manager/Contractor shall provide and maintain adequate weathertight lockable storage facilities, raised above the ground, with sides and top enclosed.
- B. Construction Manager/Contractor shall replace materials improperly stored and damaged by weathered conditions.
- C. Construction Manager/Contractor shall remove storage facilities when materials can be stored within the structure in a weathertight condition.
- D. Construction Manager/Contractor shall provide for temporary freeze protection as needed.

#### 3.5 SIGNS

- A. Other signs permitted at the site:
  - 1. Warning signs.
  - 2. Directional signs.
  - 3. Identification signs at field offices.
  - 4. Emergency medical services sign.
- B. Construction Manager/Contractor shall allow no other signs to be displayed at the project site, unless authorized by the Owner.

### 3.6 BARRIERS, CONSTRUCTION FENCE AND SECURITY

- A. Construction Manager/Contractor shall:
  - 1. Provide temporary barricades on all portions of the site adjacent to the construction and accessible to the public;
  - 2. Provide approved barriers around trees and plants designated to remain. Protect against vehicular traffic, stored materials, dumping, chemically injurious materials, water puddling and continuous running water;

TEMPORARY FACILITIES AND CONTROLS 01 50 00 - 2

- 3. Provide a minimum 6 feet-0 inch high chain link construction fence around the area of the kettle or pump trailer and any stored dangerous material for the duration of the construction period. Said temporary construction fence shall have lockable access gates necessary to adequately access the area in order to execute the project. Access gates shall be locked at the completion of each days' work; and
- 4. Determine if and when watchmen are necessary for protection of the Work, and provide such services when necessary. Neither the provision of watchmen nor the failure to provide watchmen shall relieve the Construction Manager/Contractor of responsibility in event of injury to persons or damage to property.

# 3.7 CLEANING

- A. Trash Removal: Construction Manager/Contractor shall clear the site of trash at least once a week. When rapid accumulation occurs, make more frequent removals. Remove highly combustible trash such as paper and cardboard daily. Dumpsters will not be allowed to overflow and should be emptied on a regular basis.
- B. Disposition of Debris: Construction Manager/Contractor shall remove debris from site and make legal disposition. Locations for disposal shall be of the Construction Manager's/Contractor's choice within the above restriction. No debris nor material may be buried or burned at the site, without the permission of authorities having jurisdiction. Take necessary precautions to prevent accidental burning of materials by avoiding large accumulations of combustible materials.

# 3.8 TEMPORARY FIRST AID FACILITIES

- A. Construction Manager/Contractor shall:
  - 1. Provide first aid equipment and supplies, with qualified personnel continuously available to render first aid at the site; and
  - 2. Provide a sign, posted at the field office telephone or job site construction manual, listing the telephone numbers for emergency medical services: Physicians, ambulance services and hospitals.

### 3.9 TEMPORARY FIRE PROTECTION

- A. Construction Manager/Contractor shall provide a fire protection and prevention program for employees and personnel at the site; and provide and maintain fire extinguishing equipment ready for instant use at all areas of the Project and at specific areas of critical fire hazard.
- B. Equipment:
  - 1. Hand extinguishers of the types and sizes recommended by the National Board of Fire Underwriters to control fires from particular hazards.
  - 2. Barrels of water with buckets designated for fire-control purposes.
  - 3. Water hoses connected to an adequate water pressure and supply system as needed.
  - 4. Construction period use of permanent fire protection system.
- C. Enforce fire-safety discipline:
  - 1. Store volatile materials in an isolated, protected location.
  - 2. Avoid accumulations of flammable debris and waste in or about the Project.
  - 3. Prohibit smoking in the vicinity of hazardous conditions.
  - 4. Closely supervise welding and torch-cutting operations in the vicinity of combustible materials and volatile conditions.

TEMPORARY FACILITIES AND CONTROLS 01 50 00 - 3

- 5. Supervise locations and operations of portable heating units and fuel.
- D. Construction Manager/Contractor shall maintain fire extinguishing equipment in working condition, with current inspection certificate attached to each extinguisher.

### 3.10 CONSTRUCTION AIDS

- A. Construction Manager/Contractor shall:
  - 1. Provide construction aids and equipment required to assure safety for personnel and to facilitate the execution of the work, i.e. scaffolds, staging, ladders, stairs, ramps, runways, platforms, railings, hoists, cranes, chutes and other equipment applicable and required for the performance of the work; and
  - 2. Maintain all equipment in a first-class, safe condition.

# 3.11 PARKING FACILITIES

- A. Construction Manager/Contractor shall:
  - 1. Coordinate location of parking for personnel and employees at the facility to avoid interference with traffic, walks, work and storage areas, or with materials-handling equipment.

# END OF SECTION 01 50 00

# SECTION 01 73 29 - CUTTING AND PATCHING

### PART 1 - GENERAL

### 1.1 DESCRIPTION OF REQUIREMENTS

- A. Definition: "Cutting and Patching" includes cutting into existing construction to provide for the installation or performance of other work and subsequent fitting and patching required to restore surfaces to their original undamaged condition, including original fire rating of fire rated construction.
  - 1. Cutting and patching is performed for coordination of the work for access or inspection, to obtain samples for testing, as indicated or required, to remove/replace defective work or work not conforming to the contract documents, to permit alterations to be performed, or for other similar purposes.
  - 2. Cutting and patching performed during the manufacture of products or during the initial fabrication, erection, or installation processes is not considered to be "cutting and patching" under this definition. Drilling of holes to install fasteners and similar operations are also not considered to be "cutting and patching".
- B. Refer to other Sections of these Specifications for specific cutting and patching requirements and limitations applicable to individual units of work.
  - 1. Unless otherwise specified, requirements of this Section also apply to mechanical and electrical work.

# 1.2 QUALITY ASSURANCE

A. Visual requirements - Do not cut and patch work exposed on the building's exterior or in its occupied spaces, in a manner that would, in the Architect's opinion, result in lessening the building's aesthetic qualities. Do not cut and patch work in a manner that would result in substantial visual evidence of cut and patchwork. Remove and repair or replace work judged by the Architect to be cut and patched in a visually unsatisfactory manner

### 1.3 RELATED WORK

A. All Sections of Work requiring cutting and patching, including electrical requirements.

# 1.4 SUBMITTALS

- A. Procedural Proposal for Cutting and Patching Where prior approval of cutting and patching is required, submit proposed procedures for this work well in advance of the time work will be performed and request approval to proceed. Include the following information, as applicable, in the submittal.
  - 1. Describe nature of the work and how it is to be performed, indicating why cutting and patching cannot be avoided. Describe anticipated results of the work in terms of changes to existing work, including structural, operational, and visual changes as well as other significant elements.
  - 2. List products to be used and firms including their qualifications that will perform the work. Also, provide cost proposals when applicable.
  - 3. Give dates when work is expected to be performed.
  - 4. List utilities that will be disturbed or otherwise be affected by work, including those that will be relocated and those that will be disconnected or out-of service temporarily. Indicate how long utility service will be disrupted.

### **PART 2 - PRODUCTS**

# 2.1 MATERIALS

- A. General Except as otherwise indicated or as directed by Architect, use materials for cutting and patching that are identical to materials being cut and patched. If identical materials are not available, or cannot be used, use materials that match existing adjacent surfaces to the fullest extent possible with regard to visual effect. Use materials for cutting and patching that will result in equal or better performance characteristics.
  - 1. The use of trade name and supplier's name and address is to indicate a possible source of the material or product. Product of the same type from other sources shall not be excluded provided they possess like physical and functional characteristics, except where specified as no substitutions allowed or where a material or product is specified as the basis of specification and no other approved manufacturers are listed.
  - 2. Use materials, products, and devices to maintain integrity of fire rating of existing fire rated construction which comply with the requirements of authorities having jurisdiction.

# PART 3 - EXECUTION

# 3.1 EXAMINATION

- A. Before starting work, examine the surfaces to be cut and patched and the conditions under which the work is to be performed. If unsafe or otherwise unsatisfactory conditions are encountered, take corrective action before proceeding with the work.
  - 1. Before the start of cutting work, meet at the work site with all parties involved in cutting and patching, including mechanical and electrical trades. Review areas of potential interference and conflict between the various trades. Coordinate layout of the work and resolve potential conflicts before proceeding with the work.
  - 2. After uncovering work, examine conditions affecting installation of product or performance of work.
  - 3. Report unsatisfactory or questionable conditions to Architect in writing; do not proceed with work until Architect has provided further instructions.

### 3.2 PREPARATION

- A. Provide temporary support to prevent failure of the work to be cut.
- B. Protect other work during cutting and patching to prevent damage. Provide protection from adverse weather conditions of that part of the Project that may be exposed during cutting and patching operations.
- C. Take precautions not to cut existing pipe, conduit, ducts, or wires serving the building, but scheduled to be removed or relocated until provisions have been made to bypass them.

### 3.3 PERFORMANCE

- A. General Employ only skilled workmen to perform the cutting and patching work. Except as otherwise indicated or as approved by Architect, proceed with cutting and patching at the earliest feasible time and complete the work without delay.
- B. Cut the work using methods that are least likely to damage work to be retained or adjoining work. Where possible, review proposed cutting and patching procedures with the original installer and comply with original installer's recommendations.
  - 1. In general, where cutting is required, use hand or small power tools designed for sawing or grinding, not hammering and chopping. Cut through concrete and masonry using a cutting machine such as a Carborundum saw or core drill to insure a neat hole. Cut holes and slots neatly to size required with minimum

CUTTING AND PATCHING 01 73 29 - 2 disturbance of adjacent work. To avoid marring existing finished surfaces, cut or drill from the exposed or finished side into concealed surfaces. Temporarily cover openings when not in use.

- 2. Comply with requirements of other applicable sections where cutting and patching requires excavating and backfilling.
- 3. By-pass utility services such as pipe and conduit, before cutting, where such utility services are shown or required to be removed, relocated, or abandoned. Cut-off conduit and pipe in walls or partitions to be removed. After by-passing and cutting, cap, valve, or plug and seal tight remaining portion of conduit and pipe to prevent entrance of moisture, vermin, or other foreign matter.
- C. Patching Patch with seams which are durable and as invisible as possible. Comply with specified tolerance, if any, for the work.
  - 1. Where feasible, inspect and test patched areas to demonstrate integrity of work.
  - 2. Restore exposed finishes of patched areas and where necessary extend finish restoration into retained adjoining work in a manner which will eliminate evidence of patching and refinishing.
  - 3. Where removal of walls or partitions extends one finished area into another finished area, patch and repair floor, wall, and ceiling surfaces in the new space to provide an even surface of uniform color and appearance. If necessary to achieve uniform color and appearance, remove existing floor and wall coverings or materials, and ceiling finish materials and replace with new materials.
    - a. Where patch occurs in a smooth painted surface, extend final paint coat over entire unbroken surface containing patch, after patched area has received prime and base coats.
  - 4. Patch, repair, or re-hang existing ceilings as necessary to provide an even plane surface of uniform appearance.
  - 5. Fit work airtight to pipes, sleeves, ducts, conduit and other penetrations through non-fire-rated floors and walls, and through finished surfaces.
- D. Fire Rated Construction Where cutting and patching is necessary in existing fire rated construction, use sealant and other fire resistive materials, products, and devices as required and acceptable by the authorities having jurisdiction to repair, patch, and otherwise restore original fire rating and integrity of construction.

# 3.4 CLEANING

A. Thoroughly clean area and spaces where work is performed or used as access to work. Remove completely: paint, mortar, cement, oils, putty, sealant, and items of similar nature. Thoroughly clean piping, conduit, and similar features before painting or other finishes are applied. Restore damaged pipe covering to its original undamaged condition.

# END OF SECTION 01 73 29

## SECTION 01 77 00 - CLOSEOUT PROCEDURES

## PART 1 - GENERAL

## 1.1 PRE-CLOSEOUT MEETING

A. Pre-Closeout Meeting: Schedule and convene Pre-Closeout Meeting with Owner and Architect in accordance with Section 01 31 19, Project Meetings.

## 1.2 SUBSTANTIAL COMPLETION

- A. The items listed in Document CB, Supplementary Conditions, Paragraph 9.8 and the following items shall be completed before Substantial Completion will be granted:
  - 1. Contractor's Completion List (Punch List): Submit a thorough list of items to be completed or corrected, along with a written request for Substantial Completion and for review of the Work or portion of the Work. The Architect/Engineer's Project Representative, at their discretion, may attend and assist in the preparation of the Contractor's Punch List.
  - 2. Architect's Supplemental Punch List: The Architect/Engineer, along with the Owner at the Owner's discretion, will inspect the Work utilizing the Contractor's prepared Punch List, noting completed items and incomplete items, and will prepare a supplemental list of items that have been omitted or incomplete items that were not previously noted.
  - 3. Operations and Maintenance Manuals: Submit as described in paragraph 1.3.
  - 4. Final Cleaning: Provide final cleaning and adequate protection of installed construction as described in paragraph 1.6 and 1.7.
  - 5. Starting of systems: Start up equipment and systems as described in paragraph 1.8.
  - 6. Testing and balancing: Testing and balancing of systems must be performed and completed by Owner's forces, and the report submitted and accepted by Architect/Engineer and Owner, as described in the Contract Documents. Make adjustments to equipment as required to achieve acceptance.
  - 7. Demonstrations: If required by individual specification sections or by Owner, provide demonstrations and instructions for use of equipment as described in paragraph 1.9.
- B. Date of Substantial Completion: Complete or correct items identified on Punch List and confirm that all items have been corrected prior to Architects re-inspection. Architect/Engineer, along with the Owner, will re-inspect the corrected work to establish the Date of Substantial Completion. Incomplete items remaining will be appended to the Certificate of Substantial Completion (AIA G704). The Date of Substantial Completion represents day one (1) of the closeout period, and represents the date of commencement of the Contractors correctional period and all warranty periods as described and required by the Contract Documents, except as amended in the Certificate of Substantial Completion and elsewhere in the Contract Documents.
- C. Certificate of Substantial Completion: When the Work or designated portion thereof is substantially complete; Architect will prepare the Certificate of Substantial Completion to be executed by the Owner and Contractor. Items on the appended Punch List shall be completed or corrected within the time limits established in the Certificate.

## 1.3 OPERATIONS AND MAINTENANCE MANUAL

- A. As a requirement for Substantial Completion, the final Operation and Maintenance Manual shall be submitted to, and reviewed and accepted by the Architect prior to issuance of the Certificate.
- B. Prepare 3-ring D-slant binder cover and spline with printed title "OPERATIONS AND MAINTENANCE MANUAL", title of project, and subject matter of binder when multiple binders are required.
- C. Submit one (1) copy of preliminary Operations and Maintenance Manuals to respective consultants (Civil, MEP, Structural, *etc.*) for review of conformance with contract requirements prior to submitting final to Architect. Allow time for proper review.
- D. Internally subdivide binder contents with permanent page dividers, logically organized as described below; with tab titling clearly printed under reinforced laminated plastic tabs.
- E. Drawings: Provide with reinforced punched binder tab. Bind in with text; fold larger drawings to size of text pages.
- F. Contents: Prepare Table of Contents for each volume, with each product or system description identified, typed on white paper, in three parts as follows:
  - 1. Part 1: Directory, listing names, addresses, and telephone numbers of Architect/Engineer, Contractor, Subcontractors, and major equipment suppliers.
  - 2. Part 2: Operation and Maintenance, arranged by system and subdivided by specification section. For each category, identify names, addresses, and telephone numbers of Subcontractors and suppliers. Identify the following:
    - a. Significant design criteria.
    - b. List of equipment.
    - c. Parts list for each component.
    - d. Equipment start-up instructions
    - e. Operating instructions.
    - f. Maintenance instructions for equipment and systems.
    - g. Maintenance instructions for finishes, including recommended cleaning methods and materials, and special precautions identifying detrimental agents.
  - 3. Part 3: Project documents and certificates, including the following:
    - a. Product data.
    - b. Air and water balance reports.
    - c. Photocopies of warranties, certificates and bonds. Submit originals with Closeout Documents as specified below.
- G. Submit one (1) final original and two (2) copies to Architect.

## 1.4 PROJECT CLOSEOUT

- A. Final Payment will not be authorized by the Architect until the Architect finds the Work acceptable under the Contract Documents, subject to the completion and acceptance of the following requirements and other applicable Contract requirements:
  - 1. Close-out Documents: Provide bound closeout documents as described in paragraph 1.5. Refer to Document CB, Supplementary Conditions, Paragraph 9.10 for additional information.
  - 2. Record Documents: Submit as described in paragraph 1.10.
  - 3. Extra materials: Provide extra stock, materials, and products as described in paragraph 1.11 when required by individual specification sections.

CLOSEOUT PROCEDURES 01 77 00 - 2

- 4. Locks: Make final changeover of permanent locks and transmit keys to the Owner. Advise the Owner's personnel of changeover in security provisions.
- 5. Temporary Facilities: Discontinue and remove temporary facilities from the site, along with mockups, construction aids, and similar elements.
- 6. Warranties, Certificates and Bonds: Execute and assemble transferable warranty documents, certificates, and bonds from subcontractors, suppliers, and manufacturers as described in paragraph 1.12.
- 7. Final Inspection and Acceptance by Architect is achieved as described in paragraph 1.13.

## 1.5 CLOSEOUT DOCUMENTS

- A. Coordinate the following items with the requirements of Document CB, Supplementary Conditions of the Contract.
- B. Prepare 3-ring D-slant binder cover and spline with printed title "CLOSEOUT DOCUMENTS", title of project, and subject matter of binder when multiple binders are required. Submit one (1) original and two (2) copies.
- C. Internally subdivide binder contents with permanent page dividers, logically organized as described below; with tab titling clearly printed under reinforced laminated plastic tabs.
- D. The close-out documents shall be neatly organized and easily useable as determined by the Architect and Owner. Separate Close-out Documents binders from Operations and Maintenance Manuals. Documents identified as "affidavit" shall be notarized.
- E. Contents: Prepare Table of Contents for each volume, with each item description identified, typed on white paper, in five (5) parts as follows:
  - 1. Part 1: Directory, listing names, addresses, and telephone numbers of Architect/Engineer, Contractor, Subcontractors, and major equipment suppliers. All General Contractor's vendors/suppliers and subcontractors that provided materials or performed any work related to this project must be listed on this form. Submit Final List of Subcontractors.
  - 2. Part 2: Closeout Documents and Affidavits, include the following:
    - a. AIA G707 Consent of Surety to Final Payment;
    - b. AIA G706 Contractor's Affidavit of Payment of Debts and Claims;
    - c. AIA G706A Contractor's Affidavit of Release of Liens;
    - d. Subcontractor's Release of Lien: Include contractor's, subcontractor's and direct material and equipment supplier's separate final releases. Submit on attached **Close-out Form "A"** Affidavit of Subcontractor's Release of Lien.
  - 3. Part 3: Project documents and certificates, including the following:
    - a. Copy of Certificate of Substantial Completion (AIA G704);
    - b. Copy of All Permits;
    - c. Copy of Final Utility Bill or letter of transfer;
    - d. Copy of Certificate of Occupancy;
    - e. Certification of Project Compliance: Submit on attached Close-out Form
      "B". Owner and Architect will initiate form and forward to Contractor for signature once Substantial Completion is established;
    - f. Hazardous Material Certificate: Submit on attached **Close-out Form "C"**. Affidavits from Contractor, Subcontractors and General Contractor's vendors or suppliers stating that no hazardous materials/products have been used or installed in this project.
  - 4. Part 4: Warranties, compile sequentially based on specification sections:
    - a. General Contractor's Warranty: Submit on company letterhead as described below. This Warranty shall state all sections of Work

CLOSEOUT PROCEDURES 01 77 00 - 3 performed by General Contractor's own forces, and warranty period for each section of Work;

- b. Subcontractor's Warranty: notarized, and submitted on attached **Closeout Form "D"**. This Warranty shall state all sections of Work performed by the subcontractor and warranty period;
- 5. Part 5: Receipts:
  - a. Extra Stock: Provide original receipts for delivery of "Extra Stock" items as described below, (if applicable). Receipts must be signed by an authorized Owner's representative;
  - b. Keys: Provide original receipts for delivery of "Keys", (if applicable). Receipts must be signed by an authorized Owner's representative.
- F. In addition to the three (3) required close-out binders listed above, provide Architect with one (1) separate binder for their records containing the following:
  - 1. Directory, listing names, addresses, and telephone numbers of Architect/Engineer, Contractor, Subcontractors, and major equipment suppliers;
  - 2. all MSDS sheets for the project;
  - 3. all warranties from Contractor, subcontractors, direct suppliers, and manufacturers.
- G. Failure to complete and close-out project after substantial completion may result in liquidated damages being assessed to the Contractor. Refer to Conditions of the Contract for additional requirements and liquidated damages.

## 1.6 FINAL CLEANING

- A. Execute final cleaning prior to final project inspection and acceptance.
- B. Clean interior and exterior glass, and surfaces exposed to view; remove temporary labels, stains and foreign substances, polish transparent and glossy surfaces, vacuum carpeted and soft surfaces, mop hard floor surfaces.
- C. Remove smudges, marks, stains, fingerprints, soil, dirt, spots, dust, lint, and other foreign materials from finished and exposed surfaces
- D. Clean equipment and fixtures to sanitary condition with cleaning materials appropriate to surface and material being cleaned.
- E. Clean and replace filters of operating equipment as required by Contract Documents
- F. Clean debris from roofs, gutters, downspouts, and drainage systems.
- G. Clean site; sweep paved areas, rake clean landscaped surfaces.
- H. Remove waste and surplus materials, rubbish, and temporary construction facilities from site.

## 1.7 PROTECTING INSTALLED CONSTRUCTION

- A. Protect installed Work and provide special protection where specified in individual specification sections until Work is accepted by Architect and Owner.
- B. Provide temporary and removable protection for installed products. Control activity in immediate work area to prevent damage.
- C. Provide protective coverings at walls, projections, jambs, sills, and soffits of openings.

- D. Protect finished floors, stairs, and other surfaces from traffic, dirt, wear, damage, or movement of heavy objects, by protecting with durable sheet materials.
- E. Prohibit traffic or storage upon waterproofed or roofed surfaces. When traffic or activity is necessary, obtain recommendations for protection from waterproofing or roofing material manufacturer.
- F. Prohibit traffic from landscaped areas.

## 1.8 STARTING OF SYSTEMS

- A. Coordinate schedule for start-up of various equipment and systems.
- B. Notify Architect/Engineer and Owner 48 hours prior to start-up of each item.
- C. Verify each piece of equipment or system has been checked for proper lubrication, drive rotation, belt tension, control sequence, and for conditions which may cause damage.
- D. Verify tests, meter readings, and specified electrical characteristics agree with those required by equipment or system manufacturer.
- E. Verify wiring and support components for equipment are complete and tested.
- F. Execute start-up under supervision of Contractors' personnel, and installer in accordance with manufacturers' instructions.
- G. When specified in individual specification sections or required by manufacturer, require manufacturer to provide authorized representative to be present at site to inspect, check, and approve equipment or system installation prior to start-up, and to supervise placing equipment or system in operation.
- H. When specified in individual specification sections or required by Owner or Architect/Engineer, submit a written report in accordance with Section 01300, Submittal Procedures, that equipment or system has been properly installed and is functioning correctly.

## 1.9 DEMONSTRATION AND INSTRUCTIONS

- A. Demonstrate operation and maintenance of products to Owner's personnel a minimum of 48 hours prior to date of Final Completion in accordance with Owner's requirements.
- B. Demonstrate Project equipment instructed by qualified manufacturer's representative who is knowledgeable about the Project and equipment.
- C. For equipment or systems requiring seasonal operation, perform demonstration for other season within six (6) months.
- D. Utilize maintenance manuals as basis for instruction. Review contents of manual with Owner's personnel to explain all aspects of operation and maintenance.
- E. Demonstrate start-up, operation, control, adjustment, trouble-shooting, servicing, maintenance, and shutdown of each item of equipment.
- F. Prepare and insert additional data in maintenance manuals when need for additional data becomes apparent during instruction.

G. Review and verify proper star-up and operation of equipment prior to scheduling demonstrations with Owner.

## 1.10 PROJECT RECORD DOCUMENTS

- A. Record Documents, as described in Section 01 78 39, shall be submitted at Project Closeout. Final Payment will not be authorized by the Architect until final review and acceptance by Architect and Engineers is achieved in accordance with the Owners requirements.
- B. At the Contractors request, and with associated fee, Architect may provide electronic versions of the construction drawing and specification files for Contractors use, subject to the terms and conditions of Architects standard electronic document transfer agreement.
- C. Submit reproducible to respective consultants (Civil, Structural, MEP, *etc.*) for review. Consultant will mark-up corrections and return to Contractor for final revisions. Make final revisions prior to submitting to Architect.
  - 1. Format: One (1) set of film positive reproducibles and two (2) sets bluelines of approved reproducibles.
  - 2. In addition, provide the Owner with one (1) set of Record Drawings on a non-rewritable CD in AutoCAD<sup>®</sup> latest release.

## 1.11 EXTRA STOCK, MATERIALS AND MAINTENANCE PRODUCTS

- A. Furnish extra stock, maintenance, and extra products in quantities specified in individual specification sections.
- B. Deliver to Project site and place in location as directed by Owner; obtain signed receipt from Owner's authorized representative prior to final application for payment. Delivery of materials to, or obtaining receipt from anyone other than Owner's authorized representative may constitute breach of this requirement and may require delivery of additional materials at no cost to the Owner if original materials are misplaced.
- C. Include signed receipts for delivery of extra stock and materials, including keys, with Closeout Documents.

## 1.12 WARRANTIES, CERTIFICATES AND BONDS

- A. Definitions:
  - 1. Standard Product Warranties: preprinted written warranties published by individual manufacturers for particular products and are specifically endorsed by the manufacturer to the Owner.
  - 2. Special Warranties: written warranties required by or incorporated in the Contract Documents, either to extend time limits provided by standard warranties or to provide coverage of specific defects, or both.
- B. In accordance with the general warranty obligations under Paragraph 3.5 of the General Conditions as amended by the Supplementary Conditions, the General Contractor's warranty shall be for a period of one (1) year following the date of Substantial Completion, hereinafter called the one-year warranty period. The Contractors one-year general warranty shall include all labor, material and delivery costs required to correct defective material and installation. This warranty shall not limit the Owner's rights with respect to latent defects, gross mistakes, or fraud.
- C. The Contractor's one-year warranty shall run concurrently with the one (1) year period for correction of Work required under Paragraph 12.2 of the General Conditions.

- D. In addition to the Contractors one-year warranty, Special Warranties as described in individual specifications sections, shall extend the warranty period for the period specified without limitation in respect to other obligations which the Contractor has under the Contract Documents.
- E. Manufacturer's disclaimers and limitations on product warranties do not relieve the Contractor of the warranty on the Work that incorporates the products, nor does it relieve the suppliers, manufacturers, and subcontractors required to countersign special warranties with the Contractor.
- F. Warranty Requirements:
  - 1. When correcting warranted Work that has failed, remove and replace other Work that has been damaged as a result of such failure or that must be removed and replaced to provide access for correction of warranted Work.
  - 2. When Work covered by a warranty has failed and been corrected by replacement or reconstruction, reinstate the warranty by written endorsement. The reinstated warranty shall be equal to the original warranty with an equitable adjustment for depreciation.
  - 3. Upon determination that Work covered by a warranty has failed, replace or rebuild the Work to an acceptable condition complying with requirements of Contract Documents. The Contractor is responsible for the cost of replacing defective Work regardless of whether the Owner and benefited from use of the Work through a portion of its anticipated useful service life.
  - 4. Written warranties made to the Owner are in addition to implied warranties, and shall not limit the duties, obligations, rights and remedies otherwise available under the law, nor shall warranty periods be interpreted as limitations on time in which the Owner can enforce such other duties, obligations, rights, or remedies.
  - 5. The Owner reserves the right to refuse to accept Work for the Project where a special warranty, certification, or similar commitment is required on such Work or designated portion of the Work, until evidence is presented that entities required to countersign such commitments are willing to do so.
- G. Compile copies of each required warranty properly executed by the Contractor and the subcontractor, supplier, or manufacturer. Verify documents are in proper form, contain full information, and are notarized. Co-execute warranties, certificates and bonds when required and include signed warrantees with Closeout Documents submitted to the Architect.

## 1.13 FINAL COMPLETION AND FINAL PAYMENT

- A. Final Notice and Inspection:
  - 1. When all items on the Punch List have been corrected, final cleaning has been completed, and installed work has been protected, submit written notice to the Architect that the Work is ready for final inspection and acceptance.
  - 2. Upon receipt of written notice that the Work is ready for final inspection and acceptance, the Architect and Engineer will make final inspection.
- B. Final Change Order: When the Project Closeout items described above are successfully completed and the Work is found acceptable to Architect/Engineer and Owner, a Final Change Order will be executed. This Change Order will include any Allowance adjustments as required by the Contract Documents.
- C. Final Application for Payment: When all of the above items are successfully complete, submit to the Architect a final Application for Payment and request for release of retainage.

CLOSEOUT PROCEDURES 01 77 00 - 7 D. Release of Retainage: Release of retainage will not be authorized by the Architect until Contractor completes all requirements for close-out to the satisfaction of the Owner and Architect as described herein.

## 1.14 TERMINAL INSPECTION

- A. Immediately prior to expiration of the one (1) year period for correction of the Work, the Contractor shall make an inspection of the work in the company of the Architect and the Owner. The Architect and the Owner shall be given not less than ten (10) days notice prior to the anticipated date of terminal inspection.
- B. Where any portion of the work has proven to be defective and requires replacement, repair or adjustment, the Contractor shall immediately provide materials and labor necessary to remedy such defective work and shall execute such work without delay until completed to the satisfaction of the Architect and the Owner, even if the date of completion of the corrective work may extend beyond the expiration date of the correction period.
- C. The Contractor shall not be responsible for correction of work which has been damaged because of neglect or abuse by the Owner nor the replacement of parts necessitated by normal wear in use.

## PART 2 - PRODUCTS

Not Used

## PART 3 - EXECUTION

Not Used

## END OF SECTION 01 77 00

**CLOSE-OUT FORM "A"** 

	SUBCONTRACTOR'S	AFFIDAVIT OF REL	EASE OF LIEN
STAT	E OF		
COU	NTY OF		
KNO\	WALL MEN BY THESE PRESENTS:		
		, being first	t duly sworn, deposes and says:
1.	That he / she is the who supplied, installed, and  /or erect authorized to make this Affidavit and		, the subcontractor d below, and that, he /she is duly se:
	Project: <u>Multi-Campus Building Envel</u> Owner: <u>Galveston Independent Scho</u> Work Performed:	ol District Archite	ct: BEAM
2.	That all work required under the subject subcontractor of the subject construction project has been performed in accordance with the terms thereof, that all material men, sub-subcontractors, mechanics, and laborers have been paid and satisfied in full and that there are no outstanding claims of any character arising out of the performance of said subcontractor which have not been paid and satisfied in full.		
3.	resulting from injury or death to any e	employees, sub-subco ontract, or any suits o	are no unsatisfied claims for damages ontractors, or the public at large arising or claims for any other damages of any on the property of the Owner.
4.	That he / she has received full payment of all sums due him / her for materials furnished and services rendered by the undersigned in connection with the performance of said subcontract and has and does hereby release the Owner and the Architect and his consultants and the Contractor from any and all claims of any character arising out of or in any way connected with performance of said subcontract.		
ATTE	ST (If Corporation)		
···· · · ( • • · • • · • · · · )		Name of Subcontractor	
	Secretary	(Ву)	(Title)
STAT	E OF	JURAT	
	NTY OF		
	n to and subscribed before me on this _	day of	, 20
(Seal)	)	(Notary P	ublic Signature)

## CLOSE OUT FORM "B"

## CERTIFICATION OF PROJECT COMPLIANCE

Completion of this form is required under the provisions of (1.1036)(c)(3)(F) TAC for all public school district construction projects. Instructions for completion of this form can be found on page 2.

1. PROJECT INFORMATION

Facility: Address: DISTRICT: Galveston ISD ARCHITECT/ENGINEER: BEAM CONTRACTOR/CM:

City: Galveston, TX, 77550

CONTRACT DATE:

# BRIEF DESCRIPTION OF PROJECT:

DATE DISTRICT AUTHORIZED PROJECT:

## 2. CERTIFICATION OF DESIGN AND CONSTRUCTION

The intent of this document is to assure that the school district has provided to the architect/engineer the required information and the architect/engineer has reviewed the School Facilities Standards as required by the State of Texas, and used his/her reasonable professional judgment and care in the architectural/engineering design and that the contractor has constructed the project in a quality manner in general conformance with the design requirements and that the school district certifies to project completion.

**3. The District** certifies that the educational program and the educational specifications of this facility along with the identified building code to be used have been provided to the architect/engineer.

DISTRICT:	BY:	DATE:
4. The Architect/Engineer certifies the above in the building(s) were designed in accordance with been designed to meet or exceed the design crit educational adequacy, and construction quality a adopted by the Commissioner of Education, Jun	nformation was received from the n the applicable building codes. F teria relating to space (minimum as contained in the School Facilit	e school district, and that Further, the facility has square footage), ies Standards as

ARCHITECT/ENGINEER:	BY:	DATE:	
5. The Contractor/CM certifies that thi	is project has been constru	ucted in general conformance with the	
construction documents as prepared b	v the architect/engineer lis	sted above.	

CONTRACTOR/CM: BY: DATE:
--------------------------

6. The District certifies completion of the project (as defined by the architect/engineer and contractor).

CLOSEOUT PROCEDURES
01 77 00 - 10

DISTRICT:

BY:

DATE:

INSTRUCTIONS FOR COMPLETION OF "CERTIFICATION OF PROJECT COMPLIANCE" FORM

Section 1. Identify the following:

- name and address of the school facility

- name of the school district
- the Architect/Engineer and Contractor
- the date of execution of the construction contract
- the date that the school district authorized the superintendent to hire an architect/engineer
- scope of the project.

Section 2. This section outlines the intent of the document. No action required.

Section 3. This section is to be executed by the school district upon transmittal of the information (as listed) to the architect/engineer and is to remain in the custody of the school district throughout the entire project.

Section 4. This section is to be executed by the architect/engineer upon completion of the plans and specifications and in conjunction with the completion of the plan review for code compliance (ref. 19 TAC §61.1033 or §61.1036, School Facilities Standards) and returned to the school district's files.

Section 5. This section is to be executed by the contractor upon substantial completion of the project and retained in the school district's files.

Section 6. This section is to be executed by the school district upon acceptance and occupancy of the project.

NOTE: DO NOT SUBMIT THIS DOCUMENT TO THE TEXAS EDUCATION AGENCY. The school district will retain this document in their files indefinitely until review and/or submittal is required by representatives of the Texas Education Agency.

**CLOSE-OUT FORM "C"** 

SUBCONTRACTOR	R HAZARDOUS MATERIAL CERTIFICATE
THE STATE OF	PROJECT: Multi-Campus Building Envelope Repairs
	(Package 4)
COUNTY OF 0	OWNER: Galveston Independent School District
ŀ	ARCHITECT: BEAM
S	SPECIFICATION SECTION(S):
KNOW ALL MEN BY THESE PRESENTS	S:
is the	, being first duly sworn, deposes and says that he / she
	the subcontractor / supplier, s) of work referenced above, and that he / she is duly authorized
to certify to the best of his / her informati products have been incorporated into the	ion, knowledge, and belief <u>no asbestos, lead or PCB containing</u>
ATTEST (If Comparation)	
ATTEST (If Corporation)	Name of Subcontractor / Supplier
(Title)	Secretary (By)
(The)	
	JURAT
THE STATE OF	
COUNTY OF	
Sworn to and subscribed before me on th	is day of, 20
(Seal)	(Notary Public Signature)

## **CLOSE-OUT FORM "D"**

## SUBCONTRACTOR WARRANTY

STAT	TE OF				
COU	NTY OF				
KNO	W ALL MEN BY THESE PRESENTS:				
		, being first duly sworn, deposes and says:			
1.	That he / she is the Subcontractor (or the subcontractor) who supplied, insta he / she is duly authorized to make th	alled, and / or erected the w	ork described below, and that,		
	Project: <u>Multi-Campus Building Envelo</u> Owner: <u>Galveston Independent Schoo</u> Architect: <u>BEAM</u> Work Performed: Specification Section(s):	ol District			
2.	The undersigned Contractor warrants to the Owner and Architect that materials and equipmen furnished under the Contract are of good quality and new except where otherwise required or permitted by the Contract Documents, that the Work is free from defects not inherent in the quality required or permitted, and that the Work conforms with the requirements of the Contract Documents. Work not conforming to these requirements, including substitutions not properly approved and authorized, may be considered defective. The Subcontractor's warranty excludes remedy for damage or defect caused by abuse, modifications not executed by the Subcontractor improper or insufficient maintenance, improper operation, or normal wear and tear under norma usage.				
3.	In the event of failure of materials, periods, the Subcontractor shall take of the defective items, whether notifier	appropriate measures to	assure correction or replacement		
4.	The Subcontractor warrants the work performed for a period of months from the date of Substantial Completion, except as follows:				
ATTE	EST (If Corporation)	Name of Subcontractor			
	Secretary	(By)	(Title)		
		JURAT			
STAT	TE OF				
COU	NTY OF				
Swor	n to and subscribed before me on this	day of	, 20		
	(Seal)	(Notary	Public Signature)		

## SECTION 01 78 39 - PROJECT RECORD DOCUMENTS

## PART 1 - GENERAL

## 1.1 PROJECT RECORD DOCUMENTS

- A. Maintain on site one set of the following record documents; record actual revisions to the Work:
  - 1. Drawings.
  - 2. Specifications.
  - 3. Addenda.
  - 4. Architects/Engineers written responses to Minor Change directives, Change Proposal Requests, and other supplemental instructions.
  - 5. Change Orders and other modifications to the Contract.
  - 6. Reviewed Shop Drawings, Product Data, and Samples.
  - 7. Manufacturer's instruction for assembly, installation, and adjusting.
- B. Ensure entries are complete and accurate, enabling future reference by Owner. Architect will review documents for general conformance but will not be responsible for completeness or accuracy of the recorded information.
- C. Do not use record documents for construction purposes. Store record documents separate from documents used for construction. Protect record documents from deterioration and loss in a secure, weather-tight location in accordance with Section 01 50 00, Temporary Facilities.
- D. Record information concurrent with construction progress, not less than weekly. Provide access to record documents for Architect's reference during normal working hours.
- E. Give particular attention to information on concealed products and installations that would be difficult to identify or measure and record later.
- F. Mark record sets in red erasable colored pencil; use other colors to distinguish between changes for different categories of the Work at the same location.
  - 1. Mark important additional information which was either shown schematically or omitted from original Documents.
  - 2. Note construction change directive numbers, alternate numbers, Change Order numbers and similar identification.
  - 3. Where feasible, the individual or entity who obtained record data, whether the individuals or entity is the installer, subcontractor, or similar entity, is required to prepare the mark-up on record documents.
    - a. Accurately record information in an understandable drawing technique.
    - b. Record data as soon as possible after it has been obtained. In the case of concealed installations, record and check the mark-up prior to concealment.
  - 4. Sign or initial and date each mark-up.
- G. Upon completion of the Work, submit Project Record Documents to Architect for the Owner's records in accordance with Section 01 77 00, Closeout Procedures.

## 1.2 RECORD SPECIFICATIONS

A. Record Specifications: Maintain one complete copy of the Project Manual including addenda and modifications issued. Legibly mark and record at each product section a

description of actual products installed and variations in actual Work performed in comparison with products specified. Include the following:

- 1. Manufacturer's name and product model and number.
- 2. Product substitutions or alternates utilized.
- 3. Changes made by addenda and modifications.
- 4. Related record drawing information and Product Data.
- 5. Other information necessary to provide a record of selections made and to document coordination with record Product Data submittals and maintenance manuals.

## 1.3 RECORD DRAWINGS

- A. Record Drawings: Maintain one complete blackline copy of the Contract Drawings and Shop Drawings. Mark the set to show the actual installation where the installation varies from the Work as originally shown.
  - 1. Legibly mark each item to record actual construction including, but not limited to the following:
    - a. Measured depths of foundations in relation to project finish floor datum.
    - b. Measured horizontal and vertical locations of underground utilities and appurtenances, referenced to permanent surface improvements.
    - c. Measured locations of internal utilities and appurtenances concealed in construction, referenced to visible and accessible features of the Work.
    - d. Field changes of dimension and detail.
    - e. Details not on original Contract drawings.
    - f. Revisions to details shown on the drawings.
    - g. Dimensional changes to the drawings.
    - h. Actual equipment locations.
    - i. Duct size and routing.
    - j. Changes made by Change Order; include change order number.
- B. Mark completely and accurately record prints of Contract Drawings or Shop Drawings, whichever is most capable of showing actual physical conditions. Where Shop Drawings are marked to reflect changes in the Work, record a cross-reference at the corresponding location on the Contract Drawings.
- C. Preparation of Transparencies: Prepare a full set of corrected reproducible Contract Drawings and Shop Drawings.
  - 1. Incorporate changes and additional information previously marked on print sets. Erase, redraw, and add details and notations where applicable. Identify and date each Drawing; include the printed designation "PROJECT RECORD DRAWINGS" in a prominent location on each Drawing.
  - 2. Remove Architects and Engineer's seal from drawings and specifications prior to issuance to Architect for approval.
  - 3. Refer instances of uncertainty to the Architect for resolution.

## 1.4 RECORD PRODUCT DATA

A. Maintain one copy of each Product data submittal for record document purposes. Mark Product Data to indicate the actual product installation. Include significant changes in the product delivered to the site, and changes in manufacture's instructions and recommendations for installation.

## 1.5 RECORD SAMPLE SUBMITTAL

A. Immediately prior to date of Substantial Completion, meet with the Architect, and Owner, at the Owner's discretion, at the site to determine which of the Samples maintained during the

construction period shall be transmitted to Owner for record purposes. Comply with the Architect's instructions for packaging, identification marking, and delivery to Owner's Sample storage space. Dispose of other Samples in manner specified for disposal of surplus and waster materials.

## 1.6 MISCELLANEOUS RECORD DOCUMENTS

- A. Refer to other Specification Sections for miscellaneous record-keeping requirements and submittals in connection with various construction activities. Immediately prior to Substantial Completion, complete miscellaneous records and place in good order, properly identified and bound or filed, ready for use and reference. Categories of requirements resulting in miscellaneous records, include, but are not limited to the following:
  - 1. Ambient and substrate condition tests.
  - 2. Changes requested by Owner's consultants.
  - 3. Inspections and certifications by governing authorities.
  - 4. Inspection and testing by Owner's inspection agency.
  - 5. Fire resistance and flame spread test results.

## 1.7 CERTIFICATION

A. By submittal of Project Record Documents, Contractor certifies, that to the best of his knowledge, informational and belief the documents are a true and complete representation of the actual construction of the Work of this Project.

## PART 2 - PRODUCTS

Not Used

## PART 3 - EXECUTION

Not Used

## END OF SECTION 01 78 39

## SECTION 02 50 00 - SELECTIVE DEMOLITION

## PART 1 - GENERAL

## 1.1 SECTION INCLUDES

- A. Partial demolition of existing building(s) as required to accommodate work including, but not limited to:
  - 1. New roofing, flashing and related work.
  - 2. Removal of existing mechanical, electrical, and plumbing items and associated utilities indicated or required, except for electrical work performed by Owner.

## 1.2 SUBMITTALS

- A. Submit the following items.
  - 1. Itemized Demolition Schedule.
  - 2. Detail all demolition methods to be used.

## 1.3 PERMITS

A. Procure and pay for all necessary permits or certificates required to complete the work specified. Make any and all required notifications and comply with all applicable Federal, State and local ordinances.

## 1.4 QUALITY ASSURANCE

- A. Provide at least one person who shall be present and in charge of the Demolition Work at all times and who shall be thoroughly familiar with all phases of all work performed under this section.
- B. Comply with all pertinent codes and regulations applying to this work. Where cutting or modifications are to be made to existing fire-rated construction, provide temporary closures of fire-resistive materials as required to maintain fire rating until such time as permanent fire-rated improvements are completed.

## 1.5 JOB CONDITIONS

- A. Use all means necessary to prevent the spread of dust during performance of this work. Provide additional clean filters for the existing air handling system serving those areas to remain to protect them from construction dust.
- B. Use all means necessary to protect the existing building to remain from all types of damage, including fire, water damage, and unnecessary interruption of utility services. In the event of damage of any kind, immediately make all repairs and replacements necessary to the approval of the Owner at no additional cost to the Owner.
- C. Motor driven equipment shall have functional mufflers.
- D. Visit the site and examine the existing structure. Note all conditions as to the character and extent of work involved.

## PART 2 - PRODUCTS

2.1. GENERAL

- A. Provide all barricades, shoring, and bracing necessary to protect the tenants, workmen, and Public from danger. Barricades shall be sufficiently designed to protect and or exclude the public from all hazards.
- B. All other materials not specifically described but required for proper completion of Work of this Section, shall be as selected by the Contractor subject to the approval of the Owner.
- C. Summary: Provide selective demolition as required.
  - 1. Demolish existing construction, finishes and building components designated to be removed.
  - 2. Protect portions of building adjacent to or affected by selective demolition, providing temporary shoring and other support as required to prevent unwanted collapse or movement or other damage.
  - 3. Remove and legally dispose of demolished materials off-site.
  - 4. Notify Owner at least 24 hours prior to shut-off of existing utilities. Cap off utilities, if any, that are discontinued in use.

## 2.2. DEMOLITION WORK

- A. Perform demolition work as required to complete the new construction.
- B. Perform demolition work in manner so as to allow Owner's use of existing facility.
- C. Perform demolition work in order to maintain Owner's construction schedule.

## 2.3. REMOVAL OF ROOF MATERIAL AND DECKING

- A. Roofing debris shall not be permitted to fall on adjoining roof deck in masses to exceed safe carrying capacity of decks. Existing roofs and decks shall be properly protected with plywood under area to be demolished.
- B. Structural or load-supporting members shall not be cut or removed adjacent to existing structures to remain until all loads carried by members have been removed or adequately supported.
- C. The Contractor shall take all precautions necessary to ensure the safety of the building occupants and workers.

## PART 3 - EXECUTION

## 3.1 DEMOLITION

- A. Before commencing the Work of this Section, verify with the Owner that all items to be removed by the Owner have been removed. Schedule the work in a careful manner with all necessary consideration for the Public and the Owner. The owner reserves the right to salvage any items scheduled to be removed. All items of existing equipment and materials or any other item of value shall be brought to the Owners attention prior to demolition.
- B. All material removed under this Contract, which is not to be salvaged or reused, shall become the property of the Contractor and be promptly removed from the site. At all times use movable debris boxes, covered, to convey the material through the building. Do not store or permit debris to accumulate on the site.

SELECTIVE DEMOLITION 02 50 00 - 2

- C. Conduct operations so as not to interfere with adjacent occupied spaces, roads, streets, drives, walks, service lines and the like.
- D. Disconnect electric, telephone, gas, water, steam, or other lines as required in accordance with rules and regulations of authorities having jurisdiction, as specified, or as directed by the Architect. Coordinate all disruptions in utility services with Owner. Verify that the utility services to the existing building to remain will continue operation. Relocate and reconnect existing utilities as needed to maintain operation of the existing facility.
- E. Remove all debris from the building premises and leave the construction site "Clean" each day. All debris shall be dumped in an approved disposal facility and all fees for this shall be paid by the Contractor.
- F. Dumpsters shall not overflow and shall be emptied on a regular basis.
- G. Contractor may retain any materials he desires if the Owner has not requested them to be salvaged. Contractor is responsible for completely removing all demolished materials from the site and disposing of them in accordance with all local, State and Federal Regulations.
- H. Keep all pedestrian areas clear for passage at all times.

## 3.2 MAINTAINING TRAFFIC

- A. Do not close or obstruct streets, sidewalks, parking lots, drives, trash truck passageways, without obtaining Owner's permission. Do not store materials in streets, drives, or outside of construction limits.
- B. Conduct operations with minimum interference with streets, driveways, sidewalks, and adjacent facilities.
- C. Provide, erect, maintain lights, barriers, fences as required to maintain strict security at construction site and prevent unauthorized access to area of construction site. Refer to Section 01501, Temporary Facilities.

## 3.3 UTILITY LINES

A. Until acceptance, maintain, preserve existing utilities traversing premises.

## 3.4 PROTECTION OF STRUCTURES, PROPERTY

- A. Execute demolition work to insure adjacent property against damage which might occur from falling debris or other causes.
- B. Take precautions to guard against movement, settlement, or be liable for such movement, settlement, or collapse; repair promptly such damage when so ordered.
- C. Repair damage to Owner's property or any other person or persons on or off premises by reason of required work.

## 3.5 DEBRIS

A. Remove, as it accumulates, debris, except as otherwise specified, resulting from demolition operations. Do not store or permit debris to accumulate on site. If Contractor fails to remove debris promptly, Owner reserves the right to have it be removed at Contractor's expense.

## END OF SECTION 02 50 00

## SECTION 03 47 13 - TILT-UP CONCRETE PANELS

## PART 1 - GENERAL

## 1.1 SECTION INCLUDES

- A. Tilt-up, site cast concrete wall panels, load and non-load bearing, erected from mold to final position.
- B. Supports, devices, load bearing supports, if any, and attachments.
- C. Perimeter and intermediate joint seals.
- D. Grouting under panels.

## 1.2 RELATED SECTIONS

- A. Section 01 45 23 Testing and Inspecting Services
- B. Section 03 30 00 Cast-in-Place Concrete: placement of anchorage or embedding of panel base bearing plates in concrete floor edge or top of foundation as shown or required.
- C. Division 5: Building structural frame and anchorage or welding to building structural components.
- D. Division 7: Roofing Sections
- E. Section 07 62 00 Sheet Metal Flashings, Gutters, Downspouts and Trim
- E. Section 07 84 00 Firestopping and Fire Safing
- F. Section 07 92 00 Joint Sealants: Perimeter joints with sealant and backing.
- G. Section 08 11 13 Hollow Metal Doors and Frames: Door and window frames in tilt-up units.
- H. Section 08 80 00 Glazed Systems: Windows and glass site installed in tilt-up units.
- I. Section 09 91 00 Painting and Staining

## 1.3 **REFERENCES**

- A. American Concrete Institute (ACI)
  - 1. C4, Tilt-Up Construction
  - 2. 301, Specifications for Structural Concrete for Buildings
  - 3. 304, Recommended Practice for Measuring, Mixing, Transporting and Placing Concrete
  - 4. 318, Building Code Requirements for Reinforced Concrete
- B. American Society for Testing and Materials (ASTM)
  - 1. A36, Standard Specification for Carbon Structural Steel
  - 2. A123, Standard Specification for Zinc (Hot-Dipped Galvanized) Coatings on Iron and Steel Products

- 3. A185, Standard Specification for Steel Welded Wire Fabric, Plain, for Concrete Reinforcement
- 4. A307, Standard Specification for Carbon Steel Bolts and Studs, 60 000 PSI Tensile Strength
- 5. A325, Standard Specification for Structural Bolts, Steel, Heat Treated, 120/105 ksi Minimum Tensile Strength
- 6. A416, Standard Specification for Steel Strand, Uncoated Seven-Wire for Prestressed Concrete
- 7. A615, Standard Specification for Deformed and Plain Billet-Steel Bars for Concrete Reinforcement
- 8. C31, Standard Practice for Making and Curing Concrete Test Specimens in the Field
- 9. C33, Standard Specification for Concrete Aggregates
- 10. C94, Standard Specification for Ready-Mixed Concrete
- 11. C143, Standard Test Method for Slump of Hydraulic Cement Concrete
- 12. C150, Standard Specification for Portland Cement
- 13. C260, Standard Specification for Air-Entraining Admixtures for Concrete
- 14. C330, Standard Specification for Lightweight Aggregates for Structural Concrete
- 15. C979, Standard Specification for Pigments for Integrally Colored Concrete
- C. American Welding Society (AWS)
  - 1. D1.1, Structural Welding Code, Steel

## 1.4 DESIGN REQUIREMENTS

- A. Design units to withstand erection forces and design loads as calculated in accordance with building code authorities having jurisdiction. Calculate structural properties of units in accordance with ACI 318.
- B. Design units to withstand actual loads such as wind, suction, deflection, and thermal movement loads.
- C. Design units to accommodate construction tolerances, deflection of building structural members, and clearances of intended openings.
- D. Design component connections to accommodate building movement and thermal movement. Provide adjustment to accommodate misalignment of structure without unit distortion or damage.
- E. The minimum clear distance between parallel bars shall be not less than 1-1/2 times the nominal maximum size of the aggregate, the nominal bar diameter, or one (1) inch.
- F. Tensile stresses in the concrete during handling, computed on the basis of an un-cracked section, shall be less than 75% of the modulus of rupture of the plain concrete at that time. Modulus of rupture of concrete at various ages shall be computed form the results of flexural strength tests performed in accordance with ASTM C78.

## 1.5 SUBMITTALS FOR REVIEW

- A. Shop Drawings: Indicate layout, tilt-up unit locations, configuration, unit identification marks, reinforcement, connection details, support items, location of lifting devices, dimensions, openings, and relationship to adjacent components.
- B. Samples: Submit two (2) panels, four (4) feet x four (4) feet in size illustrating surface finish, color and texture.

## 1.6 SUBMITTALS FOR INFORMATION

A. Section 01 45 23 – Inspecting and Testing Services: Submit proposed mix design before starting work.

## 1.7 QUALITY ASSURANCE

- A. Perform Work in accordance with ACI C4, ACI 318, and authorities having jurisdiction.
- B. Welding: AWS D1.1.
- C. Maintain one (1) copy of each document on site.
- D. Fabricator: Company specializing in performing the Work of this Section with minimum five (5) years experience.
- E. Design units under direct supervision of a Professional Structural Engineer experienced in design of this work and licensed in the State of Texas.
- F. Welder: Qualified within previous 12 months in accordance with AWS D1.1 and AWS D1.4.

## 1.8 MOCK-UP

- A. Fabricate and erect at site, where directed by Architect, one (1) full size of each panel, illustrating each shape, lifting device, and attachment points, and finish in accordance with approved sample.
- B. Include mockup panel with sealants, and cast-in items, if any.
- C. Approved mockup may remain as part of the Work. Remove unapproved mockup from site and replace until approved at no additional expense to Owner.

## 1.9 PRE-INSTALLATION MEETING

- A. Convene one (1) week prior to commencing Work of this Section.
- B. Include Construction Manager, Contractor, superintendents of fabricator, and erector, Owner's Representative, Architect, Structural Engineer, and all other interested parties regarding work built in, supporting or adjacent to tilt-up units.

## 1.10 DELIVERY, STORAGE, AND PROTECTION

- A. Handling Tilt-up Units: Lift units to position, consistent with their shape and design. Lift and support only from support points.
- B. Blocking and Lateral Support During Erection: Clean and non-staining, without causing harm to exposed surfaces.
- C. Provide temporary lateral support to prevent bowing, warping, or cracking.
- D. Protect units from staining, chipping, or spalling.

## PART 2 - PRODUCTS

2.1 MATERIALS

- A. Cement: ASTM C150, Type I Normal Type IA Air Entraining Portland type; grey color, unless shown or directed otherwise.
- B. Aggregate Materials: ASTM C33; sand, unless shown or directed otherwise.
- C. Reinforcing Steel: ASTM A615, deformed steel bars, unfinished, strength and size commensurate with tilt-up unit design and as shown on drawings.
- D. Air Entrainment Admixture and Other Admixtures: As specified in Section 03300, Cast-In-Place Concrete or as directed or shown on drawings.
- E. Pigment: If shown or required, shall be mineral oxide, lightfast, lime-proof, water-resistant type conforming to ASTM C979 in color(s) selected by Architect.
- F. Grout: Non-shrink, Non-metallic, minimum 10,000 psi, 28 day strength, unless shown otherwise.

## 2.2 FORMWORK AND FORM LINERS

- A. Contractor may use any of the formwork materials specified in Section 03300, Cast-In-Place Concrete, as long as materials are appropriate and of sufficient strength to support concrete work, materials will not stain or impart any undesirable texture, i.e. wood grain, where such texture would be objectionable in an exposed location, and the finished work will exhibit the finish or design characteristics, i.e. smooth, textured, ribbed, etc., detailed by the Architect for exposed locations.
- B. Form Liners:
  - 1. #311A Random Striated Rib
  - 2. #404 Medium Sand Finish
- C. Configuration: As shown on Drawings.

## 2.3 SUPPORT DEVICES

- A. Connecting and Support Devices: ASTM A36 weldable steel.
- B. Bolts, Nuts, and Washers: ASTM A307 or ASTM A325, high strength steel as recommended or required to suit application.
- C. Primer: Zinc rich rust inhibiting type.

## 2.4 ACCESSORIES

- A. Bearing Pads: Type and size recommended by manufacturer to suit application.
- B. Recessed Reglets: Galvanized steel, shaped and flanged to remain in place once cast, foam plastic filled to eliminate wet concrete intrusion.
- C. Sealant: Type recommended for use as specified in Section 07900, Building Sealants.

## 2.5 MIX

A. Concrete: 4,000 psi 28 day compressive strength, or as shown on drawings.

B. Slump for all concrete shall be uniform at 4 inches with maximum variation of 1 inch plus or minus, in accordance with ACI Code.

## PART 3 - EXECUTION

## 3.1 EXAMINATION

- A. Verify existing conditions prior to beginning work.
- B. Verify that building structure, anchors, devices, and openings are ready to receive Work of this Section.
- C. Bring discrepancies to the attention of the Architect and do not start work until discrepancies are corrected.
- D. Start of work will imply acceptance of existing conditions.

#### 3.2 **PREPARATION**

A. Provide for erection procedures and induced loads during erection. Maintain temporary bracing in place until final support is provided.

## 3.3 SITE FABRICATION

- A. Maintain environmental records and quality control program during production of tilt-up units. Make records available upon request.
- B. Use rigid molds, constructed to maintain tilt-up units uniform in shape, size and finish. Cast logo, rustication, and finish as indicated.
- C. If required, utilize form liners in accordance with manufacturer's instructions.
- D. Maintain consistent quality during manufacture.
- E. Fabricate connecting devices, plates, angles, items fit to steel framing members, inserts, bolts, and accessories. Fabricate to permit initial placement and final attachment.
- F. Embed reinforcing steel, anchors, inserts plates, angles, and other cast-in items as indicated.
- G. Place reinforcement accurately to dimensions shown. Bend bars cold and straighten in a manner which will not injure material.
- H. Space and position bars as shown. Unless otherwise noted, place, splice, and locate reinforcement in accordance with recommendations of the Concrete Reinforcing Steel Institute.
- I. Place recessed flashing reglets provided by Section 07620, Sheet Metal Flashings, Gutters, Downspouts and Trim continuous and straight.
- J. Locate hoisting devices to permit removal after erection.
- K. Cure units to develop concrete quality, and to minimize appearance blemishes such as non-uniformity, staining, or surface cracking.

L. Minor patching is acceptable, providing structural adequacy and appearance of units is not impaired.

## 3.4 FINISH - TILT-UP UNITS

- A. Ensure exposed-to-view finish surfaces of tilt-up units are uniform in color and appearance. Also free of blemishes and surface irregularities.
- B. Remove cement mortar from areas affected by aggregate retarder, if any, in accordance with retarder manufacturer's instructions.
- C. If shown or required, sand blast exterior exposed-to-view tilt-up unit surfaces to achieve aggregate surface exposure as selected by Architect. Protect adjacent surfaces.
- D. Interior finish of all tilt-up panels shall be floated, unless directed or shown otherwise.

## 3.5 FINISH - SUPPORT DEVICES

- A. Clean surfaces of rust, scale, grease, and foreign matter.
- B. Prime paint in one (1) coat, except surfaces in direct contact with concrete or requiring field welding.

## 3.6 SITE FABRICATION TOLERANCES

- A. Maximum Out of Square: 1/8 inch in 10 feet, non-cumulative.
- B. Variation from Dimensions Indicated on Shop Drawings: Plus or minus 1/8 inch.
- C. Maximum Misalignment of Anchors, Inserts, Openings: 1/8 inch.
- D. Maximum Bowing of Units: Length of bow/360.
- E. Location of Reglets: 1/4 inch from true position.

## 3.7 FIELD QUALITY CONTROL

- A. Refer to Section 03 30 00, Cast-in-place Concrete for concrete mix testing, concrete compressive tests, slump tests, and air entrainment tests to be tested in accordance with Section 01 45 23, Testing and Inspecting Services.
- B. Except as noted below, all inspection and testing related to concrete placement, including reinforcing and embedded items, shall be the responsibility of the Owner. The Owner will directly engage the services of a qualified Testing and Inspection Laboratory, however, the Contractor shall provide access to the Owner's consultant, and, if required, the Contractor shall provide patching and repairing of surfaces removed to facilitate testing and inspection.
- C. Should the strength of concrete fall below the minimum, then additional tests, including load tests, may be required. These tests, if required, shall be made at the Contractor's expense and shall be in accordance with ASTM C42 and ACI 318. If tests do not meet the applicable requirements, then the structure, or any part of the structure, shall be removed and replaced at the Contractor's expense.
- D. Any concrete testing requested by the Contractor for early formwork or shoring removal, etc., shall be at the Contractor's expense.

## 3.8 ERECTION

- A. Erect units without damage to shape or finish. Replace or repair damaged panels. Do not lift prior to 75 percent of 28 day strength.
- B. Erect members level and plumb within allowable tolerances. Shore and brace in place to insure stability throughout construction.
- C. Align and maintain uniform horizontal and vertical joints, as erection progresses.
- D. When members require adjustment beyond design or tolerance criteria, discontinue affected work; advise Architect/Engineer.
- E. Fasten and weld units in place as shown or required. Perform welding, including tack welds, in accordance with AWS D1.1.
- F. Touch-up field welds and scratched or damaged primed painted surfaces.
- G. Set units dry, without grout, attaining joint dimension with lead or plastic spacers. Grout pack base of unit.
- H. Exposed Joint Dimension: 1/2 inch, unless shown otherwise.
- I. Patch holes, cut-off anchors, surface defects, and damaged corners to match panel with epoxy/cement paste adhesive.
- J. Seal perimeter and intermediate joints in accordance with Section 07900, Building Sealants.

## 3.9 ERECTION TOLERANCES

- A. Maximum Variation from Plane of Location: 1/4 inch in 10 feet and 3/8 inch in 100 feet and non-cumulative.
- B. Maximum Offset from True Alignment between Two (2) Connecting Units: 1/4 inch.
- C. Joint Tolerance: Plus or minus 1/4 inch.

## 3.10 ADJUSTING

A. Adjust units so that joint dimensions are within tolerances.

## 3.11 PROTECTION OF FINISHED WORK

- A. Protect units from damage.
- B. Provide non-combustible shields during welding operations.

## END OF SECTION 03 47 13

## SECTION 05 50 00 - METAL FABRICATIONS

## PART 1 – GENERAL

#### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.

## 1.2 SECTION INCLUDES

- A. Miscellaneous metal items and their related components which are not necessarily individually described shall be furnished and installed in accordance with the intent of the drawings and specifications and as required to complete the work.
- B. The Work of this Section is governed by Section 05 12 00, Structural Steel, except where more stringent requirements are contained herein or on the Structural Drawings. If a conflict exists, notations on the Structural Drawings take precedence.

## 1.3 **REFERENCES**

- A. Conform to the following reference standards as applicable to the work:
  - 1. American Institute of Steel Construction (AISC), Code of Standard Practice for Steel Buildings and Bridges, latest edition.
  - 2. American Institute of Steel Construction (AISC), Specifications for the Design, Fabrication, and Erection of Structural Steel for Buildings, latest edition.
  - 3. American Iron and Steel Institute (AISI)
  - 4. ASTM International (ASTM)
    - a. A36, Standard Specification for Carbon Structural Steel
    - b. A108, Standard Specification for Steel Bar, Carbon and Alloy, Cold-Finished
    - c. A123, Standard Specification for Zinc (Hot-Dip Galvanized) Coatings on Iron and Steel Products
    - d. A126, Standard Specification for Gray Iron Castings for Valves, Flanges, and Pipe Fittings
    - e. A307, Standard Specification for Carbon Steel Bolts and Studs 60,000 psi Tensile Strength
    - f. B209, Standard Specification for Aluminum and Aluminum-Alloy Sheet and Plate
    - g. B210, Standard Specification for Aluminum and Aluminum-Alloy Drawn Seamless Tubes
    - h. B221, Standard Specification for Aluminum and Aluminum-Alloy Extruded Bars, Rods, Wire, Profiles, and Tubes
  - 5. American Welding Society (AWS)
    - a. D1.1 Structural Welding Code
  - 6. Steel Structures Painting Council (SSPC)
    - a. Painting Manual, Volume 1, Good Painting Practice
    - b. Painting Manual, Volume 2, Systems Specifications
  - 7. National Ornamental & Miscellaneous Metals Association (NOMMA)
    - a. Guideline 1 Joint Finishes.

## 1.4 SUBMITTALS

- A. Shop Drawings:
  - 1. For off-the-shelf items: Show all layouts, sizes, methods of construction and installation, including sizes and types of all fastening devices.

- 2. For custom fabricated items: Submit design calculations for the materials and their connections designed by the Contractor, prior to or with the shop drawings. Calculations shall bear the seal of a Registered Professional Engineer, licensed in the State of Texas. Shop drawings containing connections for which calculations have not been received will be returned unchecked as an incomplete submittal. Design Calculations will be retained for the Architect's file, and will not be approved or returned.
- B. Samples: As noted.

## 1.5 CONTRACTOR'S RESPONSIBILITIES

- A. As scope and performance documents, the Drawings and Specifications do not necessarily indicate or describe all the work required for the performance and completion of the Work. Contracts will be let on the basis of such documents with the understanding that the Contractor shall furnish and install the items required for proper completion of the Work without adjustment to price or schedule. Work shall be of sound, quality construction and the Contractor shall be solely responsible for the inclusions of adequate labor and materials to cover the proper and timely fabrication and installation of the miscellaneous metal items indicated, described, or implied.
- B. As a performance specification, the criteria for the solution of structurally sound miscellaneous metal items indicated on the Drawings or specified herein are the sole purpose of defining the design intent and performance requirements. The details shown are intended to emphasize the acceptable profiles and performance requirements for this Project. To avoid any misunderstanding or lack of interpretation, the Contractor is hereby advised that the responsibility for the miscellaneous metal items are totally his and that designs and resolutions proposed in the Contractor's shop drawings, structural calculations, and related documentation shall be demonstrated throughout the Work and warranty period specified or required.
- C. Design proposal submissions which follow exactly the details indicated on the Drawings, will not relieve the Contractor of his responsibility for the design, fabrication, erection, or performance of the Work of this Section.
- D. In the event of a controversy over the design, the decision of the Architect will take precedence.

## PART 2 - PRODUCTS

## 2.1 MATERIALS

- A. Structural Steel: Comply with ASTM A36.
- B. Welding:
  - 1. Comply with American Welding Society (AWS) Code.
  - 2. Comply with National Ornamental & Miscellaneous Metals Association (NOMMA Guideline 1: Joint Finishes) for joint finishes in the following locations:
    - a. Finish #1: Interior handrails and guardrails
    - b. Finish #2: Typical ornamental metals exposed to view, interior and exterior, except as indicated above. (canopies, exterior railings and guardrails, non-accessible sculptural fabricated metal, etc.)
    - c. Finish #3: Miscellaneous metals exposed interior non-public spaces, and exterior miscellaneous metals, except as indicated above. (ladders, bollards, etc.)

- d. Concealed welding: no requirement, unless otherwise indicated or required for safety or conformance with requirements of other portions of the work.
- C. Bolts:
  - 1. Comply with ASTM A307.
  - 2. Size: 3/4 inch, unless otherwise noted.
- D. Anchors:
  - 1. Expansion bolts:
    - a. 1/4 inch or less, Rawl Calk-Ins or Arrow Series 4000.
    - b. Greater than 1/4 inch: Rawl Multi-Calks. Top shall be 1/2 inch below concrete surface.
  - 2. Molly screw anchors:
    - a. In walls 1/16 inch to 5/8 inch thick, use "S" length
    - b. In walls 5/8 inch to 1/4 inch thick, use "L" length
    - c. In walls 1-1/4 inch to 1-3/4 inch thick, use "XL": length.
  - 3. Nelson stud anchors:
    - a. Comply with ASTM A108.
      - b. Concrete Embed: Headed, low carbon steel, non-threaded, galvanized, standard ferrule as required
- E. Shop Priming:
  - 1. Shop coat any ungalvanized ferrous metal with primer, except for those to receive application of spray-applied fireproofing shall be free of primer and paint.
  - 2. Clean iron and metal to be primed of scale, dirt and dust by steel scrapers, wire brushers or sandblasting. Remove oil and grease with petroleum naphtha.
  - 3. Thoroughly work paint into all joints by brush. Overall application of brush or spray coat of red oxide primer in accordance with SSPC Paint 25.
  - 4. Give any painted built-in portions one field coat of primer on all abraded parts after installation.
- F. Galvanized Metal:
  - 1. Comply with ASTM A123.
  - 2. General: Galvanize all steel sections which are fully or partially exposed to weather, regardless if they are scheduled to receive a finish coat of paint or not.
  - 3. Galvanized items to be painted shall be primed as outlined in Painting and Staining Section.
  - 4. Hot-dip galvanized after fabrication.
  - 5. Silicone protective coating shall not be used at galvanized items scheduled to receive paint.
  - Galvanizing Repair Paint: ZRC cold galvanizing compound or Galvilite manufactured by ZRC Worldwide, Marshfield, MA; Galvax Zinc-rich Cold Galvanizing Coating manufactured by Alvin Products, Inc., Lawerence, MA; or paint complying with military specification MILP-21035A, Type I or II. Apply repair paint in accordance with ASTM A780.
- G. Stainless Steel:
  - 1. General: Comply with ASTM Standards as applicable to the work.
  - 2. Type: Type 302 or 304 as applicable to the work.
  - 3. Finish:
    - a. Concealed: No. 2D finish
    - b. Exposed: No. 4, unless noted otherwise.
- H. Aluminum:
  - 1. General: Comply with ASTM Standards as applicable to work.
  - 2. Type: 6061 or 6063 as applicable to work.

- 3. Finish:
  - a. Concealed: Mill finish
  - b. Exposed: Mill finish, or Anodized or Kynar 500 or Hylar 5000 finish a specified in color selected by Architect from manufacturer's standard colors.

## 2.2 MISCELLANEOUS METAL ITEMS

- A. The following is a list of the principal miscellaneous metal items to be furnished under this Section. This list is offered only as a guide and Contractor shall thoroughly check drawings for other miscellaneous metals. All items exposed to the exterior shall be hot-dip galvanized after fabrication.
  - 1. Steel Ladders: Fabricate from 2-1/2 inch by 3/8 inch flat bar steel stringer with 3/4 inch steel rod rungs let into stringers, welded and ground smooth. Provide all angle supports and anchoring devices for bolting to wall, floor, or structure as required. Hot-dip galvanize after fabrication.
  - 2. Aluminum Ladders: Fabricate from 6061-T6 aluminum. Provide all angle supports and anchoring devices for bolting to wall, floor, or structure as required
- B. Downspout Boots: kinetic architectural products
  - 1. Construction to be 1/8" aluminum (6063-T52) powder coated by electric gun with DFT of 2.0-4.0mils with 5 year warranty.
- C. Other Miscellaneous Items: Miscellaneous metal items and their related components are not necessarily individually described. Miscellaneous items not described shall be furnished and installed in accordance with the intent of the drawings and specifications and as required to complete the work.

## PART 3 - EXECUTION

## 3.1 INSTALLATION

- A. Separate all dissimilar metals.
- B. Welded Joint Finishes: Where welding is exposed to view, welds shall be executed neatly then ground smooth. Pits and blemishes are not acceptable. Provide joints as stated above in accordance with NOMMA Guideline 1.
- C. For manufactured items, adhere to printed manufacturer's installation instructions.
- D. Refer to painting section for items that are to receive paint.

## END OF SECTION

## SECTION 06 10 00 - ROUGH CARPENTRY

## PART 1 - GENERAL

#### 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.

## 1.2 SECTION INCLUDES

- A. All rough carpentry items including, but not limited to:
  - 1. Wood blocking for support of items supported on or recessed into wood framing or requiring wood blocking for support, including but not limited to casework, LCD mounts, smart panel mounts, marker boards, metal shelving, toilet partitions, toilet accessories and window blinds, door strikes, wall stops, wall panels, and interior aluminum door frames.
  - 2. Wood cants, nailers, curbs, and other items associated with roofing work.
  - 3. Miscellaneous framing items and plywood sheathing.

## 1.3 RELATED WORK

A. All Sections of Work supported on or recessed into wood framing or requiring wood blocking for support, such as wall trim, wall cabinets, handrails, lockers, toilet compartments, toilet and bath accessories, markerboards, tackboards, smartboards N.I.C.(not in contract) in all classrooms, projection screens, fire extinguisher cabinets, etc., as applicable to the Project.

#### 1.4 SUBMITTALS

A. Product Data: Manufacturer's data on wood treatment materials.

## 1.5 STANDARDS AND GRADING

- A. All lumber used structurally shall be graded and marked with grade and trademark of a lumber grading organization approved by the Architect, except that a certification of grade from such a grading organization may be accepted in lieu of grade and trademarks when approved by Architect. Trademark of manufacturer shall also appear on each piece.
- B. Each piece of plywood used shall carry the American Plywood Association trademark.
- C. Grading Rules: Conform with all applicable requirements of American Lumber Standards "Simplified Practice Recommendations R-16" and to grading rules of manufacturer's association under whose rules the lumber is produced.
- D. Reference Standards: Conform with all requirements.
  - 1. U.S. Dept. of Commerce Product Standards (PS)
    - 2. American Plywood Association (APA)
      - a. Standards and Construction Guide
    - 3. American Wood Preservers Association (AWPA)
      - a. Standards, as they apply.
    - 4. Architectural Woodwork Institute (AWI)
      - a. "Quality Standards"
    - 5. National Woodwork Manufacturers' Association (NWMA) a. Standards
    - 6. Western Wood Products Association (WWPA)

## a. Manual

## PART 2 - PRODUCTS

## 2.1 MATERIALS

- A. Lumber:
  - Treated No. 2, S4S Southern Yellow Pine, #1 kiln dried.
    - a. Comply with NWMA Standards
    - b. Use for blocking, stripping, grounds, cants and miscellaneous wood items in contact with concrete, roofing, or exposed to the weather.
  - 2. No. 2, S4S Southern Yellow Pine: Use for framing, blocking, stripping and miscellaneous concealed interior lumber not exposed to concrete, roofing weather or moisture, when FRS lumber is not required by building code.
  - 3. Fire Retardant No, 2, S4S Southern Pine: Refer to Fire Retardant Treatment below. Use for framing, plates and blocking in all walls and partitions where required by building code or noted on drawings.
- B. Plywood:
  - 1. General: Comply with APA Standards.
  - 2. APA A-D, Group 1 Interior used where appearance of only one side is exposed to view for interior locations. Use for wall liner at MDF/IDF closets and telephone boards in mechanical and telephone rooms where shown or required. 3/4 inch thick unless required or shown otherwise. Paint as scheduled in Section 09 91 00.
  - 3. Exterior plywood, Group 1, APA rated sheathing. Use where miscellaneous plywood is exposed to concrete, weather, or at roof construction as sheathing.
  - 4. Fire Retardant Treated Plywood: Refer to Fire Retardant Treatment below. Use when required by building code or noted on drawings.
  - 5. Underlayment: If shown or required, APA rated Sturdi-floor, exterior grade, tongue and groove edges.
- C. Rough Hardware:
  - 1. Nails, Spikes, and Staples: Galvanized for exterior locations, high humidity locations, and treated wood; plain finish for other interior locations: Size and type to suit application. Do not use to resist "pull-out" loads.
  - 2. Bolts, Nuts, Washers, Lags, and Screws: Medium carbon steel; size and type to suit application. Galvanize for exterior locations, high humidity locations, and treated wood. Plain finish for other interior locations.
  - 3. Fasteners: Toggle bolt type for anchorage to hollow masonry. Expansion shield and lag bolt type for anchorage to solid masonry and concrete. Bolts or power activated type for anchorage to steel.
- D. Wood Treatment:
  - 1. Preservative Treatment (Concealed Conditions):
    - a. Micronized Copper Azole: Pressure impregnate preservative to net retention of 0.23 (PCF)., in plant licensed by manufacturer in accordance with the following standards:
      - 1) Preservative Treatment Standard: AWPA P5\
      - 2) Structural Lumber Treatment Standard: AWPA C31
      - 3) Plywood Treatment Standard: AWPA C9
    - b. Brush one (1) coat of preservative on bored or sawn surfaces of treated lumber.
    - c. Provide Quality Mark Stamp or end tag identifying third party inspection agency on treated wood for identification.

- d. Concealed conditions mean conditions that are interior, above ground that are not exposed to direct standing water, in contact with natural grade, or exposed to weather.
- e. ACQ and CCA preservatives not permitted.
- f. Acceptable Manufacturers: MicroPro "Lifewood"; Universal Forest Products " Prowood Borate"; or Architect approved equal.
- 2. Fire Retardant Treatment:
  - a. Lumber shall be pressure-impregnated with non-combustible fire retardant chemicals in accordance with U.L. FRS Fire Hazard Classification. All lumber must be dried following treatment in accordance with AWPA Standard C20.
  - b. Plywood shall be pressure-impregnated with non-combustible fire retardant chemicals in accordance with U.L. FRS Fire Hazard Classification. All plywood must be dried following treatment in accordance with AWPA Standards C27.
- E. Adhesives for gluing (furring and sleepers) to Concrete or Masonry: Formulation complying with ASTM D 3498 that is approved for use indicated by adhesive manufacturer.
  - 1. Adhesives shall have a VOC content of 70 g/L or less when calculated according to 40 CFR 59, Subpart D (EPA Method 24).

## PART 3 - EXECUTION

- A. Wood Framing:
  - 1. Framing and blocking shall be accurately cut and fitted true to line and levels, avoiding shims and wedges.
  - 2. Spiking and nailing shall be done using largest size spikes and nail practicable.
  - 3. Unless otherwise shown, use 2 inch by 4 inch wood studs spaced 16 inches o.c. with 4 inch face perpendicular to direction of wall or partition. Provide single bottom plate and double-top plates 2 inches thick by width of studs.
  - 4. Bolt nailers and blocking to steel, masonry or concrete members with bolts or proportionate strength of members attached from each end, except as otherwise noted on plans.
  - 5. Provide blocking, bucks and framing as necessary and for other trades as required.
  - 6. Drill lumber accurately for bolts and fit all bolts with suitable washers.
  - 7. Contractor to coordinate all locations where blocking is required prior to installation of gypsum board. Locations include but not limited to the following: wall stops as specified by Hardware Consultant, marker boards, tack boards, panels, etc.
- B. Plywood:
  - 1. Install plywood over framing in accordance with instruction of American Plywood Association Construction Guide Form No. E30C.
  - 2. Install underlayment plywood as shown in accordance with instructions of American Plywood Association. Space panel joints and edges 1/32 inch. Fill and sand panel edge joints, surface roughness, and damaged or open areas. Nail with 4d ring-shank nails spaced at six (6) inches at edges and eight (8) inches in field each way.

## END OF SECTION 06 10 00

## SECTION 07 19 00 - WATER REPELLENTS

## PART 1 - GENERAL

## 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and Division 01 Specification Sections, apply to this Section.

## 1.2 SUMMARY

- A. Section Includes: Requirements including but not limited to:
  - 1. Penetrating water repellent treatment for masonry and concrete surfaces.
  - 2. Film forming water repellent treatments for masonry and concrete surfaces.
  - 3. Accessories necessary for a complete installation.

## 1.3 **PERFORMANCE REQUIREMENTS**

- A. Performance: Water repellents shall meet the following performance requirements as determined by testing on standard substrates representing those indicated.
- B. Water Absorption: Minimum 90 percent reduction of water absorption after 24 hours for treated compared to untreated specimens when tested according to the following:
  - 1. Cast Stone: ASTM C 1195.
  - 2. Concrete Masonry Units: ASTM C 140.
  - 3. Clay Brick: ASTM C 67.
  - 4. Portland Cement Plaster (Stucco): ASTM D 6532.
- C. Water-Vapor Transmission: Comply with one or both of the following:
  - 1. Maximum 10 percent reduction water-vapor transmission of treated compared to untreated specimens, according to ASTM E 96/E 96M.
  - 2. Minimum 80 percent water-vapor transmission of treated compared to untreated specimens, according to ASTM D 1653.
- D. Water Penetration and Leakage through Masonry: Minimum 90 percent reduction in leakage rate of treated compared to untreated specimens, according to ASTM E 514/E 514M.
- E. Durability: Maximum 5 percent loss of water repellent performance after 2500 hours of weathering according to ASTM G 154 compared to water repellent-treated specimens before weathering.

## 1.4 SUBMITTALS

A. Product Data: Technical data including include performance data, DFT per coat, spreading rate, number of coats for each type of substrate, application procedures, and available colors.

## 1.5 QUALITY ASSURANCE

- A. Applicator Qualifications: An employer of workers trained and approved by manufacturer.
- B. Preinstallation Conference: Conduct conference at site.

## 1.6 **PROJECT CONDITIONS**

- A. Limitations: Proceed with application when existing and forecasted weather and substrate conditions permit water repellents to be applied according to manufacturers' written instructions and warranty requirements:
  - 1. Concrete surfaces and mortar have cured for not less than 28 days.
  - 2. Building has been closed in for not less than 30 days before treating wall assemblies.
  - 3. Ambient temperature is above 40 degrees F (4.4 degrees C) and below 100 degrees F (37.8 degrees C) and will remain so for 24 hours.
  - 4. Substrate is not frozen and substrate surface temperature is above 40 degrees F (4.4 degrees C) and below 100 degrees F (37.8 degrees C).
  - 5. Rain or snow is not predicted within 24 hours.
  - 6. Not less than 24 hours have passed since surfaces were last wet.
  - 7. Windy conditions do not exist that might cause water repellent to be blown onto vegetation or surfaces not intended to be treated.

## 1.7 WARRANTY

A. Warranty: Written warranty signed by manufacturer in which manufacturer and applicator agree to repair or replace materials that fail to maintain water repellency within specified warranty period.
 1. Warranty Period: Two years from date of Substantial Completion.

## PART 2 - PRODUCTS

## 2.1 PENETRATING WATER REPELLENTS

- A. Silane, Penetrating Water Repellent: Clear, containing **20** percent or more solids of alkyltrialkoxysilanes; with alcohol, mineral spirits, water, or other proprietary solvent carrier; and with 400 g/L or less of VOCs.
  - 1. Manufacturers: Subject to compliance with requirements, provide products by one of the following:
    - a. Advanced Chemical Technologies, Inc.
    - b. BASF Corporation; Construction Systems.
    - c. Dayton Superior.
    - d. Euclid Chemical Company (The); an RPM company.
    - e. PROSOCO, Inc.
    - f. Tnemec, Inc.
    - g. Vexcon Chemicals Inc.

## PART 3 - EXECUTION

## 3.1 EXAMINATION

- A. Examine substrates, areas, and conditions for compliance with requirements and conditions affecting performance of the Work.
  - 1. Verify that surfaces are clean and dry according to water-repellent manufacturer's requirements. Check moisture content in minimum of three representative locations by method recommended by manufacturer.
  - 2. Verify that there is no efflorescence or other removable residues that would be trapped beneath the application of water repellent.
  - 3. Verify that required repairs are complete, cured, and dry before applying water repellent.
- B. Test pH level according to water-repellent manufacturer's written instructions to ensure chemical bond to silica containing or siliceous minerals.

C. Proceed with installation after correcting unsatisfactory conditions.

## 3.2 PREPARATION

- A. New Construction and Repairs: Allow concrete and other cementitious materials to age before application of water repellent, according to repellent manufacturer's written instructions.
- B. Cleaning: Before application of water repellent, clean substrate of substances that could impair penetration or performance of product according to water repellent manufacturer's written instructions:
  - 1. Cast Stone and Concrete Unit Masonry: Remove oil, curing compounds, laitance, and other substances that inhibit penetration or performance of water repellents according to ASTM E 1857.
  - 2. Clay Brick Masonry: ASTM D 5703.
  - 3. Natural Stone: [ASTM C 1515.] [ASTM D 5107.]
  - 4. Portland Cement Plaster (Stucco): ASTM E 1857.
- C. Protect adjoining work, including mortar and sealant bond surfaces, from spillage or blow over of water repellent. Cover adjoining and nearby surfaces of aluminum and glass if there is the possibility of water repellent being deposited on surfaces. Cover live vegetation.
- D. Coordination with Mortar Joints: Do not apply water repellent until pointing mortar for joints adjacent to surfaces receiving water-repellent treatment has been installed and cured.
- E. Coordination with Sealant Joints: Do not apply water repellent until sealants for joints adjacent to surfaces receiving water repellent treatment have been installed and cured.
  - 1. Water repellent work may precede sealant application if sealant adhesion and compatibility have been tested and verified using substrate, water repellent, and sealant materials identical to those required.

## 3.3 APPLICATION

- A. Manufacturer's Field Service: Engage a factory authorized service representative to inspect the substrate before application of water repellent and to instruct Applicator on the product and application method to be used.
- B. Apply coating of water repellent on surfaces to be treated using 15 psi (103 kPa) pressure spray with a fan type spray nozzle, roller, or brush to the point of saturation. Apply coating in dual passes of uniform, overlapping strokes. Remove excess material; do not allow material to puddle beyond saturation. Comply with manufacturer's written instructions for application procedure unless otherwise indicated.
  - 1. Precast Concrete and Cast Stone: At Contractor's option, first application of water repellent may be completed before installing units. Mask mortar and sealant bond surfaces to prevent water repellent from migrating onto joint surfaces. Remove masking after repellent has cured.
- C. Apply second saturation coating, repeating first application. Comply with manufacturer's written instructions for limitations on drying time between coats and after rainstorm wetting of surfaces between coats. Consult manufacturer's technical representative if written instructions are not applicable to Project conditions.

## 3.4 FIELD QUALITY CONTROL

- A. Testing of Water Repellent Material: Owner reserves the right to invoke the following procedure at any time and as often as Owner deems necessary during the period when water repellent is being applied:
  - 1. Owner will engage the services of a qualified testing agency to sample water repellent material being used. Samples of material delivered to site will be taken, identified, sealed, and certified in presence of Contractor.
  - 2. Testing agency will perform tests for compliance of water-repellent material with product requirements.
  - 3. Owner may direct Contractor to stop applying water repellents if test results show material being used does not comply with product requirements. Contractor shall remove noncomplying material from Project site, pay for testing, and correct deficiency of surfaces treated with rejected materials, as approved by Architect.
- B. Coverage Test: In the presence of Architect, hose down a dry, repellent-treated surface to verify complete and uniform product application. A change in surface color will indicate incomplete application.
  - 1. Notify Architect **seven** days in advance of the dates and times when surfaces will be tested.
  - 2. Reapply water repellent until coverage test indicates complete coverage.

## 3.5 CLEANING

- A. Immediately clean water repellent from adjoining surfaces and surfaces soiled or damaged by water-repellent application as work progresses. Correct damage to work of other trades caused by water-repellent application, as approved by Architect.
- B. Comply with manufacturer's written cleaning instructions.

# END OF SECTION 07 19 00

## SECTION 07 56 00 - ACRYLIC FLUID APPLIED ELASTOMERIC COATING SYSTEM

#### PART 1 - GENERAL

#### 1.1 SECTION INCLUDES

- A. Provide a seamless, fluid applied acrylic membrane system designed for application over existing sheet membrane roof substrates.
- B. Provide manufacturer's ten (10) year limited system warranty.

## 1.2 RELATED WORK

- A. Section 07 62 00 Flashing and Sheet Metal
- B. Section 07 72 00 Roof Accessories

## 1.3 QUALITY ASSURANCE

- A. Contractor Qualifications:
  - 1. Contractor shall be approved by the coatings manufacturer, and shall have a minimum of three (3) years experience in the application of acrylic elastomeric roof coatings.
  - 2. Contractor shall provide a list of project references, including contact names and telephone numbers.
- B. Manufacturer Qualifications:
  - 1. Manufacturer shall have a proven 20 year track record utilizing elastomeric acrylic technology.
- C. Testing and Labeling:
  - System must be Underwriters' Laboratories, Inc. (UL) 790 classified as Class A for maintenance and repair of existing Class A, B, or C roofing construction. All containers must bear the UL label and be subject to the UL follow-up service. The coating shall be listed by Factory Mutual (FM) as an acceptable recoating system over existing roof substrates.
  - 2. Labels must include the Manufacturer's name, product name, type and class of material, UL sticker with classification number, FM logo, batch number, mixing and application instructions, and precautions.

#### 1.4 SUBMITTALS

- A. Product Data:
  - Manufacturer's specifications, literature, and other data needed to prove compliance with specified requirements, including, but not limited to, material physical properties (including American Society for Testing and Materials (ASTM) methods), estimated application rate for dry mil thickness in accordance with project specifications. Include data on coating, as well as primers, sealers, reinforcement, etc., and all other data and information to satisfy requirements of manufacturer on warranty needs.
  - 2. Color chart of Manufacturer's available colors for Architect's selection.
  - 3. Provide Manufacturer's application instructions and MSDS.
- B. Certifications: Provide copy of Approved Applicator's letter or certificate as issued by Coating Manufacturer.
- C. Warranty:

- Provide copy of Coating Manufacturer's warranty in accordance with project 1. requirements.
- 2. Provide letter from Manufacturer signed by agent authorized to do so stating acceptance of warranty as specified and detailed.

#### 1.5 PRODUCT DELIVERY, STORAGE AND HANDLING

- Α. Materials shall be delivered to jobsite in manufacturer's original, sealed Deliverv: containers.
- Β. Storage: Materials shall be stored in accordance with manufacturer's recommendations.
- C. Handling: Materials shall be handled and installed in accordance with manufacturer's instructions.
- D. Damaged Materials: Contaminated, damaged or unsealed materials, or materials not conforming to the specified requirements shall be removed immediately from the jobsite and replaced at Contractor's expense.

#### 1.6 **ENVIRONMENTAL CONDITIONS**

- Α. Install all materials in accordance with Manufacturer's published safety and weather precautions.
- Β. Do not apply elastomeric acrylic coating system components when the ambient and/or surface temperature is below 50 degrees F (10 degrees C) or above 110 degrees F (43 degrees C), if any surface moisture is present, when the dew point is within 5 degrees F (3 degrees C) of the surface temperature or when there is a possibility of temperatures falling below 32 degrees F (0 degrees C) within a 24 hour period. Do not spray apply if the wind velocity exceeds 10 MPH without taking appropriate precautions to eliminate overspray.
- C. Take all measures necessary to protect unrelated surfaces from coating overspray or spillage.

#### 1.7 FIELD QUALITY CONTROL

- The overall weather conditions, including surface temperature, surface moisture, ambient Α. temperature, relative humidity and wind velocity shall be recorded by the Contractor, at designated time intervals, on the Daily Quality Control Report Form if so requested by the Architect or Owner.
- Β. Verification of Protective Coating Thickness: The wet film thickness shall be measured and recorded daily, along with the quantity, batch numbers and total square feet applied, on the Daily Quality Control Report Form.

## **PART 2 - PRODUCTS**

#### 2.1 **APPROVED MANUFACTURERS**

- Specifications are based on components of "GacoFlex A4600" water-borne acrylic roof Α. coating.
  - GAF Top Coat Wayne NJ (973) 628-3000 1.
  - ER Systems, Inc., Loretto, MN; (800) 403-7747
  - 2. 3. IPC Pearland, Texas (800) 237-8759
  - 4. Uniflex Industrial Roof Coating: Cleveland, OH (888) 321-3539
  - SEALOFLEX Inc. Charleston, SC: (800) 770-6466 5.
  - Johns Manville Denver, CO (800) 654-3103 6.

- 7. Firestone Indianapolis, IN. (800) 428-4442
- B. Fluid applied waterproofing materials such as cementitious coatings, ceramic-filled coatings, asphaltic based materials, moisture-cured urethanes, Kraton based rubbers, Hypalons and butyls are not considered substitutes for materials specified herein.
- C. Ceramic Coated Roofing Granules.

## PART 3 - EXECUTION

## 3.1 SURFACE INSPECTION

- A. Apply only when air, material, and surface temperatures are between 50°F and 110°F. Mix until A46 is homogenous before application. Apply product in the morning to allow for maximum cure time during the daylight hours. If roof temperature is over 100 °F, a light mist of water may be used to increase working time. surfaces shall be clean, dry, structurally sound, stable and well secured.
- B. The roof system shall be free of excessive ponding water. Roof surfaces which pond water 48 hours after a rain shall be considered unacceptable. All water shall be allowed positive drainage from the roof.
- C. Inspect condition of flashing details adjacent to protrusions, penetrations, roof mounted equipment, curbs, walls, parapets, drains and roof edge to ensure that details are acceptable and will maintain a weathertight installation after being properly reinforced and coated.
- D. Determine moisture content of existing substrate, insulation and deck. A moisture content of ten (10) percent or greater indicates a potential problem. Work shall not proceed until the cause of high moisture content is verified and condition is corrected.

## 3.2 SURFACE PREPARATION

- A. All surfaces shall be clean and dry, and free of dirt, dust, gravel, oil, surface chemicals or other contaminates which may interfere with optimum adhesion.
- B. Unsound areas in the roof deck or insulation, including blisters, delamination deterioration, excessive moisture content, etc., shall be repaired or replaced at Contractor's expense. After removing gravel from ballasted existing sheet membrane single-ply systems, spot ballast with sand bags to avoid excessive fluttering of the membrane.
- C. All existing membrane surfaces, whether new or existing, shall be cleaned using cleaning concentrate. Dilute cleaning concentrate at the rate of one (1) part concentrate to ten (10) parts water. Apply the dilute mixture under low pressure spray at the rate of 200 square feet per gallon. After allowing to sit for 15 to 20 minutes, rinse thoroughly with fresh water under pressure (minimum 2,000 psi) to remove the solution from the roof. Heavy deposits of dirt or contamination may require agitation with a stiff-bristle broom or similar mechanical scrubber. If existing sheet membrane exhibits significant deterioration, utilize a low pressure, 500 psi rinse with extra attention given to "birdbaths" and other low areas. Significant deterioration is defined as large areas of exposed reinforcement scrim and/or severe chalking or alligatoring of the membrane. Allow the roof to dry thoroughly.

- D. Tighten or re-secure all terminations and caulk termination bars and reglet counterflashings with caulk.
- E. On all mechanically fastened as well as fully adhered existing sheet membrane single-ply systems, remove and reinstall all fasteners which are backed out or "tented", no more than six (6) inches from its original location. Use FMRC-approved stress plates and fasteners when replacing defective or worn fasteners. Apply fabric adhesive and primer liberally to the affected seam and surrounding area with a brush or roller. While the fabric adhesive and primer is still wet, embed a strip of six (6) inch reinforcing tape, centered over the seam. Work mesh into the fabric adhesive and primer, applying additional material as necessary to totally encapsulate the reinforcing tape.
- F. Repair all loose, torn or open seams in the existing sheet membrane using fabric adhesive and primer and reinforcing tape (mesh). Apply fabric adhesive and primer liberally to the affected seam and surrounding area with a brush or roller. While the fabric adhesive and primer is still wet, embed a strip of six (6) inch reinforcing tape, centered over the seam. Work mesh into the fabric adhesive and primer, applying additional material as necessary to totally encapsulate the reinforcing tape.
- G. Repair all tears, breaks, holes (from fastener relocation or protruding fasteners), cut out wrinkles or other openings in the existing sheet membrane by applying using fabric adhesive and primer and reinforcing tape in a similar manner as described above.
- H. Reinforce detail areas with either reinforcing tape and fabric adhesive and primer, or with single package polyurethane fluid applied reinforcing mastic, or single package acrylic fluid applied reinforcing mastic. If utilizing specified reinforcing tape, encapsulate the fabric with fabric adhesive and primer as described above around the base of all vents, stacks, fans and other protrusions, around all drains and scuppers, and around the base of all HVAC units and other roof-mounted equipment. If utilizing single package polyurethane fluid applied reinforcing mastic or single package acrylic fluid applied reinforcing mastic, use a brush to apply 60 to 80 dry mils of coating liberally around these detail areas. If using single package acrylic fluid applied reinforcing mastic, apply the material in a minimum of two (2) separate applications, allowing ample dry time between coats.

## 3.3 COATING APPLICATION

- A. Install per the manufacturer's instructions.
- B. Per the manufacturer's instructions, Contractor to apply ceramic coated roofing granules to prevent consumption from birds.

# 3.4 CLEANUP

- A. Maintain work and work areas in a clean, safe condition at all times during coating installation. Remove excess materials, trash and debris from the jobsite daily.
- B. At the completion of the project, clean area of all spills and containers and clean up all roofing debris leaving jobsite in a clean and orderly condition.

## 3.5 WARRANTY

A. Upon completion of the roof coating system, the Coating Manufacturer's Representative, Owner's Representative, Architect, and Applicator shall make a final inspection to determine the dry film thickness of the fluid applied acrylic membrane and to verify that the system meets the Manufacturer's requirements for warranty. The Contractor shall notify all interested parties in advance of said inspection.

- B. As a condition of the project's completion and acceptance, deliver to the Owner a copy of the fully executed Warranty from the Coating Manufacturer, in accordance with project specifications.
- C. Provide a ten (10) year system "no dollar limit/no penal sum" labor and material warranty, with full replacement cost. No exclusion for ponding water and will guarantee coating system will not leak for warranty period.

# END OF SECTION

# SECTION 07 63 00 - ROOF RELATED SHEET METAL

## PART I - GENERAL

## 1.1 SECTION INCLUDES

- A. It is the intent of this Section that the Work shall:
  - 1. conform to all applicable building code requirements and of authorities having jurisdiction;
  - 2. include all shop and field formed sheet metal work shown on drawings, specified or required, including, but not limited to:
    - a. Roof penetration sleeves and hood and umbrella counterflashing
    - b. Metal counterflashing
    - c. Expansion joint
    - d. Roof drains
    - e. Scuppers
    - f. Metal perimeter edge
    - g. Gutters, Downspouts, Splash Blocks and Splash Pans
    - h. One-way roof moisture relief vents
    - i. Metal gravity vents
    - j. Metal heat exhaust vents
    - k. Sanitary vent pipes
    - I. Pipe box
    - m. Copings, trim and miscellaneous sheet metal accessories.
  - 3. be part of the Work of Section 07 54 19, Single Ply Membrane; and
  - 4. be performed by a roof membrane contractor.

## 1.2 RELATED WORK

- A. Section 07 54 19 Mechanically Attached Thermoplastic Single Ply Membrane Roofing System
- B. Section 07 72 00 Roof Accessories
- C. All Sections of Work relating to or affecting the roofing system, including mechanical, plumbing and electrical items.

## 1.3 REFERENCES

- A. American Society for Testing and Materials (ASTM)
  - 1. A525, Specification for General Requirements for Steel Sheet, Zinc-Coated (Galvanized) by the Hot-Dip Process
  - 2. A526, Specification for Steel Sheet, Zinc-Coated (Galvanized) by the Hot-Dip Process, Commercial Quality
  - 3. A527, Specification for Steel Sheet, Zinc-Coated (Galvanized) by the Hot-Dip Process, Lock-Forming Quality
  - 4. A167, Specification for Stainless and Heat-Resisting Chromium-Nickel Steel Plate, Sheet, and Strip
  - 5. B32, Specification for Solder Metal
  - 6. C1107, Standard Specification for Packaged Dry, Hydraulic-Cement Grout (Nonshrink)
- B. ASCE 7
- C. Federal Specifications (FS)
  - 1. QQ-L-201

- D. FM Global
  - 1. FM Global Property Loss Prevention Data Sheets: 1-49, Perimeter Flashing
- E. National Association of Architectural Metal Manufacturers (NAAMM)
- F. National Roofing Contractors Association (NRCA)
  - 1. Roofing and Waterproofing Manual
- G. Sheet Metal and Air Conditioning Contractors National Association, Inc. (SMACNA)
  1. Architectural Sheet Metal Manual
- H. ANSI / SPRI ES-1: Fabricate and install sheet metal edge flashings and copings to comply with requirements of ANSI / SPRI ES-1 for 160 MPH wind speed zone and wind resistance loads.
- I. Texas State Board of Insurance

## 1.4 SUBMITTALS

- A. Product Data:
  - 1. Manufacturer's specifications and other data needed to prove compliance with specified requirements.
  - 2. Manufacturer's installation instructions.
- B. Shop Drawings: Indicating sizes, configurations, details of attachment to related and adjacent work, materials, and finishes.
- C. Samples:
  - 1. Full range of finish colors for Architect's selection.
  - 2. 12 inch long sample of each specified item with approved finish.
  - 3. Provide full size mockup of all shop built assemblies.

## 1.5 QUALITY ASSURANCE

- A. Single Source Responsibility: Fabricator and installer of roof-related flashing and accessories shall be the same as the membrane roof installer.
- B. Comply with governing codes and regulations of authorities having jurisdiction.
- C. ANSI / SPRI ES-1: Fabricate and install sheet metal edge flashings and copings to comply with requirements of ANSI/SPRI ES-1 for 90 MPH wind speed zone and wind resistance loads.

## 1.6 INSTALLATION CONFERENCE

A. Refer to Section 01 31 13, Project Coordination.

## 1.7 DELIVERY, STORAGE AND HANDLING

- A. Deliver, handle and store materials in accordance with manufacturer's instructions.
- B. Handle and store materials and equipment in such a manner as to avoid damage.
- C. No storage of materials shall be permitted on roof areas other than those materials that are to be installed the same day. Any exception must be in written form. Do not place materials or equipment in such a manner as to overload structure.

BEAM Professionals Project No. 18267R

# 1.8 WARRANTIES

- A. Manufacturer's Product Warranty:
  - 1. Manufacturer's standard 20 year Kynar 500 or Hylar 5000 Finish warranty signed by the manufacturer, guaranteeing covering failure of the fluoropolymer finish during the warranty period.
  - 2. Failure is defined to include, but not be limited to:
    - a. Deterioration of finish, such as fading, discoloring, peeling, cracking, corroding, etc.
    - b. Leaking water within building or construction.
  - 3. Correction may include repair or replacement of failed product.
- B. Roofing Contractor's Warranty:
  - 1. Contractor shall warrant the sheet metal work and related work to be free from defects in workmanship and materials, and that the metal flashings will be and remain watertight, for a period of five (5) years from date of Substantial Completion.
  - 2. Defects shall include, but not be limited to:
    - a. Leaking water or bitumen within building or construction.
    - b. Becoming loose from substrate.
    - c. Loose or missing parts.
    - d. Finish failure as defined above.

## PART 2 - PRODUCTS

## 2.1 APPROVED MANUFACTURERS

- A. Manufacturers named within specification are approved for use on the Project providing:
  - 1. their products meet or exceed the specifications;
  - 2. company has a minimum of five (5) years experience manufacturing products of the type specified;
  - 3. products have been tested in conjunction with roofing membrane system as an assembly and as such has obtained the same approval and rating as the roofing membrane system; and
  - 4. products are approved for use by the roofing membrane manufacturer.
- B. Substitutions shall be in accordance with Division 1 requirements regarding substitutions.

## 2.2 SHEET METAL MATERIALS

- A. General Requirements: Roofing sheet metal system shall have been tested in conjunction with roofing membrane system as an assembly and have the same approval and rating as the roofing membrane system.
- B. Prefinished Aluminum Sheet:
  - 1. Precoated type, aluminum conforming to Fed. Spec. QQ-A-250, ASTM B209.
  - 2. Finish: Kynar 500, color as selected by Architect from manufacturer's standard colors.
  - 3. Thickness: Minimum 0.040 inch, except as otherwise indicated.
- C. Membrane Clad Galvalume Sheet Steel:
  - 1. Commercial quality with 0.20 percent copper, conforming to ASTM A526, except ASTM A527 for lock forming, with G-90 hot-dip galvanized coating designation.
  - 2. Thickness: Except as otherwise indicated, minimum 24 gauge. SMACNA recommendations shall govern.
  - 3. Finish: PVC coating as per Membrane Manufacturer's requirements.

- D. Sheet Lead:
  - 1. Comply with FS QQ-L-201, Grade B
    - a. Four (4) pound minimum for use at roof drains and soil stacks.
- E. Stainless Steel: ASTM A167, Type 302/304 Soft Temper, No. 2D finish. Minimum thickness 24 gauge, except as otherwise noted.

## 2.3 FASTENERS

- A. Same metal as flashing/sheet metal or other non-corrosive metal or as noted below.
- B. Exposed fasteners shall be self-sealing and gasketed for weathertight installation. (ZAC type)
- C. Match finish of exposed heads with material being fastened.
- D. Mechanical Fasteners:
  - 1. Nails: Stainless Steel Ring shank, minimum 1-1/2 inches in length with 1/2 inch diameter head.
  - 2. Washers: Steel washers with bonded rubber sealing gasket.
  - 3. Screws: Self-tapping sheet metal type of stainless steel or compatible with material being fastened, with integral EPDM washers.
  - 4. Rivets: Stainless steel and cadmium plated material, closed end type of sizes recommended by sheet metal manufacturer to suit application.
- E. Clips:
  - 1. Continuous Cleat (fascia): Minimum .050 aluminum. Match material of fascia and provide one (1) gauge heavier.

# 2.4 RELATED MATERIALS

- A. Solder: ASTM B32, alloy grade 58, 50 percent tin, 50 percent lead.
- B. Flux:
  - 1. Phosphoric acid type, manufacturer's standard.
    - a. For Use with Steel or Copper: Rosin flux
    - b. For Use with Stainless Steel: Acid-chloride type flux, except use rosin flux over tinned surfaces.
- C. Waterproofing Membrane: 40 mil thick SBS modified bituminous product of self-adhering type with non-stick surface conforming to "Carlisle WIP 300HT" manufactured by Carlisle, or Architect approved equal.
- D. Adhesives: Type recommended by flashing sheet manufacturer seaming and adhesive application of flashing sheet to ensure adhesion and watertightness.
- E. Metal Accessories: Sheet metal clips, straps, anchoring devices, clamps and similar accessories required for the complete installation of work, matching or compatible with material being installed, non-corrosive, size and gauge recommended by installer to suit application and performance.
- F. Sealant:
  - 1. Type A:
    - a. Type: One-part, non-sag, moisture-curing polyurethane sealant.
    - b. Approved Products / Manufacturers: "Chem-Calk 900" manufactured by Bostik Construction Products Division, "Vulkem 921" manufactured by Mameco International, Inc., "Dynatrol I" manufactured by Pecora

ROOF RELATED SHEET METAL 07 63 00 - 4 Corporation, "NP 1" manufactured by Sonneborn Building Products, or approved equal.

- 2. Type B:
  - a. Type: One-part, neutral-curing, medium-modulus silicone sealant for sealing metal to metal surfaces, i.e. metal edge, cover plates, etc.
  - b. Approved Products / Manufacturers: "Chem-Calk 1200" manufactured by Bostik Construction Products Division, "795 Silicone Building Sealant" manufactured by Dow Corning Corporation, "895 Silicone" manufactured by Pecora Corporation, "Omniseal" manufactured by Sonneborn Building Products, "Spectrem 2" manufactured by Tremco Incorporated, or approved equal.
- G. Grout Pitch Pans:
  - 1. Type: Quick-setting, non-shrink, non-metallic, high strength formula complying with ASTM C1107.
  - 2. Approved Products/Manufacturers: "Sure Grip High Performance Grout" manufactured by Dayton Superior Corporation, "Premier Quick-Trim" manufactured by L & M Construction Chemicals, Inc., "Masterflow" manufactured by Master Builders, Inc., "Sonnogrout 10K" manufactured by Sonneborn Building Products, or approved equal.
- H. Pitch Pan Filler:
  - 1. Type: Pourable polyurethane sealer, approved by roofing system manufacturer.
  - 2. Approved Products/Manufacturers: "SPM Pourable Sealer" manufactured by Johns Manville, or approved equal.
- I. Termination Bar:
  - 1. Material: Stainless steel or extruded aluminum bar with lipped profile
  - 2. Size: 0.090 inch thick by 3/4 inch wide with 3/16 inch lip width and a 45 degree lip angle, factory punched 1/4 inch x 3/8 inch oval holes spaced six (6) inches on center.
  - 3. Approved Product/Manufacturer: "LIPTB 06" manufactured by Olympic Manufacturing Group, Inc., or approved equal.
- J. Pipe Hangers and Supports: Refer to Section 07 72 00, Roof Accessories.
- K. Retro-fit Drains: Refer to Section 07 72 00, Roof Accessories.

## 2.5 FABRICATION

- A. Except as otherwise indicated, fabricate work in accordance with SMACNA Architectural Sheet Metal Manual and other recognized industry practices and reviewed shop drawings. Form all flashings, receivers and counterflashings in accordance with standards set forth in the NRCA roofing manual and SMACNA.
- B. Comply with manufacturer's installation instructions and recommendations.
- C. Unless noted otherwise, fabricate perimeter edge/fascia, scuppers, gutters, downspouts, copings, and trim from pre-finished aluminum sheet.
- D. Shop fabricate work to greatest extent possible. Fabricate inside and outside corners for metal edges, counterflashing, and coping caps.
- E. Fabricate items to size and dimensions as indicated on the drawings. Limit single-piece lengths to ten (10) feet.

- F. Fabricate for waterproof and weather-resistant performance; with expansion provisions for running work sufficient to permanently prevent leakage, damage or deterioration of the work.
- G. Integrate flashing in a manner consistent with detailing. Form work to fit substrates.
- H. Make angle bends and folds for interlocking metal with full regard for expansion and contraction to avoid buckling or fullness in metal after installation.
- I. Fabricate items with straight lines, sharp angles, smooth curves, and true levels. Avoid tool marks, buckling, and oil canning.
- J. Fold back edges on concealed side of exposed edge to form hem.
- K. Unless noted otherwise, lap joints minimum three (3) inch. Lap joints to have sealant installed as per details, to maintain watertight condition, inside and outside corners and elevation changes to be riveted and soldered.
- L. Seams:
  - 1. Wherever possible, fabricate non-moving seams in sheet metal with flat-lock seams and end joints.
  - 2. Pre-finished Steel Metal: Seal pre-finished metal seams with rivets and silicone sealant.
  - 3. Metal Other than Aluminum: Tin edges to be seamed, form seams, and solder.
- M. On Kynar 500 or Hylar 5000 pre-finished metal, surface sand metal flanges prior to applying any primers. Prime all metal in contact with bituminous material.
- N. Backpaint all concealed metal surfaces with bituminous paint where expected to be in contact with cementitious materials or dissimilar metals.
- O. Expansion Provisions: Where lapped or bayonet type expansion provisions in work cannot be used or would not be sufficiently waterproof or weatherproof, form expansion joints of intermeshing hooked flanges, not less than one (1) inch deep filled with mastic sealant concealed within joints.

# 2.6 FABRICATED ITEMS

- A. Metal Flashings:
  - 1. Through wall Receiver Tray: Minimum 24 gauge stainless steel, through wall receivers shall not extend past the face of the exterior veneer more than  $\frac{3}{4}$ ".
  - 2. Counterflashing: Minimum .040 gauge aluminum steel where visible from ground. Unless otherwise shown on drawings, use 24 gauge stainless steel.
- Wind Clips: Minimum 0.040 gauge aluminum to match material of counterflashing, one (1) inch wide by length to engage counterflashing a minimum of 1/2 inch. To be installed at all wall flashings and at curb flashing lengths longer than 5 feet.
- C. Roof Penetrations:
  - 1. Umbrella Counterflashing: Two-piece construction of minimum, .040 aluminum metal fabricated in accordance with drawings or project requirements.
  - 2. Pitch Pans:
    - a. 24 ga. membrane clad galvanized metal.
    - b. Fabricate to provide installed minimum clear inside perimeter dimension of two (2) inches on each side of penetrating element.

- c. Fabricate pans to at least six (6) inches above the finished roof membrane and with 1/4 inch hem at top edge and with four (4) inch flanges. Round all corners of flange.
- d. Fabricate metal bonnets for all pans, NO EXCEPTIONS. Fabricate bonnets with metal compatible with metal to which bonnet is to be attached. On beams and other steel, weld in place bonnets fabricated from 1/4 inch steel plate. Draw band bonnets fabricated from 24 gauge stainless steel may be used on circular projections.
- D. Metal Edge:
  - Minimum 24 gauge membrane clad galvanized metal with 0.040 gauge aluminum pre-finished metal cover fascia formed in maximum ten (10) foot lengths, with six (6) inch wide cover plates of same profile, four (4) inch flange, maximum seven (7) inch fascia, including a 3/4 inch gravel stop.
  - 2. Provide expansion slip joints at maximum 20 feet on center.
  - 3. Shop fabricate all interior and exterior corners. Fabricate exterior corners with 18 inch minimum to four (4) foot maximum legs. Lap, rivet, and seal prior to delivery to jobsite.
  - 4. Fabricate to sizes and dimensions as indicated on drawings with a minimum one (1) inch coverage past top of wall. Refer to SMACNA Fig. 2-5A.
  - 5. Provide mock-up for Architect's approval prior to fabrication.
- E. Vent Hoods, Sleeves, Penetration Flashings, and Accessories: Minimum 24 gauge stainless steel, or as shown or directed otherwise.
- F. Angle Termination Bar: One (1) inch x one (1) inch 24 gauge galvanized steel.
- G. Vent Pipe Flashing: Four (4) pound lead. Provide proper size to fold down inside of pipe a minimum of one (1) inch.
- H. Pipe Box Cover: 0.040 gauge pre-finished aluminum metal
- I. Heat Exhaust Curbs and Hoods: 22 gauge stainless steel.

# PART 3 - EXECUTION

## 3.1 EXAMINATION

- A. Verify substrates are smooth and clean to extent required to perform sheet metal work.
- B. Verify roof openings, curbs, pipes, sleeves, ducts or vents through roof are solidly set in place.
- C. Verify that reglets, nailers, cants, and blocking to receive sheet metal are in place and free of concrete and soil.
- D. Do not start work until conditions are satisfactory.

## 3.2 **PREPARATION**

- A. Field measure site conditions prior to fabrication work.
- B. Install starter and edge strips and cleats before starting installation.

# 3.3 INSTALLATION

- A. Install sheet metal with lines, arises, and angles sharp and true, and plane surfaces free from objectionable wave, warp, or buckle. Exposed edges of sheet metal shall be folded back to form 1/4 inch hem on concealed side from view. Finished work shall be free from water retention and leakage under all weather conditions. Pre-fabricated corners or transitions are required at changes in direction, elevation, or plane and at intersections. Locate field joints not less than 12 inches, nor more than three (3) feet from actual corner. Laps shall be one (1) inch, riveted and soldered at following locations:
  - 1. Pre-fabricated corners;
  - 2. transitions;
  - 3. changes in direction, elevation, and plane; and
  - 4. at intersections.
- B. Anchor units of work securely in place to prevent damage or distortion from wind or buckling. Provide for thermal expansion of metal units; conceal fasteners wherever possible; and set units true to line and level as indicated. Install work with laps, joints, and seams which are permanently watertight and weatherproof.
- C. Install fabricated sheet metal items in accordance with manufacturer's installation instructions and recommendations and with SMACNA Architectural Sheet Metal Manual.
- D. Separations: Provide for separation of metal from dissimilar metal or corrosive substrates by coating concealed surfaces with zinc chromate, bituminous coating, or other permanent separation at locations of contact as recommended by manufacturer or fabricator. Do not use materials which are incompatible with roofing system.
- E. Continuous Cleat: At exposed edges of perimeter edge, fascias, cap flashings, and where required, attach continuous cleat at six (6) inches on center with appropriate fasteners. At a distance of 10 feet from each direction of all corners, install fasteners spaced at 3 inches on center. Install cleat so fascia extends a minimum of 1 inch below top of exterior wall finish.
- F. Gravel Guard / Fascia:
  - 1. Install with expansion joints 10 feet o.c., 1/2 inch expansion leeway, with cover plate.
  - 2. Fasten into nailer at 3 inches o.c. staggered.
  - 3. Install fascia cover over raised membrane cladded cleat and crimp.
- G. Counterflashing:
  - 1. Do not use surface mount counterflashing.
  - 2. Set in through wall with receiver and spring lock counterflashing, as detailed in drawings and to NRCA roofing manual, SMACNA standards.
  - 3. Coordinate installation of through-wall flashing with the masonry contractor.
  - 4. Seal through-wall in conjunction with masonry wall waterproofing.
  - 5. Install wind clips 30 inches o.c. at all counterflashing over five (5) feet in length.
- H. Pitch Pans:
  - 1. Apply sealant under pitch pan flange at least 1/2 pound per linear foot.
  - 2. Clean all projections enclosed in pitch pans in any manner suitable and coated with a rust inhibitive coating as approved by the Architect. Coating shall be allowed to dry prior to pitch pan fill.
  - 3. Fill base of pitch pans with grout or cementitious binder and allow to cure.
  - 4. Top Finish Fill: Self-leveling, one-part urethane; at least two (2) inches to top of pitch pan sides.
  - 4. Strip in flange with specified stripping plies with PVC Membrane extending 6 inches from the outer edge of the flange and butt base of pitch pan.
- I. Pipe Box:

- 1. Construct using membrane clad 24 ga galvanized metal.
- 2. Height shall be six (6) inches minimum above roof surface.
- 3. Install with flanges fastened to deck.
- 4. Fill pans with grout to a height of 3/4 inch of the total pan height.
- 5. Fill remaining height of pitch pans with specified pitch pan filler.
- 6. Install hood over pan securing to each side with self-tapping screws.
- 7. Install face plate to cover box opening around pipe penetrations.
- 8. Strip in with membrane material.

## 3.4 CLEANING AND PROTECTION

- A. Remove flux and residual acid immediately by neutralizing with baking soda and washing with clean water. Leave work clean of stains.
- B. Remove scraps and debris and leave work area clean.
- C. Clean exposed metal surfaces, removing substances which might cause corrosion of metal or deterioration of finishes. Paint areas where finish is damaged on pre-finished metal by painting with a compatible paint in color to match undamaged finish.
- D. Prime soldered area of phosphatized metal after cleaning to prevent rusting.
- E. Paint metal flashings that have been soiled with bitumen with aluminized paint.
- F. Clean other work damaged or soiled by Work of this Section.
- G. Protect finished work from damage.

## END OF SECTION

## SECTION 07 92 00 - JOINT SEALANTS

## PART 1 - GENERAL

## 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.

#### 1.2 SECTION INCLUDES

- A. Exterior sealants and sealants for moving joints, except for joints in those systems listed under Related Work.
- B. Interior sealants and caulking at all dissimilar materials.

#### 1.3 RELATED WORK

A. Division 7 - Roofing Sections: Sealants used in conjunction with roofing.

## 1.4 **REFERENCES**

- A. ASTM International (ASTM)
  - 1. C717, Standard Terminology of Building seals and Sealants
  - 2. C793, Standard Test Method for Effects of Accelerated Weathering on Elastomeric Joint Sealants
  - 3. C794, Standard Test Method for Adhesion-in-Peel of Elastomeric Joint Sealants
  - 4. C834, Standard Specification for Latex Sealants
  - 5. C920, Standard Specification for Elastomeric Joint Sealants
  - 6. C1193, Standard Guide for Use of Joint Sealants
- B. Sealant, Waterproofing and Restoration Institute (SWRI)
  - 1. The Processional's Guide

## 1.4 SUBMITTALS

- A. Product Data:
  - 1. Manufacturer's specifications and other data needed to prove compliance with specified requirements.
  - 2. Manufacturer's installation instructions
- B. Sample: On site sample for Architect's approval of colors.
- C. Certification: Manufacturer's affidavit that materials used in Project contain no asbestos.
- D. VOC Content: Penetration firestopping sealants and sealant primers shall comply with the following limits for VOC content when calculated according to 40 CFR 59, Subpart D (EPA Method 24):
  - 1. Sealants: 250 g/L.
  - 2. Sealant Primers for Nonporous Substrates: 250 g/L.
  - 3. Sealant Primers for Porous Substrates: 775 g/L.

- E. Submit environmental data in accordance with Table 1 of ASTM E2129 for products provided under work of this Section.
- F. Certificates and Reports:
  - 1. Product Certificates: Manufacturer's product certificate for each kind of joint sealant and accessory.
  - 2. Sealant, Waterproofing, and Restoration Institute (SWRI) Validation Certificate: For each sealant specified to be validated by SWRI's Sealant Validation Program.
  - 3. Product Test Reports: Based on evaluation of comprehensive tests performed by a qualified testing agency, indicating that sealants comply with requirements.
  - 4. Preconstruction Compatibility and Adhesion Test Reports: From sealant manufacturer, indicating the following:
    - a. Materials forming joint substrates and sealant backings have been tested for compatibility and adhesion with sealants.
    - b. Interpretation of test results and written recommendations for primers and substrate preparation needed for adhesion.
  - 5. Preconstruction Field Adhesion Test Reports: Indicate which sealants and joint preparation methods resulted in optimum adhesion to joint substrates based on testing specified.
  - 6. Field Adhesion Test Reports: For each sealant application tested.

## 1.5 QUALITY ASSURANCE

- A. Installer Qualifications: experienced in building sealant installation whose work has resulted in sealant installations with a record of successful performance.
- B. Source Limitations: unless specifically indicated, obtain each type of building sealant through one source from a single manufacturer.
- C. Pre-construction Field Adhesive Testing: Prior to installation of building sealants, field test their adhesion to joint substrates in accordance with manufacturer's instructions. Perform test in locations indicated by Architect. Perform test for each type of building sealant and each substrate as required by Architect. If required by Architect, arrange for tests to be performed with sealant manufacturer's representative present. Follow-up review by Architect and manufacturer may be required to observe sealant performance over time and may result in re-application of sealant or replacement.
- D. Cleaning: Facade sealants that have collected dirt at the time of Substantial Completion shall be cleaned over the entire facade prior to acceptance by the Owner. 11 months after final completion of the building, if the sealant joints show dirt, they shall again be cleaned over the entire façade.

## 1.6 **PROJECT CONDITIONS**

- A. Environmental Conditions: Do not proceed with installation of sealants when joint substrates are wet or when ambient temperature conditions are above limits permitted by sealant manufacturers or are below 40 degree F.
- B. Joint Width Conditions: Do not proceed with installation of joint sealants where joint widths are less than those allowed by sealant manufacturer for applications indicated.
- C. Joint-Substrate Conditions: Do not proceed with installation of sealants until contaminates which may interfere with adhesion are removed from substrates.

## 1.7 PRE-INSTALLATION CONFERENCE

- A. Refer to Section 01 31 13 Project Coordination.
- B. In addition, refer to information above concerning Field Adhesive Testing.

## 1.8 WARRANTY

- A. Installer's Warranty: Written warranty, signed by Installer agreeing to repair or replace elastomeric joint sealant work which has failed to provide a weathertight system within specified warranty period.
  - 1. Warranty Period: Five years from date of Substantial Completion.
- B. Warranties: Written warranties (weatherseal and stain resistance), signed by sealant manufacturer agreeing to furnish joint sealants to repair or replace those that fail to provide airtight and watertight joints, or fail in adhesion, cohesion, abrasion resistance, stain resistance, weather resistance, durability, or appear to deteriorate in manner not specified in the manufacturer's data as an inherent quality of the material within specified warranty period.
  - 1. Warranty Period: 2 years from date of Substantial Completion.
- C. Warranties specified exclude deterioration or failure of sealants from:
  - 1. Movement of the structure caused by structural settlement or errors attributable to design or construction resulting in stresses on the sealant exceeding sealant manufacturer's written specifications for sealant elongation and compression.
  - 2. Disintegration of joint substrates from natural causes exceeding design specifications.
  - 3. Mechanical damage caused by individuals, tools, or outside agents.
  - 4. Changes in sealant appearance caused by accumulation of dirt or other atmospheric contaminants.

## PART 2 - PRODUCTS

## 2.1 APPROVED MANUFACTURERS

- A. Specifications are based on the products or materials of the named manufacturer, otherwise selection may be made from any manufacturer listed below whose products meet or exceed the specifications. Other manufacturers must have a minimum of five (5) years experience manufacturing the products meeting or exceeding the specifications and comply with Division 1 requirements regarding substitutions to be considered.
  - 1. Typical Sealants:
    - a. Bostik
    - b. Pecora, Inc.
    - c. Sonneborn
    - d. Sika
    - e. Tremco Incorporated
  - 2. Silicone Sealants at Appliances:
    - a. Pecora, Inc.
    - b. General Electric Co.
    - c. Dow Corning

# 2.2 MATERIALS

- D. Silicone Sealant: ASTM C 920, Type S, Grade NS, Class 50, for Use NT.
  - 1. Use: Wetglazing
  - 2. Properties: Performance: Nonstain, nonbleed, nonstreaking to sealed and adjacent substrates.
  - 3. Cure System and Oil Content: Neutral Cure System specifically manufactured with controlled oil content to eliminate oil migration into sealed substrates and residue rundown over and onto adjacent substrates.
  - 4. Product and Manufacturer:
    - a. Dow Corning Corporation; 795
    - b. Or approved equal
- E. Polyurethane Sealants: ASTM C920, Type M, Grade NS, Class 25; use NT, M, A and O.
  - 1. Use: Typical 1" or smaller wall joints (One part polyurethane sealants).
  - 2. Properties: Performance: Nonstain, nonbleed, nonstreaking to sealed and adjacent substrates.
  - 3. Products and Manufacturers:
    - a. BASF Building Systems; Sonolastic NP-1.
    - b. Or approved equal
- F. Two Part Polyurethane Sealants: ASTM C920, Type M, Grade NS, Class 50; use NT, M, A and O.
  - 1. Use: Typical 2" or smaller wall joints (Two Part Polyurethane Sealants). For 3" or larger install Emseal Colorseal.
  - 2. Properties: Performance: Nonstain, nonbleed, nonstreaking to sealed and adjacent substrates. The minimum pli value after 7 day immersion shall not be less than 13 when tested in strict accordance with ASTM C794 Adhesion in Peel.
  - 3. Products and Manufacturers: One of the following:
    - a. BASF Construction Chemicals; NP 2.
      - b. Or approved equal
- G. Sealant Backing: Provide sealant backings that are nonstaining; compatible with joint substrates, sealants, primers, and joint fillers; and approved for applications indicated by sealant manufacturer based on field experience and laboratory testing.
  - 1. Cylindrical Sealant Backings: Preformed, compressible, resilient, nonstaining, nonwaxing, nonextruding backings of flexible plastic foam complying with ASTM C 1330, and of type indicated below. Select shape and density of cylindrical sealant backings in consultation with the manufacturer for proper performance in specific condition of use in each case.
  - 2. Type C: Closed cell polyethylene foam material with surface skin, nonabsorbent to liquid water and gas, nonoutgassing in unruptured state; one of the following:
    - a. HBR Closed Cell Backer Rod; Nomaco, Inc.
    - b. Sonolastic Closed-Cell Backer-Rod; BASF Construction Chemicals.
  - A. Primers, Cleaners, Top Coats: Use only materials listed as suitable in resistance to staining, compatibility and durability before proceeding.
  - B. Expanded Polyethylene Joint Filler: Provide flexible, compressible, closed-cell, polyethylene of not less than 10 psi compression deflection (25 percent); except provide higher compression deflection strength as may be necessary to withstand installation forces and provide proper support for sealants, surface water absorption of not more than 0.1 pounds per square foot, as manufactured by Sonneborn, or pre-approved equal.

- C. Sealant Backer Rod: Provide compressible rod stack of polyethylene foam, polyurethane foam, polyethylene jacketed polyurethane foam, butyl rubber foam, neoprene foam or other flexible, permanent, durable, non-absorptive material as recommended by sealant manufacturer for back-up of and compatibility with sealant. Where used with hot-applied sealant, provide heat-resistant type, which will not be deteriorated by sealant application temperature as indicated.
- D. Bond Breaker Tape: Provide polyethylene tape or other plastic tape as recommended by sealant manufacturer, to be applied to sealant-contact surfaces where bond to substrate or joint filler must be avoided for proper performance of sealant. Provide self-adhesive tape where applicable.

## PART 3 - EXECUTION

## 3.1 APPLICATION

- A. Temperatures: Do not install sealants when air temperature is under 40 degrees F. Sealants may be warmed to ease installation when recommended by the manufacturer.
- B. Tooling: Tool exposed joints to a slightly concave surface using slicking materials recommended by the manufacturer. The tooling procedure shall press sealant against the sides of the groove. No materials shall be left "feathered" out or smeared on the abutting materials. If necessary, protect adjacent surfaces with tape. Completed joints shall have a uniform professional appearance. Use an anti-tack compound on sealant that does not set up fast enough to avoid dust collection.
- C. Sealant Back-Up: Provide back-up filler where groove depth is too great to fill with sealant. Review joint design with Architect.
- D. Compressive Filler: Seal vertical expansion joints with fillers. Provide compressible filler twice the width of the joint and with a depth of 1-1/2 times the compressed width. Lap ends 2 inch minimum.
- E. Seal ends together in such a manner to allow natural drainage. Install filler by compressing material and sliding into joint. Align filler on one face of the joint before it expands to the full joint width.

## END OF SECTION 07 92 00

## SECTION 09 91 00 - PAINTING

## PART 1 - GENERAL

## 1.1 RELATED DOCUMENTS

A. Drawings and general provisions of the Contract, including General and Supplementary Conditions and other Division 01 Specification Sections, apply to this Section.

## 1.2 SECTION INCLUDES

- A. Surface preparation and field application of paints and stains on interior substrates where shown or required.
- B. Surface preparation and field application of paints and stains on exterior substrates where shown or required.
- C. Surfaces not included, as applicable to the project, include, but are not limited to the following:
  - 1. Areas above finished ceiling or scheduled "unpainted" on the Finish Schedule of the drawings, except for those items within those spaces scheduled to painted.
  - 2. Exposed concrete, unless noted otherwise.
  - 3. Shop coat of paint on metal, except for damaged shop primer touch-up (unless noted otherwise.
  - 4. Structural steel and related items scheduled to receive sprayed fireproofing or encased in concrete.
  - 5. Face brick, unless otherwise noted.
  - 6. Natural stone.
  - 7. Aluminum and copper items, unless noted otherwise.
  - 8. Factory finished items other than prime painted to be field paint finished.
  - 9. Glass and glass masonry.
  - 10. Sealants of types which should not be painted and to which paint will not adhere.
  - 11. Plastic laminate items, such as doors, countertops, casework, etc.
  - 12. Acoustical ceiling and grid work, unless noted otherwise.
  - 13. Acoustical panels
  - 14. Piping of copper, aluminum, and stainless steel.
  - 15. Stainless steel items.
  - 16. Ceramic or tile of any kind.
  - 17. Valves and controls.
  - 18. Sprinkler heads.
  - 19. Name plates on equipment.
  - 20. Fire rating labels, including those on fire rated doors and frames.
  - 21. Finish hardware, except that which is factory primed and designated as "BHMA 600", if any, in Finish Hardware Section.
  - 22. Materials not noted to be refinished or to receive a finish, except as noted, including, but not limited to the following:
    - a. Roofing
    - b. Asphalt paving, except for parking and lane striping, fire lane, and "wheel chair" handicapped access parking spaces.
    - c. Concrete paving and curbs, except for parking and lane striping, fire lane, and "wheel chair" accessible parking spaces.
    - d. Flooring, except those to receive game striping, or as noted.

# 1.3 **REFERENCES**

- A. American Society for Testing and Materials (ASTM)
  - 1. D16, Standard Terminology for Paint, Related Coatings, Materials and Applications.
- B. MPI Architectural Painting Specification Manual (MPI Manual)

## 1.4 SUBMITTALS

- A. Material lists. Give the supplier's name, product name, number and generic description of each proposed product and its use. Provide product data sheets if so requested.
- B. Samples. Submit full range of colors, patterns, textures and finishes available for selection, including the following:
  - 1. Color Chips: Provide complete duplicate sets of color chips for color selection.
  - 2. Small Applied Samples: Provide pieces of actual material on which paint will occur with minimum dry mil thickness of specified paint. Provide painted 12 inch x 12 inch actual gypsum wallboard samples with approved textures for Architect's approval. Approved samples will become standard for which all work will be judged.
  - 3. Sheen Samples: Provide full range of varying sheens when sheens are controllable by intermixing.
- C. Installed Samples. Provide large size samples for approval. Approved samples may be left in place as part of the work.
- D. One room and/or area, as selected by the Architect, shall be painted with materials specified or accepted and applied directly from container, un-thinned. After acceptance by Architect, room and/or area shall be standard of quality of entire project.
- E. Certification. Furnish a letter certifying that materials submitted are truly equivalent or better than those called out in the finish schedule.

## 1.5 **RESPONSIBILITY OF COORDINATION**

- A. Coordinate the work specified herein with the following work:
  - 1. Provide information to preceding trades for proper preparation of substrate.
  - 2. Inspect substrate before proceeding to verify proper preparation.
  - 3. Notify Architect of any item to receive paint which may not be covered by a scheduled finish type. Architect will furnish appropriate specification.

## 1.6 QUALITY ASSURANCE

- A. Materials:
  - 1. Delivery and Storage: Products shall be delivered to jobsite in unopened containers bearing manufacturer's labels intact and legible at time of use. Storage shall be in designated areas away from excessive heat and open flames and in accordance with manufacturer's recommendations.
  - 2. Quality or Grade:
    - a. Paints and coatings shall be the manufacturer's highest professional quality material of types specified and shall be applied directly from containers in which material is purchased, except where thinning is recommended by manufacturer and approved by Architect to suit

PAINTING 09 91 00 - 2 intended use, i.e. painting acoustical tile or panels without destroying their acoustical properties.

- b. Primers and other undercoat paints shall be those produced by same manufacturer as finish coats.
- c. Thinners shall be those recommended by paint manufacturer's printed instructions.
- 3. Equipment:
  - a. Spray Equipment: Shall be the type recommended for the application and shall be maintained clean and in proper working order.
  - b. Brushes, Rollers, etc.:
    - 1) Shall be new of the various sizes and types recommended for each application.
    - 2) Shall be properly cleaned and stored in accordance with manufacturer's instructions at the end of each days' use.
    - 3) Shall be replaced as often as necessary to attain the best finish quality in the Work.
- 4. Application:
  - a. Applicator:
    - 1) Shall be person(s) or entity specializing in application of paints and coatings of types specified with minimum five (5) years experience.
    - 2) Shall provide Owner and Architect a notarized certification that paint used is as specified.
  - b. Application:
    - Shall not proceed on surfaces which are not suitable to be painted, until such surfaces have been corrected. Notify Architect in writing of which surfaces need to be corrected and their locations. Surfaces shall be corrected by the responsible trades. Surfaces not suitable for painting shall include, but not be limited to:
      - a) Damaged surfaces.
      - b) Oily, greasy, dusty or excessively soiled surfaces.
      - c) Non-dressed welds which will be exposed to view.
      - d) Lack of touch-up where specified.
      - e) Rusted or excessively deteriorated shop-prime painted surfaces.
    - 2) Number of coats of each of several finishes shall be in accordance with detailed specifications, which will produce first quality finish if properly applied. If number of coats specified fails to produce a finish acceptable to Architect, this Contractor shall apply additional coat or coats at his own expense until acceptable finish is achieved

## 1.7 **PRODUCT HANDLING**

- A. Store only approved materials at the jobsite, and store only in a suitable and designated area restricted to the storage of paint materials and related equipment.
- B. Temperature in the storage area shall be between 40 degrees F and 110 degrees F. Open and mix all materials in the storage area.
- C. Use all means necessary to protect materials before, during, and after application and to protect the installed work and materials of all other trades.

- D. Apply water-base paints only when temperature of surfaces to be painted and surrounding air temperatures are between 50 degrees F (10 degrees C) and 90 degrees F (32 degrees C), unless otherwise permitted by paint manufacturer's printed instructions.
- E. Apply solvent-thinned paints only when temperature of surfaces to be painted and surrounding air temperatures are between 45 degrees F (7 degrees C) and 95 F (35 degrees C), unless otherwise permitted by paint manufacturer's printed instructions.
- F. Do not paint in snow, rain, fog or mist, or when relative humidity exceeds 85 percent, or to damp or wet surfaces, unless otherwise permitted by paint manufacturer's printed instructions. Painting may be continued during inclement weather if areas and surfaces to be painted are enclosed and heated within temperature limits specified by paint manufacturer during application and drying periods.

# 1.8 EXTRA STOCK

A. Upon completion of the work of this section, deliver to the Owner, an extra stock equaling ten (10) percent or a minimum of one (1) gallon, whichever is greater, of each color, type, and gloss of paint used in the work, tightly sealing each container and clearly labeling contents and location where used.

# 1.9 WARRANTY

- A. The undertaking of a painting subcontract will indicate that the subcontractor will warrant the work specified herein for two (2) years against becoming unserviceable or causing an objectionable appearance resulting from either defective or nonconforming materials and workmanship.
- B. Defects shall include by not be limited to the following:
  - 1. Discoloring noticeably by yellowing, streaking, blooming, changing color or darkening
  - 2. Mildewing
  - 3. Peeling, cracking, blistering, alligatoring or releasing from the substrate
  - 4. Chalking or dusting excessively
  - 5. Changing sheen in irregular fashion
  - 6. Softening or becoming tacky
  - 7. Bubbling
- C. In the event of damage, immediately make all repairs and replacements necessary for approval of the Architect and at no additional cost to the Owner.

# PART 2 - PRODUCTS

# 2.1 ACCEPTABLE MANUFACTURERS

- A. All paint materials selected for coating systems for each type of surface shall be the product of a single manufacturer and shall, as a system, have flame spread, fuel contribution, and smoke density test results less than 25.
- B. Paint materials listed herein, unless otherwise designated in the "Painting Schedule", are the products of The Sherwin-Williams Company, 101 Prospect Avenue N.W., Cleveland, OH 44115, (800) 321-8194 and require no further approval as to manufacturer or catalogue number.

- C. Similar firstline material of one of the following manufacturers may be used subject to approval by the Architect for items indicated to be coated with Glidden Professional materials:
  - 1. Paints:
    - a. Benjamin Moore & Co.
    - b. Glidden Professional (formerly ICI Dulux Paints)
    - c. Pittsburgh Paints, PPG Industries, Inc., Professional & High Performance Coatings
    - d. Pratt & Lambert Paints (Div. of Sherwin Williams)
    - e. The Sherwin Williams Co.
    - f. Tnemec Company Inc.
  - 2. Stains:
    - a. Olympic Stains, PPG Industries, Inc., Professional & High Performance Coatings
    - b. Samuel Cabot, Inc.
- D. VOC Content of Field-Applied Interior Paints and Coatings, Primers, and Stains, and Transparent Finishes: Provide products that comply with the following limits for VOC content, exclusive of colorants added to a tint base, when calculated according to 40 CFR 59, Subpart D (EPA Method 24) and Green Seal Environmental Standard GS-11 for interior paint VOC thresholds; these requirements do not apply to paints and coatings that are applied in a fabrication or finishing shop:
  - 1. Flat Paints, Coatings, and Primers: VOC content of not more than 50 g/L.
  - 2. Nonflat Paints, Coatings, and Primers: VOC content of not more than 150 g/L.
  - 3. Anti-Corrosive and Anti-Rust Paints Applied to Ferrous Metals: VOC content not more than 250 g/L.
  - 4. Flat Topcoat Paints: VOC content of not more than 50 g/L.
  - 5. Nonflat Topcoat Paints: VOC content of not more than 150 g/L.
  - 6. Anti-Corrosive and Anti-Rust Paints Applied to Ferrous Metals: VOC content not more than 250 g/L.
  - 7. Primers, Sealers, and Undercoaters: VOC content of not more than 200 g/L.
  - 8. Dry-Fog Coatings: VOC content of not more than 400 g/L.
  - 9. Zinc-Rich Industrial Maintenance Primers: VOC content of not more than 340 g/L.
  - 10. Pre-Treatment Wash Primers: VOC content of not more than 420 g/L.
  - 11. Clear Wood Finishes, Varnishes: VOC content not more than 350 g/L.
  - 12. Clear Wood Finishes, Lacquers: VOC content not more than 550 g/L.
  - 13. Shellacs, Clear: VOC content not more than 730 g/L.
  - 14. Stains: VOC content not more than 250 g/L.
- E. Chemical Components of Field-Applied Interior Paints and Coatings: Provide topcoat paints and anti-corrosive and anti-rust paints applied to ferrous metals that comply with the following chemical restrictions; these requirements do not apply to paints and coatings that are applied in a fabrication or finishing shop:
  - 1. Aromatic Compounds: Paints and coatings shall not contain more than 1.0 percent by weight of total aromatic compounds (hydrocarbon compounds containing one or more benzene rings).
  - 2. Restricted Components: Paints and coatings shall not contain any of the following:
    - a. Acrolein.
    - b. Acrylonitrile.
    - c. Antimony.
    - d. Benzene.
    - e. Butyl benzyl phthalate.
    - f. Cadmium.

- g. Di (2-ethylhexyl) phthalate.
- h. Di-n-butyl phthalate.
- i. Di-n-octyl phthalate.
- j. 1,2-dichlorobenzene.
- k. Diethyl phthalate.
- I. Dimethyl phthalate.
- m. Ethylbenzene.
- n. Formaldehyde.
- o. Hexavalent chromium.
- p. Isophorone.
- q. Lead.
- r. Mercury.
- s. Methyl ethyl ketone.
- t. Methyl isobutyl ketone.
- u. Methylene chloride.
- v. Naphthalene.
- w. Toluene (methylbenzene).
- x. 1,1,1-trichloroethane.
- y. Vinyl chloride.

# 2.2 MATERIALS

- A. Paint and Coatings: Ready mixed, except for field catalyzed coatings; having good flow and brushing properties and consistent drying or curing behavior, free of sags and streaks.
- B. Accessory Materials: Linseed oil, turpentine, paint thinners and other materials recommended by paint and coatings manufacturer as necessary to achieve finishes specified.
- C. Patching and Surface Preparation: Latex fillers as recommended by paint and coatings manufacturer.

## 2.3 COLORS

- A. Colors shall be as selected by Architect. Different colors may be selected for each room, and more than one (1) color may be selected in each room.
- B. In Gymnasia, all Corridors, Collaboration Areas, Band, Choir, Orchestra, Drill Team, Cafeteria, Main Street, Library, Kitchen, Commons, locker and dressing areas, and all other areas allow for six (6) colors on each wall as selected by Architect. Provide masking or other means to provide finish with clean lines free of runs, sags, holidays, or other defects. Note that color separation on bands or stripes on interior CMU walls shall be between raked horizontal joints as specified in Section 04 20 00, Unit Masonry.

## PART 3 - EXECUTION

## 3.1 EXAMINATION

- A. Verify that site environmental conditions are appropriate and substrates are in proper condition to receive Work of this Section.
- B. Verify that shop applied primers are compatible with specified finish coats.

- C. Measure moisture content of surfaces using an electronic moisture meter. Do not begin application of coatings unless moisture content of surfaces is below the following maximum values:
  - 1. Gypsum soffits: 12 percent.
  - 2. Plaster: 12 percent.
  - 3. Masonry surfaces: 12 percent.
  - 4. Wood surfaces: 15 percent.
  - 5. Vertical concrete surfaces: 12 percent.
  - 6. Horizontal concrete surfaces: 8 percent.

# 3.2 ITEMS TO RECEIVE PAINT

- A. Generally, all new items that are normally painted in any typical building, including but not limited to the following list:
  - 1. All ferrous metal
  - 2. All exterior galvanized metal excluding exterior lintels
  - 3. All exterior wood
  - 4. All interior exposed structural steel, joists, and metal roof deck above Natatorium ceiling.
  - 5. All interior wood
  - 6. All prime coated hardware
  - 7. All exposed conduit, outlet boxes and electrical cabinets, including those located in mechanical rooms.
  - 8. All exposed pipe, plumbing, and ductwork, including those located in mechanical rooms and outdoors.
  - 9. All metal grilles, except aluminum, unless otherwise indicated.
  - 10. All exposed gypsum board surfaces, including all mechanical rooms.
  - 11. All exposed concrete masonry units (CMU), including all mechanical rooms.
  - 12. Miscellaneous other items which normally require painting or are scheduled to be painted.
  - 13. Consult plans, finish schedule, details and specifications for other trades as all items usually field painted or finish will be considered as part of the Contract.
  - 14. All exposed mechanical equipment and electrical equipment.
  - 15. Mechanical Room piping and interior and exterior condenser water piping whether insulated or not insulated.
  - 16. All spray applied fireproofing in Mechanical Rooms.
  - 17. All cementitious wood fiber decks and exposed structure scheduled or noted to receive paint. Protect exposed wood structure from overspray.
  - 18. Color coding of mechanical room, central plant, and service yard pipes.
- B. All work where a coat of material has been applied must be inspected and approved by Architect before application of succeeding specified coat, otherwise no credit for coat applied will be given. Notify Architect when a particular coat has been completed for inspection and approval. Apply coats of material in strict accordance with manufacturer's specifications except where requirements of these specifications are in excess of manufacturer's requirements. Paint all sight exposed pipe and plumbing only after all mechanical work and tests have been completed.

# 3.3 PREPARATION

- A. General: Surface must be clean to insure adhesion. Remove oil and grease with paint thinner. Wash off dirt with warm soapy water and rinse with clean water. Remove rust by wire brushing or sanding.
- B. Wall surfaces must be dry before painting. Verify with moisture meter.

#### PAINTING 09 91 00 - 7

- C. Unfinished Surfaces
  - 1. Wood: Sand smooth and apply one (1) coat of Primer Undercoat. After primer has dried overnight, putty nail holes and cracks, then spot-prime putty with primer. Again, allow the primer to dry over- night, sand lightly and topcoat.
  - 2. Masonry and Concrete: Remove efflorescence or cement dust on masonry and concrete by etching with a 10 percent solution or muriatic (Hydrochloric) acid. Flush off surface after etching with clean water, and paint while still damp. On surface where muriatic acid cannot be used to neutralize the efflorescence, remove the efflorescence by sanding, scraping or wire brushing and apply a coat of Masonry Conditioner before painting. If efflorescence is not present, no primer is necessary on concrete and masonry surfaces. Fill voids and pores in concrete and haydite blocks with Latex Block Filler and allow to dry overnight before topcoating.
  - 3. Iron and Steel: Prime with Metal Primer and allow to dry overnight before topcoating.
  - 4. Galvanized Metal: Prime with galvanized metal primer and allow to dry overnight before topcoating.

# 3.4 APPLICATION

- A. General: Surfaces to be finished must be clean, dry and free of dirt, oils, loose paint or any other contamination that would adversely affect adhesion, protective properties or appearance of the coating.
- B. Paint Thickness: Provide the following minimum dry film thickness per coat unless noted otherwise:
  - 1. Enamels on Metal: 1 mil
  - 2. Latex Paints: 1 mil
  - 3. Metal Primers: 1.5 mils
  - 4. Undercoats: 1.5 mils
  - 5. Oil Paints: 1.5 mils
  - 6. Epoxy Coating: 2.0 mils
  - 7. Thickness test: Use observation gauge that measures "V" shape scratch.
- C. Allow exterior paints to dry 72 hours between coats and interior paint to dry 24 hours between coats. Allow all enamels and varnishes to dry 24 hours between coats. If enamel and varnishes are tacky after 24 hours, allow additional time until finish is dry.
- D. Leveling: Apply with proper consistency and quality so paint flows out to a level surface free of brush and roller marks, bubbles, dust, runs, sags, and holidays. Spread evenly.
- E. Appearance: Uniform color, texture and sheen.
- F. Neatness: Paint shall not be smeared, spattered or run over adjoining colors or materials. Cut-on lines shall be straight.
- G. Contractor to provide supplemental work lighting at areas being painted. Workers must be able to see clearly in the opinion of the Owner.

## 3.5 CLEANING AND PROTECTION

A. Keep project premises free of painting-related debris. Collect material that may constitute a fire hazard, place in closed metal containers, and remove daily from site.

B. Protect work adjacent to painting operations from paint spatters and spills. Immediately remove paint that falls on finished surfaces not scheduled to receive paint, using materials and techniques that will not damage affected surfaces.

#### 3.6 SCHEDULE

- Α. The following is a schedule of typical painted items and does not specifically include every item that is to receive paint but should establish type and quality of finish for all items normally included in a complete paint job.
- Β. Color of first finish coat shall be different from specified color of final finish coat in order to facilitate inspection of work.
- C. Exterior Surfaces: Note: Exterior surfaces are divided into two (2) different categories, based upon color and level of graffiti resistance required. System 1 will be used when standard earthtone colors or neutral colors are specified, and System 2 will be used when bright colors (primary reds, yellows, and oranges) are specified and/or when a graffiti resistant coating is required.
  - Galvanized Metal: 1.
    - Surface Preparation: Acid etch galvanized surfaces that have not a. weathered at least six (6) months prior to beginning painting operations.
    - b. Primer: One (1) coat DTM Primer/Finish (B66W1)
    - Finish: Two (2) coats DTM Acrylic Gloss (B66 Series) C.
  - 2. Un-galvanized Metal:
    - Primer: One (1) coat Kem Kromik Universal Primer (B50Z Series) a.
    - Finish: Two (2) coats Industrial Enamel (B54) b.
  - 3. Concrete and CMU:
    - Primer: One (1) coat PrepRite Block Filler (B25W25) a.
    - Finish: Two (2) coats A-100 Acrylic Gloss (A8 Series) b.
  - 4. Wood (Includes plywood siding and wooden trim):
    - Primer: One (1) coat A-100 Latex Wood Primer (B42W41) а
    - Finish: Two (2) coats A-100 Acrylic Gloss (A8 ser.) h
  - 5. Fiber-Cement Materials:
    - Primer: One (1) coat Loxon Masonry Primer (A24W300) а
    - Finish: Two (2) coats A-100 Acrylic Gloss (A8 Series) b.
  - Parking Line and Driveway Paint: Setfast Waterborne Yellow (TM225) (meets 6. Federal Specification (FS) TTP-1952-B)
  - Exterior uninsulated piping at cooling tower (colors as listed in item 8 below) 7. Primer: One (1) coat Devran 205 Universal Epoxy Primer
    - a.
    - Finish: Two (2) coats Devthane 379UVA Aliphatic Urethane Gloss b. Enamel
  - 8. All piping in mechanical rooms shall be painted in their entirety, in the following colors:

HVAC SYSTEM	SHERWIN WILLIAMS /	SHERWIN WILLIAMS /
	COLOR NUMBER	COLOR NAME
Chilled Water from Chiller	SW-4056 (Dark Blue)	Blueprint
Chilled Water to Chiller	SW-4054 (Light Blue)	Basin
Hot Water from Boiler	SW-6871 (Dark Red)	Positive Red
Hot Water to Boiler	SW-6858 (Light Red)	Zany Pink
Condenser Water from Tower	SW-4070 (Light Green)	Generator Green
Condenser Water to Tower	SW-4071 (Dark Green)	Rain Forest
Natural Gas Piping	SW-4084 (Safety Yellow)	Safety Yellow

D. Interior Surfaces:

- 1. Galvanized Metal:
  - a. Primer: One (1) coat DTM Acrylic Primer/Finish (B66W1)
  - b. Finish: Two (2) coats ProClassic Waterborne Acrylic Gloss (B31 Series)
- 2. Gypsum Wallboard:
  - a. Primer: One (1) coat PrepRite 200 Latex Primer (B28W200)
  - b. Finish: Two (2) coats ProClassic Waterborne Acrylic Semi-Gloss (B31 Series)
- 3. Concrete and CMU: (Enamel)
  - a. Primer: One (1) coat PrepRite Block Filler (B25W25)
  - b. Finish: Two (2) coats ProClassic Waterborne Acrylic Semi-Gloss (B31 Series)
- 4. Wood: (Painted)
  - a. Primer: PrepRite Classic Latex Primer (B28W101)
  - b. Finish: ProClassic Waterborne Semi-Gloss (B31 Series)
- 5. Wood: (Stained)
  - a. Stain: Wood Classics Interior Stain (A49-200 Series)
  - b. Finish (First Coat): Wood Classics Polyurethane Varnish (A67 Series)
  - c. Finish (Second Coat): Wood Classics Polyurethane Varnish (A67 Series)
- 6. Gypsum Wallboard: (Epoxy) Kitchens, bathrooms, laboratories, etc.
  - a. Primer: One (1) coat PrepRite 200 Latex Primer (B28W200)
  - b. Finish: Two (2) coats Water-Based Catalyzed Epoxy (B70/B60)
- 7. CMU: (Epoxy) Kitchens, bathrooms, laboratories, etc.
  - a. Primer: Two (2) coats Heavy Duty Block Filler (B42W46)
  - b. Finish: Two (2) coats Water-Based Catalyzed Epoxy (B70/B60)
- 8. Exposed structural steel columns, joists, roof deck above the acoustical ceiling in the Natatorium.
  - a. Prime: Tnemec Series 90-97 Tneme-Zinc applied at 2.5 to 3.5 dry mils.
  - b. Second Coat: Tnemec Series 115 Uni-Bond DF applied at 3.0 to 4.0 dry mils.
  - c. Finish Coat: Tnemec Series 115 Uni-Bond DF applied at 3.0 to 4.0 dry mils.

## END OF SECTION 09 91 00